# Examining sex differences in cognitive ageing and lifetime risk factors for dementia

Louisa Patricia Needham

Thesis submitted for the degree of Doctor of Philosophy (PhD)

MRC Unit for Lifelong Health and Ageing University College London (UCL)

2024

I, Louisa Patricia Needham, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

## Abstract

**Background:** Mechanisms of increased female dementia risk are currently unclear given a paucity of research explicitly examining sex differences in cognitive and brain ageing.

**Methods:** Using data from a mostly cognitively unimpaired cohort, the 1946 British Birth Cohort and its neuroscience sub-study, Insight 46, sex differences in life course cognitive performance and neuroimaging indicators of ageing-, vascular-, and Alzheimer's Disease pathology-related brain health at age ~70 are examined (Chapter 3). A cumulative measure of lifetime modifiable dementia risk factor exposures is derived and tested for associations with later-life cognition and brain health in males and females (Chapter 4). The extent to which the female-specific menopause transition associates with later-life cognitive and brain health outcomes is examined (Chapter 5). Multivariable regression analyses test the extent to which life course socioeconomic, health, lifestyle, and genetic (*APOE*- $\varepsilon$ 4) variables contribute to associations in both sexes.

**Results:** Females showed cognitive performance advantages across a range of assessments from childhood to later-life; socioeconomic and educational disparities suppressed female advantages. At age 70, males had smaller relative brain volumes and females had greater levels of cerebral small vessel disease (cSVD) pathology (Chapter 3). Modifiable dementia risk factor exposures, greater in males, negatively associated with later-life cognitive performance in both sexes and with smaller brain volumes in males, independently of early cognitive, socioeconomic, and *APOE*- $\varepsilon$ 4 predictors of risk exposures (Chapter 4). Later menopause age associated with better later-life cognitive and brain health outcomes in women (e.g. larger brain volumes), with associations partly explained by childhood cognition and health-related variables, respectively (Chapter 5).

**Conclusions:** Sex differences in cognitive resilience to brain ageing and cSVD, and in the life course sociocultural and biological mechanisms underlying cognitive resilience, were shown. Findings advocate for the importance of sex- and gender-based analyses to understand life course pathways to dementia in males and females.

#### Impact statement

Evidence from this thesis that demonstrated sex differences in cognitive performance and brain health reinforces initiatives which advocate the value of sex- and gender-based analyses (SGBA) in health research. For example, the Medical Science Sex and Gender Equity (MESSAGE) initiative aims to publish a sex and gender policy framework for the UK health research sector.<sup>1</sup> The findings in this thesis also provide evidence to support improved research on women's health issues, given that menopause timing was associated with later-life cognitive performance and brain health. Evidence from these analyses and the wider scientific literature indicates that female-specific reproductive factors, including but not limited to menopause, hormone therapy use, and pregnancy, could contribute to sex differences in later-life outcomes. The work presented in this thesis supports recommendations for improved data collection regarding women's health issues in cohort studies, which is often insufficient to facilitate meaningful analyses.

This thesis also advocates for policies to provide better support for women, which could potentially reduce the burden of women's health issues on later-life outcomes. An example of such policy is the Women's Health Strategy for England, which aims to improve the ways in which the health and care system listens to women, providing improved access to relevant information and healthcare support. Demonstration that gendered socioeconomic and educational inequalities typically favouring males could limit female opportunities to develop and maintain protective structural and functional brain resources also highlights how policies to reduce such inequalities could benefit female outcomes. Such policies might include improving educational access, supporting women to return to work after having children, and encouraging men to take on greater child rearing responsibilities.

The need for public health policies aimed at reducing dementia risk factor exposures is also reiterated through evidence that greater lifetime exposures to modifiable risks associated with adverse later-life cognitive and brain health outcomes. Sex-stratified analyses demonstrated that greater risk exposures could accelerate brain ageing particularly in males, while brain ageing in females was more strongly associated with female-specific reproductive ageing. The development of effective public health policies to prevent or delay dementia onset in both sexes will require careful consideration for the specific variables which increase

dementia risk in each sex and when in the life course interventions might be most beneficial, which could differ between males and females.

Findings from this thesis have been disseminated through national and international conference poster presentations (Alzheimer's Association International Conference/AAIC Neuroscience Next 2020, Wellcome Longitudinal Studies 2021, AAIC 2021, Organisation for the Study of Sex Differences 2023, AAIC 2023) and various internal seminar talks, including participation in the 2023 UCL Institute of Cardiovascular Science 3-minute thesis competition. Analyses from Chapter 5, specifically the examination of menopause age associations with later-life cognitive performance measures, have been published in a peer-reviewed academic journal (*Maturitas*, 2023). By invitation, findings from Chapter 4 (examining cumulative dementia risk factor exposures) will be presented in international talks for the Alzheimer's Association Sex and Gender Professional Interest Area and the UCL Dementia Research Centre in April and June 2024, respectively.

# Inclusion of published works

# UCL Research Paper Declaration Form

referencing the doctoral candidate's own published work(s)

**1.** For a research manuscript that has already been published (if not yet published, please skip to section 2)

## a) What is the title of the manuscript?

A comprehensive assessment of age at menopause with well-characterized cognition at 70 years: A population-based British birth cohort

### b) Please include a link to or doi for the work

- https://doi.org/10.1016/j.maturitas.2023.01.009
- c) Where was the work published?

Maturitas

d) Who published the work? (e.g. OUP)

Elsevier

- e) When was the work published?
- 1<sup>st</sup> April 2023
- f) List the manuscript's authors in the order they appear on the publication

Louisa P Needham, Kirsty Lu, Jennifer M Nicholas, Jonathan M Schott, Marcus Richards, Sarah-Naomi James

g) Was the work peer reviewed?

Yes

h) Have you retained the copyright?

Yes

i) Was an earlier form of the manuscript uploaded to a preprint server? (e.g. medRxiv). If 'Yes', please give a link or doi)

No

If 'No', please seek permission from the relevant publisher and check the box next to the below statement:

 $\boxtimes$  I acknowledge permission of the publisher named under **1d** to include in this thesis portions of the publication named as included in **1c**.

## For a research manuscript prepared for publication but that has not yet been published (if already published, please skip to section 3) N/A

- a) What is the current title of the manuscript?
- b) Has the manuscript been uploaded to a preprint server? (e.g. medRxiv; if 'Yes', please give a link or doi)
- c) Where is the work intended to be published? (e.g. journal names)
- d) List the manuscript's authors in the intended authorship order
- e) Stage of publication (e.g. in submission)
- **3.** For multi-authored work, please give a statement of contribution covering all authors (if single-author, please skip to section 4)

Louisa P Needham conceived and designed the study, performed the analyses, wrote the first draft of the manuscript, contributed to the interpretation of the results and critically revised the manuscript.

Kirsty Lu contributed to the interpretation of the results and critically revised the manuscript.

Jennifer M Nicholas supervised the statistical analyses, contributed to the interpretation of the results and critically revised the manuscript.

Jonathan M Schott contributed to the interpretation of the results and critically revised the manuscript.

Marcus Richards conceived and designed the study, contributed to the interpretation of the results and critically revised the manuscript.

Sarah-Naomi James conceived and designed the study, contributed to the interpretation of the results and critically revised the manuscript.

#### 4. In which chapter(s) of your thesis can this material be found?

Chapter 5i

5. e-Signatures confirming that the information above is accurate (this form should be cosigned by the supervisor/ senior author unless this is not appropriate, e.g. if the paper was a single-author work)

*Candidate* Louisa Needham 12.04.2024

Supervisor/ Senior Author (where appropriate) Dr Sarah-Naomi James 10.04.2024

# Acknowledgements

I would like to thank my supervisors, Dr Sarah-Naomi James and Professor Marcus Richards. I have learnt so much from you and I am grateful for how you have shared your knowledge and enthusiasm, challenged and stretched my thinking, and provided valuable support and encouragement throughout the PhD.

Thank you also to colleagues within the LHA and the Insight 46 team for your approachability, for offering advice and expertise, and for creating a positive, friendly environment to learn and work in. To the yoga group – it has been so valuable to have time set aside each week to destress, thank you for sharing in that and for keeping the sessions going.

The work in this thesis would not have been possible without the range of data obtained from NSHD study members over several decades. My thanks go to those involved in data collection and management, to the MRC as my funders, and to the study members who have shown incredible generosity and dedication to research.

The enduring support from my friends and family has been fundamental in getting me to this point. You have all kept believing in me when I doubted myself and have provided welcome distractions from PhD life.

I particularly want to thank my parents. I am incredibly grateful for everything you have done and continue to do for me. You have always encouraged me to follow my interests and have provided me with the opportunities to do so. I would not be where I am without your love and support.

Finally, to my fiancé, Max. Thank you for being there for me and having my back throughout this process, for celebrating the highs, helping me through the lows, and for always giving me reason to smile. I simply could not have done this without you.

# Table of Contents

Abstract
Impact statement
Inclusion of published works6
Acknowledgements
List of Tables
List of Figures
Abbreviations19
1.0. Introduction
1.1. The life course approach21
1.2. Defining sex and gender22
1.3. Defining cognition23
1.4. Defining dementia23
1.4.1. Dementia diagnosis25
1.5. Measuring brain health26
1.6. Resilience, reserve, and resistance27
1.7. Sex differences in cognition27
1.7.1. Summary29
1.8. Sex differences in the brain
1.8.1. Summary
1.9. Sex differences in dementia31
1.9.1. Summary
1.10. Dementia risk factors
1.10.1. Individual characteristics
1.10.2. Health and lifestyle risks
1.10.3. Sex differences in dementia risk factors

1.10.4. Summary		39
1.11. The role of me	nopause in female cognitive and brain ageing	39
1.11.1. The menopa	ause transition	40
1.11.2. Menopause	and cognition	41
1.11.3. Menopause	and brain health	42
1.11.4. Menopause	and dementia risk	44
1.11.5. Effects of ho	ormone therapy	45
1.11.6. Summary		46
1.12. Summary and	identification of gaps in the literature	47
2.0. Data and analytical st	trategy	51
2.1. Aims and objectives	s	51
2.2. Introduction to the	datasets	53
2.2.1. NSHD		53
2.2.2. Insight 46		54
2.3. Measures		55
2.3.1. Sex		55
2.3.2. Cognitive perfo	ormance	55
2.3.3. Brain health		62
2.3.4. Covariables		64
2.4. Analyses		65
2.4.1. Descriptive ana	alyses	65
2.4.2. Regression mod	delling	65
3.0. Empirical section A: S	Sex differences in lifetime cognitive performance and late	r-life
neuroimaging outcomes i	in NSHD and Insight 46	67
3.1. Introduction		68
3.1.1. Objectives and	research questions	69

3.1.2. Hypotheses	70
3.2. Analytic method	71
3.2.1. Analytic sample	71
3.2.2. Cognitive outcome measures	71
3.2.3. Neuroimaging outcome measures	72
3.2.4. Covariables	72
3.2.5. Statistical analyses	73
3.3. Results	75
3.3.1. Participant characteristics	75
3.3.2. Cognitive performance in childhood and young adulthood	84
3.3.3. Cognitive performance ages 43 to 69	87
3.3.4. Insight 46 wave I cognitive measures (age ~70)	90
3.3.5. Insight 46 wave I neuroimaging measures (age ~70)	92
3.4. Discussion	104
3.4.1. Key findings	104
3.4.2. Interpretation of findings	104
3.4.3. Strengths and limitations	109
3.4.4. Summary	110
4.0. Empirical section B: Examining sex differences in cumulative lifetime	e dementia risks
and their association with later-life cognitive function and brain health.	111
4.1. Introduction	112
4.1.1. Objectives and research questions	114
4.1.2. Hypotheses	115
4.2. Analytic method	116
4.2.1. Analytic sample	116
4.2.2. Lifetime cumulative risks score (CRS)	116

4.2.3. Cognitive outcomes	123
4.2.4. Neuroimaging outcomes	
4.2.5. Early-life and genetic covariables	
4.2.6. Statistical analyses	
4.2.7. Secondary analyses	126
4.3. Results	126
4.3.1. Participant characteristics	126
4.3.2. Lifetime cumulative risks score (CRS) distribution and characteristic	cs127
4.3.3. Predictors of risk score	131
4.3.4. Associations of lifetime CRS with later-life cognitive performance a	nd brain health
measures	137
4.3.5. Use of risk-modifying treatments	143
4.3.6. Secondary analyses	145
4.4. Discussion	149
4.4.1. Key findings	149
4.4.2. Interpretation of findings	149
4.4.3. Strengths and limitations	155
4.4.4. Summary	158
5.0. Empirical section C: Menopause and female later-life cognition and brai	n health 159
5i. Menopause and later-life cognition	160
5i.1. Introduction	161
5i.2. Analytic method	163
5i.3. Results	170
5i.4. Discussion	175
5ii. Menopause and later-life brain health	179
5ii.1. Introduction	

5ii.2. Analytic method	
5ii.3. Results	
5ii.4. Discussion	
5.1. Summary	201
6.0. Discussion	202
6.1. Summary of key findings	202
6.2. Interpretation of key findings	207
6.2.1. Sex differences in reserve and resilience	207
6.2.2. Brain ageing vs. cerebrovascular health	
6.2.3. Midlife and menopause as a sensitive period for female	es209
6.2.4. The role of <i>APOE</i> genotype	210
6.2.5. Early-life and sociocultural influences	212
6.2.6. Pathways to greater female dementia risk	214
6.3. Methodological limitations	215
6.3.1. Multiple testing	215
6.3.2. Non-dementia outcomes	216
6.3.3. Data availability and biases	217
6.3.4. Generalisability	219
6.3.5. Sex and gender interrelationships	
6.4. Policy and research implications	223
6.4.1. Sex- and gender-based analysis (SGBA) research policie	s223
6.4.2. Women's health	
6.4.3. Reducing dementia risk exposures	225
6.5. Future directions	226
6.5.1. Cognitive resilience and risk factor exposure interrelation	onships226
6.5.2. Addressing risk-modifying treatment use	

6.5.3. Integrating female reproductive health228
6.5.4. Longitudinal follow-up229
6.5.5. Examining survivor bias230
6.5.6. Varied cohorts231
6.6. Conclusions231
References 233
Appendix A 267
Appendix B (Chapter 4 supplementary) 268
Appendix C (Chapter 5 supplementary) 274
5i. Supplementary material274
Model outputs from section 5i sensitivity analyses excluding women with ACE-III total
<82
Model outputs from section 5i sensitivity analyses excluding women who had surgical
menopause via bilateral oophorectomy285
5ii. Supplementary material287
Model outputs from section 5ii sensitivity analyses adjusting for WMHV296
Model outputs from section 5ii sensitivity analyses excluding women with clinically
diagnosed neurological conditions298
Model outputs from section 5ii sensitivity analyses excluding women with early
menopause (age <45 years)

# List of Tables

<b>Table 1</b> . (Chapter 2). Summary of the abilities tests and functional cognitive assessmentsadministered during NSHD data collection waves.56
<b>Table 2.</b> (Chapter 2). Summary of tests and outcomes derived from the Insight 46 wave Icognitive test battery, assessed at ages 69-7160
Table 3. (Chapter 3). Maximal samples at each timepoint71
Table 4. (Chapter 3). Summary of model adjustments at each timepoint.       75
Table 5. (Chapter 3). Descriptive statistics for each cognitive assessment complted by NSHDand Insight 46 males and females.77
Table 6. (Chapter 3). Descriptive statistics for each Insigt 46 neuroimaging outcome measure.
<b>Table 7.</b> (Chapter 3). Summary of male and female participant characteristics for covariablesmeasured at different timepoints across the life course.80
<b>Table 8.</b> (Chapter 3). Estimated effects of sex on cognitive performance and neuroimagingoutcome measures
<b>Table 9.</b> (Chapter 3). Estimated interactive effects of sex and <i>APOE</i> -ε4 genotype on cognitive performance and neuroimaging outcome measures101
Table 10. (Chapter 3). Estimated effects of sex on cognitive performance and neuroimagingoutcome measures, stratified by APOE-ε4 status103
Table 11. (Chapter 4). Description of how each risk factor included in the cumulative risksscore (CRS) was measured, defined, and scored.118
Table 12. (Chapter 4). Participant characteristics for NSHD participants with available datafor at least one cognitive outcome measure at age 69.129
Table 13. (Chapter 4). Participant characteristics for the Insight 46 sub-sample130
Table 14. (Chapter 4). Estimated effects of early-life covariables on cumulative risks score, inmales and females.132
Table 15. (Chapter 4). Estimated effects of cumulative risks score on cognitive performanceand neuroimaging outcomes, in males and females.139
<b>Table 16.</b> (Chapter 4). Interactive effects of cumulative risks score with <i>APOE</i> -ε4 status on cognitive performance and neuroimaging outcome measures, in males and females140
Table 17. (Chapter 4). Estimated effects of cumulative risks score on cognitive performance

 Table 17. (Chapter 4). Estimated effects of cumulative risks score on cognitive performance

 and neuroimaging outcome measures in males and females, adjusting for covariables.....144

<b>Table 18.</b> (Chapter 4). Estimated effects of midlife cumulative risks score on cognitiveperformance and neuroimaging outcome measures, in males and females.146
<b>Table 19.</b> (Chapter 4). Estimated effects of non-weighted lifetime cumulative risks score with cognitive performance and neuroimaging outcome measures, in males and females148
Table 20. (Chapter 5i). Characteristics for women with available data on age at menopauseand cognitive home assessment at age 69 (ACE-III), and Insight 46 sub-studyneuropsychology assessment (PACC).165
<b>Table 21.</b> (Chapter 5ii). Participant characteristics for women with available age at naturalmenopause and neuroimaging outcome measures.187
<b>Table 22.</b> (Chapter 5ii). Estimated effects of menopause age on cortical thickness regions of interest, stratified by <i>APOE</i> -ε4 status

# List of Figures

<b>Figure 1.</b> (Chapter 3). Unadjusted (M0) model estimates and 95% confidence intervals assessing sex differences in cognitive assessments administered during childhood (ages 8, 11, and 15) and young adulthood (age 26)
<b>Figure 2.</b> (Chapter 3). Unadjusted (M0) model estimates and 95% confidence intervals assessing sex differences in cognitive assessments administered at ages 43, 53, 60-64, and 69.
<b>Figure 3.</b> (Chapter 3). Unadjusted (M0) model estimates and 95% confidence intervals assessing sex differences in cognitive assessments administered at Insight 46 wave I (age ~70)
<b>Figure 4.</b> (Chapter 3). Model estimates and 95% confidence intervals assessing sex differences in total brain volume, hippocampal volume, and white matter hyperintensity volume outcomes
<b>Figure 5.</b> (Chapter 3). Model estimates and 95% confidence intervals assessing sex differences in amyloid standardised uptake value ratio and odds ratio for being amyloid positive
<b>Figure 6.</b> (Chapter 3). Model estimates and 95% confidence intervals assessing sex differences in normal appearing white matter fractional anisotropy and mean diffusivity measures
<b>Figure 7.</b> (Chapter 3). Model estimates and 95% confidence intervals assessing sex differences in cortical thickness regions of interest
<b>Figure 8.</b> (Chapter 4). Distribution of lifetime cumulative dementia risk scores in males (N=740) and females (N=769), in the whole-NSHD analytical sample
<b>Figure 9.</b> (Chapter 4). Distribution of raw scores per risk factor, in males (N=740) and females (N=769) in the whole-NSHD analytical sample
<b>Figure 10.</b> (Chapter 4). Distribution of lifetime cumulative dementia risk scores in males (N=201) and females (N=188), in the Insight 46 analytical sample135
<b>Figure 11.</b> (Chapter 4). Distribution of scores per risk factor, in males (N=201) and females (N=188) in the Insight 46 analytical sample
<b>Figure 12.</b> (Chapter 4). Predicted associations of cumulative risks score with standardised cognitive test performance at age 69 in males and females141
<b>Figure 13.</b> (Chapter 4). Predicted associations of cumulative risks score with total brain volume (TBV) and hippocampal volume at age ~70, in Insight 46 males (n=239) and females (n=229).
Figure 14. (Chapter 5i). Flow chart demonstrating the sample selection for analyses164

**Figure 16.** (Chapter 5i). Model estimates and 95% confidence intervals for the effect of 1year increase in age at menopause on Preclinical Alzheimer's Cognitive Composite (PACC; total score and sub-tests) at age 69 to 71 in the Insight 46 sample......173

**Figure 18.** (Chapter 5ii). Predicted values and 95% confidence intervals for the non-linear association of age at natural menopause with amyloid standardised uptake value ratio. ...189

# Abbreviations

Αβ	Beta-amyloid
ACE-III	Addenbrooke's Cognitive Examination – 3 <sup>rd</sup> edition
AD	Alzheimer's disease
APOE	Apolipoprotein E
CRS	Cumulative risks score
cSVD	Cerebral small vessel disease
CVD	Cardiovascular disease
DSST	Digit-symbol substitution test
FA	Fractional anisotropy
FNAME	Face-name associative memory examination
GHQ-28	28-item general health questionnaire
GMV	Grey matter volume
HT	Hormone therapy
MCI	Mild cognitive impairment
MD	Mean diffusivity
MICE	Multiple imputation by chained equations
MMSE	Mini-mental state examination
MRC	Medical research council
MRI	Magnetic resonance imaging
NART	National adult reading test
NAWM	Normal appearing white matter
NSHD	National survey of health and development
PA	Physical activity
PACC	Preclinical Alzheimer's cognitive composite
PAF	Population attributable fraction
PET	Positron emission tomography
QC	Quality control
RCT	Randomised controlled trial
ROI	Region of interest
SEP	Socioeconomic position
SUVR	Standardised uptake value ratio
ТВІ	Traumatic brain injury
TBV	Total brain volume
ΤΙV	Total intracranial volume
WMHV	White matter hyperintensity volume

## 1.0. Introduction

Dementia is more common in women than in men; two-thirds of people with Alzheimer's Disease (AD; the most common form of dementia) are female.<sup>2</sup> Sex differences in dementia features (e.g. symptoms, biomarkers, progression)<sup>3</sup> suggest more complex explanations for increased female risk than higher female life expectancy alone, but the mechanisms underlying greater female risk are not yet well understood.

Traditionally, much dementia research has included only males or under-represented females, limiting our knowledge of dementia in females. More recently, the introduction of policies to include both sexes in research has increased female inclusion, but most studies simply adjust for sex in their analyses and do not explicitly examine potential sex differences. With the advent of emerging dementia treatments, understanding how dementia differs between males and females will be useful for treatment effectiveness and for identifying individuals who might benefit most. Additionally, identifying groups with the greatest risk can inform targeted strategies for dementia prevention, which may prolong improved quality of life and reduce the need for costly treatments which can induce side-effects. There is increasing recognition that while dementia primarily occurs at older ages, underlying pathology develops over several decades before symptoms emerge,<sup>4</sup> hence a life course approach to understanding dementia processes and risks is necessary.

Given the long preclinical phase in dementia, when there are pathological brain changes but preserved cognitive and functional abilities, subtle changes in performance on cognitive assessments and across multi-modal neuroimaging indicators of brain health could help us to understand the pathways leading to diagnosable dementia. This thesis examines data from a generally cognitively unimpaired cohort – the 1946 British birth cohort (the MRC National Survey of Health and Development/NSHD) and its neuroscience sub-study, Insight 46 – up to age ~70, when dementia diagnoses remain rare. NSHD is an age homogenous cohort studied since birth which is geographically representative of the UK and is comprised of ~50% females. A vast array of sociocultural, biological, and cognitive measures is available across multiple timepoints, including multiple indicators of brain health obtained using combined PET-MRI neuroimaging at age ~70 (Insight 46 only).

In this thesis, a life course approach is used to examine whether different biological and lifestyle factors measured throughout life associate with later-life cognition and brain health in males and females. Sex differences in the longitudinal trajectories of cognitive change over time are not examined given that such analyses have previously been conducted in NSHD; females showed slower decline in processing speed performance between ages 43 and 69, while there were no sex differences in verbal memory performance trajectories.<sup>5</sup> Lifetime factors which both males and females can be exposed to are considered and examined for sex differences in their associations with later-life outcomes, alongside consideration of sexspecific reproductive factors, namely menopause, which might contribute to female cognitive and brain ageing.

#### 1.1. The life course approach

With acknowledgement that dementia, while typically presenting at older ages, develops throughout life, a life course approach is important for elucidating how sex differences in dementia occur. Some key concepts in life course epidemiology are the sensitive period and accumulation models.<sup>6</sup>

The sensitive period model outlines that exposures within a specific time frame have stronger effects than outside that period. This is distinct from a critical period, where exposures have little to no influence outside of a limited time window.<sup>6</sup> Many, but not all, critical periods are developmental. For instance, maternal health behaviours (e.g. smoking, diet) during pregnancy influence foetal brain development,<sup>7</sup> with lasting impacts throughout life. Relevant to this thesis is the concept that midlife may be a sensitive period, particularly for females. As will be discussed in Section 1.11, the female menopause transition has been proposed as a neurological transition state which could represent a period of vulnerability for female brains to develop dementia pathology.<sup>8</sup>

The accumulative model contrasts from sensitive period perspectives in that exposures at any stage of life are proposed to influence health outcomes, and such exposures have additive effects on outcomes.<sup>6</sup> There can be a 'chain of risk' whereby one exposure leads to another and so on, such that the number of risk (or protective) exposures continually increases.<sup>6</sup> For example, poorer socioeconomic circumstances during childhood could lead to poorer diet and

reduced educational attainment, increasing the risk for adverse health behaviours such as smoking and physical inactivity, which then further increase the risk for cardiovascular disease. The effects of an exposure in midlife thereby represent the summed effects of all previous exposures up until that point.<sup>9</sup>

In reality, sensitive period and accumulation models are not mutually exclusive. For example, early-life as a sensitive period for brain development may set in motion a subsequent chain of events which accumulate throughout life; poor brain development *in utero* or during early infancy could limit later educational attainment, reducing occupational opportunities in adulthood. In midlife, menopause could increase female sensitivity to cardiovascular risks (e.g. female hypertension risk increases at the time of menopause),<sup>10</sup> which may trigger a cascade of further adversities (e.g. reduced physical activity, social isolation) beyond the menopause transition.

The sensitive period and accumulative models of life course epidemiology are inherently difficult to test. Support for a chain of risks would require causal evidence for each step in the chain. The sensitive period model can only be proven if the assumption holds that, apart from the timing of exposure, all else is equal including the level of exposure and consistent covariables. While not explicitly tested, the concepts of sensitive periods of vulnerability to dementia risks, accumulation of risks throughout the life course, and their interrelationships are drawn on throughout this thesis.

#### 1.2. Defining sex and gender

The terms sex and gender are often used interchangeably, and there are ongoing debates around how sex and gender should be defined, particularly regarding whether biological sex is in fact binary<sup>11</sup> and how gender can be measured.<sup>12</sup> For the purposes of this thesis, where the focus is on sex differences, sex refers to biologically defined characteristics (e.g. anatomical features, sex chromosomes, hormones) which are traditionally considered to distinguish males from females. Gender is self-identified and refers to the social constructs of masculinity and femininity, which lie on a spectrum.<sup>13,14</sup> Many sociocultural exposures are gendered and may interact with biological sex. For example, traditional gender roles are such that women take on childrearing, caregiving, and domestic responsibilities, while men work to earn money and provide for their families. Gendered sociocultural factors can vary across cultures and generations.

#### 1.3. Defining cognition

Cognition refers to a complex set of functions giving rise to knowledge, skills, thoughts, and behaviours. The main cognitive domains are perception, motor control, memory, attention, executive function, and language.<sup>15</sup>

Cognitive ageing is the decline in cognition experienced in late adulthood, as part of normal ageing. Crystalised abilities (knowledge and skills retained from previous experiences)<sup>16</sup> typically remain intact with advancing age. Tests of such abilities include assessments of general knowledge, vocabulary and reading comprehension (e.g. National Adult Reading Task/NART). Fluid abilities (those facilitating perception, interpretation, and responses to environmental stimuli)<sup>16</sup> tend to decline from midlife onwards.<sup>17,18</sup> Fluid ability assessments typically involve problem solving or rapid responses (e.g. mental rotation and choice reaction tasks).

## 1.4. Defining dementia

Dementia is a syndrome, typically occurring approximately after age 65 years, whereby cognitive function declines faster than expected for normal ageing and causes functional impairment.<sup>19</sup> Dementia is caused by various conditions affecting the brain, including disease and injury, with dementia sub-types having different causes and underlying pathology. AD, responsible for 60-70% of world-wide dementia cases,<sup>19</sup> is characterised by the pathological hallmarks of extracellular beta-amyloid (Aβ) plaques and intracellular neurofibrillary tau tangles. These pathologies, which can disrupt neuronal functioning and contribute to brain atrophy,<sup>20</sup> typically first appear in the hippocampi and entorhinal cortex. Memory impairment is therefore often, although not always, an initial symptom of AD. Cognitive impairment can first present as mild cognitive impairment (MCI), when patients experience cognitive decline beyond expected with normal ageing but daily living is not significantly disrupted. Amnestic MCI primarily affects memory, while non-amnestic MCI affects other cognitive domains.<sup>21</sup> In

dementia, as damage spreads throughout the brain cognitive impairment becomes more severe and more cognitive functions are affected.<sup>22</sup>

The presentation of cognitive impairment may differ across dementia sub-types, including frontotemporal dementia (FTD), Parkinson's Disease Dementia (PDD), Lewy Body Dementia (LBD), and vascular dementia. FTD, characterised by progressive symptoms which vary according to variant (e.g. executive function impairments with behavioural variant, speech expression difficulties with progressive non-fluent aphasia), often has a younger age at onset than most other dementias and has greater levels of genetic heritability.<sup>23</sup> PDD is associated with a movement disorder caused by a loss of dopaminergic neurons in the substantia nigra, leading to symptoms including bradykinesia, rigidity, resting tremor, gait disturbance, and speech difficulties.<sup>24</sup> LBD, which can co-occur with Parkinson's Disease, is characterised by abnormal accumulation of  $\alpha$ -synuclein (Lewy Body) protein and, in addition to cognitive decline, can include symptoms such as visual hallucinations, depression, and Parkinsonism motor symptoms.<sup>25</sup>

Vascular dementia is caused by reduced blood flow to the brain, with symptoms varying dependent on the location of brain damage.<sup>26</sup> Large vessel disease (e.g. atherosclerosis, occlusion/stroke) causes neuronal injury and grey matter infarcts (tissue death due to reduced blood supply).<sup>27</sup> Small vessel disease (e.g. microbleeds, pathologic protein deposition) leads to axonal injury and white matter infarcts.<sup>27</sup> White matter hyperintensities, regions of intense signal on MRI, are examined in this thesis as indicators of axonal loss or ischaemic damage related to cerebral small vessel disease (cSVD).<sup>28,29</sup> There is a growing understanding of the vascular contributions to cognitive impairment and dementia (VCID), with cerebrovascular comorbidities often found across dementia sub-types.<sup>30</sup> For example, vascular and AD pathology comorbidities include reduced cerebral blood flow associated with Aβ deposition, impaired Aβ clearance with cerebrovascular changes, and increased risk of AD with greater levels of white matter hyperintensities.<sup>30,31</sup> While the directionality of these associations is unclear, it is important to understand how cerebrovascular changes might contribute to the development of dementia pathology and clinical symptoms.

#### 1.4.1. Dementia diagnosis

There are three stages of dementia: preclinical, prodromal, and diagnosable dementia.<sup>31</sup> Preclinical dementia is defined by the presence of dementia pathology (e.g. Aβ) but the absence of symptoms. For AD, there is a long preclinical phase whereby pathological proteins are detected several years or decades prior to symptom onset.<sup>4</sup> The prodromal phase is MCI, which increases dementia risk, but some individuals with MCI never progress to dementia and cognition can return to normal.<sup>21</sup> Diagnosable dementia, according to the DSM-V (Diagnostic and Statistical Manual of Mental Disorders, fifth edition), requires significant decline in one or more cognitive domain, impacting the individual's ability to independently complete everyday activities.<sup>32</sup>

Dementia is typically diagnosed by a GP or specialists at a memory clinic. A clinical history is obtained, including informant (e.g. family member) accounts, to understand the patient's symptoms and their impact on daily living.<sup>32</sup> Clinical tests are also administered, such as the General Practitioner Assessment of Cognition (GPCOG),<sup>33,34</sup> to examine the patient's cognitive abilities across cognitive domains. Brain scans are not typically used in dementia diagnosis, particularly if dementia is the likely diagnosis based on clinical tests, but MRI can help to identify dementia sub-type (e.g. atrophy in the frontal and temporal lobes can indicate FTD) or to rule out other causes (e.g. brain tumour).<sup>35</sup> Indeed, dementia is a diagnosis of exclusion, whereby it must be determined that other neurological conditions (e.g. depression) do not better explain the clinical symptoms.<sup>32</sup>

Existing methods to detect pathological proteins, such as PET neuroimaging or cerebrospinal fluid (CSF) measures of Aβ and tau, for example, are not widely used in clinical settings due to cost and practicality limitations.<sup>32</sup> Work is ongoing to develop effective blood-based biomarkers of dementia;<sup>36</sup> plasma phosphorylated tau (p-tau), for example, shows promising associations with AD.<sup>37</sup> The feasibility of using plasma p-tau in clinical settings is being investigated, with potential benefits for detecting early dementia and for identifying individuals who might benefit from new AD treatments.

#### 1.5. Measuring brain health

Historically, brain health was measured only *post-mortem*, but the development of neuroimaging and other (e.g. CSF and blood-based biomarkers) techniques now facilitate *in vivo* assessments of brain health and the presence of dementia pathologies.

Structural MRI measures of total brain volume (TBV) and hippocampal volume are some of the most well-established neuroimaging indicators of brain health. Smaller volumes indicate greater levels of brain atrophy, which gradually occurs with age and is accelerated by neurodegenerative processes linked with dementia.<sup>38</sup>

Cortical thickness also reduces with age and cortical atrophy accelerates with dementia, with the pattern of thinning across brain regions varying depending on dementia type.<sup>18</sup> MRI signatures of AD have been derived, comprising cortical regions of interest across which thinning is characteristic of AD.<sup>19,20</sup>

White matter hyperintensities on MRI, which indicate cSVD and associate with increased dementia risk,<sup>28,29</sup> were previously manually counted but can now be measured using an automated pipeline to generate white matter hyperintensity volume (WMHV).<sup>39</sup> There is also increasing recognition that changes in the microstructural integrity of normal appearing white matter (NAWM) could be a precursor to the development of white matter hyperintensities,<sup>40</sup> with changes in NAWM also linked with ageing<sup>41</sup> and cognitive decline.<sup>42</sup> Measures of NAWM are relatively new and asses axonal integrity using MRI techniques such as diffusion tensor imaging (DTI) which measures the diffusion properties of water molecules within NAWM; greater diffusivity indicates reduced microstructural integrity.<sup>42</sup>

PET neuroimaging uses radioactive tracers to detect the presence of different molecules within the brain. Tracers which bind to  $A\beta$  or tau proteins have been relatively recently introduced, facilitating *in vivo* neuroimaging which can detect preclinical levels of AD pathology.<sup>43</sup>

With advances in neuroimaging techniques, multimodal methodologies such as combined MRI and PET imaging can now be used to examine the underlying disease processes of dementia, including how vascular and other pathologies develop across different dementia phases and sub-types.<sup>44</sup> These techniques will be useful for detecting the early, preclinical phases of dementia; with the advent of new treatments aimed at reducing levels of

pathological proteins in the brain, early dementia screening could identify individuals who are most likely to benefit from such treatments. Additionally, neuroimaging methodologies sensitive to subtle brain changes can be valuable endpoints in clinical trials aiming to slow or reverse dementia processes.

#### 1.6. Resilience, reserve, and resistance

The overlapping concepts of resilience, reserve, and resistance, although an evolving field, can offer a framework through which individual differences, including sex differences, in cognitive abilities and susceptibility to brain ageing and dementia may be understood.<sup>45</sup> Resistance refers to the ability to avoid pathologies; some individuals may show low levels of pathology despite high levels of risk.<sup>46</sup> In contrast, resilience refers to the brains' ability to maintain cognitive function in spite of brain ageing and pathology, encompassing various mechanisms through which individuals may be better able to manage the impacts of pathology: cognitive reserve, brain reserve, and brain maintenance.<sup>45</sup> Cognitive reserve describes the adaptability of functional brain processes while brain reserve refers to structural brain characteristics (e.g. number of neurons and synapses) which can protect against the effects of pathology. Brain maintenance reflects the plasticity of the brain, whereby genetics and lifestyle have the potential to enhance brain structure and function. Therefore, across the life course there are opportunities for risk and protective factors to limit or enhance the ability of the brain to avoid pathology (resistance) and to cope with pathology (maintain cognitive function and brain structure; reserve) when it does emerge.

## 1.7. Sex differences in cognition

Whether males and females differ in their cognitive abilities has traditionally been a topic of scientific and popular interest. A wealth of studies have been conducted, although limitations, particularly in older studies, include the impact of researcher bias and failure to consider societal, gender-based influences such as stereotypical beliefs about men and women's abilities.<sup>47,48</sup>

Nonetheless, the most consistent sex differences reported are male advantages in spatial abilities and female advantages in verbal abilities. Males typically outperform females on mental rotation tasks with moderate effect sizes across the life course, but other spatial assessments (e.g. spatial visualisation, spatial perception) show only small male advantage.<sup>49</sup> The most consistently reported female advantage is in verbal fluency, although effect sizes are also typically small, and meta-analyses do not support statistically significant sex differences in crystallised verbal abilities such as vocabulary or reading comprehension.<sup>47</sup> The description of variations between male and female cognitive task performance as sex differences should be interpreted with caution, given that most 'differences' refer to mean performance levels while there is still a large amount of overlap in performance scores for males and females.<sup>47</sup> Additionally, the extent to which biological sex explains observed differences in cognitive performance between males and females is debateable, with evidence showing that when the gendered context of task instructions is altered, sex differences in task performance can be removed. For instance, describing a mental rotation task as assessing stereotypically female skills (e.g. handicraft abilities) removed the male advantage in mental rotation task performance.<sup>50</sup>

There is also a lack of consensus regarding whether males and females vary in their rates of cognitive decline with advancing age. Across five US cohorts<sup>51</sup> (median baseline age=58 years, median follow-up=7.9 years) no sex differences in the rate of memory decline were detected but overall cognitive state and executive function declined faster in women. Conversely, in the English Study of Longitudinal Ageing (ELSA; mean baseline age men=64.6 years, women=65.0 years), cognitive state, executive function and memory declined slower in women, over 8 years.<sup>52</sup> Data from NSHD also showed slower decline in women on repeated search speed measures between ages 43 and 69 years, although there were no sex differences in rates of verbal memory decline.<sup>5</sup> A systematic review of longitudinal research on cognition in people aged 60 years and over (follow-up range 5-17 years) concluded that men and women have similar rates of age-related cognitive decline, although it remains unclear whether sex differences in cognitive decline are present after age 80.<sup>53</sup> Inconsistent findings across studies could reflect variations in sample age ranges, since rates of age-related cognitive decline might not be linear. Additionally, there may be cultural differences in gendered exposures between samples; in the US women might be less likely to work than

28

women in the UK, which might contribute to varied sex differences in cognitive decline between the two countries.

Societal gender roles also vary over time, reflected by generational variations in socioeconomic opportunities, including educational access which has traditionally been limited for women but has increased over time.<sup>46</sup> Such generational changes have implications for sex differences in cognitive performance. For example, in the ELSA and Whitehall II UK-based cohorts, secular improvements in women's access to education associated with better female performance on verbal memory and verbal fluency tasks, and females in more recent generations (with greater access to education) outperformed males on verbal fluency while males outperformed females in earlier-born generations.<sup>54</sup> The impact of societal influences on cognition demonstrates that cognitive performance differences between males and females are not wholly biologically determined, and that observed sex differences may not be consistent across cultures or generations, given variations in gendered sociocultural exposures.

#### 1.7.1. Summary

Overall, findings demonstrate that men and women have similar abilities in most cognitive domains, in agreement with the gender similarities hypothesis<sup>55</sup> which also states that observed differences can vary dependent on factors such as the context of performance measurement. When sex differences in mean performance are seen they tend to be subtle with external, gendered societal influences offering some explanation for the differences. It is also unclear whether males and females differ in their rate of age-related cognitive decline. Most longitudinal research addresses cognitive change from late adulthood onwards; it is unclear whether there are sex differences in cognitive change at midlife, a time when many risk factors for dementia are present.<sup>56</sup> Further, the cognitive tests typically assessed in epidemiological studies may not be sensitive to detect subtle, early dementia-related cognitive decline or potential sex differences in such changes.

#### 1.8. Sex differences in the brain

After accounting for brain size, which is typically larger in males throughout life, sex explains only 1% of the variance in regional brain volumes.<sup>57</sup> Historical reports of greater hippocampal volume in males than females, when brain size was not accounted for,<sup>47</sup> do not therefore represent a robust sex difference in hippocampal volumes.

Some evidence does, however, indicate sex differences in levels of brain activation, even when cognitive task performance is equal. For instance, females are reported to have greater global cerebral blood flow, measured using SPECT (single photon emission computed tomography) and PET imaging techniques, during both rest and cognitive activities including when no significant sex differences in cognitive task performance are detected.<sup>58,59</sup> Females are also reported to have greater levels of cerebral glucose metabolism than males,<sup>58</sup> although evidence has shown variations in females' regional glucose metabolism across menstrual cycle phases,<sup>60</sup> indicating a potential role of sex hormones on brain activation. Despite such evidence though, most sex differences research does not report or consider how menstrual cycle phases in females of reproductive age might contribute to observed sex differences or similarities.

Although sex differences in Aβ are not consistently reported in people with and without cognitive impairment or dementia,<sup>61</sup> sex differences in other neuroimaging indicators of dementia-related pathology have been identified. For instance, females are found to have poorer NAWM microstructural integrity than males,<sup>62-64</sup> and females reportedly have greater WMHV,<sup>65</sup> particularly at older ages when women are post-menopausal.<sup>66</sup> Additionally, in the longitudinal Rotterdam study (n=5,286, mean baseline age=64.4 years, mean follow-up=5.2 years), males started to show increases in WMHV at younger ages than females,<sup>67</sup> further indicating that trajectories of cSVD differ between the sexes. Importantly, health and lifestyle behaviours which can show gendered differences in prevalence (e.g. physical inactivity is typically greater in women, alcohol consumption is higher in men) have been linked with indicators of cerebrovascular disease; sociocultural exposures may therefore contribute to observed sex differences in brain health.<sup>68</sup>

Males in the Rotterdam study have also shown earlier and faster decline in global and regional grey matter volumes than females, although significant sex-by-age interactions were not

detected.<sup>67</sup> The Baltimore Longitudinal Study of Ageing (n=617, mean baseline age=71.2 years, up to 20 years follow-up) did, however, find a significant sex difference in the annual rates of decline in TBV and grey matter volume (GMV) which were faster in males than females.<sup>69</sup> Similar to discrepancies in cognitive sex differences (Section 1.7.), these variations across studies might be explained by non-linear age-related trajectories of brain health, and by sociocultural differences between samples. For example, variations in educational access and occupations might contribute to differential levels of brain maintenance between males and females.

#### 1.8.1. Summary

Although evidence for sex differences in regional brain volumes is weak, there is support for sex differences in some neuroimaging indicators of dementia-related pathology and in rates of brain ageing. Despite greater levels of brain activation than males, females show poorer cerebrovascular health (indicated by reduced NAWM microstructural integrity and greater WMHV). Males show earlier and faster rates of WMHV increases and brain atrophy, but it is possible that at older ages and post-menopause, female trajectories of brain ageing accelerate. Sociocultural exposures which differ between genders, including health and lifestyle behaviours, are important to consider when elucidating the potential mechanisms underlying brain health sex differences.

#### 1.9. Sex differences in dementia

While overall rates of dementia are more frequent in females than males, largely driven by greater AD prevalence in females,<sup>2</sup> sex differences can vary by dementia type; LBD, PDD, and behavioural variant FTD are more frequent in males, and primary progressive aphasia FTD is more prevalent in females.<sup>70</sup> Variations in dementia symptoms and rate of progression between sexes are also recognised. Female PDD patients are more likely to have visuospatial difficulties, while males more commonly experience verbal fluency and emotion recognition problems.<sup>24</sup> In AD dementia, females typically experience greater cognitive impairment than males, with cognitively normal female advantages in verbal memory, for example, reversed in AD patients.<sup>61</sup> Female AD patients also show faster rates of cognitive decline than male AD

patients, although this could be partially explained by females typically receiving AD diagnoses at later disease stages than males.<sup>61</sup> Such sex differences demonstrate that there are likely variations and nuances in dementia development processes and mechanisms between males and females. Although these differences are not yet well understood, some theories have been proposed.

A common perspective is that females have greater dementia risk due to their greater longevity – dementia incidence and prevalence rates increase with age and females live longer than males,<sup>71</sup> even with AD diagnoses.<sup>72</sup> When accounting for age, studies in the US have not found sex differences in dementia rates,<sup>73,74</sup> giving some support to the theory of female longevity. However, several studies in Europe and Asia have found higher dementia rates in females even when accounting for age,<sup>75</sup> meaning that female longevity did not fully explain increased female dementia risk in these populations. Societal influences which vary across cultures are therefore important considerations, including gendered differences in education, occupations, caregiving responsibilities, and health and lifestyle behaviours (e.g. diet, smoking, physical activity). The role of survivor bias must also be considered and forms part of the debate as to whether female longevity explains dementia sex differences; cardiovascular disease increases dementia risk, but mortality rates from cardiovascular disease higher in males than females, meaning that males surviving to older ages have better cardiovascular health and therefore lower cardiovascular risks for dementia.<sup>61,71</sup>

Interestingly, longevity has been linked with sex chromosomes across species, with organisms carrying two X chromosomes surviving for longer.<sup>76</sup> A study using AD mouse models demonstrated that genetic sex (XX[female], XY[male]) had a greater impact on survival than phenotypic sex (expression of testes or ovaries), and that this was driven by the presence of two X chromosomes rather than absence of the Y chromosome.<sup>77</sup> Females are potentially afforded some level of resilience from having two X chromosomes, possibly explaining greater female longevity, even with AD. However, females can experience gene dosage complications. Across different female cells, one X chromosome is randomly inactivated to regulate gene dosage, but some genes escape inactivation.<sup>78,79</sup> While this is an evolving field of research, it is intriguing that many genes found to consistently escape inactivation are immune related.<sup>79</sup>

The immune system has a role in dementia, with neuroinflammation contributing to dementia pathology. In AD, microglia (the brain's immune cells) are activated in response to  $A\beta$  but

become less efficient at Aβ clearance after prolonged periods of activation. In a downward spiral, microglia are chronically activated and release toxins which damage neurons, leading to neurodegeneration.<sup>80</sup> Widely reported sex differences in immune responses, with females having stronger immune responses and higher incidence of autoimmune conditions than males,<sup>81</sup> has led to the theory that variations in the immune system could underly dementia sex differences. Indeed, microglia activation is found to mediate the relationship of Aβ with tau deposition only in females,<sup>82</sup> demonstrating sex differences in immune-related mechanisms for the development of dementia pathologies.

Immune responses have also been linked with sex hormones, with stronger responses in females observed only after puberty, when both males and females experience a surge in sex hormone levels.<sup>81</sup> Given that variations in sex hormone levels are among the biological features distinguishing males from females (males have higher androgen levels, females have higher oestrogen and progestogen levels), much literature (discussed further in Section 1.11.) has focused on sex hormone contributions to dementia sex differences. Briefly, sex hormone receptors are found throughout the body, including the brain, and oestrogen in particular is known to have protective effects in the brain and the cardiovascular system.<sup>83,84</sup> Additionally, females have more marked hormonal changes than males throughout the life course including the menstrual cycle, pregnancy, and menopause, demonstrating female-specific processes which could contribute to female brain health and dementia risk. The menopause transition, as will be discussed in Section 1.11., has received much attention; falling levels of neuroprotective oestrogen during menopause are proposed to increase the vulnerability of the ageing female brain to the development of dementia pathologies.<sup>8</sup> Women going through the menopause transition can also experience a range of symptoms including 'brain fog', hot flushes, sleep disturbance, and low mood. These can have implications for women's work lives, social relationships and lifestyle behaviours including engagement with physical activity, which may further contribute to female dementia risk.<sup>85</sup>

Finally, while much sex differences literature focuses on why dementia risk is increased in females, another approach is to consider why males have reduced risk, drawing on the concept of resilience (Section 1.6.). Cognitive reserve can be built via various lifetime experiences including education, occupational complexity, and SEP, which are traditionally higher in men, meaning that men generally have greater opportunities to build cognitive

33

reserve than women.<sup>86</sup> It has also been shown that, for each additional unit of AD pathology (measured *post-mortem* in 141 members of the Religious Orders Study), the odds of clinical AD increased 3-fold in males compared with 20-fold in females, demonstrating greater male resilience to AD pathology.<sup>87</sup> Additionally, in the Alzheimer's Disease Neuroimaging Initiative (ADNI) cohort, increases in CSF AD biomarkers (Aβ-42 and total tau) associated with longitudinal hippocampal atrophy and decline in executive function performance (average 2.5 year follow-up) to a greater extent in females than males, and this was exacerbated among females with low education levels.<sup>88</sup> While some evidence indicates greater cognitive resilience to AD pathology in males, further sex stratified research is needed to understand how males develop greater resilience and whether interventions to improve female opportunities to build cognitive and brain reserve could reduce overall female dementia risk.<sup>86</sup>

#### 1.9.1. Summary

Various theories for dementia sex differences have been proposed, which are unlikely to be mutually exclusive. For example, immune responses are linked with sex chromosomes and hormonal changes.<sup>81</sup> Theories highlight biological distinctions between males and females, indicating that some sex-specific processes (e.g. menopause in females) could contribute to dementia sex differences. However, there is also a role for sociocultural, health, and lifestyle factors which both sexes can be exposed to, with gendered variations in such exposures possibly contributing to sex differences in resilience mechanisms.

#### 1.10. Dementia risk factors

Several factors have been identified which can increase individuals' risk of developing dementia. Some are non-modifiable, while others have the potential to be modified.

#### 1.10.1. Individual characteristics

Genetics can influence dementia risk; *APOE* is strongly associated with dementia, particularly AD.<sup>89-91</sup> *APOE* codes for apolipoprotein E (ApoE), a lipid transporter protein involved in maintenance of the central nervous system (CNS) and A $\beta$  clearance, and has three alleles –

ε2, ε3, and ε4.<sup>92</sup> One copy of the *APOE* ε4 allele (ε4 heterozygous) increases AD risk 2- to 3fold, while two copies (ε4 homozygous) increases risk 12-fold, compared with ε3homozygotes.<sup>89-91</sup> While genetic risk is traditionally non-modifiable, gene therapy techniques are a potential route through which the AD risk associated with *APOE*-ε4 could be modified. An ongoing clinical trial is investigating whether a viral vector carrying the ε2 allele, which is protective against AD pathology, can safely increase expression of the ApoE ε2 protein in ε4homozygotes with Aβ pathology and MCI or moderate dementia diagnoses.<sup>93</sup>

Other individual characteristics linked with dementia risk include age and sex. Age is the greatest dementia risk factor (risk doubles every five years after age 69),<sup>94</sup> and female sex associates with increased risk, given greater dementia rates in females than males.<sup>2</sup> Further demonstrating exacerbated dementia risk in females,  $\varepsilon$ 4 heterozygous females are also found to have a higher AD risk than heterozygous males between ages 65 and 75 years,<sup>95</sup> highlighting that there are interrelationships between different risk factors.

#### 1.10.2. Health and lifestyle risks

The 2020 Lancet commission on dementia prevention, intervention and care<sup>56</sup> reviewed existing literature and identified twelve modifiable factors across the life course which were most convincingly associated with dementia risk. The authors estimated that if all twelve risks were removed, up to 40% of worldwide dementia cases could be delayed or prevented.

Low education was identified as an early-life (<45 years) risk. Midlife (age 45-65 years) risks are: hearing impairment, traumatic brain injury (TBI), hypertension, high alcohol consumption, and obesity. Later-life (>65 years) risks are: smoking, depression, social isolation, physical inactivity, diabetes, and exposure to high levels of air pollution.<sup>56</sup> Some risk factors, such as low education, hearing impairment, and social isolation, represent how reduced opportunities for cognitive stimulation could limit the development and maintenance of cognitive reserve which is protective against dementia.<sup>45,96,97</sup> Others reflect the impact of cardiovascular health on brain health. For instance, persistent high blood pressure in midlife associates with reduced brain volumes and greater WMHV,<sup>98</sup> and low physical activity associates with increased dementia risk particularly among individuals with existing cardiovascular comorbidities.<sup>99</sup> Some risks reflect trauma and insults to the brain

which could initiate neuroinflammatory processes. For example, TBI is associated with neuroinflammation and tau deposition, and smoking and air pollution could lead to toxic particles impacting the brain, also contributing to neuroinflammation.<sup>56,100</sup>

While the Lancet risks are based on an extensive review of available literature, it is worth noting that the twelve risks identified are not exhaustive. It is probable that other health and lifestyle factors contribute to dementia risk, but further research is required to provide convincing evidence for an association and clarity on how interventions could reduce risk. For example, sleep disturbance has been associated with various indicators of dementia-related pathology including neuroinflammation and tau and A $\beta$  deposition.<sup>56,101</sup> Both short (<5 hours) and long (>10 hours) sleep duration associate with dementia risk, but the mechanisms for how sleep disturbance contributes to dementia processes are currently unclear and evidence for the effects of sleep disturbance medications (e.g. hypnotics, benzodiazepines) is mixed.<sup>56,102</sup>

There is evidence that reducing modifiable risk factors can have cognitive benefits. The Finnish Geriatric Intervention Study to Prevent Cognitive Impairment and Disability (FINGER) is a randomised controlled trial (RCT) of a 2-year multidomain lifestyle intervention comparing regular health advice (control group) with a combination of dietary, exercise, cognitive training and vascular risk monitoring interventions (intervention group), among older-aged (60-77 years at baseline) individuals at increased risk of dementia.<sup>103</sup> Over two years from baseline, the intervention group showed a 25% greater improvement in neuropsychological test battery performance than the control group,<sup>104,105</sup> showing that multidomain lifestyle interventions can be beneficial in maintaining or improving cognitive performance in old age, which could protect against cognitive decline.

Further recognising interrelationships between dementia risk factors, dementia risk scores have been derived to quantify the impact of multiple risk factors on overall dementia risk and to predict individuals' level of dementia risk. Some scores, such as the lifestyle for brain health index (LIBRA),<sup>106</sup> include only modifiable risk factors. Others, such as the cardiovascular risk factors, ageing, and dementia (CAIDE) score,<sup>107</sup> include modifiable and non-modifiable risks. While increased dementia risk scores have been associated with poorer cognitive performance and with poorer neuroimaging measures of brain health, including greater WMHV and smaller brain volumes,<sup>108</sup> existing risk scores are not found to accurately predict
future dementia.<sup>109</sup> Several predictive dementia risk scores have been created, but to develop effective scores with useful research and clinical applications, there needs to be consideration for whether the risks included can be measured practically and, ideally, incorporate modifiable risks such that interventions might reduce dementia risk.<sup>110</sup>

# 1.10.3. Sex differences in dementia risk factors

A recent study quantified midlife exposures to eight of the twelve Lancet risks in an online survey sample, finding that males had more risk exposures but that the adverse effect of increasing risks on a composite measure of cognitive performance was greater in females than males.<sup>111</sup> Additional studies assessing whether the impact of multiple lifetime risk factor exposures differs between the sexes are, however, currently lacking.

There are established sex differences in the prevalence of some individual risk factors. For example, depression and obesity are more common in women,<sup>112-116</sup> while smoking, high alcohol consumption and hypertension are more prevalent in men.<sup>117-121</sup> However, such differences are not always consistent across the life course. For example, males have greater cardiovascular risks and develop cardiovascular disease (CVD) at younger ages than females, who show increases in CVD at midlife and post-menopause such that the incidence of CVD in later-life is similar between males and females.<sup>122,123</sup>

Aside from sex differences in prevalence, it is also useful to understand whether the dementia risk associated with each factor is equal in males and females. Some studies suggest that hypertension-associated dementia risk differs by sex and age at hypertension onset. In one study (n=5,646, 45.2% female; mean follow-up=15.3 years), mid-adulthood (mean age=44 years) hypertension associates with increased female all-cause dementia risk, while early adulthood (mean age=33 years) hypertension does not associate with risk in either sex.<sup>124</sup> However, others report that mid-adulthood (<65 years) hypertension associates with increased male AD risk, while late adulthood ( $\geq$ 65 years) hypertension does not associate with AD risk in either sex (n=848,505, 42.2% female; mean baseline age women=53.6 years, men=51.9 years; follow-up=14 years).<sup>125</sup> Variations in the type of dementia examined and sample ages might explain discrepancies between studies, but also highlight how sex differences in risks are not necessarily consistent across the life course and highlight the value of considering nuances in risks associated with different dementia types.

Indeed, sex differences in diabetes associations with dementia-related outcomes differ across studies examining different dementia types and across cognitively impaired and cognitively intact cohorts. In a pooled analysis of over 2 million individuals, diabetes-associated vascular dementia risk was 19% higher in women than in men, but non-vascular dementia diabetes-related risk did not differ by sex.<sup>126</sup> Within the ADNI cohort (n=911, 46% female; average baseline age women=72.2 years, men=73.9 years; average follow-up=4.1 years), no sex differences in prediabetes-associated dementia risk (not specified by dementia sub-type) were found, while in MCI patients only women showed a prediabetes-associated decline in glucose metabolism, and executive function and language performance.<sup>127</sup> Conversely, in a cognitively intact cohort, no significant associations of diabetes status with global or domain-specific cognitive performance were found in either sex.<sup>128</sup> In Insight 46, a mostly cognitively unimpaired cohort, hyperglycaemia (a marker of diabetes) associated dementia and cognitive outcomes are worse for women than men, but the associated risks may vary according to type of cognitive impairment and whether baseline cognition is intact.

Some studies have examined sex differences in risk profiles. In the Italian Longitudinal Study of Ageing (n=2,501, 56.3% female; mean baseline age=71.3 years), sex-stratified analyses indicated that baseline older age, lower education, heart failure, Parkinsonism, mild depressive symptoms, and family history of dementia associated with increased male dementia incidence at follow-up (median 7.8 years). Female dementia incidence associated with older age, lower education, current smoking, lower BMI, and mild and severe depressive symptoms.<sup>130</sup> In a French population-based cohort (n=6,892; mean baseline age women=74.3 years, men=74.0 years), male risks for conversion from MCI to dementia over 4 years of follow-up were *APOE*-ɛ4, stroke, low education, age and loss of instrumental activities of daily living (IADL), although IADL loss is a requirement for dementia ascertainment. Female risks were similar, except stroke was not a risk while subclinical depression and anticholinergic drug use were.<sup>131</sup> In the UK Biobank (n=502,226, 54.4% female; mean baseline age women=56.3 years, men=56.7 years; median follow-up=11.8 years), being overweight or obese associated with higher dementia risk in women than in men, as did increased systolic

blood pressure.<sup>132</sup> These sex-stratified analyses indicate that dementia risk profiles may differ between sexes, although which factors are included in male and female risk profiles remains unclear. Additionally, given that these studies only assessed risk factors present in mid- or later-adulthood, the relevance of risk factor timing in the life course remains unclear and it is difficult to determine whether the factors identified indicate dementia risks or prodromal features of dementia.

#### 1.10.4. Summary

Several dementia risks are identified, but there is currently insufficient evidence to fully understand sex-by-risk interactions on later-life cognitive, dementia and brain health outcomes. The 2020 Lancet commission outlined twelve modifiable risk factors spanning early-, mid-, and later-life, but most existing research only considers mid- or late-adulthood risks, omitting earlier risks, and it is unclear whether sex interacts with the age at which dementia risks are present. The prevalence of some individual risk factors differs by sex, indicating how sociocultural norms around health and lifestyle behaviours (e.g. smoking, alcohol consumption, physical activity) could contribute to dementia sex differences. Risks also interact with one another, and while several dementia risk scores have been created to quantify multiple risk exposures, there is debate over which risk factors should be included, with no consensus on how risk profiles might differ by sex, and existing scores do not perform well in predicting dementia.

# 1.11. The role of menopause in female cognitive and brain ageing

While variations in the prevalence and influences of lifetime risk factors which both sexes can be exposed to can provide some insight into dementia sex differences, the role of sex-specific exposures are also of interest. Marked hormonal events in females (e.g. menarche, pregnancy, menopause) represent female-specific experiences. Menopause, which typically occurs in midlife<sup>133</sup> and can involve neurological symptoms,<sup>8</sup> has received much attention regarding female dementia risk.

#### 1.11.1. The menopause transition

Menopause marks the end of a woman's reproductive period and is reached after 12 consecutive months without menstrual bleeding.<sup>134</sup> It usually occurs between age 45-55 years,<sup>133</sup> but is considered premature before age 40 years, early before age 45, and late after age 55.<sup>135-137</sup>

The Stages of Reproductive Ageing Workshop (STRAW) criteria<sup>138</sup> outline five broad stages of reproductive ageing characterised by changes in menstrual cycle length and hormonal levels: late reproductive, early menopausal transition, late menopausal transition, early postmenopausal, and late postmenopausal. Perimenopause refers to the early and late menopausal transition; the time between the start of the transition through to menopause being reached which can last 2 to 8 years, with a mean duration of 5 years.<sup>139</sup> During perimenopause, oestrogen levels gradually decline and remain low post-menopause.<sup>134</sup> As the body adjusts to reduced hormone levels, women may experience symptoms including hot flushes, sleep disturbance, depressive affect, mood fluctuations and cognitive impairment.<sup>8,140</sup> Such symptoms can have adverse impacts on social relationships, performance at work, and the ability or motivation to engage in physical activities.<sup>85</sup> Hormone therapy (HT) can help to alleviate symptoms by replacing falling endogenous hormones with hormonal medications, although there is debate around the risks and benefits of HT, which will be discussed in Section 1.11.5.

While natural menopause involves a gradual decline in sex hormone levels over several years, menopause can be surgically induced resulting in a more acute fall in oestrogen levels. Reasons for surgery might include treatment or prevention of certain cancers, or to reduce symptoms of conditions such as endometriosis.<sup>141</sup> Surgeries which can induce menopause include hysterectomy (removal of the uterus) and oophorectomy (removal of one or both ovaries), which may be combined or performed independently. When both ovaries are conserved, oestrogen decline is more gradual than when one or both ovaries are removed.<sup>142</sup>

#### 1.11.2. Menopause and cognition

## 1.11.2.1. Cognitive symptoms during the menopause transition

Around two-thirds of women experience subjective cognitive impairments ('brain fog') during menopause,<sup>143</sup> including concentration and memory problems.<sup>144</sup> Objective measures of cognitive performance correlate with subjective reports, although women typically still perform within normative ranges throughout the menopause transition.<sup>143,145-147</sup>

Additional impacts of the menopause transition on performance at work and social relationships, coupled with a general lack of awareness for how menopause can impact women, could induce additional stressors during the transition,<sup>85</sup> further affecting cognitive function. Cognitive difficulties might also be secondary to other neurological menopausal symptoms (e.g. sleep disturbance),<sup>146,148</sup> which typically subside post-menopause.<sup>149</sup> Suggesting that cognitive difficulties during menopause are transient, perimenopausal women are shown to experience more cognitive difficulties than pre- or post-menopausal women,<sup>146</sup> and perimenopausal women are shown to have poorer ability to learn from previous experiences (with fewer practise effects on repeated verbal memory and processing speed tasks over 4 years) than pre- and post-menopausal women.<sup>145</sup> Conversely, another study with 14 years of follow-up reported worsening verbal memory throughout successive menopause stages (independently of age, race, BMI, education and baseline cognition), contradicting the argument that cognition 'recovers' post-menopause.<sup>147</sup> Unlike studies which do not find poorer cognition post-menopause, this study did not include women taking hormonal medication - a difference which might explain discrepant findings given some evidence for cognitive differences between hormonal contraceptive or HT users and nonusers.145

## 1.11.2.2. Long-term cognition post-menopause

Long-term cognitive outcomes post-menopause are generally found to be poorer in women who undergo surgical rather than natural menopause,<sup>150,151</sup> but indications for surgical menopause may confound such associations. For example, cardiovascular risk factors associated with later-life cognition<sup>152</sup> can also increase the likelihood of surgically induced menopause.<sup>153</sup> Surgical menopause, by definition, occurs at earlier ages than natural menopause, hence it can be difficult to parse the effects of surgery from menopause timing, and many studies examining the effects of menopause timing on long-term outcomes post-menopause exclude women who had a surgical menopause. A review of 13 studies (4 of which included surgical menopause) found that, overall, earlier age at menopause associated with poorer cognitive performance in later-life, although no cognitive domains were consistently reported across studies.<sup>154</sup>

Most research relies on retrospective self-reports of menopause timing, which may induce error; women with poorer cognition or cognitive impairment might have difficulty recalling their age at menopause. Kuh et al. (2018)<sup>155</sup> tested associations between prospectively reported age at menopause and cognitive performance within the 1946 British birth cohort. Age at natural or surgical menopause did not associate with processing speed at any age between 43-69 years, but later age at natural menopause associated with better verbal memory performance across all timepoints. Associations remained after accounting for prospectively measured covariables including BMI, education, occupation, and childhood cognition, although there was some attenuation with childhood cognition which is shown to associate with menopause age<sup>156,157</sup> and with later life cognition.<sup>158,159</sup> Childhood cognition, which may reflect upstream developmental processes and represent a proxy indicator of lifetime oestrogen exposure,<sup>156</sup> is therefore an important confounder of menopause-cognition associations, but most studies are unable to account for this.

# 1.11.3. Menopause and brain health

Brinton et al. (2015)<sup>8</sup> propose that perimenopause, while primarily a reproductive transition, is also a neurological transition, given the neural nature of some menopausal symptoms (i.e. sleep circadian rhythm disruption, thermoregulation difficulties). Disruption in the brain oestrogen system during menopause is hypothesised to impair neural metabolism, leaving the brain vulnerable to further neural decline. However, determining whether hormonal changes lead to brain changes or vice versa is difficult given the synergistic roles of the brain and reproductive organs in hormone production via the hypothalamus-pituitary-gonadal (HPG) axis.<sup>160</sup> A small cross-sectional study (n=40)<sup>161</sup> found that some regional grey matter volumes were smaller in post- than pre-menopausal women, with volumes positively

correlated with oestradiol levels in both groups, demonstrating potential hormonal mechanisms for volumetric differences.<sup>161</sup> However, most studies do not include hormone measurements.

There is evidence of cerebral, structural, metabolic, and functional changes across menopause stages. Over 3 years of follow-up, peri- and post-menopausal women showed greater increases in A<sup>β</sup> levels, alongside faster rates of memory performance decline, than pre-menopausal women and age-matched males.<sup>162</sup> Levels of glucose metabolism and some grey matter volumes are also found to be reduced in peri- and post-menopausal women compared with pre-menopausal women and age-matched males, although over a 2 year follow-up levels of decline stabilised post-menopause and GMV in the precuneus increased.<sup>163</sup> UK Biobank research (excluding women with hysterectomy or bilateral oophorectomy, HT use or stroke) also showed that TBV and hippocampal volumes were greater in post- than in pre-menopausal women,<sup>164</sup> supporting 'recovery' of or improvements in brain volume post-menopause. However, older postmenopausal UK Biobank women also showed faster TBV decline over time than older premenopausal women,<sup>164</sup> suggesting that menopause contributes to accelerated brain ageing. This could have implications for female health in older age, with accelerated reductions in structural brain reserve potentially increasing female vulnerability to brain pathologies and functional impairments. The timing of health and lifestyle dementia risk factor exposures (Section 1.10.2) in relation to menopause are therefore important. For example, the female brain could be more sensitive to the adverse impacts of hypertension during menopause than during other life stages. Indeed, evidence associating cardiovascular risks with increased markers of cerebrovascular disease in females was strongest for midlife cardiovascular risks.<sup>40</sup>

UK Biobank data also shows that later menopause age associated with smaller TBV and hippocampal volumes,<sup>164</sup> seemingly contradicting evidence suggesting that later menopause is beneficial for postmenopausal cognition and that early menopause might increase dementia risk.<sup>151,154</sup> However, these neuroimaging measures were not analysed alongside cognitive outcomes, hence conclusions on whether the reported associations are linked with cognition in a beneficial or detrimental manner cannot be inferred.

Overall, neuroimaging research suggests that the menopause transition associates with changes in GMV and neurodegenerative markers such as Aβ load and glucose metabolism.

Few studies have linked brain changes with hormonal measures, making conclusions about the potential mechanisms driving brain changes difficult. Additionally, neuroimaging studies do not consistently assess cognitive performance; we cannot infer whether menopauseassociated brain changes underlie menopause-associated cognitive changes. Further, most neuroimaging samples discussed have strict inclusion and exclusion criteria, generally limiting small samples to illness-free individuals (e.g. diabetes, CVD, neuropsychiatric conditions excluded), restricting the generalisability of results.

#### 1.11.4. Menopause and dementia risk

A large meta-analysis<sup>95</sup> found increased female AD risk in the years following menopause (<75 years) compared with men of similar age. This difference in risk disappeared at older ages (to age 85), suggesting that late midlife to early old age is a time of vulnerability (or sensitive period) for AD risk in women. However, research assessing dementia risk associated with menopause age and type does not draw clear conclusions.

A review and meta-analysis did not find overall support for an association of menopause age with dementia risk, although several studies showed significant associations indicating increased risk with earlier age at menopause.<sup>154</sup> Another meta-analysis did not detect an overall association of surgical menopause and dementia risk, but there was evidence for increased dementia risk in women who underwent surgical menopause at an early age ( $\leq$ 45 years).<sup>151</sup> The association of early surgical menopause with dementia risk could be confounded by other health conditions, although most studies do not ascertain the reason for surgical menopause, and by social disadvantage which is linked with earlier menopause age.<sup>165,166</sup> Heterogeneity across methodologies, with variations in the covariates accounted for and in inclusion and exclusion criteria, might also disguise relationships between menopause and dementia risk present in some groups but not others.

An additional challenge is the time lag between the menopause transition and the age of dementia onset. Post-menopause, other health and lifestyle factors may further influence dementia risk. For example, women may retire from work, take up new hobbies or interests, or take on additional caring responsibilities (e.g. for elderly parents). Additionally, menopause has been consistently associated with cardiovascular risk factors such as obesity, diabetes,

and hypertension.<sup>10,167,168</sup> In females, CVD is rare before age 50 but cases rapidly increase in later years.<sup>167</sup> As a major transition occurring at midlife, menopause might contribute to female midlife cardiovascular risk, particularly since falling oestrogen levels might induce metabolic changes (e.g. reduced glucose metabolism) and changes in body fat distribution.<sup>167</sup> Social disadvantage could, however, confound menopause-cardiovascular associations, being linked with earlier age at menopause<sup>165,166</sup> and increased risk of heart disease.<sup>169</sup>

# 1.11.5. Effects of hormone therapy

There are discrepancies between observational studies and RCTs examining the effects of HT on cognitive performance and dementia risk. Observational studies tend to find benefits of HT use,<sup>170</sup> while RCTs typically report null or adverse effects.<sup>171</sup>Observational HT studies could have a healthy user bias, since HT users typically have better health than non-users.<sup>172</sup> RCTs can overcome this issue through random allocation of HT or placebo, but RCT participants are likely to be healthier than the general population due to strict exclusion criteria.

Perhaps the most well-known HT RCT is the Women's Health Initiative (WHI) trial of conjugated equine oestrogen (CEE) plus medroxyprogesterone acetate (MPA) in healthy postmenopausal women (mean baseline age 63.3 years), which was stopped early after 5.2 years when risks (primarily invasive breast cancer) were found to outweigh the benefits.<sup>173</sup> A second arm of the trial assessing unopposed CEE in healthy postmenopausal women without a uterus (mean baseline age=63.6 years) was also stopped early, due to increased stroke risk.<sup>174</sup> WHI follow-up sub-studies have found evidence for faster cognitive decline,<sup>175,176</sup> increased dementia risk,<sup>177,178</sup> and frontal lobe and hippocampal atrophy, particularly among women with poorer cognition at baseline,<sup>179,180</sup> in HT compared with placebo groups. However, the association of HT with dementia risk reported in the WHIMS (WHI Memory Study) sub-study was only detected for the CEE plus MPA arm; unopposed CEE was not significantly associated with dementia risk.<sup>178</sup>

The WHI trials assessed HT initiated in women 10 years, on average, post-menopause, meaning that results are not generalisable to women initiating HT closer to menopause. Indeed, a sub-group analysis of WHI participants aged 50-55 at trial medication initiation (mean time since final menstrual period/FMP 6.5 years) did not find effects of HT on cognitive

performance 7.2 years after discontinuation of trial medication.<sup>181</sup> Additionally, no differences in cognitive outcomes between HT and placebo groups were found in another trial (the Kronos Early Estrogen Prevention Cognitive and Affective Study; KEEPS) which examined the effects of HT use over 4 years in healthy, early post-menopausal women without hysterectomy (mean baseline age 52.6, average time since FMP 1.4 years).<sup>182</sup> Observationally, evidence from Cache County (a population-based dementia study of elderly residents) demonstrated a proximal effect whereby HT use associated with reduced AD risk if taken within 5 years of menopause.<sup>171</sup> The effects of HT on cognitive and dementia outcomes therefore seem to differ by when treatment is initiated in relation to menopause timing, supporting the sensitive period hypothesis which postulates that HT is beneficial if taken within a window of opportunity close to menopause.<sup>183,184</sup> Complimentary to this, the healthy cell bias theory proposes that exogenous oestrogen can have beneficial effects if cells are well-functioning and healthy, but may be harmful if cell functioning has already started to decline<sup>185,186</sup> (e.g. due to neuroprotective oestrogen decline during menopause).<sup>187</sup>

Many different forms of HT are available, with different formulations, dosages, and administration methods, and women may vary in how long they take HT for.<sup>188</sup> Such variations in HT use, which are rarely recorded with sufficient detail in observational studies and which vary across RCTs, mean that there is currently an overall lack of consensus regarding HT effects on cognition, dementia risk and brain health. Indeed, a recent examination of available data in the UK Biobank revealed a nuanced picture of HT associations with MRI outcomes; longer HT duration associated with older predicted brain ages (brains appeared older than expected for chronological age), greater WMHV, and smaller hippocampal volumes, and HT users with hysterectomy or oophorectomy history showed younger predicted brain ages than those without such surgeries.<sup>189</sup>

## 1.11.6. Summary

Associations between female hormones and later life cognitive and brain health outcomes are complex and poorly understood. Biological plausibility for such associations is mainly focused on the role of oestrogen in the brain. Little attention is given to the effects of nonoestrogen hormones (e.g. progesterone, luteinising hormone, follicle-stimulating hormone) which also fluctuate throughout life. There is also a lack of prospective research utilising cognitive, neuroimaging and hormonal measures in combination, hence the mechanisms of associations with female hormonal events are unclear.

HT could modify associations between female hormonal events (i.e. menopause type and timing) and later-life outcomes in ways that we do not yet understand. Most studies do not record detailed information of HT use, meaning that analyses have generally not accounted for variations in dosage, formulation, duration, or timing of initiation. Those which have examined some of these variations demonstrate complex relationships of HT use with later-life outcomes.

Most research assessing the associations of menopause with later-life cognition and dementia risk is also unable to account for childhood cognition, which is shown to be an important confound, potentially representing a proxy measure of lifetime oestrogen exposure.<sup>156</sup> Beyond hormonal mechanisms though, the menopause transition and its associated symptoms can have health and lifestyle implications for women, which may contribute to the accumulation of dementia risk factor exposures (such as those outlined in Section 1.10.2.) during and post-menopause. Reducing symptoms, for example through HT medications, might limit subsequent accumulation of risk exposures. Given the time lag between menopause in midlife and later-life cognition or dementia, other health, lifestyle, and sociocultural factors during and after midlife could mediate or otherwise explain associations.

# 1.12. Summary and identification of gaps in the literature

Sex differences are reported in dementia prevalence, symptoms, and progression rates, with variations across dementia types.<sup>3</sup> Several theories of dementia sex differences have been proposed which consider the influence of biological factors specific to each sex and the role of sociocultural and lifestyle factors which can have gendered patterns and potentially differential impacts on cognitive ageing and dementia processes between males and females. However, the mechanisms underlying sex differences in cognitive ageing, brain ageing, and dementia pathology are not yet fully understood, due to an overall paucity of research explicitly testing sex differences or conducting sex-stratified analyses; only recently have policies been introduced to encourage the inclusion of sex as a biological variable in research.

There has been interest in studying sex differences in cognitive functions and brain structure, but historically studies have suffered from researcher bias, and have been used to justify or reinforce gender stereotypes. While understanding sex differences in cognition and the brain is useful for determining how males and females might differ regarding cognitive and brain ageing processes, such research must be treated with caution so as not to encourage potentially negative gender stereotypes. Additionally, there must be acknowledgement that the identification of sex differences does not necessarily represent a dichotomy between males and females; differences typically refer to statistically significant differences in mean values, while there may still be a degree of overlap in the distribution of values attributed to males and females.<sup>47</sup> Evidence also shows that cognitive sex differences can be modified by the nature of task instructions (e.g. whether tasks are presented within a stereotypically male or female context)<sup>50</sup> and by secular changes in access to education,<sup>54</sup> demonstrating that there are societal influences on cognitive sex differences and that such differences are not biologically pre-determined, highlighting that there are interrelationships between sex and gender. While sex differences in cognitive performance have been examined at various ages, there is no clear narrative for how patterns in cognitive performance sex differences across domains might change across the life course. Creating such a narrative will be useful to identify potential societal, lifestyle, or biological influences on cognitive sex differences, which could inform how brain ageing and dementia processes might vary between sexes.

Advances in technologies which can detect and measure *in vivo* biomarkers of dementia have led to the concept of preclinical dementia, whereby early dementia pathologies develop prior to the onset of noticeable cognitive and functional symptoms. Relatively new multimodal neuroimaging approaches, such as combining PET and MRI, facilitate a holistic approach to understanding the processes underlying dementia. Examining whether and how various *in vivo* indicators of brain health differ between the sexes will be valuable in elucidating the possible differential pathways to dementia in males and females. Indeed, neuroimaging studies have demonstrated poorer cerebrovascular health in females than in males, with females demonstrating poorer NAWM microstructural integrity and greater WMHV (indicative of greater cSVD) than males.<sup>62-64</sup> However, it is currently unclear how such sex differences in brain health relate to differences in cognitive functions, with few studies assessing sex differences in neuroimaging indicators of brain health alongside differences in cognitive performance. Considering cognitive and brain health sex differences in parallel is necessary for understanding how males and females might differ in the functional changes associated with brain ageing or dementia-related pathology, particularly in relation to cognitive resilience.

Several dementia risk factors have been identified, some of which may limit opportunities for the development or maintenance of reserve (e.g. low education, social isolation), while others can impact cerebrovascular health (e.g. hypertension, obesity).<sup>56</sup> Dementia risk scores have been created which acknowledge that there are interrelationships between risks, but they do not accurately predict future dementia<sup>109</sup> and do not incorporate all possible modifiable and non-modifiable risks for dementia. Notably, existing risk scores typically focus on risks present at one point in time, primarily during midlife, and do not quantify cumulative risk exposures throughout the life course. Additionally, few studies have explicitly examined sex differences in the associations of dementia risk scores with later-life cognitive, brain health or dementia outcomes. Existing risk scores also fail to incorporate gendered sociocultural variations in risk factors (e.g. alcohol consumption is greater in men, physical inactivity is greater in women), or sex-specific risks (e.g. female reproductive factors).

As a female-specific transition, menopause has received much attention as a potential contributor of female cognitive and brain ageing. It is hypothesised that reductions in oestrogen, which has neuroprotective properties, during menopause leaves the female brain vulnerable to the effects of neural ageing and acts as a 'tipping point' for the development of dementia-related pathologies.<sup>8</sup> Evidence has shown that, overall, earlier ages at menopause associate with poorer cognitive performance in later-life,<sup>154</sup> possibly reflecting prolonged effects of reduced oestrogen levels. However, few studies have examined associations of menopause age with neuroimaging indicators of brain health and dementia-related pathology, so the neural mechanisms through which menopause timing associates with cognitive performance remain unclear. There is also heterogeneity across studies assessing menopause timing and later-life outcomes, with variations in the cognitive domains examined, such that no cognitive domains have been consistently associated with menopause age across studies. Many studies also exclude women who underwent surgical menopause or who took HT, possibly reflecting the current lack of understanding for how HT associates with later-life brain health outcomes given variations in HT dosages, formulations, and durations of

use, for example, which are often poorly recorded in observational studies. Additionally, most research examining associations of menopause timing rely on retrospective reports of menopause age, which could induce error; women who experience cognitive impairments could be more likely to incorrectly recall their age at menopause. Further, evidence has shown that early-life factors are important contributors to menopause timing and could confound associations with later-life outcomes; better childhood cognition associates with later menopause age<sup>156,157</sup> and improved later-life cognition,<sup>158,159</sup> and attenuates associations of menopause age with later-life cognition.<sup>155</sup> However, most studies assessing menopause timing are unable to account for such early-life covariables due to unavailability of life course data.

# 2.0. Data and analytical strategy

# 2.1. Aims and objectives

The overall aim of this PhD is to examine sex differences in lifetime risks for dementia, within the context of a mostly cognitively unimpaired cohort (the MRC NSHD and its neuroscience sub-study, Insight 46). The purpose is to develop an understanding for how different socioeconomic, lifestyle, and biological factors throughout the life course might contribute to the increased risk of developing dementia observed in females over males.

Sex differences in dementia risk factors which both males and females can be exposed to are examined, alongside consideration of how female-specific exposures (namely, timing of the menopause transition) associate with cognition and brain health in later-life.

Cognitive and brain health outcomes are examined at an age (69-71 years) when pathological brain health is expected to be accumulating while diagnosable, symptomatic dementia is rare. Assessing both cognition and neuroimaging measures is useful to help understand the possible pathological pathways underlying cognitive differences.

Although cognitive performance measures are available across the life course, this thesis focuses on cross-sectional rather than longitudinal analyses. This is because true repeated measures, that is cognitive assessments comparable across time points, are the assessments of processing speed and verbal memory administered between ages 43 and 69 (see Section 2.3.2.). As described in Section 1.7., sex differences in these cognitive performance trajectories have already been examined by others.<sup>5</sup>

This thesis consists of three empirical sections, broadly asking:

- A. Are there sex differences in cognitive performance and brain health measures across the life course? (Chapter 3)
- B. Do associations of lifetime dementia risk factors with later-life cognition and brain health differ between males and females? (Chapter 4)
- C. To what extent does menopause timing contribute to later-life cognitive and brain health outcomes in females? (Chapter 5)

The objectives for each section are as follows, with greater detail and hypotheses outlined in the relevant chapters.

**Empirical section A (Chapter 3):** Generate a descriptive overview of cognitive and brain health sex differences and similarities across different life stages within the NSHD and Insight 46 cohorts. The purpose is to inform about which cognitive domains and brain health measures might show male or female advantages and to highlight when in the life course any sex differences are more evident. Sex differences are examined accounting for socioeconomic, educational, and lifestyle factors, to determine the extent to which sex differences can be explained by societal and environmental influences. Additionally, since the dementia risk associated with *APOE*- $\varepsilon$ 4 is greater in females,<sup>95</sup> whether sex differences are modified by *APOE*- $\varepsilon$ 4 status is examined to determine whether *APOE* genotype also associates with greater adverse cognitive performance and brain health measures in females throughout the life course.

**Empirical section B (Chapter 4):** A cumulative risks score (CRS) is derived from life course data, quantifying lifetime exposures to each of the twelve modifiable risk factors outlined in the Lancet commission (Section 1.10.2.). The objective is to examine the extent to which early-life socioeconomic and genetic factors predict lifetime exposures to modifiable dementia risks, and whether cumulative risk exposures associate with later-life cognitive performance and brain health differently in males and females. This is to inform whether the associations of exposures to non-sex-specific risk factors with later-life outcomes are greater in one sex over the other.

**Empirical section C (Chapter 5):** The objective is to examine how prospectively reported age at menopause, as a female-specific hormonal transition, associates with cognitive performance across a range of domains and with multi-modal neuroimaging markers of brain health at age ~70 years, and whether associations can be explained by relevant early cognitive, sociodemographic, reproductive, and health-related covariables. The aim is to build an understanding of menopause as a possible contributor to increased female dementia risk and variations in cognitive and brain ageing between women.

# 2.2. Introduction to the datasets

## 2.2.1. NSHD

The MRC National Survey of Health and Development (NSHD) began as a maternity survey to investigate the costs of childbirth and to assess maternity services. All 16,695 births in England, Wales, and Scotland during one week in March 1946 were included. A nationally representative cohort of 5,362 (2,547 females), stratified by occupational social class, was selected for follow-up from single births to married mothers; all babies whose fathers had agricultural or non-manual occupations were selected, and a random 1 in 4 sample of babies whose fathers had a manual occupation were included.<sup>190</sup> There have been over 25 data collections, using a combination of questionnaires and clinical assessments to generate a rich dataset including prospective socioeconomic, health, and cognitive variables spanning the life course. Questionnaires were completed by mothers, teachers, school nurses, and doctors prior to 1966, and completed by study members themselves thereafter. In childhood, follow-up data collections took place at ages 26, 36, 43, 53, 60-64, and 69.

For this thesis, data up to age 70 was used. This included a postal questionnaire at age 68, followed by a home visit at age 69. A total of 2,638 (1,318 female) completed at least one of these, representing 94% of the target sample (n=2,816). Participation rates did not differ by sex, but higher participation rates were associated with having a non-manual occupation, higher educational attainment, and higher cognitive ability in childhood.<sup>191</sup> Of the 2,546 study members who were not contacted for this data collection, 18% were known to have died, 12% had previously withdrawn, 11% had emigrated, and 7% had been untraceable for over 5 years.<sup>191</sup>

#### 2.2.1.1. Women's Health in the Middle Years

During midlife, 1,752 female NSHD study members took part in the Women's Health in the Middle Years survey.<sup>192</sup> Women completed 9 postal questionnaires between ages 43-54 years, providing detailed prospective data on women's reproductive health at midlife, relevant for Chapter 5 analyses. As will be detailed in Section 5.0., variables of interest included age at

menopause, whether menopause occurred naturally or if it was surgically induced, and whether women had used menopausal HT.

#### 2.2.2. Insight 46

Insight 46 is the NSHD neuroscience sub-study which aims to identify brain changes associated with healthy ageing and to detect brain changes which might predict who is at increased risk of dementia. The third wave of data collection is currently underway, but this thesis uses data from the first wave which assessed 502 (49% female) study members between ages 69 and 71. Wave I incorporated an enhanced cognitive test battery, with assessments sensitive to cognitive changes linked with dementia (Section 2.3.2.), and simultaneous MRI and PET neuroimaging which provided well established indicators of brain health including TBV, GMV, WMHV, and Aβ load through validated pipelines (Section 2.3.3.).<sup>193</sup>

To maximise the life course data available for analyses alongside Insight 46 data, NSHD study members were eligible to take part in Insight 46 if they had attended a clinic visit at age 60-64 and had available relevant data across childhood and adulthood. Relevant variables included assessments of cognitive function, mental health, educational attainment, cardiovascular function, and physical activity levels, amongst others, as further detailed in Lane et al. (2017).<sup>193</sup> Of the 1,322 eligible study members, 779 indicated a willingness to attend a clinic visit in London and were invited to take part. Once data collection was underway, eligibility criteria were relaxed to ensure that the target sample of 500 participants was reached; an additional 62 participants who did not have a previous measure of lung function, smoking, or physical exercise were invited. The final sample of 502 participants represented 60% of the 841 NSHD study members invited; 24% refused, 3% temporarily refused, 1% did not respond, 0.4% died, 3% cancelled a visit, and 8% were excluded due to PET-MRI contraindications including claustrophobia and metal implants.<sup>194</sup>

Higher educational attainment, non-manual socioeconomic position, higher cognitive abilities, not smoking, and higher self-rated health were associated with increased likelihood of Insight 46 participation, across all stages of recruitment (i.e. eligibility, willingness to attend a London clinic, attending clinic).<sup>194</sup> Insight 46 participants therefore have greater SEP,

educational attainment, cognitive function, and general health than the wider NSHD cohort. Insight 46 participation was not, however, predicted by sex or *APOE*-ε4 status.<sup>194</sup>

# 2.3. Measures

A range of health, lifestyle, genetic, and socioeconomic variables measured throughout the life course are included in the analyses of this thesis and are detailed in the relevant analytical chapters. Several variables, including cognitive and neuroimaging outcome measures, are common across the analyses and are outlined below.

### 2.3.1. Sex

As outlined in Section 1.2., sex refers to the biological characteristics which distinguish males from females. Participant sex within NSHD is defined as sex recorded at birth and is a binary measure: male or female. Throughout the analyses in this thesis which include both sexes, male is the reference category.

# 2.3.2. Cognitive performance

The intellectual and cognitive abilities of NSHD study members have been assessed at several timepoints across the life course. Tests of verbal and non-verbal abilities were administered at ages 8, 11, and 15. Functional cognitive abilities were assessed at ages 43, 53, 60-64, and 69. Table 1 summarises the tests administered at each timepoint, and further details can be found at https://closer.ac.uk/cross-study-data-guides/cognitive-measures-guide/nshd-cognition/

Insight 46 participants completed an additional comprehensive neuropsychological test battery during wave I of data collection (age ~70). A range of domains including memory, executive function, visuospatial function, and cognitive state were assessed in tests designed to detect subtle cognitive changes associated with dementia, outlined in Table 2.

In Chapter 3 sex differences in performance across all cognitive assessments are examined. In Chapters 4 and 5, later-life cognitive outcomes (those measured at age 69 and Insight 46 wave I) are used as cognitive outcome measures. **Table 1**. Summary of the abilities tests and functional cognitive assessments administered during NSHD data collection waves. Prior to age 43 years only verbal and nonverbal abilities were assessed; functional cognitive abilities have been assessed from age 43 onwards. Further details for each test can be found at https://closer.ac.uk/cross-study-data-guides/cognitive-measures-guide/nshd-cognition/

Age	Test name	Test description	Domains	Outcome	
(year)			assessed	measures	
8 (1954)	Picture intelligence	60 items across 3 tasks: a) Choose the image which is the odd-one out of a series b) Select which image, from a choice of 5, completes a series c) Select, from a choice of 5, an image which corresponds to a rule (e.g. foot is to shoe as head is to hat).	Non-verbal reasoning	One point per correct answer (0-60)	
	Sentence completion	Choose, from a selection of 5 words, the correct word to complete each sentence (35 sentences).	Verbal ability, reading comprehension	One point per correct sentence (0- 35)	
	Word reading	Participants read a list of 50 words aloud.	Verbal ability, pronunciation	One point per word correctly pronounced (0-50)	
	Vocabulary	Participants were asked the meaning of each of the 50 words presented in the word reading task.	Verbal ability, word comprehension	One point per correct response (0- 50)	
	Overall cognitive performance	Summary measure of performance across the four tests, devised by National Foundation for Educational Research (NFER), <sup>195</sup> administered at age 8.	Verbal and non- verbal abilities	Z-score average performance across all four tests	
11 (1957)	General ability test	Choose the correct word or shape/symbol to complete a series.	Verbal and non- verbal reasoning	One point per correct series (0-80)	
	Arithmetic test	20 mechanical sums and 30 problem questions assessing the ability to add, subtract, multiply, and divide.	Verbal (problem questions) and non-verbal (mechanical sums) abilities	One point for each correct solution (0-50)	
	Word reading	As at age 8			
	Vocabulary	As at age 8			
	Overall cognitive performance	Summary measure of performance across the four	Verbal and non- verbal abilities	Z-score average performance	

		tests, devised by NFER, <sup>195</sup>		across all four
		administered at age 11.		tests
15 (1961)	The Alice Heim Group Ability Test (AH4) <sup>196,197</sup>	A 130-item test of verbal and non-verbal abilities including: series completion, mental arithmetic, vocabulary, and reasoning.	Verbal and non- verbal ability	One point per correct answer (0- 130)
	Watts-Vernon Reading Test <sup>196,198</sup>	Choose, from a selection of 5 words, the correct word to complete each sentence (35 sentences).	Verbal ability, reading comprehension	One point per correct sentence (0- 35)
	Mathematics test <sup>196</sup>	A 47-item test assessing arithmetic, geometry, trigonometry, and algebra.	Mathematical abilities	One point per correct item (0-47)
	Overall cognitive performance	Summary measure of performance across the three tests administered at age 15.	Verbal and non- verbal abilities	Z-score average performance across all three tests
26 (1972)	Watts-Vernon Reading Test <sup>198</sup>	As at age 15, but with 10 additional items of increased difficulty.	Verbal ability, reading comprehension	One point per correct sentence (0- 45)
43 (1989)	Peg placement <sup>199</sup>	Time to move 10 pegs in a wooden peg board from one hole to an adjacent one. Five trials per hand.	Motor speed and praxis	Mean speed across 5 trials, per hand
	Picture recall <sup>200</sup>	Participants were given 30 seconds to look at 5 cards, each with a unique picture, and were asked ~20 minutes later to recall what was on the cards.	Visual, non- verbal memory	One point for each correctly recalled picture (0-5)
	Verbal learning/word list recall test <sup>201</sup>	15 words presented (one word every 2 seconds). Participants write down as many words recalled as possible. A different word list was given to each half of the cohort.	Verbal memory	One point for every correctly recalled word, over three trials (0-45)
	Timed letter search/letter cancellation test <sup>200</sup>	Cancellation task with 2 different target letters embedded among non-target letters in three blocks. Participants were asked to cross out as many target letters as possible, as quickly as possible, within a block over 1- minute. There were three 1- minute trials.	Processing speed	Number of letters scanned in each 1-minute trial
53	Verbal	As at age 43, with the addition	Verbal memory	One point for
(1999)	learning/word	of a delayed recall (90 seconds)		every correct
(/	list recall test <sup>201</sup>	condition. Participants were		word recalled

		presented a different word list		(0-45	
		to the one seen at age 43.		immediate	
				recall; 0-15	
				delayed recall)	
	Timed letter	As at age 43 but only one trial.	Processing speed	Number of	
	search/letter	with letters covering a full	0 1	letters	
	cancellation	nage		scanned	
	test <sup>200</sup>	puge.		Scannea	
	National Adult	Ability to read aloud and	Verbal ability	Number of	
	Reading Test	sorractly propounce E0	verbarability,	orrors	
		irregular words of increasing	reauing	inverted such	
	(NART)	difficultur		thet bigh or	
		difficulty.		that higher	
				scores	
				indicate	
				better	
				performance	
				(0-50)	
	Verbal fluency	Category fluency: name as	Verbal fluency	Total number	
	animal naming	many different animals as		of animals	
	test <sup>203</sup>	possible in one minute.		named	
	Prospective	Participants were informed	Prospective	3-point	
	memory	that, later in the interview, they	verbal memory	ordinal score:	
	,	would be given an envelope	,	3=both	
		and asked to write a name and		actions	
		address on it. On receipt of the		completed	
		envelope they were to		without	
		remember to turn it over soal		prompting	
		it and write their initials on it		prompting 2-One action	
		It, and write their mitials on it.			
		when handed the envelope,		completed	
		participants were asked to		without	
		write "John Brown, 42 West		prompting	
		Street, Bedford".		1=No actions	
				completed	
				without	
				prompting	
	Verbal				
	learning/word	As at age 53			
	list recall test <sup>204</sup>				
	Timed letter				
	search/letter	0	F2		
	cancellation	As at age 53			
~~ ~ ~	test <sup>205</sup>				
60-64	Reaction time <sup>206</sup>	Simple RT: press a key (using	Simple reaction	Mean reaction	
(2006-		one finger only) as quickly as	time	time of	
2010)		nossible when '0' or '8'	Choice reaction	correct trials	
		anneared on the screen 20	time	for both tasks	
		triale			
		Choice $PT$ : numbers (1) (2) (2)			
		(4') would appear on			
		and 4 would appear on			
		screen, participants were to			
		press the corresponding keys as	1		

		quickly as possible, using both hands. 40 trials.		
	Verbal learning/word list recall test <sup>207</sup>	As at age 53		
	Timed letter search/letter cancellation test <sup>207</sup>	As at age 53		
69 (2015)	Addenbrooke's Cognitive Examination- third edition (ACE-III) <sup>208,209</sup>	A validated screening test of overall cognitive state, comprising tests of: attention and orientation, language, memory, verbal fluency, visuospatial function. NSHD participants were administered the ACE-III by iPad (ACEMobile http://www.acemobile.org). A paper version was used when iPad testing was not possible.	Cognitive state (verbal and non- verbal ability) Attention & orientation Language Memory Verbal fluency Visuospatial function	One point per correct item. Total score (max 100) is a sum of scores across all sub- domains: attention & orientation (0- 18), language (0-26), memory (0- 26), verbal fluency (0-14), visuospatial function (0- 16)
	Finger tapping <sup>210</sup>	With palm down and fingers extended, participants were asked to tap a lever with their index finger as fast as possible for 10 seconds; both hands	Psychomotor speed	Number of taps per hand

**Table 2.** Summary of tests and outcomes derived from the Insight 46 wave I cognitive test battery, assessed at ages 69-71.

Task	Description	Domains	Outcome measures	
		assessed		
delayed recall, DSST, FNAME-12A)				
Mini-Mental State	Test of cognitive state	Overall	Total score out of 30	
Examination	including tests of	cognitive state		
(MMSE) <sup>158,193,211</sup>	attention, orientation in			
	space and time, memory,			
	language, executive			
	function, and visuospatial			
	function.			
Logical Memory	Administrator reads a	Episodic	Total score out of 25	
delayed	short story aloud;	memory		
recall <sup>158,193,212</sup>	participants recall what			
	they can after an			
	approximate 20-minute			
	delay.			
Digit-Symbol	Participants fill in a	Processing	Total number of symbols	
Substitution Test	worksheet as quickly and	speed	completed	
(DSST) <sup>158,193,213,214</sup>	accurately as possible	Associative		
	within 90 seconds, using a	learning		
	code table pairing	Attention		
	numbers with different	Psychomotor		
	symbols.	speed		
		Executive		
		function		
Face-name	Participants learn and	Associative,	I otal names recalled plus	
associative	recall twelve face-name	episodic	total occupations recalled	
memory	and face-occupation pairs.	memory		
(FINAIVIE- 17A)158,193,215				
12A) / /	Additional computer	rised cognitive tas		
Matrix	Participants are presented	Non-verbal	Total number of correct trials	
reasoning <sup>158,193,216</sup>	with grids of geometric	reasoning		
	shapes with a section			
	missing. They then select			
	the missing piece from a			
	choice of five.			
Choice reaction	Participants are presented	Choice	Intra-individual variability in	
time <sup>193,217</sup>	with words or arrows	Reaction	reaction times (IIVrt), correct	
	indicating the direction	Time;	trials only: (SD / mean RT)	
	they should respond	Executive	(Z-score inversed so that	
	(left/right) as quickly as	function	higher scores indicate better	
	possible, using a button		performance, i.e. less RT	
	box.		variability)	
Response	Participants are cued to	Executive	Proportional difference in RT	
inhibition <sup>193</sup>	respond, as guickly as	function:	between correct congruent	

Viewench	possible using a button box, to the direction indicated by words or arrows (left/right) presented simultaneously (congruently or incongruently).	response inhibition	and incongruent trials: ((mean RT incongruent trials – mean RT congruent trials) / mean RT congruent trials) (Z-score inversed so that higher scores indicate better performance, i.e. less slowing of RT for incongruent vs. congruent trials)
Visuomotor integration <sup>40,193</sup>	Participants use a stylus to trace a circle on a tablet screen as quickly and accurately as possible, under direct (hand and tablet visible) and indirect (hand and tablet hidden, circle viewed on vertical screen) visual feedback conditions. Administered as a dual-task, alongside serial subtraction.	Executive functions: visuomotor integration, error detection and correction Psychomotor speed Attention	Mean number of errors per rotation across dual-task trials (z-score inversed so higher scores indicate better performance, i.e. fewer errors) Mean number of rotations across dual-task trials Mean subtraction rate (numbers per second) across dual-task trials
"What was where?" visual working memory <sup>193,218</sup>	Either one or three fractal objects are presented in random locations on a touch screen. Participants choose which of two fractals they have previously seen and drag it to the location they think it was presented.	Visual memory Visual memory binding Attention Visuospatial function	Identification rate: proportion of fractals correctly identified across all trials Mean localisation error: distance from the correct target location, across all trials (z-score inversed so higher scores indicate better performance, i.e. smaller localisation error) Swap error rate: percentage of correctly identified fractal trials in which a distractor fractal location is chosen (z- score inversed so higher scores indicate better performance, i.e. lower error rate)

PACC=Preclinical Alzheimer's Cognitive Composite; MMSE=mini-mental state examination; DSST=digit-symbol substitution test; FNAME=face-name associative memory examination; RT=reaction time; IIVrt=intra-individual variability in reaction times; SD=standard deviation

#### 2.3.3. Brain health

Brain health measures were derived from simultaneous PET-MRI neuroimaging completed by Insight 46 wave I participants at age 69-71. Throughout this thesis, the multimodal neuroimaging measures examined include those which represent AD- (Aβ, hippocampal volume, cortical thickness), non-specific ageing- (TBV), and vascular-related (WMHV, NAWM metrics) pathways to dementia.

These neuroimaging variables were derived, using validated pipelines, by neuroimaging experts at the UCL Dementia Research Centre, who provided guidance in using these variables within this thesis. The methodologies for deriving the relevant measures for this thesis are described below.

#### 2.3.3.1. Imaging protocol

Of the 502 Insight 46 participants, 471 completed PET-MRI scanning.<sup>191</sup> The full imaging protocol has been previously described.<sup>193</sup> In brief, a single Biograph mMR 3T PET-MRI scanner (Siemens Healthcare, Erlangen) was used, with simultaneous acquisition of dynamic PET-MRI data, including volumetric (1·1 mm isotropic) T1-weighted and T2-weighted Fluid Attenuated Inversion Recovery (FLAIR) sequences. PET data were acquired continuously in list mode, during and following injection of 370 MBq <sup>18</sup>F florbetapir (Avid Radiopharmaceuticals, Philadelphia, PA).

#### 2.3.3.2. Volumetric measures

Volumetric T1-weighted and FLAIR images underwent visual quality control (QC) before processing using automated pipelines, previously summarised in Lane et al. (2017),<sup>193</sup> which generated, among other measures, total brain volume (TBV; cm<sup>3</sup>), total intracranial volume (TIV; cm<sup>3</sup>) and hippocampal volume (cm<sup>3</sup>). For these analyses, hippocampal volume was taken as the mean volume of the left and right hippocampi.

#### 2.3.3.3. White matter hyperintensity volume (WMHV)

The Bayesian Model Selection (BaMoS) algorithm,<sup>39</sup> followed by visual QC, was used to segment white matter hyperintensities from 3D T1 and FLAIR images to generate a global

WMHV which includes subcortical grey matter but excludes infratentorial regions.<sup>98</sup> For analyses assessing WMHV as an outcome, participants who failed BaMoS QC were excluded (n=3).

#### 2.3.3.4. Beta-amyloid

Amyloid (A $\beta$ ) burden was assessed over a 10-minute period, around 50 minutes after injection of florbetapir PET tracer. Global standardised uptake value ratio (SUVR) was calculated from cortical regions of interest, normalised to eroded subcortical white matter, providing a continuous measure of cerebral A $\beta$  levels. SUVR in Insight 46 is non-normally distributed (Appendix A Figure 1), with most individuals showing low levels of A $\beta$ . A binary indicator of amyloid load was therefore also examined as an outcome variable, to compliment continuous SUVR analyses; positive or negative A $\beta$  status was determined using a Gaussian mixture model applied to SUVRs, taking the 99th percentile of the lower Gaussian as the cut-off (0.6104).<sup>98</sup>

## 2.3.3.5. Normal appearing white matter (NAWM)

As described in James et al. (2023),<sup>40</sup> multi-shell diffusion MRI (dMRI) was used to derive diffusion maps indicating NAWM microstructural integrity. For the purposes of these analyses, mean diffusivity (MD) and fractional anisotropy (FA) maps were of interest. MD measures the level of diffusion in all directions, with greater MD representing more water dispersal and therefore reduced microstructural integrity.<sup>26</sup> FA represents the directionality of diffusion, with lower FA indicating more streamlined movement of water molecules along axons, showing better microstructural integrity.<sup>26</sup> All images underwent visual QC, with images failing QC excluded from analyses (n=59). Diffusion maps were converted to z-score maps, via comparison with a sub-set of participants with limited WMHV (<1ml), and a mean z-score over each NAWM mask was calculated for each diffusion metric (i.e. MD, FA).

#### 2.3.3.6. Cortical thickness

Cortical thickness estimation was performed using a modified version of the standard automated pipeline in FreeSurfer version 6.0.<sup>219</sup> Regions of interest (ROI) were the surface area-weighted averages of the frontal, temporal, parietal, and occipital lobes, and the Harvard composite signature of AD cortical thinning (ADsig Harvard).<sup>220</sup> The ADsig Harvard comprises

areas of the frontal, temporal and parietal regions: entorhinal cortex, parahippocampus, inferior parietal lobe, pars opercularis, pars orbitalis, pars triangularis, inferior temporal lobe, temporal pole, precuneus, supramarginal gyrus, superior parietal lobe, and superior frontal lobe.<sup>220</sup>

#### 2.3.4. Covariables

The specific covariables included in each analysis are detailed in the relevant chapters. Some covariables common across analyses are outlined below.

#### 2.3.4.1. APOE genotype

Blood samples taken at age 53 were analysed to ascertain *APOE* genotype from the single nucleotide polymorphisms (SNPs) rs429358 and rs7412. Genotype was categorised according to *APOE*-ε4 allele carrier status: ε4 carriers (ε4 homozygous or heterozygous) or non-carriers.

#### 2.3.4.2. Educational attainment

Educational attainment was recorded as the highest educational qualification achieved by age 26 and by age 43. Categories were derived from the Burnham Scale:<sup>221,222</sup> no qualifications, sub-GCE or sub-Burnham C (vocational only), GCE O level or Burnham C (GCSE) or equivalent, GCE A level or Burnham B or equivalent, and degree or higher. These categories were further collapsed for use in these analyses, as detailed in the relevant chapters.

#### 2.3.4.3. Socioeconomic position (SEP)

SEP was categorised according to the UK Registrar General social classifications:<sup>223</sup> unskilled, partly skilled, skilled manual, skilled non-manual, intermediate, professional. Binary coding grouped the first three categories as manual (lower SEP) and the remaining categories as non-manual (higher SEP). Childhood SEP was determined according to the father's occupational social class at ages 4, 11, or 15, and thereafter SEP was derived from participants' own occupational social class from age 15 to 53. Overall adulthood SEP was derived by using SEP at age 53, or 43, and updating it with previous SEP if missing.

#### 2.3.4.4. Childhood cognitive ability

Childhood cognitive ability, which is a predictor of later-life cognitive ability,<sup>159</sup> is a key covariable included in these analyses and represents a strength of the NSHD cohort, since other datasets typically lack prospective measures of cognitive ability in childhood. As shown in Table 1, childhood cognitive ability is derived for ages 8, 11, and 15 as an averaged, standardised z-score across all assessments completed at the respective age. In this thesis, cognitive ability at age 8, as the earliest indicator of cognitive ability, is taken to indicate childhood cognitive ability, with data from age 11 or 15 used if missing at age 8.

# 2.4. Analyses

Analyses were conducted in Stata (versions detailed in relevant chapters), and the details of each analytical sample and analytical method are outlined in each empirical section of this thesis (Chapters 3, 4, 5). Some methodology was shared between chapters, which is briefly outlined here.

#### 2.4.1. Descriptive analyses

Variables in each analytical dataset were checked for normality and outliers by visually inspecting histograms. Where outcome variables were non-normally distributed, bootstrapping with 1000 replications was applied to the analyses for that outcome, to calculate non-parametric, bias-corrected confidence intervals.

Continuous variables were summarised using mean and standard deviation statistics. Categorical variables were described using frequencies and percentages. Descriptive statistics were compared between males and females using t-tests and chi-squared tests for continuous and categorical variables, respectively, with p<0.05 taken to indicate a statistical sex difference.

## 2.4.2. Regression modelling

Multivariable regression modelling is the main statistical method used throughout this thesis to test associations of life course variables with cognitive performance and brain health outcomes. An inferential framework was adopted to identify relevant confounding, mediating, and competing exposure variables, with models adjusted for covariables accordingly. Linear regression was used for continuous outcome variables, and logistic regression for binary outcomes (e.g. A $\beta$  status). The estimated associations were described using regression coefficients or odds ratios, and 95% confidence intervals, using bias-corrected confidence intervals for bootstrapped analyses. Non-linearity was assessed using quadratic terms, and effect modifications were examined by including relevant interaction terms, deemed significant at the p<0.10 level.

To maximise the use of available data, maintain statistical power, and reduce bias due to missing data, multiple imputation by chained equations (MICE) was used across analyses to account for missing covariable data, under the assumption that data were missing at random. Each MICE analysis included 50 imputations, except for analyses combining bootstrapping with MICE in Chapter 5. Imputation models included variables which might predict missingness (auxiliary variables), and all outcome, covariate, and predictor variables relevant to each analytical model. Continuous variables were imputed using Gaussian normal regression, and categorical variables using logistic or ordinal regression techniques.

# 3.0. Empirical section A: Sex differences in lifetime cognitive performance and later-life neuroimaging outcomes in NSHD and Insight 46

The purpose of this chapter is to provide a descriptive narrative of sex differences in cognitive performance across the life course and of sex differences in a range of neuroimaging indicators of brain health measured in later-life, within the NSHD and Insight 46 cohorts. Multivariable regression modelling was used to compare outcome measures in females with measures in males, while considering the potential role of socioeconomic, education, and lifestyle factors, and whether sex differences were modified by *APOE*- $\varepsilon$ 4 status. Females tended to perform better than males on most cognitive assessments throughout life and into older age, despite showing poorer cerebrovascular brain health (specifically, increased markers of cSVD) aged ~70, demonstrating female cognitive resilience to cerebrovascular pathology. There was evidence that sociocultural factors contributed to cognitive sex differences, with higher education level being especially beneficial for cognitive advantages in females. Primarily at later ages, there were also larger adverse associations of *APOE*- $\varepsilon$ 4 with cognitive performance in females. Overall, these analyses suggest that females could be more susceptible than males to some sociocultural and genetic factors linked with dementia risk and that, in females, cognitive function persists despite subtle brain changes.

#### 3.1. Introduction

As discussed in Section 1.7., sex differences in cognition have been extensively examined and there are discrepancies in the findings, but the most consistent differences reported are male advantages in spatial abilities and female verbal memory advantages.<sup>47</sup> Evidence for sex differences in brain volumes is also mixed (Section 1.8.), but *in vivo* MRI demonstrates that males typically have larger brains throughout life, explaining the majority of variance in regional volumes.<sup>57,58</sup> There are reports of greater brain activation in females, indicated by greater cerebral blood flow,<sup>58</sup> who also reportedly show greater WMHV<sup>65,66</sup> and poorer NAWM<sup>62,63</sup> microstructural integrity than males, particularly at older ages. Conversely, sex differences in the AD hallmark, Aβ, are not found.<sup>61</sup>

Evidence demonstrates that cognitive sex differences are not robust against environmental influences, with task training or the gendered context of a task shown to remove sex differences in cognitive task performance, 50.224 indicating that differences are not necessarily biologically pre-determined. Indeed, lifestyle factors – including smoking and physical activity (PA), more common and frequent in males<sup>225,226</sup> – have been associated with cognitive and neuroimaging measures.<sup>227–229</sup> SEP and education, typically lower in females,<sup>230,231</sup> are also positively associated with cognitive performance and neuroimaging outcomes.<sup>232,233</sup> Sex differences in genetic risk for dementia might also contribute to cognitive and brain health sex differences. Despite an equal distribution of the *APOE*- $\epsilon$ 4 genetic risk allele for AD between the sexes, the dementia risk associated with  $\epsilon$ 4 is greatest in females<sup>90,95</sup> and, compared with male  $\epsilon$ 4 carriers, female carriers have greater levels of brain pathology, poorer cognitive performance, and faster cognitive decline over time.<sup>95,234</sup> Understanding whether and how males and females differ in cognitive abilities and measures of brain health across the life course can help to identify how socioeconomic, lifestyle, and biological factors present at each life stage potentially underlie sex differences in later-life dementia risk.

Some, but not all, cognitive measures assessed throughout the life course in NSHD and Insight 46 (Section 2.3.2.) have been examined for sex differences. For instance, in 1968, Douglas et al.<sup>235</sup> reported a consistent female advantage in verbal skills from age 8 to 15. No sex differences in non-verbal abilities at ages 8 and 11 were found, but at 15 males showed non-verbal and mathematics advantages.<sup>235</sup> In another study looking at predictors of cognitive trajectories from age 43 to 69, females consistently outperformed males on repeated

measures of verbal memory and search speed across timepoints. While no sex differences in rates of verbal memory performance decline over time were detected, females showed slower decline in search speed task performance.<sup>5</sup> Within Insight 46 at age ~70, studies examining predictors of cognitive performance across a range of tasks found that females performed better than males on the PACC and each of its four sub-tests,<sup>158</sup> although no sex differences in matrix reasoning<sup>158</sup> or reaction time task performance<sup>217</sup> were detected. Another study primarily aiming to assess the effects of *APOE*-ε4 and Aβ pathology on visual working memory within Insight 46 did detect a dissociation whereby females showed better object identification and poorer recall of object locations than males.<sup>218</sup>

Insight 46 females have also shown younger predicted brain ages than males, based on volumetric neuroimaging measures.<sup>236</sup> When excluding individuals with neurological conditions, women had poorer NAWM integrity than men, but this was largely explained by differences in WMHV,<sup>237</sup> suggesting that alterations in NAWM and white matter hyperintensity may be part of an overlapping pathological process. However, in Insight 46 participants without neurological conditions, there was limited evidence for sex differences when looking at total WMHV itself.<sup>158</sup> In the same analyses, no sex differences in Aβ load were detected.<sup>158</sup>

While these NSHD and Insight46 studies have reported sex differences in papers with a wider scope, not specifically aiming to examine sex differences, sex differences across cognitive and imaging metrics have not yet been collated into one space to provide a comprehensive narrative of sex differences across the life course. Additionally, no direct assessments of socioeconomic and lifestyle covariable roles or potential *APOE*- $\epsilon$ 4 effect modifications on sex differences have been completed.

# 3.1.1. Objectives and research questions

This work tests for sex differences in cognitive performance measures assessed throughout life in the NSHD and Insight 46 cohorts, and for sex differences in neuroimaging indicators of brain health at age ~70 within Insight 46. The aim is to inform which, if any, cognitive domains and brain health measures show male or female advantages, and when in the life course any cognitive sex differences are more evident. A descriptive overview of cognitive and brain

health sex differences across different life stages within the NSHD and Insight 46 cohorts is generated.

The following research questions are addressed:

- Are there sex differences in cognitive performance measured across the life course and in brain health outcomes measured aged ~70?
- 2. Are sex differences explained by socioeconomic, education, and lifestyle covariables?
- 3. Are sex differences modified by APOE-E4 status?

# 3.1.2. Hypotheses

Given previous reports outlined in Section 1.0., males and females were hypothesised to perform similarly across most cognitive domains, with subtle male advantages in spatial tasks<sup>49</sup> and female advantages in verbal ability assessments, namely verbal fluency.<sup>238</sup> Patterns of cognitive sex differences were expected to be similar across the life course, given that childhood cognition predicts later-life cognitive performance,<sup>159</sup> and previous NSHD evidence for consistent female processing speed and verbal memory advantages between ages 43 and 69.<sup>5</sup> Analyses were expected to replicate previous Insight 46 findings of no sex differences in Aβ or WMHV,<sup>158</sup> poorer NAWM integrity in females,<sup>237</sup> and larger brain volumes in females (given younger female predicted brain ages derived from volumetric MRI measures).<sup>236</sup>

Socioeconomic, education, and lifestyle factor adjustments were expected to influence estimated sex differences in cognitive performance and brain health measures, reflecting positive effects of education, SEP, and PA (all typically greater in males) and negative effects of smoking (greater in males) on outcomes.<sup>54,228,229</sup>

Sex-by-*APOE* interactions were expected to be evident in later-life, when subtle cognitive and brain health changes related to dementia could be beginning to emerge. Female *APOE*- $\epsilon$ 4 carriers were expected to show poorer outcomes than male carriers, particularly for outcome measures associated with AD: memory domains, performance on a cognitive composite designed to detect subtle pre-clinical cognitive changes associated with AD, and A $\beta$  pathology.

# 3.2. Analytic method

# 3.2.1. Analytic sample

At each timepoint, NSHD males and females were included in analyses if they had available data for at least one cognitive outcome at that timepoint. Insight 46 participants were included if they had available data for at least one cognitive assessment completed at wave I of data collection (age 69-71) and, for neuroimaging outcomes, if they underwent wave I PET-MRI neuroimaging and had available data for at least one neuroimaging outcome of interest. The maximal samples available at each timepoint are presented in Table 3.

Age	Max N	% Female
8	4,269	48.4
11	4,032	47.8
15	4,019	47.8
26	3,713	49.5
43	3,237	50.1
53	2,956	51.0
60-64	2,216	52.3
69	2,140	51.0
Insight 46 wave I cognition (~70)	502	49.0
Insight 46 wave I neuroimaging (~70)	470	48.7

1

1

 Table 3. Maximal samples at each timepoint.

# 3.2.2. Cognitive outcome measures

As outlined in Section 2.3.2., NSHD participants were administered tests of verbal and nonverbal abilities at ages 8, 11, and 15, while functional cognitive abilities were assessed at ages 43, 53, 60-64, and 69, summarised in Table 1. Insight 46 participants completed an additional comprehensive neuropsychological test battery aged 69-71 (Table 2), assessing a range of domains including memory, executive function, visuospatial function, and cognitive state.

All cognitive outcome measures were standardised to the analytical sample for each outcome, generating z-scores to facilitate comparison across different cognitive domains. Where necessary (i.e. for measures of error rates), z-scores were inverted such that higher z-scores represented better cognitive performance, across tasks.

#### 3.2.3. Neuroimaging outcome measures

As outlined in Section 2.3.3., Insight 46 wave I participants underwent combined PET-MRI neuroimaging, generating a range of brain health and dementia-related pathology measures at age ~70. For these analyses, sex differences in TBV, hippocampal volume, WMHV, continuous amyloid SUVR, A $\beta$  positivity status, NAWM microstructural integrity measures (FA and MD) and cortical thickness across ROI (frontal, occipital, parietal, temporal, the Harvard AD signature region)<sup>220</sup> are examined.

For analyses assessing WMHV as an outcome, participants who failed BaMoS QC were excluded (n=15; 53.3% female). Participants whose scans failed NAWM QC (n=59; 49.2% female) were excluded from NAWM analyses.

## 3.2.4. Covariables

Previous literature has demonstrated that socioeconomic, education, and lifestyle factors can differ between males and females. For example, men typically had greater levels of education,<sup>230</sup> higher occupational SEP,<sup>231</sup> greater levels of PA,<sup>225,239,240</sup> and higher levels of smoking<sup>226,241</sup> than women in this era. Evidence has also linked such factors with cognitive outcomes and with neuroimaging measures.<sup>227–229,232,233</sup> Therefore, in these analyses sociodemographic (education, SEP) and lifestyle (PA, smoking) covariables prospectively recorded throughout the life course were adjusted for, to examine the potential contribution of such factors in cognitive and neuroimaging sex differences.

Highest educational attainment by ages 26 and 43 (Section 2.3.4.2.) was categorised as: no qualifications, ordinary qualifications (O levels or equivalent), or advanced (A levels or higher).

SEP (Section 2.3.4.3.) was categorised as: unskilled, partly skilled, skilled manual, skilled nonmanual, intermediate, professional. Overall childhood SEP reflects the most recent social class recorded at age 15, or at ages 11 or 4 if missing at 15. Overall adulthood SEP was derived at age 53 and 43, taking the most recent social class recorded and updating with previous SEP if missing at 53 or 43.

Participation in leisure time PA was recorded at ages 43, 53, 60, and 69 using the EPIC Physical Activity Questionnaire-2,<sup>242</sup> which assessed the frequency of participation in sports, vigorous leisure activities or exercise in the previous month (age 43) or the previous 4 weeks (ages 53,
60-64, 69). PA was not assessed during childhood or at age 26. As in previous work in this cohort,<sup>227</sup> responses at each age were categorised as: not active (no PA/month), moderately active (1-4 times/month), or most active ( $\geq$ 5 times/month).

Smoking pack years – a measure determined by multiplying the number of cigarette packs smoked per day by the number of years smoked – was based on self-reported frequencies at ages 26, 43, 53, and 60-64. The measure was not available during childhood or at age 69.

APOE genotype (Section 2.3.4.1.) indicated ε4 carrier or non-carrier status.

#### 3.2.5. Statistical analyses

Analyses were conducted using Stata version 17.0.

Continuous raw cognitive performance scores, neuroimaging outcomes, and covariables were summarised by calculating the range, mean, and standard deviation (SD) for each variable, for males and females separately. For categorical variables, the percentage of participants in each category was calculated for males and females. Independent samples t-tests and chi-squared tests examined sex differences in continuous and categorical covariables, respectively.

For all cognitive outcome measures, binary sex (male/female) was included as the predictor variable in linear multivariable regression models, with male as the reference category. Initial models (M0) were completely unadjusted for NSHD cognitive assessments from age 8 through to 69, while models assessing Insight 46 (age 69-71) cognitive outcomes were adjusted for age at cognitive testing due to the two-year range required to complete Insight 46 data collection.

For neuroimaging outcomes, linear multivariable regression models with sex as the predictor examined sex differences in TBV, hippocampal volume, continuous SUVR, NAWM measures of microstructural integrity (MD, FA), and all cortical thickness ROI. Multivariable logistic regression was used to examine the odds ratio for being Aβ positive in females compared with males. Sex differences in WMHV, which has a skewed distribution, were examined using a generalised linear model with gamma distribution log link. All models were adjusted for age at scan and, for volumetric outcomes only (TBV, hippocampal volume, WMHV), for TIV.

Associations between sex and outcome measures were firstly examined in minimally adjusted models (M0). Next, models were individually adjusted for SEP (M0+SEP), education

(M0+education), smoking (M0+smoking), and PA (M0+PA) covariables available at each timepoint, as outlined in Table 4. Adjustments were recorded either concurrent to the outcome or at the most recent timepoint prior to the outcome measure, and SEP models for outcomes measured at age 43 onwards were also adjusted for an overall measure of childhood SEP.

Finally, for participants with available APOE- $\varepsilon$ 4 data, effect modification by APOE- $\varepsilon$ 4 status was examined by including a sex-by-APOE interaction term in minimally adjusted models. If the interaction was significant at the p<0.1 level, subsequent analyses were stratified by APOE- $\varepsilon$ 4 carrier status.

To account for missing SEP, education, smoking, and PA covariable data, multiple imputation was applied to the analytical sample at each timepoint (Section 2.4.2.) The imputation models included sex and the outcome measures relevant to the sample at each timepoint, in addition to other covariables included in the analytical models. Auxiliary variables included BMI and smoking status measured closest to the outcome timepoint, and the Townsend deprivation index measured in 1999.

Where cognitive outcome measures were not normally distributed, bootstrapping was applied (Section 2.4.1.) to the analyses of minimally adjusted models, including *APOE* interaction models. When adjusting for covariables, changes in effect sizes were of interest so to avoid the need to combine bootstrapping with multiple imputation, which is computationally intensive, bootstrapping was not applied to adjusted models.

Age (year)	M0	M0+SEP	M0+Education	M0+PA	M0+Smoking
8 (1954)	Unadjusted	Paternal SEP, age 4	-	-	-
11 (1957)	Unadjusted	Paternal SEP, age 11	-	-	-
15 (1961)	Unadjusted	Paternal SEP, age 15	-	-	-
26 (1972)	Unadjusted	Paternal SEP, up to age 15	Education, age 26	-	Smoking, age 26
43 (1989)	Unadjusted	Paternal SEP, up to age 15 + SEP 15-43	Education, age 43	PA, age 43	Smoking, age 43
53 (1999)	Unadjusted	Paternal SEP, up to age 15 + SEP 15-53	Education, age 43	PA, age 53	Smoking, age 53
60-64 (2006-10)	Unadjusted	Paternal SEP, up to age 15 + SEP 15-53	Education, age 43	PA, age 60-64	Smoking, age 60-64
69 (2015)	Unadjusted	Paternal SEP, up to age 15 + SEP 15-53	Education, age 43	PA, age 69	Smoking, age 60-64
69-71 (2015-18) Cognition	Age at cognitive testing	Paternal SEP, up to age 15 + SEP 15-53	Education, age 43	PA, age 69	Smoking, age 60-64
69-71 (2015-18) Neuroimaging	Age at scan + TIV [TBV, hippocampal volume, WMHV only]	Paternal SEP, up to age 15 + SEP 15-53	Education, age 43	PA, age 69	Smoking, age 60-64

## Table 4. Summary of model adjustments at each timepoint.

SEP=socioeconomic position; PA=physical activity; TIV=total intracranial volume; TBV=total brain volume; WMHV=white matter hyperintensity volume

# 3.3. Results

# 3.3.1. Participant characteristics

Mean performance scores on each cognitive assessment across timepoints are summarised for males and females in Table 5 and mean neuroimaging measures from Insight 46 wave I (aged ~70 years) are summarised in Table 6.

As shown in Table 7, which presents summary data for the covariables measured at each timepoint, females had overall higher occupational classifications in adulthood. However, the highest classifications (professional and intermediate) had greater proportions of males than

females. Educational attainment was significantly greater in males than females, and significantly more males than females obtained further qualifications in adulthood between ages 26 and 43 (n=228 males vs. 163 females;  $\chi^2$ =21.43; p<0.01). Females had significantly greater levels of physical inactivity than males at age 43, but not at subsequent ages. At younger ages, males had greater levels of smoking than females, but this diminished over time and was no longer statistically significant at age 60-64, possibly explained by an overall decline in the number of smokers (41.54% of respondents were current smokers at age 26, compared with 12.13% at age 60-64). There was a trend (p<0.1) for a greater proportion of *APOE*- $\varepsilon$ 4 carriers in males than females. For Insight 46 participants, there was no difference in age at visit or age at scan between males and females, but males had significantly greater TIV than females.

**Table 5.** Mean, standard deviation, and range of raw performance scores for each cognitive assessment completed by NSHD and Insight 46 males and females at several ages throughout the life course.

		Males Females									
Age	Cognitive assessment	Ν	Mean	SD	Min	Max	N	Mean	SD	Min	Max
8yrs	Picture intelligence	2,203	40.17	9.50	0	60	2061	40.21	9.47	0	59
	Sentence completion	2,196	13.73	8.02	0	34	2,062	14.71	7.48	0	34
	Word reading	2,196	16.37	10.56	0	49	2,062	17.71	9.91	0	45
	Vocabulary	2,196	16.52	6.04	0	36	2,062	16.01	5.93	0	40
	Age 8 overall cognition (z-score)	2,197	-0.02	0.87	-2.61	2.68	2059	0.02	0.85	-2.61	2.52
11yrs	Word reading	2,101	35.96	11.08	0	50	1924	36.94	9.82	0	50
	Vocabulary	2,101	30.30	7.48	0	47	1924	29.64	7.41	0	49
	Arithmetic	2,102	26.15	12.05	0	50	1921	26.66	11.39	0	50
	General ability	2,106	43.76	15.94	0	80	1924	46.38	15.72	0	78
	Age 11 overall cognition (z-score)	2,101	-0.03	0.90	-3.01	1.74	1921	0.03	0.86	-2.91	1.96
15yrs	Watts-Vernon	2,097	24.67	6.69	0	35	1915	24.25	6.57	0	35
	General ability	2,097	74.14	20.52	0	125	1918	73.97	19.62	5	125
	Mathematics	2,098	15.83	10.82	0	46	1915	12.55	9.20	0	42
	Age 15 overall cognition (z-score)	2,094	0.05	0.93	-2.93	2.17	1912	-0.08	0.85	-2.78	2.07
26yrs	Watts-Vernon	1,874	33.11	8.21	0	45	1839	32.48	8.31	0	45
43yrs	Word learning	1,528	23.89	6.25	0	42	1531	25.55	6.43	6	41
	Search speed	1,577	332.90	77.16	127	450	1573	354.71	74.70	25	450
	Peg speed	1,575	104.12	16.80	77	516	1588	104.87	14.35	78	362
	Picture memory	1,608	4.34	0.82	0	5	1616	4.47	0.75	0	5
53yrs	Word learning	1,408	22.97	6.22	4	40	1478	24.84	6.25	3	41
	Search speed	1,438	272.67	75.42	91	591	1493	289.24	75.84	64	591
	NART	1,369	34.41	9.66	2	50	1455	34.21	9.42	1	50
	Animal naming	1,442	23.74	6.73	1	52	1506	23.40	7.09	1	62
60-64vrs	Word learning	1,023	23.04	5.92	6	41	1127	25.36	6.08	4	43
	Search speed	1,039	261.14	71.39	98	591	1143	271.78	71.72	112	591

	MRT	1,033	287.53	71.03	41	836	1134	284.84	66.07	185	849
	llVrt	1,033	0.23	0.12	0	2	1134	0.22	0.11	0	1
69yrs	Word learning	1,011	21.14	5.98	0	39	1063	23.15	5.99	6	40
	Search speed	1,038	256.37	74.32	70	591	1073	268.02	73.56	60	591
	Finger tapping	1,005	49.92	10.91	15	87	1041	44.94	10.20	11	74
	ACE-III total	849	91.33	5.84	53	100	913	91.69	6.17	59	100
	ACE-III verbal fluency	1,028	10.86	2.17	1	14	1073	11.14	2.05	1	14
	ACE-III language	849	25.28	1.15	19	26	916	25.26	1.18	16	26
	ACE-III attention	865	16.86	1.71	6	18	921	16.59	1.97	5	18
	ACE-III memory	865	23.17	2.98	7	26	922	23.73	2.71	12	26
	ACE-III visuospatial	861	15.13	1.19	9	16	917	14.98	1.37	4	16
69-71yrs	PACC total z-score	256	-0.17	0.71	-3.49	1.72	246	0.18	0.71	-3.48	1.67
(Insight 46)	MMSE z-score	256	-0.07	1.03	-7.17	0.74	246	0.08	0.97	-6.18	0.74
	DSST z-score	255	-0.15	1.00	-2.76	3.03	246	0.16	0.98	-2.28	3.31
	Logical memory delayed z-score	256	-0.22	1.00	-3.10	2.57	246	0.23	0.95	-3.10	3.11
	FNAME total z-score	256	-0.23	0.99	-2.91	1.51	244	0.25	0.96	-3.40	1.62
	Matrix reasoning z-score	256	-0.02	1.04	-4.07	1.42	246	0.02	0.96	-3.87	1.62
	llVrt	256	0.12	0.03	0.04	0.23	245	0.13	0.03	0.06	0.22
	Response inhibition	256	0.13	0.11	-0.15	0.82	244	0.14	0.12	-0.09	0.88
	Visuomotor rotations	249	0.99	0.80	0.00	6.16	234	1.00	1.02	0.00	12.04
	Visuomotor errors	249	5.45	3.60	0.63	29.16	234	5.18	4.11	0.60	35.84
	Visuomotor subtraction rate	249	0.55	0.17	0.19	1.01	234	0.45	0.15	0.12	1.08
	Visual working memory ID rate	246	0.82	0.09	0.46	1.00	240	0.83	0.09	0.58	1.00
	Visual working memory localisation error	246	6.55	1.95	3.18	12.94	240	7.03	2.00	2.75	13.83
	Visual working memory swap error rate	246	0.20	0.11	0.00	0.50	240	0.19	0.11	0.00	0.54

SD=standard deviation; NART=National Adult Reading Test; MRT=mean reaction time; IIVrt=intra-individual variability in reaction times; ACE-III=Addenbrooke's Cognitive Examination 3<sup>rd</sup> edition; PACC=pre-clinical Alzheimer's Cognitive Composite; MMSE=mini-mental state examination; DSST=digit-symbol substitution test; FNAME=face-name associative memory examination; ID=identification

			Males					Females		
Neuroimaging measure	Ν	Mean	SD	Min	Max	N	Mean	SD	Min	Max
TBV (cm <sup>3</sup> )	239	1151.35	86.69	945.90	1493.86	229	1046.75	82.49	818.60	1265.16
Hippocampal Volume (cm <sup>3</sup> )	239	3.25	0.33	2.42	4.27	229	3.00	0.30	2.06	3.72
WMHV (cm <sup>3</sup> )	234	4.77	5.09	0.27	33.67	221	5.47	5.78	0.35	32.78
SUVR	235	0.57	0.08	0.45	0.87	227	0.56	0.07	0.47	0.85
Amyloid positive [N(%)]			47 (20%)				3	39 (17.18%)		
Amyloid negative [N(%)]			188 (80%)				1	88 (82.82%)		
NAWM FA (z-score)	218	0.00	0.26	-0.92	0.72	206	-0.11	0.24	-0.76	0.48
NAWM MD (z-score)	218	0.16	0.37	-0.60	1.77	206	0.23	0.34	-0.55	1.23
CT Harvard ADsig (mm)	239	2.68	0.08	2.39	2.87	229	2.68	0.08	2.42	2.88
CT Frontal (mm)	239	2.76	0.09	2.18	2.95	229	2.75	0.09	2.36	2.92
CT Occipital (mm)	239	2.20	0.10	1.90	2.46	229	2.18	0.09	1.94	2.38
CT Parietal (mm)	239	2.46	0.08	2.13	2.67	229	2.48	0.08	2.19	2.67
CT Temporal (mm)	239	2.87	0.08	2.55	3.15	229	2.85	0.09	2.57	3.09

*Table 6.* Mean, standard deviation, and range of neuroimaging measures in Insight 46 males and females measured at age ~70 years.

SD=standard deviation; TBV=total brain volume; WMHV=white matter hyperintensity volume; SUVR=standardised uptake value ratio; NAWM=normal appearing white matter; FA=factional anisotropy; MD=mean diffusivity; CT=cortical thickness; ADsig=Alzheimer's Disease signature region

	Males		Females		Test of difference	
Variable	N	%	N	%	t/χ²	p-value
Father's social class age 4	2,354		2,144		1.32	0.93
Professional	136	5.78	125	5.83		
Intermediate	387	16.44	361	16.84		
Skilled non-manual	432	18.35	392	18.28		
Skilled manual	732	31.1	689	32.14		
Partly skilled	505	21.45	437	20.38		
Unskilled	162	6.88	140	6.53		
High (professional to non-manual)	955	40.57	878	40.95	0.07	0.79
Low (manual to unskilled)	1,399	59.43	1,266	59.05		
Father's social class age 11	2,039		1,883		4.86	0.43
Professional	125	6.13	112	5.95		
Intermediate	409	20.06	354	18.80		
Skilled non-manual	314	15.40	289	15.35		
Skilled manual	683	33.50	650	34.52		
Partly skilled	371	18.20	374	19.86		
Unskilled	137	6.72	104	5.52		
High (professional to non-manual)	848	41.58	755	40.10	0.90	0.34
Low (manual to unskilled)	1,191	58.41	1,128	59.90		
Father's social class age 15	1,945		1,798		3.20	0.67
Professional	135	6.94	114	6.34		
Intermediate	458	23.55	391	21.75		
Skilled non-manual	274	14.09	264	14.68		
Skilled manual	618	31.77	583	32.42		
Partly skilled	351	18.05	349	19.41		
Unskilled	109	5.60	97	5.39		
High (professional to non-manual)	867	44.58	769	42.77	1.24	0.27
Low (manual to unskilled)	1,078	55.42	1,029	57.23		

**Table 7.** Summary of male and female participant characteristics for covariables measured at different timepoints across the life course.

Childhood social class	2,452		2,221		2.38	0.80
Professional	155	6.32	135	6.08		
Intermediate	483	19.70	424	19.09		
Skilled non-manual	385	15.70	344	15.49		
Skilled manual	808	32.95	746	33.59		
Partly skilled	449	18.31	433	19.50		
Unskilled	172	7.01	139	6.26		
High (professional to non-manual)	1,023	41.72	903	40.66	0.54	0.46
Low (manual to unskilled)	1,429	58.28	1,318	59.34		
Overall social class age 15 to 43	2,119		2,039		871.56	<0.01
Professional	228	10.76	28	1.37		
Intermediate	795	37.52	618	30.31		
Skilled non-manual	202	9.53	801	39.28		
Skilled manual	640	30.20	159	7.80		
Partly skilled	192	9.06	337	16.53		
Unskilled	62	2.93	96	4.71		
High (professional to non-manual)	1,225	57.81	1,447	70.97	78.31	<0.01
Low (manual to unskilled)	894	42.19	592	29.03		
Overall social class age 15 to 53	2,144		2,056		798.02	<0.01
Professional	246	11.47	36	1.75		
Intermediate	769	35.87	638	31.03		
Skilled non-manual	213	9.93	767	37.31		
Skilled manual	644	30.04	170	8.27		
Partly skilled	204	9.51	334	16.25		
Unskilled	68	3.17	111	5.40		
High (professional to non-manual)	1,228	57.28	1,441	70.09	74.37	<0.01
Low (manual to unskilled)	916	42.72	615	29.91		
Educational attainment by age 26	2,308		2,124		126.54	<0.01
None attempted	930	40.29	835	39.31		
Vocational or GCSE-level	484	20.97	732	34.46		
A-level or higher	894	38.73	557	26.22		
Educational attainment by age 43	2,401		2,216		91.95	<0.01

None attempted	797	33 10	782	35.29		
Vocational or GCSE-level	605	25.20	787	35.51		
	005	25.20 41.61	647	20 20		
Physical activity ago 42	1 624	41.01	1 6 2 7	29.20	10.45	<0.01
Physical activity age 45	1,054	40.50	1,027		19.45	<0.01
	794	48.59	904	33.50		
1-4 times	386	23.62	367	22.56		
5 or more times	454	27.78	356	21.88		
Physical activity age 53	1,465		1,520		3.80	0.15
None	705	48.12	772	50.79		
1-4 times	273	18.63	245	16.12		
5 or more times	487	33.24	503	33.09		
Physical activity age 60-64	1,047		1,141		1.39	0.50
None	683	65.23	717	62.84		
1-4 times	137	13.09	162	14.20		
5 or more times	227	21.68	62	22.96		
Physical activity age 69	1,260		1,341		3.16	0.21
None	761	60.40	811	60.48		
1-4 times	142	11.27	178	13.27		
5 or more times	357	28.33	352	26.25		
APOE-e4 status	1,335		1,351		3.88	0.05
Non-carrier	909	68.09	967	71.58		
Carrier	426	31.91	384	28.42		
Smoking nack years age 26 (Mean(SD)[range])	1 568	2.33(2.81)	1 551	1.65(2.33)	7 36	<0.01
	1,500	[0, 19.50]	1,551	[0, 15.90]	7.50	
Smoking pack years age 43 (Mean(SD)[range])	1,579	2.38(3.68)	1,590	1.79(2.93)	5.04	<0.01
		2.80(4.76)		2.05(3.79)		
Smoking pack years age 53 (Mean(SD)[range])	1,434	[0, 25.00]	1,491	[0, 31.25]	4.75	<0.01
Smoking pack years age 60-64 (Mean(SD)[range])	1,410	1.33(3.21) [0, 22.50]	1,402	1.11(2.67) [0, 20.00]	1.97	0.05
Insight 46 age at visit (Mean(SD)[range])	256	70.64(0.69) [70.56, 70.73]	246	70.66(0.68) [70.57, 70.4]	-0.28	0.78
Insight 46 age at scan (Mean(SD)[range])	241	70.66(0.68) [69.25, 71.78]	230	70.67(0.67) [69.27, 71.86]	-0.18	0.86

Insight 46 TIV (cm <sup>3</sup> )	220	1519.62(1274.26)	220	1343.16(92.47)	10 16	<0.01
(Mean(SD)[range])	239	[1274.26, 1938.77]	229	[1114.35, 1558.05]	19.10	<b>\0.01</b>

*SD*=*standard deviation; GCSE*=*general certificate of secondary education; APOE*=*apolipoprotein E; TIV*=*total intracranial volume.* 

Tests of difference (t-tests for continuous variables,  $\chi^2$  for categorical variables) are highlighted in bold when significant at the p<0.05 level.

## 3.3.2. Cognitive performance in childhood and young adulthood

Unadjusted model estimates comparing male and female performance on cognitive ability tests completed by NSHD participants at ages 8, 11, 15 and 26 are presented in Figure 1, which demonstrates that females performed better than males on most assessments at ages 8 and 11, while males performed better on most assessments at age 15 and 26.

Some notable patterns of sex differences across childhood and young adulthood include:

- i. Changing advantage in overall cognition: Significant sex differences in overall cognition z-scores at age 8 were not detected, while at 11 there was a trend for better overall cognitive performance in females than in males ( $\beta$ =0.05; 95% Cl 0.00, 0.11), and at 15 males performed better than females overall ( $\beta$ =-0.13[-0.19, -0.08]). At 11, females performed better than males on the general ability test ( $\beta$ =0.17[0.10, 0.23]) but at 15 no sex differences were detected in general ability test performance.
- ii. Changing advantage in sentence completion task and arithmetic task: At age 8, females performed better than males on the sentence completion task ( $\beta$ =0.13[0.07, 0.19]), but at 15 there was a trend for better male performance on a similar sentence completion assessment (the Watts-Vernon test:  $\beta$ =-0.06; bias-corrected 95% CI -0.13, 0.00), and at age 26, males performed significantly better than females on this task ( $\beta$ =-0.08; bias-corrected 95% CI -0.14, -0.01). As previously found by Douglas et al. (1968),<sup>235</sup> no sex difference was detected in arithmetic performance at age 11, but at 15 males showed better performance on the mathematics test than females ( $\beta$ =-0.32[-0.38, -0.26]).
- iii. Consistent female advantage on word reading task: Females consistently performed better than males on the word reading task at ages 8 (β=0.13; bias-corrected 95% CI 0.07, 0.019) and 11 (β=0.09; bias-corrected 95% CI 0.03, 0.16).
- iv. Consistent male advantage on vocabulary task: There was a consistent male advantage in vocabulary test performance at ages 8 (β=-0.09[-0.15, -0.03]) and 11 (β=-0.09[-0.15, -0.03]).

Adjustments for SEP (Table 8) did not substantially change model effect estimates for cognitive outcomes measured at ages 8, 15, and 26. However, SEP adjustment at age 11 revealed a trend for better female than male performance on the arithmetic test ( $\beta$ =0.06[0.00, 0.11]) and

the trend for better overall cognitive performance at age 11 in females became statistically significant ( $\beta$ =0.06[0.01, 0.11]), reflecting positive confounding. At 26, adjusting for education attenuated the male advantage in Watts-Vernon task performance, while accounting for smoking slightly strengthened the male advantage, demonstrating negative confounding from smoking.

No effect modification by *APOE*- $\epsilon$ 4 status was detected for cognitive outcomes measured at ages 8, 11, or 26, but at 15 a sex-by-*APOE* interaction was detected for mathematics performance (Table 9) whereby a stronger male advantage was observed in *APOE*- $\epsilon$ 4 carriers ( $\beta$ =-0.46; bias-corrected 95% CI -0.61, -0.31) than in non-carriers ( $\beta$ =-0.28; bias-corrected 95% CI -0.38, -0.18) (Table 10).



**Figure 1.** Unadjusted (M0) model estimates and 95% confidence intervals for cognitive assessments administered during childhood (ages 8, 11, and 15) and young adulthood (age 26).

\*Bootstrapping applied: confidence intervals are bias-corrected

### 3.3.3. Cognitive performance ages 43 to 69

As shown in Figure 2, unadjusted model estimates indicate that, overall, females performed better than males on most cognitive assessments from age 43 to 69.

Some notable patterns in sex differences include:

- i. *Persistent female advantages on word list recall and timed letter search tasks:* From age 43 through to age 69, females outperformed males on repeated measures of these tasks.
- ii. Changes in verbal fluency advantages: No sex differences in verbal fluency performance were detected at age 53, but at age 69 females showed advantages on the ACE-III verbal fluency sub-domain.
- iii. *No sex differences in cognitive state at age 69:* Males and females did not significantly differ in their ACE-III total performance.
- iv. *Male task advantages observed only at age 69:* Males performed significantly better than females on the finger tapping task at age 69, and on the ACE-III attention and orientation and visuospatial sub-domains.

SEP adjustments (Table 8) strengthened female advantages in word list recall and timed letter search performance at ages 43, 60-64, and 69, prospective memory performance at age 54 and ACE-III sub-domains at age 69. SEP adjustments attenuated male advantages in ACE-III attention and orientation, and visuospatial domains at age 69.

Adjusting for education (Table 8) attenuated male advantages in ACE-III attention and orientation and visuospatial sub-domain performance at 69, while female performance advantages across all ages were strengthened. At 53, adjusting for education revealed a statistically significant female advantage on the NART ( $\beta$ =0.11[0.05, 0.18]).

Smoking adjustments (Table 8) had minimal impact on effect estimates for cognitive outcomes at ages 43, 60-64 and 69, but at age 53 smoking adjustments slightly attenuated female performance advantages across tasks. Across ages and cognitive measures, adjustments for PA did not substantially alter model estimates (Table 8). *APOE*-ε4 effect modification was detected for word list recall (age 53 and 69), prospective memory (age 53), verbal fluency (age 53), ACE-III total (age 69), and ACE-III memory (age 69) (Table 9). Stratifying these models by *APOE*-ε4 status (Table 10) demonstrated that, of these memory tasks that initially showed female advantage (word list recall, prospective memory, ACE-III memory), the female advantage was stronger in ε4 non-carriers than in carriers. Conversely, the verbal fluency task at age 53 showed a male advantage only in ε4 carriers.



*Figure 2.* Unadjusted (M0) model estimates and 95% confidence intervals for cognitive assessments administered at ages 43, 53, 60-64, and 69.

\*Bootstrapping applied: confidence intervals are bias-corrected

NART=national adult reading test; MRT=mean reaction time; IIVrt=intra-individual variability in reaction times; ACE-III=Addenbrooke's cognitive examination 3<sup>rd</sup> edition

# 3.3.4. Insight 46 wave I cognitive measures (age ~70)

Females performed better than males on the PACC sub-tests, while few sex differences were detected for additional computerised tasks (Figure 3). Males had significantly greater visuomotor integration task subtraction rates ( $\beta$ =-0.59[-0.76, -0.41]) and fewer visual working memory localisation errors ( $\beta$ =-0.24[-0.42, -0.06]) than females.

With adjustments for SEP (Table 8), female advantages on the PACC total and its sub-domains increased. The male advantage for visuomotor subtraction rate was slightly strengthened with SEP adjustment, while the effect for better male object localisation was attenuated. Adjusting for education strengthened most female advantages and slightly attenuated male advantages. Model estimates were not substantially altered with smoking or PA adjustments.

Effect modification by *APOE*- $\epsilon$ 4 status was detected for PACC total, logical memory, visual working memory localisation error, and visual working memory swap error (Table 9). In stratified analyses of these models (Table 10), female advantages on the PACC total and logical memory tasks were detected only in  $\epsilon$ 4 non-carriers. Male non-carriers showed better object localisation than female non-carriers, and female  $\epsilon$ 4 carriers showed fewer visual working memory swap errors than male carriers.



**Figure 3.** Unadjusted (M0) model estimates and 95% confidence intervals for cognitive assessments administered at Insight 46 wave I (age ~70).

\*Bootstrapping applied: confidence intervals are bias-corrected

PACC=pre-clinical Alzheimer's cognitive composite; MMSE=mini-mental state examination; DSST=digit-symbol substitution test; FNAME=face-name associative memory examination; IIVrt=intra-individual variability in reaction times; VWM=visual working memory; ID=identification

#### 3.3.5. Insight 46 wave I neuroimaging measures (age ~70)

#### 3.3.5.1. Volumetric measures

With minimal adjustments, females had, on average, a 23.45cm<sup>3</sup> greater TBV, relative to TIV, than males (95% CI 12.65, 34.27). This effect was strengthened with adjustment for SEP, but adjustments for education, smoking, and PA did not substantially alter the model estimates (Table 8; Figure 4). No effect modification by *APOE*- $\epsilon$ 4 status was detected (Table 9).

No sex difference in hippocampal volume was detected, nor did model adjustments alter the effect estimates (Table 8; Figure 4). No effect modification by *APOE*-ε4 status was detected (Table 9).

#### *3.3.5.2.* White matter hyperintensity volume (WMHV)

With minimal adjustments, females had, on average, 33% greater WMHV than males (relative log change=1.33[1.02, 1.73]). This effect was slightly increased with adjustments for SEP and PA while adjustments for education and smoking did not alter the effect estimate (Table 8; Figure 4). No effect modification by *APOE*- $\epsilon$ 4 status was detected (Table 9).

### 3.3.5.3. Amyloid

No sex differences in continuous SUVR or in amyloid status were detected, nor were there substantial alterations in effect estimates with covariable adjustments (Table 8; Figure 5). No effect modifications by *APOE*- $\epsilon$ 4 status were detected (Table 9).

#### *3.3.5.4.* Normal appearing white matter (NAWM)

With minimal adjustments, females had lower FA ( $\beta$ =-0.11[-0.16, -0.06]) and greater MD ( $\beta$ =0.07[0.00, 0.14]) than males, indicating poorer NAWM microstructural integrity in females. Adjusting for covariables did not substantially alter effect estimates (Table 8; Figure 6).

Effect modifications by *APOE*- $\epsilon$ 4 status were detected for FA and MD (Table 9). Stratified analyses (Table 10) showed that, in  $\epsilon$ 4 carriers, there were no sex differences in FA or MD, while in non-carriers females had lower FA ( $\beta$ =-0.13[-0.19, -0.08]) and greater MD ( $\beta$ =0.11[0.03, 0.19]) than males.

### 3.3.5.5. Cortical thickness

As shown in Figure 7, no sex differences were detected in Harvard AD signature region or frontal cortical thickness, but there were trends for thinner occipital ( $\beta$ =-0.02[-0.03, 0.00]) and temporal ( $\beta$ =-0.02[-0.03, 0.00]) cortices and thicker parietal cortices ( $\beta$ =0.02[0.00, 0.03]) in females than males. Adjustment for SEP attenuated effects for occipital and temporal cortices while education, smoking, and PA adjustments did not alter effect estimates (Table 8; Figure 7). No effect modifications by *APOE*- $\epsilon$ 4 status were detected (Table 9).



**Figure 4.** Model estimates and 95% confidence intervals showing the difference in: A) mean total brain volume (TBV; cm<sup>3</sup>); B) hippocampal volume (cm<sup>3</sup>); C) relative log difference in white matter hyperintensity volume (WMHV) between males and females, with minimal adjustments and with additional adjustments for socioeconomic and lifestyle covariables.

SEP=socioeconomic position; PA=physical activity

M0=age at scan + total intracranial volume (TIV); SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69



**Figure 5.** Model estimates and 95% confidence intervals showing the difference in A) amyloid standardised uptake value ratio (SUVR) and B) the difference in odds ratio for being amyloid positive, between males and females, with minimal adjustments and with additional adjustments for socioeconomic and lifestyle covariables.

SUVR=standardised uptake value ratio; Aβ+=beta-amyloid positive; OR=odds ratio; SEP=socioeconomic position; PA=physical activity

M0=age at scan; SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69



**Figure 6.** Model estimates and 95% confidence intervals showing the difference normal appearing white matter A) fractional anisotropy and B) mean diffusivity measures between males and females, with minimal adjustments and with additional adjustments for socioeconomic and lifestyle covariables.

NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; SEP=socioeconomic position; PA=physical activity

M0=age at scan; SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69



**Figure 7.** Model estimates and 95% confidence intervals showing the difference cortical thickness regions of interest measures between males and females, with minimal adjustments and with additional adjustments for socioeconomic and lifestyle covariables. A) Harvard Alzheimer's Disease signature; B) Frontal cortex; C) Occipital cortex; D) Parietal cortex; E) Temporal cortex.

SEP=socioeconomic position; PA=physical activity

M0=age at scan; SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69

**Table 8.** Model estimates for the effects of sex on cognitive performance or neuroimaging measures across the life course. Estimates from minimally adjusted models (MO) and models adjusted for socioeconomic and lifestyle covariables applicable at each timepoint are presented.

Age	Outcome variable	Ν	Μ0 (β)	M0+SEP (β)	M0+Education (β)	M0+Smoking (β)	Μ0+ΡΑ (β)
	Picture intelligence	4,264	0.00	0.00			
	Sentence completion	4,258	0.13	0.13			
8	Word reading*	4,258	0.13	0.13	N/A	N/A	N/A
	Vocabulary	4,258	-0.09	-0.09			
	Overall cognition age 8	4,255	0.04	0.04			
	Word reading*	4,025	0.09	0.10			
	General ability	4,030	0.17	0.18			
11	Vocabulary	4,025	-0.09	-0.08	N/A	N/A	N/A
	Arithmetic	4,023	0.04	0.06			
	Overall cognition age 11	4,022	0.05	0.06			
	General ability	4,015	-0.01	0.00			
15	Watts-Vernon*	4,012	-0.06	-0.06	NI/A	NI/A	NI / A
15	Mathematics*	4,013	-0.32	-0.31	N/A	N/A	N/A
	Overall cognition age 15	4,006	-0.13	-0.12			
26	Watts-Vernon*	3,713	-0.08	-0.07	0.00	-0.11	N/A
	Word list recall	3,059	0.26	0.30	0.36	0.24	0.29
12	Timed letter search	3,150	0.28	0.32	0.32	0.27	0.30
43	Peg placement*	3,163	-0.05	-0.03	-0.03	-0.06	-0.04
	Picture recall*	3,224	0.16	0.16	0.19	0.16	0.18
	Word list recall	2,886	0.30	0.30	0.40	0.27	0.31
	Timed letter search	2,931	0.22	0.22	0.26	0.20	0.22
53	NART	2,824	-0.02	0.02	0.11	-0.05	-0.01
	Verbal fluency	2,948	-0.05	-0.01	0.02	-0.06	-0.04
	Prospective memory*	2,924	0.15	0.18	0.17	0.14	0.15
	Word list recall	2,150	0.38	0.45	0.47	0.37	0.37
60-64	Timed letter search	2,182	0.15	0.17	0.19	0.14	0.15
00-04	MRT*	2,167	0.04	0.08	0.07	0.04	0.03
	IIVrt*	2,167	0.07	0.06	0.08	0.07	0.07
	Word list recall	2,074	0.33	0.36	0.41	0.32	0.34
69	Timed letter search	2,111	0.16	0.17	0.19	0.15	0.16
	Finger tapping	2,046	-0.46	-0.47	-0.45	-0.47	-0.46

	ACE-III total*	1,762	0.05	0.11	0.14	0.04	0.06
	ACE-III verbal fluency*	2,101	0.13	0.16	0.19	0.13	0.13
	ACE-III language*	1,765	-0.01	0.02	0.05	-0.02	-0.01
	ACE-III attention & orientation*	1,785	-0.15	-0.07	-0.12	-0.15	-0.15
	ACE-III memory*	1,786	0.20	0.21	0.26	0.19	0.20
	ACE-III visuospatial function*	1,778	-0.12	-0.10	-0.06	-0.13	-0.12
	PACC total	502	0.35	0.41	0.39	0.35	0.34
	MMSE*	502	0.15	0.21	0.19	0.15	0.14
	DSST	501	0.31	0.36	0.34	0.30	0.30
	Logical memory	502	0.45	0.50	0.48	0.44	0.45
	FNAME	500	0.48	0.56	0.53	0.48	0.47
	Matrix reasoning*	502	0.04	0.16	0.09	0.04	0.04
Insight 46 cognition	llVrt	501	-0.13	-0.13	-0.12	-0.13	-0.13
(age ~70)	Response inhibition	500	-0.03	-0.06	-0.02	-0.04	-0.04
	Visuomotor integration rotation errors*	483	-0.01	0.07	0.01	-0.01	-0.01
	Visuomotor integration number of rotations*	483	-0.07	-0.12	-0.07	-0.07	-0.07
	Visuomotor integration subtraction rate	483	-0.59	-0.61	-0.57	-0.60	-0.59
	Visual working memory ID rate	486	0.18	0.19	0.18	0.17	0.18
	Visual working memory localisation error	486	-0.24	-0.19	-0.23	-0.24	-0.25
	Visual working memory swap errors	486	0.05	0.12	0.05	0.05	0.04
	TBV (cm <sup>3</sup> )	468	23.46	28.43	23.55	23.05	22.97
	Hippocampal volume (cm <sup>3</sup> )	468	-0.05	-0.04	-0.05	-0.05	-0.05
	WMHV (relative log change)	455	1.33	1.37	1.33	1.33	1.35
	SUVR	462	-0.01	0.00	-0.01	-0.01	-0.01
	Amyloid status [OR; ref:Aβ negative]	462	0.83	1.00	0.81	0.83	0.83
Insight 46 neuroimaging	NAWM FA (z-score)	411	-0.11	-0.11	-0.10	-0.11	-0.11
(age ~70)	NAWM MD (z-score)	411	0.07	0.08	0.07	0.07	0.07
	CT ADSig (mm)	468	0.01	0.01	0.01	0.01	0.01
	CT Frontal (mm)	468	0.00	0.00	0.00	0.00	0.00
	CT Occipital (mm)	468	-0.02	-0.01	-0.02	-0.02	-0.02
	CT Parietal (mm)	468	0.02	0.02	0.02	0.02	0.02
	CT Temporal (mm)	468	-0.02	-0.01	-0.02	-0.02	-0.02

SEP=socioeconomic position; PA=physical activity; N/A=not applicable; NART=national adult reading test; MRT=mean reaction time; IIVrt=intra-individual variability in reaction times; ACE-III=Addenbrooke's cognitive examination 3<sup>rd</sup> edition; PACC=pre-clinical Alzheimer's cognitive composite; MMSE=mini-mental state examination; DSST=digit-symbol substitution test; FNAME=face-name associative memory examination; ID=identification; TBV=total brain volume; WMHV=white matter hyperintensity volume;

SUVR=standardised uptake value ratio; OR=odds ratio; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; ADSig=Alzheimer's Disease signature

Reference category: males.

Yellow cells indicate female advantages, significant at the p<0.05 level. Blue cells indicate male advantages, significant at the p<0.05 level.

\*Outcome variable is not normally distributed. The model estimates presented in this table are from analyses which did not have bootstrapping applied. While the model estimates are interpretable, parametric 95% confidence intervals and p-values for these outcomes cannot be interpreted. Cells are highlighted if bias-corrected 95% confidence intervals from M0 analyses (see Figures 1-3) did not cross zero.

Age 8: SEP=father's social class at participant age 4

Age 11: SEP=father's social class at participant age 11

Age 15: SEP=father's social class at participant age 15

Age 26: SEP=overall childhood social class; Education=highest educational attainment up to age 26; Smoking=smoking pack years to age 26

Age 43: SEP= overall childhood social class + overall own social class from age 15 to 43; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 43; PA=frequency of PA at age 43

Age 53: SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 53; PA=frequency of PA at age 53

Age 60-64: SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 64

Age 69: SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69

Insight 46 cognition: M0=age at visit; SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69

Insight 46 neuroimaging: M0=age at scan + TIV; SEP= overall childhood social class + overall own social class from age 15 to 53; Education=highest educational attainment up to age 43; Smoking=smoking pack years to age 60-64; PA=frequency of PA at age 69

**Table 9.** Significance p-values and 95% confidence intervals for sex-by-APOE interaction terms, for each cognitive and neuroimaging outcome. Interaction terms were included in minimally adjusted models assessing the effects of sex on outcome measures (MO).

Age	Outcome variable	Lower Cl	Upper Cl	Р
	Picture intelligence	-0.04	0.30	0.13
	Sentence completion	-0.26	0.08	0.29
8	Word reading*	-0.20	0.16	-
	Vocabulary	-0.26	0.08	0.31
	Overall cognition age 8	-0.16	0.13	0.82
	Word reading*	-0.26	0.09	-
	General ability	-0.21	0.13	0.66
11	Vocabulary	-0.28	0.07	0.24
	Arithmetic	-0.27	0.08	0.27
	Overall cognition age 11	-0.23	0.07	0.30
	General ability	-0.20	0.14	0.74
15	Watts-Vernon*	-0.32	0.04	-
15	Mathematics*	-0.34	0.01	-
	Overall cognition age 15	-0.27	0.04	0.14
26	Watts-Vernon*	-0.23	0.12	-
	Word list recall	-0.27	0.08	0.27
43	Timed letter search	-0.24	0.10	0.43
	Peg placement*	-0.13	0.18	-
	Picture recall*	-0.23	0.09	-
	Word list recall	-0.36	-0.03	0.02
	Timed letter search	-0.26	0.07	0.27
53	NART	-0.27	0.07	0.26
	Verbal fluency	-0.32	0.01	0.06
	Prospective memory*	-0.31	0.02	-
	Word list recall	-0.34	0.04	0.12
60-64	Timed letter search	-0.25	0.15	0.56
00 04	MRT*	-0.22	0.14	-
	IIVrt*	-0.13	0.23	-
	Word list recall	-0.47	-0.07	0.01
	Timed letter search	-0.26	0.13	0.53
	Finger tapping	-0.12	0.27	0.47
	ACE-III total*	-0.43	0.00	-
69	ACE-III verbal fluency*	-0.27	0.11	-
	ACE-III language*	-0.22	0.20	-
	ACE-III attention & orientation*	-0.34	0.11	-
	ACE-III memory*	-0.53	-0.09	-
	ACE-III visuospatial function*	-0.29	0.18	-
	PACC total	-0.54	0.01	0.06
	MMSE*	-0.72	0.11	-
	DSST	-0.60	0.17	0.27
	Logical memory	-0.70	0.05	0.09
	FNAME	-0.63	0.13	0.19
Insight 46 cognition	Matrix reasoning*	-0.17	0.57	-
(age ~70)	llVrt	-0.70	0.06	0.10
	Response inhibition	-0.25	0.53	0.48
	Visuomotor integration rotation errors*	-0.44	0.21	-
	Visuomotor integration number of	-0.43	0.32	-
	rotations*			
	Visuomotor integration subtraction rate	-0.40	0.36	0.91

	Visual working memory ID rate	-0.29	0.49	0.61
	Visual working memory localisation error	0.04	0.81	0.03
	Visual working memory swap errors	0.06	0.84	0.02
	TBV (cm <sup>3</sup> )	-24.27	11.12	0.47
	Hippocampal volume (cm <sup>3</sup> )	-0.18	0.05	0.26
	WMHV (relative log change)	0.54	1.31	0.45
	SUVR	-0.02	0.04	0.44
	Amyloid status [OR; ref:Aβ negative]	0.50	3.59	0.57
Insight 46 neuroimaging	NAWM FA (z-score)	-0.01	0.20	0.09
(age ~70)	NAWM MD (z-score)	-0.28	0.01	0.08
	CT ADSig (mm)	-0.01	0.05	0.19
	CT Frontal (mm)	-0.01	0.06	0.15
	CT Occipital (mm)	-0.01	0.06	0.13
	CT Parietal (mm)	-0.01	0.06	0.11
	CT Temporal (mm)	-0.02	0.05	0.48

\*Bootstrapping applied; confidence intervals are bias-corrected and p-values are noninterpretable

All cognitive measures are presented as standardised z-scores.

Reference category: males.

Bold text indicates significant sex-by-APOE interactions (p<0.1 or bias-corrected confidence intervals not crossing by greater than .02).

CI=confidence interval; NART=national adult reading test; MRT=mean reaction time; IIVrt=intra-individual variability in reaction times; ACE-III=Addenbrooke's cognitive examination 3<sup>rd</sup> edition; PACC=pre-clinical Alzheimer's cognitive composite; MMSE=minimental state examination; DSST=digit-symbol substitution test; FNAME=face-name associative memory examination; ID=identification; TBV=total brain volume; WMHV=white matter hyperintensity volume; SUVR=standardised uptake value ratio; OR=odds ratio; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; ADSig=Alzheimer's Disease signature **Table 10.** Model estimates and 95% confidence intervals for the association between sex and cognitive performance or neuroimaging measures, stratified by APOE-ε4 status, from minimally adjusted models (M0).

			APOE-ɛ4 carriers					APOE-ɛ4 non-carriers				
Age	Outcome	Ν	β	Lower CI	Upper Cl	Р	N	β	Lower Cl	Upper Cl	Р	
15	Mathematics*	684	-0.46	-0.61	-0.31	-	1,609	-0.28	-0.38	-0.18	-	
53	Word list recall	787	0.16	0.02	0.30	0.03	1,818	0.35	0.26	0.44	<0.01	
	Verbal fluency	797	-0.15	-0.29	-0.02	0.03	1,856	0.01	-0.09	0.10	0.92	
	Prospective memory*	794	0.05	-0.10	0.19	-	1,841	0.20	0.09	0.29	-	
69	Word list recall	555	0.10	-0.007	0.27	0.24	1,277	0.37	0.26	0.48	<0.01	
	ACE-III total*	470	-0.17	-0.34	0.02	-	1,076	0.07	-0.05	0.19	-	
	ACE-III memory*	475	-0.06	-0.24	0.15	-	1,092	0.26	0.15	0.37	-	
Insight 46 cognition (age ~70)	PACC total	148	0.17	-0.07	0.42	0.17	352	0.43	0.29	0.58	<0.01	
	Logical memory	148	0.25	-0.06	0.55	0.11	352	0.56	0.35	0.77	<0.01	
	VWM localisation error	144	0.08	-0.26	0.41	0.64	340	-0.35	-0.56	-0.14	<0.01	
	VWM swap errors	144	0.38	0.06	0.70	0.02	340	-0.07	-0.29	0.14	0.51	
Insight 46	NAWM FA (z-score)	119	-0.04	-0.13	0.05	0.37	290	-0.13	-0.19	-0.08	<0.01	
neuroimaging	NAWM MD (z-score)	119	-0.03	-0.16	0.11	0.72	290	0.11	0.03	0.19	0.01	
(age ~70)												

(age ~70) | \*Bootstrapping applied; confidence intervals are bias-corrected and p-values are non-interpretable

All cognitive measures are presented as standardised z-scores.

Reference category: males.

APOE=apolipoprotein E; CI=confidence interval; ACE-III=Addenbrooke's cognitive examination 3<sup>rd</sup> edition; PACC=pre-clinical Alzheimer's cognitive composite; VWM=visual working memory; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity

### 3.4. Discussion

### 3.4.1. Key findings

Differences in mean performance on a range of cognitive assessments were detected between males and females, at several timepoints throughout the life course. Overall, females had better performance on tasks assessing verbal and memory domains, while males performed better on tasks of non-verbal and visuospatial domains. Female cognitive performance advantages were generally strengthened with adjustment for socioeconomic and education covariables, which tended to attenuate male cognitive performance advantages. Lifestyle covariables (smoking, PA) did not substantially contribute to cognitive performance sex differences. Most sex differences in cognitive performance were not modified by APOE- $\epsilon$ 4 status, but effect modifications were detected mainly at older ages and indicated that female memory performance advantages were stronger in  $\epsilon$ 4 non-carriers than in carriers.

Some sex differences in mean neuroimaging measures of brain health at age ~70 were detected; females had larger TBV relative to head size, but also greater WMHV and poorer NAWM microstructural integrity. Socioeconomic, education, and lifestyle covariables did not substantially contribute to sex differences in neuroimaging measures.

## 3.4.2. Interpretation of findings

Overall, these analyses align with previous reports of male advantages in non-verbal, visuospatial cognitive domains and female advantages in verbal and memory domains.<sup>47</sup> Here, and elsewhere, the reported effect sizes for such differences are small and there is substantial overlap in the distribution of male and female task performance.<sup>47,49</sup> Nonetheless, repeated replication of such differences in mean performance suggests that there are meaningful, albeit subtle, variations in non-verbal and verbal abilities between the sexes.

Interestingly, females showed an advantage in sentence completion performance at age 8 while males performed better on similar reading comprehension tasks at ages 15 and 26. While most research reports female reading comprehension advantages,<sup>243</sup> some evidence shows that boys can perform better when using phonological rather than analytical or context-driven strategies,<sup>243</sup>

highlighting how variations in cognitive strategies potentially underlie some observed cognitive sex differences. Indeed, in these analyses males did better on rule-based tasks such as mathematics, while females performed better on tasks of episodic memory which were more reliant on contextual processing. It is possible that some aspect of education, which could not be accounted for, contributed to the change in reading comprehension sex differences over time. For example, there could have been a bias in how males and females were taught during the 1950s, possibly with greater emphasis on preparing boys for employment or higher education while girls were taught more domestic skills, given prevalent traditional gender roles at that time.<sup>244</sup>

Indeed, education is repeatedly identified as an important explanatory factor in cognitive sex differences,<sup>245,246</sup> with secular improvements in women's access to education being associated with better female performance on cognitive tasks, namely verbal memory and fluency.<sup>54</sup> Being representative of post-war Britain, NSHD females unsurprisingly had lower education levels than males. Since accounting for education level strengthened female cognitive advantages while weakening male advantages, these analyses reinforce the notion that reduced female access to education can suppress female cognitive abilities. Adjusting for SEP had a similar effect; although more females than males had overall skilled non-manual occupations, females were outnumbered in the highest-level occupations. Greater engagement in cognitive stimulating activities, such as those linked with occupation and higher education levels, are thought to improve cognitive performance and reduce dementia risk by improving cognitive reserve<sup>86</sup> (Section 1.6.). It has been suggested that traditionally lower education and SEP in females could contribute to greater rates of AD in females,<sup>247</sup> emphasising the importance of understanding how and whether modifiable environmental and social factors contribute to cognitive variability between the sexes.

The current findings did not, however, support a role of smoking and PA lifestyle factors in cognitive sex differences. While such factors have previously been associated with cognitive performance,<sup>228,229</sup> associations appear to be driven by extremes (i.e. heavy smoking and physical inactivity)<sup>227,248</sup> and evidence for sex differences in how these factors link with dementia risk is mixed.<sup>247</sup> It should be noted that, within NSHD and Insight 46, sex differences in PA were only

105

detected at age 43 and that smoking sex differences diminished over time, with no significant difference at age 60-64, although this could be a result of limited power to detect sex differences given that few participants (n=291) were current smokers at age 60-64. The relative lack of sex differences in these lifestyle factors could offer some explanation for their lack of contribution to differences in cognitive performance.

The role of biological variability between males and females must also be considered. For instance, puberty typically occurs between age 8 and 14, with girls experiencing puberty at younger ages than boys.<sup>249</sup> Increasing sex hormone levels during this time could have contributed to the changing pattern of cognitive sex differences during childhood, particularly given that sex hormones are linked with pubertal changes in brain organisation.<sup>250</sup> The parietal and occipital lobes, subserving sensory and visuospatial processing,<sup>251,252</sup> tend to mature before the frontal and temporal lobes<sup>253</sup> which subserve higher level cognitive abilities such as attention, problem solving, and memory.<sup>254,255</sup> Reflecting the generally earlier age of puberty onset in girls than boys, females typically reach peak grey matter volumes at younger ages than males,<sup>253</sup> which could explain why females tended to perform better than males on cognitive tasks at younger ages during adolescence (8-11 years), but not at later ages (15-26 years). Additionally, with increasing age during adolescence, the hippocampus – rich in oestrogen receptors and involved in memory processing – is found to increase in volume only in females, while the amygdala – rich in androgen receptors and involved in attention and emotion processing - increases in volume only in males.<sup>253,256</sup> Potential advances in attentional abilities in males during adolescence, might also contribute to general improvements in male task performance at later stages of adolescence.

Hormonal changes at midlife could also have contributed to variations in verbal fluency sex differences over time. While literature consistently reports a female advantage in verbal fluency,<sup>47</sup> here a female advantage was only detected at age 69 and not at 53, although an effect modification by *APOE*-ε4 showed a male advantage at age 53 in ε4 carriers. During midlife, women experience declining oestrogen levels as they transition through menopause, typically aged 45-55.<sup>133</sup> The transition has been associated with cognitive changes, and 'brain fog' is a commonly reported symptom referring to cognitive difficulties, including word finding problems.<sup>143</sup> Word finding difficulties linked with menopause could therefore have suppressed

female verbal fluency performance during midlife. However, that female advantages in repeated measures of verbal memory were consistently detected from age 43 to 69, corroborating previous analyses in this cohort,<sup>5</sup> demonstrates that females still maintain some verbal cognitive advantages throughout midlife and into later-life.

While there is no clear explanation for better male ɛ4 carrier verbal fluency performance at 53, it is possible that any adverse effect of  $\varepsilon 4$  on verbal fluency performance was stronger in females, and that this effect could have been exacerbated during midlife. Most sex-by-APOE-E4 interactions on cognitive performance were detected in later-life, in agreement with literature generally detecting negative associations of  $\varepsilon 4$  with cognition in later-life<sup>257</sup> but null associations at younger ages.<sup>258</sup> Some positive associations of ɛ4 with cognition have, however, been reported. Insight 46 E4 carriers showed better object identification and localisation on the visual working memory task than non-carriers,<sup>218</sup> while other positive associations tend to be reported at vounger ages.<sup>259,260</sup> It has been hypothesised that ε4 has some evolutionary advantages, possibly related to inflammatory responses protective against pathogens,<sup>261</sup> and only now that human lifespan has increased are the deleterious effects of ɛ4 in later-life observed.<sup>262</sup> Thus far, sex differences in the potential antagonistic pleiotropy of APOE-E4 have not been explored, but if inflammatory processes are involved then greater immune responses in females<sup>81</sup> might contribute to sex differences. Overall, current results show that female cognitive advantages, particularly in memory domains, were stronger in ε4 non-carriers than in carriers, indicating that ε4 presence could suppress female cognitive abilities, particularly at older ages.

Conversely, only female  $\epsilon$ 4 non-carriers showed poorer microstructural integrity, although these *APOE*- $\epsilon$ 4 interactions were attenuated when models were adjusted for WMHV (log-transformed) and TIV (MD p=0.14; FA p=0.20). The volume of cSVD present therefore explains some of this relationship; those with greater cSVD could have less NAWM, with poorer microstructural integrity than those with lower cSVD burden. Previous work in this cohort showed that poorer NAWM microstructural integrity in females was largely explained by differences in WMHV, suggesting that alterations in NAWM and white matter hyperintensities indicating cSVD may be part of an overlapping pathological process.<sup>263</sup> The links between NAWM and WMHV are corroborated in the current results, given that females also show greater WMHV than males.

107

However, previous Insight 46 analyses, which excluded individuals with neurological conditions, did not find sex differences in WMHV.<sup>158</sup> While, in the current analyses, excluding scans which failed BaMoS QC did exclude participants with multiple sclerosis (n=2) from WMHV analyses, these discrepant findings highlight that the presence of neurological conditions explain greater female WMHV in this cohort. Nonetheless, the detection of sex differences in NAWM and WMHV measures linked with cSVD<sup>264</sup> suggests a role for cerebrovascular factors underlying sex differences in white matter pathology.<sup>291</sup> Indeed, midlife cardiovascular risks only associated with NAWM measures in females in Insight 46.<sup>237</sup> In agreement with existing literature,<sup>158,265</sup> no sex differences in AD-related brain pathology (A $\beta$  load, hippocampal volume) were detected within Insight 46. It is possible that cerebrovascular changes precede the development of AD pathology, as proposed by the vascular theory of AD which postulates that vascular dysfunction contributes to amyloid accumulation and neurodegeneration.<sup>266</sup>

While most neuroimaging sex differences detected seem to indicate poorer outcomes in females (i.e. females showed poorer NAWM integrity, greater WMHV, and thinner occipital and temporal cortices), females did have greater TBV which is generally thought to be beneficial, given that brain atrophy is a marker of dementia.<sup>20</sup> While literature reports mixed evidence for sex differences in TBV, meta-analysis does show overall support for greater TBV in males,<sup>267</sup> although most existing evidence is restricted to younger samples, aged 18-59.<sup>267</sup> In older, preclinical AD populations, there is suggestion that brain volume increases could precede brain atrophy.<sup>268</sup> If so, greater TBV at age ~70 could be interpretated as disadvantageous, which would align with the indication from other neuroimaging outcomes that Insight 46 females have overall poorer brain health than males aged ~70.

Despite finding support for the contribution of socioeconomic and education factors to cognitive sex differences, there did not appear to be a role for such factors in neuroimaging sex differences, nor did lifestyle covariables contribute. Education levels and SEP might influence the cognitive strategies individuals use to complete tasks, with different strategies potentially arising from the same underlying brain structures. Indeed, literature does not always correlate neuroimaging outcomes with cognitive performance measures.<sup>269</sup> For instance, James et al. did not find associations between Insight 46 PACC performance and NAWM measures.<sup>237</sup> Higher levels of

108
brain pathology could be required before cognitive functions are affected, highlighted by dementia pathology beginning up to 20 years prior to symptom onset.<sup>270</sup> In this mostly cognitively unimpaired cohort, the finding that females showed better cognitive performance but poorer cerebrovascular brain health than males at age ~70 seems counterintuitive, but potentially indicates that cognitive function persists in spite of subtle brain changes, at least in females. Better female performance on most cognitive assessments throughout the life course indicates greater cognitive reserve in females, and larger female TBV at age ~70 could reflect greater female brain reserve. Greater levels of cognitive and structural brain resources could, therefore, support female cognitive resilience to cSVD pathology.

### 3.4.3. Strengths and limitations

The main strength of these analyses is the use of an age-homogenous sample and the availability of prospective life course data collected over seventy years, facilitating an examination of cognitive sex differences at specific ages and an opportunity to observe patterns in sex differences over time. The potential contribution of age-specific socioeconomic and lifestyle covariables were also considered, a strength over many studies reliant on retrospectively reported data which may not capture any changes in such sociocultural factors throughout the life course. Additionally, cognitive sex differences were considered alongside sex differences in a range of dementiarelated neuroimaging measures in later-life. This is useful for developing an understanding of the patterns of brain ageing which potentially underlie sex differences in dementia.

However, as is the case with most longitudinal research, the impact of sample attrition over time must be acknowledged; at each successive timepoint of interest, sample sizes for both males and females declined. Given that participation withdrawal or death is greater among individuals from more disadvantaged backgrounds and with poorer general health, sample attrition induced a survivor or retention bias. Further, the results of these analyses should be considered within the generational and cultural context of the cohort, an entirely white British cohort representative of mainland Britain in 1946 which may not be generalisable to other populations. Secular changes in education policies and societal views on traditional gender roles could mean that the

contribution of sociodemographic and lifestyle factors to sex differences is not equal across generations.

There is also a need to be mindful of potential collider bias in these analyses given that education, SEP, and lifestyle variables were individually adjusted as mediators. For instance, if an unmeasured confounder associates with sex as the exposure variable, with cognitive performance as the outcome variable, and with smoking as the mediating variable, then smoking would be a shared outcome (a collider) of the unmeasured confounder and sex. Adjusting for smoking could therefore open a back door path where the unmeasured confounder explains the associations between sex and cognitive performance.

#### 3.4.4. Summary

This examination of life course cognitive and brain health sex differences demonstrates that, in NSHD, females generally perform better than males on some cognitive assessments throughout the life course, but in later-life show poorer cerebrovascular brain health, specific to cSVD. Despite greater cSVD pathology, female cognitive advantages were maintained at age ~70 years, demonstrating female cognitive resilience to pathology. Sex differences in cognitive performance were not necessarily biologically pre-determined, with evidence for a contribution of sociodemographic variables; most notably, higher education levels were especially beneficial for female cognitive advantages. Some evidence for modifying effects of *APOE*- $\varepsilon$ 4 were detected, particularly at later ages, indicating a potentially stronger adverse impact of being an  $\varepsilon$ 4 carrier for females, notably for memory task performance. Overall, these analyses reinforce that, while females typically perform better than males on some cognitive tasks, they may be more susceptible to some sociocultural and genetic factors linked with dementia risk.

# 4.0. Empirical section B: Examining sex differences in cumulative lifetime dementia risks and their association with later-life cognitive function and brain health.

The primary purpose of this chapter is to examine whether life course modifiable dementia risk exposures differentially associate with cognition and brain health in later-life between males and females. A cumulative risks score (CRS) was derived from NSHD prospective life course data pertaining to each of the twelve modifiable dementia risk factors outlined in the 2020 Lancet commission on dementia prevention, intervention, and care;<sup>56</sup> higher CRS indicated a greater number of risk factor exposures across the life course, up to age 69. Unadjusted regression modelling assessed whether early-life and genetic variables (childhood cognition, childhood SEP, puberty timing, APOE genotype) associated with CRS; higher childhood cognition and SEP, and APOE-E4 carrier (vs. non-carrier) status associated with lower CRS. Multivariable regression models accounting for these early-life predictors of CRS showed adverse associations of higher CRS with cognitive performance at age 69 in both sexes; associations with cognitive state were equal in both sexes, females showed slightly stronger associations with verbal memory, and an association with processing speed was only detected in males. Greater CRS associated with smaller total brain- and hippocampal volumes only in males, while there was no evidence for CRS relationships with AD- (AB) or cSVD-related (WMHV) pathology in either sex. There was no evidence for effect modifications by APOE genotype in either sex. Key findings were replicated when CRS was calculated up to midlife (age 53). Overall, these analyses showed that, beyond early-life and genetic (APOE) factors, greater life course exposures to modifiable dementia risks associated with later-life outcomes indicative of more advanced cognitive and brain ageing, particularly in males.

#### 4.1. Introduction

As outlined in Section 1.0., there is increasing recognition that a life course approach is required to understand dementia risks, and that risks are interrelated.<sup>111</sup> The renowned 2020 Lancet commission on dementia prevention, intervention and care convened experts to identify the most convincing modifiable risk factors spanning early- (<45 years), mid- (45-65 years) and later-life (>65 years).<sup>56</sup> The twelve risks identified are: low education, hearing impairment, traumatic brain injury (TBI), hypertension, high alcohol consumption, obesity, smoking, depression, social isolation, physical inactivity, diabetes, and exposure to high levels of air pollution.<sup>56</sup> Although the causal mechanisms underlying the association of each factor with dementia risk are not fully established, the Lancet commission provides the most comprehensive overview of life course risk factors linked with dementia, using the best available evidence to date. A range of dementia risk scores have been developed to try to quantify the effects of multiple risk factors and to predict future risk of developing dementia.<sup>106,107</sup> While such scores have poor predictive accuracy,<sup>109</sup> they have been associated with later-life cognitive and brain health outcomes; higher scores, indicating more risk exposures, associated with poorer outcomes.<sup>108,271</sup> However, sex differences in the cumulative effects of multiple dementia risks are largely understudied.

Some studies have examined sex differences in the associations of individual cardiometabolic risk factors with later-life outcomes, generally demonstrating greater adverse effects in females than males. For example, midlife hypertension had been associated with poorer cognitive performance on the MMSE and greater dementia risk in women but not in men.<sup>124,272</sup> In the UK Biobank, being overweight or obese associated with increased dementia risk, compared with being a healthy weight, only in women.<sup>132</sup> Similarly, in NSHD at age 60-64, only in women did greater BMI associate with poorer performance on the letter search task assessing processing speed.<sup>206</sup>

Only one study to date, which derived a dementia risk score based on the Lancet commission identified risks, examined sex differences in the effects of multiple risk exposures. The online survey of 22,117 18-89 year-olds asked about eight of the twelve risks (low education, hearing loss, TBI, alcohol or substance abuse, hypertension, smoking, diabetes, depression). A dose-response effect was observed in males and females whereby each additional risk associated with poorer performance on a cognitive composite comprising tests of spatial working memory,

associative memory, processing speed, and executive function including inhibitory control and set shifting.<sup>111,273</sup> Females had fewer risk exposures than males across all ages, but the negative cumulative effect of multiple risk factors on associative memory performance was greater in females than in males.<sup>111</sup> Similar results were found when risks were weighted by the population attributable fractions (PAFs) outlined in the Lancet commission.<sup>111,273</sup>

While dementia risks have been identified across the life course, only one childhood risk is included in the Lancet risks (low education). Given the time lag between early-life exposures and dementia risk in later-life, the role of early-life factors in dementia risk can be controversial. However, early-life experiences are important for developmental processes including brain development and for building cognitive and brain reserve.<sup>274</sup> Additionally, as proposed by the accumulative model of life course epidemiology<sup>6</sup> (Section 1.1.), life course risks can be downstream results of early-life and developmental factors. For example, poorer SEP in childhood is linked with greater adversities including poorer housing quality, diet, and reduced access to education and healthcare, which are associated with poorer health including increased cardiovascular risks (e.g. obesity, hypertension) and riskier lifestyle behaviours (e.g. physical inactivity, smoking, high alcohol consumption).<sup>275,276</sup> Studies examining the effects of multiple dementia risks have not incorporated such early-life predictors of dementia risks in their analyses, owing to a lack of available prospective early-life data. It is also unclear whether *APOE* genotype modifies the extent to which each modifiable risk factor contributes to dementia risk, which is important to consider given greater ε4 associated dementia risk in females than males.<sup>95</sup>

Existing studies imply that exposure to increasing numbers of modifiable dementia risk factors associates with poorer cognitive and brain health outcomes, prior to dementia onset, and there is preliminary evidence that such adverse effects of multiple risk exposures could be exacerbated in females. However, existing studies have only measured risks present at a single timepoint, not considering the potential cumulative impacts of risks experienced throughout the whole life course. Additionally, the extent to which early-life factors predict subsequent exposures to dementia risk factors and explain the associations of such risk factors with later-life outcomes remains unclear. Sex differences in how multiple risk factor exposures associate with later-life

cognitive and brain health outcomes remain largely understudied, and the mechanisms underlying risk factor associations with outcomes in males and females are unclear.

#### 4.1.1. Objectives and research questions

This work capitalises on the broad range of prospective life course data available within NSHD, which provides a unique opportunity to examine early determinants of lifetime risk exposures. A cumulative measure of lifetime modifiable dementia risk factor exposures (a cumulative risks score/CRS) spanning from birth to age 69 is derived in an age-homogenous population-based sample. CRS is based on the twelve risks outlined in the 2020 Lancet commission and weighted according to the Lancet risk-specific population PAFs.<sup>56</sup> The overall aims are to test whether cumulative risk exposures and early-life determinants of CRS differ between males and females, and whether there are sex differences in how cumulative risks associate with later-life cognitive and brain health measures, reflecting AD (A $\beta$ , hippocampal volume), vascular (WMHV), and nonspecific ageing (TBV) pathways to dementia. Whether early-life (childhood cognition, childhood SEP, puberty timing) and genetic (APOE-E4 status) factors predict CRS is examined, and the extent to which these predictors contribute to associations of CRS with later-life outcomes in males and females is assessed, including whether associations are modified by APOE-E4 status. Additionally, whether use of risk-modifying treatments (anti-hypertensive medication, hearing aids, diabetes medication, anti-depressant medication) mitigate associations is considered. Secondary analyses examine whether associations of CRS with later-life outcomes are detected when risks are measured up to midlife (age 53) rather than up to age 69, and whether there are associations of a non-weighted lifetime CRS with later-life cognitive and brain health outcomes.

The following research questions are addressed:

#### Part A – CRS distribution and predictors:

1) What is the distribution of lifetime CRS in a population-based cohort, including the contribution of individual risk factors, in males and females?

2) How are early-life factors (childhood cognition, childhood SEP, puberty timing, APOE-ε4 genotype) related with lifetime CRS in males and females?

# Part B – Associations of CRS with later-life cognitive performance and brain health

- 1) Does lifetime CRS associate with later-life cognitive performance and brain health, and do associations differ between males and females?
- 2) Are associations of CRS with later-life outcomes modified by APOE-E4 carrier status?
- 3) Do associations remain after accounting for early-life predictors of lifetime CRS?
- 4) Does the use of risk-modifying treatments (hearing aids, anti-hypertensive medications, anti-depressant medications, diabetes medications) mitigate relationships of lifetime CRS with later-life cognition and brain health?

# Part C – Secondary analyses

- 1) Do key associations of CRS with cognitive and brain health outcomes remain when using a risk score capturing risk exposures up to midlife (age 53)?
- 2) Does the pattern of findings remain when using a non-weighted CRS?

# 4.1.2. Hypotheses

Females were expected to have lower CRS than males, given previous cross-sectional quantification of some of the Lancet risks.<sup>111</sup> Associations of CRS with later-life cognitive performance and brain health measures were expected to indicate adverse effects of increased risk factor exposures, to a greater extent in females. Adverse early-life factors (e.g. poorer childhood SEP, lower childhood cognition) were expected to associate with lower CRS, since downstream effects of early-life factors, for example childhood SEP, on lifetime health and lifestyle behaviours have previously been demonstrated.<sup>275,276</sup> Additionally, given associations of early-life factors with later-life outcomes (e.g. childhood cognition predicts later-life cognitive performance),<sup>159</sup> adjustments for early-life predictors of CRS were hypothesised to attenuate associations of CRS with later-life outcomes. *APOE*-ε4 carriers were anticipated to show greater

adverse associations of increased CRS with outcomes, given that ε4 is the risk allele for dementia (Section 1.10.1.).

#### 4.2. Analytic method

#### 4.2.1. Analytic sample

Males and females from NSHD were included in analyses if they had available data for cognitive measures assessed at age 69 (ACE-III total score, word list recall, timed letter search; maximum n=2140). Insight 46 participants were included if they had completed the PACC at data collection wave I (aged ~70 years) or if they had available data for the neuroimaging outcomes of interest (TBV, hippocampal volume, WMHV, amyloid SUVR; maximum n=502).

## 4.2.2. Lifetime cumulative risks score (CRS)

A cumulative score of risk factors experienced across the life course was derived based on available data within NSHD for the twelve modifiable risk factors identified in the 2020 Lancet commission:<sup>56</sup> low education, hearing loss, TBI, hypertension, high alcohol consumption, obesity, smoking, depression, social isolation, physical inactivity, air pollution, diabetes. To ensure equal variance in the number of times each risk could be counted, the data were collapsed into three timepoints; earlier-life (up to and including age 36 years), midlife (age 43 to 53 years), and later-life (age 60 to 69 years). For each risk, except for low education and TBI, the number of timepoints in which the risk was present was counted, giving a maximum 'score' of three per risk. Low educational attainment up to age 43 was assigned a score of three, with the assumption that education level (measured according to the Burnham scale, see Section 2.3.4.2.) remained low in later-life. Low education was indicated by having 'no qualifications attempted' by age 43, or by age 26 if data were unavailable at 43 (n=2). Ever having had a TBI up to age 60 was also assigned a score of three, since it was not possible to distinguish the age at which a TBI had occurred or the number of TBIs experienced.

To calculate a weighted lifetime CRS per individual, each risk factor score was multiplied by the risk-specific PAF outlined in the 2020 Lancet commission<sup>56</sup> (Table 11) and then summed across all twelve risk factors. This value was divided by the maximum possible PAF (1.2), to generate a final CRS between the values of 0 and 1, whereby 1 indicates that all risks were present at all timepoints (early-, mid-, and later-life). CRS was calculated only for participants who had available data at early-, mid-, and later-life timepoints for each risk (whole-NSHD n=1509, 51% female; Insight 46 n=389, 48% female). Details for how each risk was measured are presented in Table 11.

**Table 11.** Description of how each risk factor included in the cumulative risks score (CRS) was measured, defined, and scored.

Timepoint	Age at	Measure	Description	Scoring	
	measurement				
Low educatio	n			Weighting (PAF): 0.07	
Life course	Up to age 43	No qualifications	Based on self-reported highest qualification level achieved by age 43, or by age 26 if data were unavailable at 43, classified according to the Burnham Scale. <sup>221</sup>	Score 3 if: - Low education up to age 43	
Hearing impa	irment			Weighting (PAF): 0.08	
Earlier-life (≤ age 36)	Childhood (6-15 years)	Doctor assessed hearing as 'poor'	Doctor's assessed children's hearing at ages 6, 7, 11, and 15, giving a rating of either poor, good, or excellent.	Score 1 if: - Hearing rated as 'poor' at age 6, 7, 11, OR 15	
Midlife (age 43-53)	43	Difficulty following a conversation	Self-reported "Do you have great difficulty following a conversation if there is background noise, for example, a TV, radio or child playing (wearing your hearing aid)?"	Score 1 if: - Difficulty following a conversation at age 43, OR	
	53	Difficulty following a conversation	Self-reported "Do you find it very difficult to follow a conversation if there is background noise (without a hearing aid)?"	<ul> <li>Difficulty following a conversation OR hearing during testing at age 53</li> </ul>	
		Difficulty hearing during testing			
Later-life (age 60-69)	60-64	Difficulty following a conversation Difficulty hearing over the phone Difficulty hearing during testing	Self-reported "In the last 12 months have you had a problem with hearing conversation in a noisy room?" Self-reported "In the last 12 months have you had a problem with hearing over the phone?" Study nurse recorded whether the study member had hearing difficulty during testing.	<ul> <li>Score 1 if:</li> <li>Difficulty following a conversation OR hearing over the phone OR hearing during testing at age 60-64, OR</li> <li>Difficulty following a conversation OR hearing over the phone OR hearing</li> </ul>	
	69	Difficulty following a conversation Difficulty hearing over the phone Difficulty hearing during testing	Self-reported "In the last 12 months have you had difficulty hearing conversation in a noisy room?" Self-reported "In the last 12 months have you had difficulty hearing over the phone?" Study nurse recorded whether the study member had difficulty hearing during testing.	during testing at age 69	
Traumatic Bra	ain Injury (TBI)			Weighting (PAF): 0.03	
Life course	Up to age 53	Ever had a TBI	Self-reported "Have you ever been knocked unconscious by a blow to the head?"	Score 3 if: - Ever reported having lost consciousness	
	Up to age 60-64	Had a TBI since age 53	Self-reported "Since 1999 have you been knocked unconscious?" Responses to this question were used to update information obtained at age 53, to indicate whether participants had ever had a TBI up to age 60-64.	after a head injury, up to age 60-64	

Hypertension	ı			Weighting (PAF): 0.02
Earlier-life (≤ age 36)	36	BP >140/90mmHg	Blood pressure was measured twice using a Hawksley random zero sphygmomanometer with regular (12×23 cm) upper arm cuff. A correction was made for arm circumference.	Score 1 if: - BP >140/90mmHg at age 36
Midlife (age	43	BP >140/90mmHg	Blood pressure was measured as at age 36.	Score 1 if:
43-53)		Self-reported high BP	"Have you had high blood pressure?"	- BP >140/90mmHg OR self-reported
	53	BP >140/90mmHg	Blood pressure was measured twice, while seated and after 5 minutes of rest, using the validated automated Omron HEM-705 (Omron Corp., Tokyo, Japan) digital oscillometric sphygmomanometer.	hypertension at age 43, OR - BP >140/90mmHg OR self-reported hypertension at age 53
		Self-reported high BP	Indicates "hypertension" in response to "Have you had any kind of blood pressure problems in the last 10 years?"	-
Later-life	60-64	BP >140/90mmHg	Blood pressure was measured as at age 53.	Score 1 if:
(age 60-69)		Self-reported Doctor diagnosed high BP	"Has a doctor told you that you have high blood pressure?"	<ul> <li>BP &gt;140/90mmHg OR self-reported hypertension at age 60-64, OR</li> </ul>
69		BP >140/90mmHg	Blood pressure was measured 3 times, while seated and after 5 minutes of rest, using the validated automated Omron 907-HEM digital oscillometric sphygmomanometer.	<ul> <li>BP &gt;140/90mmHg OR self-reported hypertension at age 69</li> </ul>
		Self-reported Doctor diagnosed high BP	Indicated "hypertension" in response to "Since 2006 have you been told by a doctor that you have blood pressure problems?"	-
High alcohol	consumption			Weighting (PAF): 0.01
Earlier-life (≤ age 36)	36	Problematic drinking	A question included in the present state examination (PSE) <sup>277</sup> asked "Is alcohol in any way a problem for you?"	Score 1 if: - Responded 'yes' to problematic drinking
		Alcohol units >21/week <sup>56</sup>	Average daily alcohol units were recorded in diet diaries. This was converted into weekly alcohol units.	OR >21 units/week at age 36
Midlife (age 43-53)	43	CAGE <sup>278</sup> score ≥2	4-item screening tool for alcohol use disorder scored from 0-4, where scores of 2 or more indicate possible problem drinking.	Score 1 if: - CAGE ≥2 OR >21 units/week at age 43, OR
		Alcohol units >21/week	Participants self-reported how many measures of spirits, wine, and beer they had consumed in the past 7 days. This was later converted to an overall measure of alcohol units consumed within the 7 days.	<ul> <li>CAGE ≥2 OR &gt;21 units/week at age 53</li> </ul>
	53	CAGE score ≥2	As at age 43.	-
		Alcohol units >21/week	As at age 43.	-
Later-life	60-64	CAGE score ≥2	As at age 43 and 53.	Score 1 if:
(age 60-69)		Alcohol units >21/week	As at age 43 and 53.	

	69	Alcohol Use Disorders Identification Test <sup>279</sup> (AUDIT) score ≥20	20-item questionnaire scored from 0-40, where scores of 20 or more indicate possible alcohol dependence.	<ul> <li>CAGE ≥2 OR &gt;21 units/week at age 60- 64, OR</li> <li>AUDIT ≥20 OR &gt;21 units/week at age 6</li> </ul>
		Alcohol units >21/week	As at age 43, 53, and 60-64.	
Obesity				Weighting (PAF): 0.01
Earlier-life (≤ age 36)	26	BMI ≥ 30.0	BMI calculated from self-reported weight (stones, later converted to kg) divided by height (feet, later converted to m) squared: kg/m <sup>2</sup>	Score 1 if: - BMI ≥ 30.0 at age 26 OR 36
	36	BMI ≥ 30.0	BMI calculated from weight (kg) and height (m) measured at study visit: kg/m <sup>2</sup>	
Midlife (age	43	BMI ≥ 30.0	As at age 36.	Score 1 if:
43-53)	53	BMI ≥ 30.0	As at age 36 and 43.	<ul> <li>BMI ≥ 30.0 at age 43 OR 53</li> </ul>
Later-life	60-64	BMI ≥ 30.0	As at age 36, 43, and 53.	Score 1 if:
(age 60-69)	69	BMI ≥ 30.0	As at age 36, 43, 53, and 60-64.	- BMI ≥ 30.0 at age 60-64 OR 69
Smoking				Weighting (PAF): 0.05
Earlier-life	20	Current smoker	Self-reported smoking status as current, ever, or never.	Score 1 if:
(≤ age 36)	25	Current smoker	As at age 20.	- Current smoker at age 20, 25, 31, OR 36
	31	Current smoker	As at age 20 and 25.	
	36	Current smoker	As at age 20, 25, and 31.	
Midlife (age	43	Current smoker	As at age 20, 25, 31, and 36.	Score 1 if:
43-53)	53	Current smoker	As at age 20, 25, 31, 36, and 43.	- Current smoker at age 43 OR 53
Later-life	60-64	Current smoker	As at age 20, 25, 31, 36, 43, and 53.	Score 1 if:
(age 60-69)	68	Current smoker	As at ages 20, 25, 31, 36, 43, and 53.	- Current smoker at age 60-64 OR 68
Depression				Weighting (PAF): 0.04
Earlier-life	Childhood (age 13 and 15)ª	Internalising rating ≥2.085 <sup>280</sup>	Sum of teacher ratings for emotional problems at 13 and	Score 1 if:
(= 050 50)	36	PSE ≥5	Present State Examination, <sup>277</sup> scored 0 to 28.	OR - PSE ≥5 at age 36
Midlife (age 43-53)	43	PSF ≥23	Psychiatric Symptom Frequency scale <sup>281</sup> (excluding suicide question), scored 0-90	Score 1 if: - PSF ≥23 at age 43, OR
	53	GHQ ≥5	GHQ-28, <sup>282</sup> scored 0-28	- GHQ ≥5 at age 53
Later-life	60-64	GHQ ≥5	GHQ-28, scored 0-28	Score 1 if:
(age 60-69)	69	GHQ ≥5	GHQ-28, scored 0-28	- GHQ ≥5 at age 60-64,
		Self-reported depression	Since 2006, have you had depression?	<ul> <li>GHQ ≥5 OR self-reported depression at age 69</li> </ul>
Social isolation	on			Weighting (PAF): 0.04
Earlier-life (≤ age 36)	26	Unmarried and living alone	Participants self-reported their marital status and how many people, including themselves, lived in their household.	Score 1 if: - Unmarried and living alone at age 26, OR - Unmarried and living alone at age 36

	36	Unmarried and living alone	As at age 26.	
Midlife (age	43	Unmarried and living alone	As at age 26 and 36.	Score 1 if:
43-53)		Small social network size	Self-reported "How many friends or relatives do you meet	<ul> <li>Unmarried and living alone OR small</li> </ul>
		(None)	and talk to socially on a regular basis?"	network at age 43, OR
	53	Unmarried and living alone	As at age 26, 36, and 43.	<ul> <li>Unmarried and living alone OR small</li> </ul>
		Small social network size	Self-reported "How many friends or relatives do you see	network at age 53
		(None)	once a month or more?"	
Later-life	60-64	Unmarried and living alone	As at age 26, 36, 43, and 53.	Score 1 if:
(age 60-69)		Small social network size	Self-reported "How many relatives or friends do you see	<ul> <li>Unmarried and living alone OR small</li> </ul>
		(None	once a month or more?"	network OR low contact frequency at
		Low social contact frequency	Self-reported "Thinking of all your relatives or friends,	age 60-64, OR
		(Never/almost never)	how often do you regularly visit or are visited by these people?"	<ul> <li>Unmarried and living alone OR small network OR low contact frequency at</li> </ul>
	69	Unmarried and living alone	As at age 26, 36, 43, 53, and 60-64.	age 69
		Small social network size	Self-reported "How many relatives and/or friends do you	-
		(None)	see once a month or more?"	
		Low social contact frequency	Self-reported "Thinking of all you friends/relatives, how	
		(Never/almost never)	often do you regularly visit or are visited by any of these	
			people?"	
Physical inact	tivity			Weighting (PAF): 0.02
Earlier-life	36	No participation in physical	Self-completed modified validated Minnesota leisure time	Score 1 if:
(≤ age 36)		activity/month	physical activity questionnaire (frequency of participation	<ul> <li>No physical activity/month at age 36</li> </ul>
			in a range of physical activities per month). <sup>227</sup>	
Midlife (age	43	No participation in physical	EPIC Physical Activity Questionnaire-2 (frequency of	Score 1 if:
43-53)		activity/month	participation in sport, vigorous leisure activities or	<ul> <li>No physical activity/month at age 36 OR</li> </ul>
			exercise in the previous month). <sup>227</sup>	53
	53	No participation in physical	EPIC Physical Activity Questionnaire-2 (frequency of	
		activity/month	participation in sport, vigorous leisure activities or	
	60.64	N	exercise in the previous 4 weeks). <sup>227</sup>	C 4:5
Later-life	60-64	No participation in physical	As at age 53.	Score 1 if:
(age 60-69)			As at any 52 and 50 54	- No physical activity/month at age 60-64
	09	No participation in physical	AS at age 53 and 60-64.	OK 69
Exposure to b	high air pollution <sup>b</sup>	activity/month		Woighting (DAE): 0.02
Exposure to r	Childhood (hirth	Howw pollution	Pollution index based on demostic local coal	Score 1 if:
Larner-line		neavy policion	consumption, categorised as beauty or low-	- High pollution index in childhood
( <u>s</u> age 50)			moderate. <sup>283,284</sup>	
Midlife (age	43	Geocoded residential NO <sub>2</sub>	1991 NO <sub>2</sub> air pollution maps, based on a land use	Score 1 if:
43-53)		estimate > 41.5 μg/m <sup>3</sup> , as	regression (LUR) model from the Automatic Urban and	<ul> <li>NO<sub>2</sub> estimate &gt; 41.5 µg/m<sup>3</sup> at age 43 OR</li> </ul>
		10, ,		

		indicated in the Lancet Commission. <sup>56</sup>		_
	53	Geocoded residential NO <sub>2</sub> estimate > 41.5 μg/m <sup>3</sup>	2001 NO <sub>2</sub> air pollution maps, based on a LUR model from the RGI project.	_
Later-life (age 60-69)	64	Geocoded residential NO <sub>2</sub> estimate > 41.5 µg/m <sup>3</sup>	2010 NO <sub>2</sub> air pollution maps, based on a LUR model from the European Study of Cohorts for Air Pollution Effects (ESCAPE).	Score 1 if: - NO₂ estimate > 41.5 µg/m³ at age 64
Diabetes <sup>c</sup>				Weighting (PAF): 0.01
Earlier-life (≤ age 36)	Up to age 36	Self-reported diabetes up to age 36	"Have you had diabetes?"	Score 1 if: - Report diabetes up to age 36
Midlife (age 43-53)	Up to age 43	Self-reported diabetes up to age 43	"Have you had diabetes?"	Score 1 if: - Report diabetes up to age 36 OR up to
	Up to age 53	Self-reported diabetes up to age 53	"Diabetes at any point up to and including 1999"	age 43 OR up to age 53
Later-life (age 60-69)	Up to age 60-64	Self-reported diabetes diagnosed by GP at age 60- 64, or self-reported at up to 53	"Diabetes diagnosed by GP"	<ul> <li>Score 1 if:</li> <li>Report diabetes up to age 36 OR up to age 43 OR up to age 53 OR up to age 60-64 OR up to age 69</li> </ul>
	Up to age 69	Self-reported Doctor diagnosed diabetes at age 69, or self-reported up to 60-64	"Ever self-reported doctor diagnosed diabetes"	_

PAF=population attributable fraction; TBI=traumatic brain injury; BP=blood pressure; PSE=present state examination; AUDIT=alcohol use disorders identification test; BMI=body mass index; PSF=psychiatric symptom frequency scale; GHQ=general health questionnaire; LUR=land use regression; AURN=automatic urban and rural network; RGI=Ruimte voor Geoinformatie

a: At ages 13 and 15, teachers were asked to complete questionnaires (a forerunner of the Rutter A scale<sup>285</sup>) giving ratings for the study members' internalising problems (anxiety, timidity, fearfulness, diffidence, avoidance of attention).<sup>280</sup> Ratings from age 13 and 15 were summed, and childhood internalising caseness was indicated by an overall rating of 2.085 or more, based on previous NSHD studies.<sup>286</sup>

b: In adulthood, estimated nitrogen dioxide (NO<sub>2</sub>) levels – identified as a key pollution measure in the Lancet commission<sup>56</sup> – were derived from land use regression (LUR) models. For NO<sub>2</sub> (µg/m<sup>3</sup>) at age 43 (1989), 1991 Great Britain-wide NO<sub>2</sub> air pollution maps (200m x 200m; based on a LUR model using data from the Automatic Urban and Rural Network/AURN, Defra) were overlaid on maps of participants residential addresses in 1989. For NO<sub>2</sub> at age 53 (1999), 2001 air pollution maps (100m x 100m) based on a LUR models from the RGI project (grant RGI-137, Dutch program Ruimte voor Geoinformatie) were overlaid on maps of participants's residential addresses in 1989. NO<sub>2</sub> concentrations for participants geocoded addresses in 2010 were estimated from a LUR model derived as part of the European Study of Cohorts for Air Pollution Effects (ESCAPE).<sup>287</sup>

c: Since diabetes is non-curable, presence of diabetes at an earlier age also indicated persistent presence of the risk at subsequent timepoints.

#### 4.2.3. Cognitive outcomes

At age 69, the cognitive outcomes used for these analyses were the ACE-III total score, word list learning (verbal memory), and timed letter search (search speed) measures. For the Insight 46 sample at age ~70, cognitive performance was assessed using the PACC total score. Raw scores for each assessment were standardised to the analytical sample, generating z-scores. Further details for each cognitive assessment have previously been outlined in Section 2.3.2. (Tables 1 and 2).

#### 4.2.4. Neuroimaging outcomes

As outlined in Section 2.3.3., Insight 46 participants underwent PET-MRI neuroimaging at age  $\sim$ 70. For these analyses, the neuroimaging outcomes of interest were TBV, hippocampal volume, WMHV, continuous A $\beta$  SUVR, and A $\beta$  positivity status. For WMHV, participants who failed BaMoS QC were excluded (n=16; 56.3% female).

#### 4.2.5. Early-life and genetic covariables

Variables hypothesised to associate with CRS and later-life outcomes are childhood cognition, childhood SEP, *APOE*- $\epsilon$ 4 status (each described in Section 2.3.4.), and puberty timing. Childhood cognition and SEP, previously associated with later-life cognitive performance,<sup>159,205</sup> have been shown to be predictive of general health and health behaviours in adulthood.<sup>275,276</sup> Given the role of sex hormones in brain organisation, neural development and the cardiovascular system, puberty timing – when sex hormone levels surge – is hypothesised to contribute to cognitive and brain reserve processes, and to variations in cardiovascular health.<sup>81,83,84</sup> The *APOE*- $\epsilon$ 4 allele is strongly associated with AD dementia,<sup>288</sup> more so in females,<sup>95</sup> and has been linked with increased risk of cardiovascular disease.<sup>289</sup>

Puberty timing was determined from age at menarche for females, grouped as very early (age  $\leq 11$  years), early (age 12), average (age 13), or late (age  $\geq 13$ ), as outlined in previous NSHD analyses.<sup>290</sup> A binary variable indicated later puberty for menarche at age 13 or older, and earlier puberty for menarche younger than age 13. Male puberty timing was ascertained via school doctor assessment at age 14-15, rating development as pre-pubertal, early puberty,

advanced puberty, or fully mature, whereby those assessed as fully mature had the earliest puberty timing.

Childhood cognition at age 8 years was standardised to the sample at the time of cognitive testing. Childhood SEP up to age 15, was categorised as manual or non-manual (Section 2.3.4.3.). *APOE*-ε4 carrier status indicated ε4 carriers and non-carriers (Section 2.3.4.1.).

Risk-modifying treatments including hearing aid use and anti-hypertensive, anti-depressant, and diabetes medications were considered as potentially mediating factors. Summary variables indicating ever or never use of each of these risk-modifying covariables were derived from life course data. At ages 11 and 15, hearing aid use was reported during doctor examinations. At ages 43, 53 and 69, participants self-reported if they wore a hearing aid. Anti-hypertensive medications, prescribed for high blood pressure or heart failure, were self-reported by participants at home visits or via postal questionnaire at ages 36, 43, 53, 60-64, and 69. Anti-depressant medications, including tricyclics, monoamine-oxidase inhibitors, and selective serotonin reuptake inhibitors were self-reported via home visits or postal questionnaires at ages 31, 36, 43, 53, 60-64, and 69. Diabetic medications, including insulin and oral diabetic agents, treatments for low blood sugar, and diabetic testing kits, were self-reported at home visits or on postal questionnaires at ages 31, 36, 43, 53, 60-64, and 69. For all medication types, use was categorised as using or not using treatment for each assessment.

#### 4.2.6. Statistical analyses

Analyses were conducted using Stata version 18.0.

As presented in Tables 12 and 13, the mean, standard deviation, and range of continuous variables were calculated for males and females. T-tests and chi-squared tests examined sex differences in continuous and categorical variables, respectively, including examination of sex differences in total CRS and in mean scores for each risk factor. All analyses were sex-stratified, to determine how associations of CRS, later-life outcomes, and early-life covariables might differ between males and females.

The association of each early-life covariable with CRS was examined using unadjusted linear regression models, in the whole-cohort and Insight 46 samples.

Associations of CRS with cognitive outcome measures, TBV, hippocampal volume, and continuous SUVR were assessed using linear multivariable regression models. Multivariable logistic regression models examined associations of CRS with Aβ status, and WMHV was assessed using generalised linear models with gamma distribution log link, due to the skewed distribution of WMHV.

Associations of CRS with later-life cognitive performance and brain health outcomes were first examined in minimally adjusted models (M0). Models assessing cognitive outcomes at age 69 were completely unadjusted, while those assessing Insight 46 outcomes were adjusted for age at cognitive testing (PACC) or age at scan (neuroimaging outcomes), given the two-year range required to complete Insight 46 data collection. Additionally, models examining volumetric neuroimaging outcomes (TBV, hippocampal volume, WMHV) were adjusted for TIV.

Potential modifying effects of *APOE*-ɛ4 genotype on associations of CRS with later-life outcomes were examined by including a CRS-by-*APOE*-ɛ4 status interaction term in minimally adjusted models, interpreting significant interactions at the p<0.10 level. To determine whether associations of CRS and later-life outcomes were independent of early-life predictors of CRS, subsequent models cumulatively adjusted for early-life covariables found to associate with CRS. Where associations of CRS with outcomes were detected, post-estimation tests of difference assessed whether model coefficients significantly differed between males and females.

For outcomes found to associate with CRS, fully adjusted models were further adjusted for the use of risk-modifying treatments individually (fully adjusted model + hearing aids; fully adjusted model + anti-hypertensive medications; fully adjusted model + anti-depressant medications; fully adjusted model + diabetes medications) and all together (fully adjusted model + all risk-modifying treatments), to identify whether such treatments impact the associations of risk factor exposures with later-life outcomes.

To address missing data in CRS and covariables, multiple imputation was used (Section 2.4.), with imputation models stratified by sex. For models assessing the ACE-III outcome, which has a skewed distribution, bootstrapping was applied (Section 2.4.); multiple imputation was not applied for ACE-III models.

125

Non-linear associations of CRS with key outcomes (ACE-III total in the whole sample; TBV in Insight 46) were assessed by including a quadratic term for CRS in fully adjusted models. No evidence for non-linear associations was found (p>0.1).

#### 4.2.7. Secondary analyses

To address the issue of reverse directionality from some risk factor variables which can be a part of the dementia prodrome (e.g. low BMI, depression)<sup>291,292</sup> and given an interest in midlife as a period of vulnerability, especially for females,<sup>8</sup> cumulative risk exposures up to midlife (age 53) were also calculated. Midlife CRS was ascertained using the same criteria as lifetime CRS, but each risk factor was scored from 0 to 2, based on data recorded in early-life (up to age 36) and midlife (ages 43 and 53) only. The PAF-weighted risk-specific scores were again summed across all twelve risk factors and divided by the new maximum possible PAF (0.8), generating a midlife CRS between the values of 0 and 1. Midlife CRS was calculated only for participants who had available data at early- and mid-life timepoints for each risk (whole-NSHD n=1,604, 52% female; Insight 46 n=420, 49% female). Fully adjusted sex-stratified multivariable models examined the associations of midlife CRS with outcome variables that were significantly associated with lifetime CRS.

While the PAF weighting applied to each risk factor allows for variances in the effects of each risk, the PAFs calculated in the Lancet commission may not be accurate and could also vary by sex. A non-weighted lifetime CRS was therefore also calculated, summing the risks present across each timepoint (early-, mid-, and later-life). This sum of risks was divided by 36 (the maximum possible risk count) to generate a value between 0 and 1. Fully adjusted sex-stratified multivariable models examined associations of non-weighted lifetime CRS with cognitive performance at age 69 (ACE-III, verbal memory, search speed), and neuroimaging outcome measures aged ~70 (TBV, hippocampal volume, WMHV, SUVR).

# 4.3. Results

# 4.3.1. Participant characteristics

Participant characteristics for the whole-cohort sample, with available data for at least one cognitive outcome at age 69 (n=2140; 51.1% female), are presented in Table 12. T-tests

demonstrated that females performed better than males on the word list learning and timed letter search tasks, but there were no sex differences in total ACE-III performance, in agreement with findings from Chapter 3 (Section 3.0.). Chi-squared tests showed that males had higher levels of anti-hypertensive medication use, while females were more likely to have taken anti-depressant medications. Puberty timing was also more likely to have been classed as later in females (first period at 12.83 years or older) than in males (pre-pubertal or in early puberty at age 14-15), although differences in how puberty is measured between males and females mean that this is not a direct comparison.

Table 13 presents participant characteristics for the 502 (48.3% female) participants in the Insight 46 sample. Unadjusted t-tests showed greater PACC performance in females than males, greater TBV, hippocampal volume and TIV in males, and no sex differences in WMHV or amyloid measures. As in the whole-cohort, chi-squared tests demonstrated greater anti-hypertensive medication use in males, greater anti-depressant use in females, and later puberty timing in females.

#### 4.3.2. Lifetime cumulative risks score (CRS) distribution and characteristics

For the whole-cohort sample, lifetime CRS was calculated for 1,509 (51.0% female) participants who had available data at all timepoints, for all risk factors. Mean CRS was 0.25 (SD 0.13), indicating that participants on average were exposed to 25% of the maximum possible lifetime modifiable dementia risk. The CRS distributions for males and females are shown in Figure 8; female mean CRS was 0.25 (SD 0.13), which did not significantly differ from that of males (mean 0.26 (SD 0.13); p=0.188; Table 12). Figure 9 shows the distribution of scores for each risk factor, where higher scores indicate a greater number of timepoints in which the risk was present. T-tests examining sex differences in mean scores per risk showed that hearing impairment, TBI, hypertension, and high alcohol consumption were more often recorded in males than females, while females reported depression and physical inactivity more than males.

For the Insight 46 sample, CRS was calculated for 389 (48.3% female) participants, with a mean score of 0.21 (SD 0.10), indicating that participants on average were exposed to 21% of the maximum possible lifetime modifiable dementia risk, which was significantly lower than in the

whole-cohort sample (one-sample t-test p<0.001). The distributions of CRS by sex are presented in Figure 10; males had significantly higher scores (mean=0.23 (SD 0.10)) than females (mean=0.20 (SD 0.11); p=0.013; Table 13). As in the whole-cohort sample, hearing impairment, TBI, hypertension, and high alcohol consumption were more common in males than females while depression was more common in females, although within Insight 46 there was no sex difference in physical inactivity (Figure 11).

**Table 12.** Participant characteristics for NSHD participants with available data for at least one cognitive outcome measure at age 69.

		Male		Female	
	1,	047 (48.9%)	1,0	093 (51.1%)	
Variable	Ν	Mean(SD);	Ν	Mean(SD);	Test p-
		range/%		range/%	value <sup>a</sup>
Lifetime cumulative risks	740	0.26(0.13);	769	0.25(0.13);	0.188
score		0.00, 0.76		0.02, 0.68	
ACE-III total score	849	91.33(5.84);	913	91.69(6.17);	0.205
		53, 100		59, 100	
Word list learning (number	1,011	21.14(5.98);	1,063	23.15(5.99);	<0.001
of recalled words)		3, 39		6, 40	
Timed letter search	1,038	256.38(74.32);	1,073	268.02(73.56);	<0.001
(number of letters scanned)		70, 591		60, 591	
Hearing aid use					
Never	918	87.7%	985	90.1%	0.072
Ever	129	12.3%	108	9.9%	
Anti-hypertensive					
medication					
Never	677	64.7%	808	73.9%	<0.001
Ever	370	35.3%	285	26.51%	
Anti-depressant medication					
Never	945	90.3%	890	81.4%	<0.001
Ever	102	9.7%	203	18.6%	
Diabetes medication					
Never	951	90.8%	1,012	92.6%	0.140
Ever	96	9.2%	81	7.4%	
Puberty staging					
Earlier	492	56.5%	351	39.1%	<0.001
Later	379	43.5%	547	60.9%	
Childhood cognition (8yrs) <sup>b</sup>	970	0.089(0.83);	1,010	0.16(0.80);	0.079
		-1.90, 2.50		-2.12, 2.47	
Childhood SEP					
Manual	526	52.7%	542	52.8%	0.956
Non-manual	473	47.3%	485	47.2%	
APOE-ɛ4 status					
Non-carrier	624	68.1%	692	71.3%	0.137
Carrier	292	31.9%	279	28.7%	

<sup>a</sup>Continuous variables: p-value from linear regression t-test. Categorical variables: p-value from Pearson chi-squared test. Bold text indicates a significant (p<0.05) sex difference.

<sup>b</sup>Standardised (z-score) to sample at the time of cognitive testing.

*SD=standard deviation; ACE-III=Addenbrooke's cognitive examination 3<sup>rd</sup> edition; SEP=socioeconomic position; APOE=apolipoprotein-e.* 

		Male		Female	
Variables	N	201 (51.7%) Mean(SD)·	N	100 (48.3%) Mean(SD)·	Test n-
Vanabics		range/%		range/%	value <sup>a</sup>
Lifetime cumulative risk	201	0.23(0.10);	188	0.20(0.11);	
score		0.04, 0.50		0.02, 0.50	0.013
ACE-III total score	214	93.30(4.08);	207	93.46(5.10);	
		79, 100		70, 100	0.722
Word list learning (number	253	22.95(5.67);	237	24.48(5.76);	
of recalled words)		5, 38		9, 40	0.003
Timed letter search (number	252	262.87(66.05);	237	271.52(72.45);	
of letters scanned)		138, 487		138, 591	0.168
PACC	256	-0.17(0.71);	246	0.18(0.71);	
<b>TD</b> ) ( (	220	-3.49, 1.72	220	-3.48, 1.67	<0.001
IBV (cm <sup>2</sup> )	239	1,151.35(86.69);	229	1,046.75(82.49);	40.001
Llippo compolyclyma (cm <sup>3</sup> )	220	945.90, 1,493.86	220	818.60, 1,265.16	<0.001
hippocampai volume (cm <sup>-</sup> )	239	3.20(0.33); 2.42.4.27	229	3.00(0.30);	<0.001
$WMHV(cm^3)$	23/	<u> </u>	221	5 47(5 78)	10.001
	234	4.77(3.05), 0.27 33.67	221	0 35 32 79	0 174
SUVR	235	0.57(0.08):	227	0.56(0.07):	0.174
	200	0.45. 0.87	/	0.47. 0.85	0.329
Amyloid status				,	
Negative	188	80.0%	188	82.8%	0.436
Positive	47	20.0%	39	17.2%	
Hearing aid use					
Never	222	86.7%	221	89.8%	0.278
Ever	34	13.3%	25	10.2%	
Anti-hypertensive					
medications					
Never	170	66.4%	193	78.5%	0.003
Ever	86	33.6%	53	21.5%	
Anti-depressant medications	242		200	94.6%	-0.001
Never	242	94.5%	208	84.6% 15.4%	<0.001
Diabetes medication	14	5.570	50	15.470	
Never	238	93.0%	230	93 5%	0 814
Ever	18	7.0%	16	6.5%	0.011
Puberty staging				0.075	
Earlier	125	55.1%	95	43.6%	0.015
Later	102	44.9%	123	56.4%	
Childhood cognition (8yrs) <sup>b</sup>	256	0.36(0.75);	246	0.42(0.73);	
		-1.60, 2.50		-1.59, 2.47	0.353
Childhood SEP					
Manual	106	41.4%	110	45.6%	0.341
Non-manual	150	58.6%	131	54.4%	
APOE-e4 status	4				
Non-carrier	171	67.3%	181	73.6%	0.126
Carrier	83	32.7%	65	26.4%	

# Table 13. Participant characteristics for the Insight 46 sub-sample.

Age at scan (years)	241	70.66(0.68);	230	70.70(0.67);	
		69.25, 71.78		69.27, 71.86	0.860
Age at cognitive testing	256	70.64(0.69);	246	70.66(0.68);	
(years)		69.23, 71.73		69.27, 71.84	0.782
TIV (cm <sup>3</sup> )	239	1,519.62(105.93);	229	1,343.16(92.47);	
		1,274.26, 1,938.77		1,114.35, 1,558.05	<0.001

<sup>a</sup>Continuous variables: p-value from linear regression t-test. Categorical variables: p-value from Pearson chi-squared test. Bold text indicates a significant (p<0.05) sex difference.

<sup>b</sup>Standardised (z-score) to sample at the time of cognitive testing.

SD=standard deviation; PACC=pre-clinical Alzheimer's cognitive composite; TBV=total brain volume; WMHV=white matter hyperintensity volume; SUVR=standardised uptake value ratio; SEP=socioeconomic position; APOE=apolipoprotein-e; TIV=total intracranial volume.

# 4.3.3. Predictors of risk score

Unadjusted associations of early-life factors with lifetime CRS are presented in Table 14.

In the whole-cohort sample, males and females showed associations of higher childhood cognition and higher childhood SEP with lower CRS. In males, *APOE*- $\epsilon$ 4 carriers had lower CRS than non-carriers. While not statistically significant, females showed the same association of *APOE*- $\epsilon$ 4 with CRS. Puberty timing was not associated with CRS in males or females.

In the Insight 46 sample, no associations of early-life factors with CRS were found in males or females.

Childhood cognition, childhood SEP, and APOE-E4 are included as confounding variables in subsequent adjusted models.

**Table 14.** Model estimates and 95% confidence intervals showing the estimated effects of early-life covariables on cumulative risks score, from unadjusted linear regression models.

Cohort	Sex	Value	Childhood cognition (8yrs)ª	Childhood SEP	Puberty timing	<i>APOE</i> -ε4 status
Whole-	Males	β	-0.040	-0.060	0.016	-0.021
cohort	(n=1047)	95% CI	-0.050,	-0.077,	-0.003,	-0.040,
			-0.029	-0.043	0.034	-0.002
		p-value	<0.001**	<0.001**	0.10	0.03*
	Females	β	-0.046	-0.050	0.009	-0.019
	(n=1093)	95% CI	-0.057,	-0.068,	-0.011,	-0.038,
			-0.034	-0.033	0.028	0.001
		p-value	<0.001**	<0.001**	0.37	0.06
Insight	Males	β	0.006	-0.004	-0.008	0.007
46	(n=256)	95% CI	-0.012,		-0.037,	-0.022,
			0.024	-0.032, 0.023	0.020	0.035
		p-value	0.49	0.75	0.56	0.65
	Females	β	-0.008	-0.015	0.002	-0.014
	(n=246)	95% CI	-0.029,		-0.031,	-0.049,
			0.013	-0.047, 0.017	0.035	0.020
		p-value	0.44	0.35	0.89	0.42

\* p<0.05 \*\* p<.01

<sup>a</sup>Childhood cognition z-score is standardised to the sample at the time of cognitive testing.

Reference categories: childhood SEP = manual, puberty timing = earlier, APOE- $\varepsilon$ 4 status = non-carrier



*Figure 8.* Distribution of lifetime cumulative dementia risk scores in males (N=740) and females (N=769), in the whole-NSHD analytical sample.



**Figure 9.** Distribution of raw scores per risk factor, in males (N=740) and females (N=769) in the whole-NSHD analytical sample. The maximum score per risk is 3. Low education 'No'=0, 'Yes'=3; TBI 'Never'=0, 'Ever'=3; diabetes 'Never'=0, 'Since late-life'=1, 'Since mid-life'=2, 'Since early-life'=3; all other risks are scored 0 to 3 where 0 means that the risk is never present and 3 means that the risk is present at three timepoints. For data protection reasons, some categories have been collapsed. **A)** Mean scores for males and females. **B)** Percentage of males and females with different levels of risk factor exposures. TBI=traumatic brain injury \*\* t-test p<0.01



*Figure 10. Distribution of lifetime cumulative dementia risk scores in males (N=201) and females (N=188), in the Insight 46 analytical sample.* 



**Figure 11.** Distribution of scores per risk factor, in males (N=201) and females (N=188) in the Insight 46 analytical sample. The maximum score per risk is 3. Low education 'No'=0, 'Yes'=3; TBI 'Never'=0, 'Ever'=3; diabetes 'Never'=0, 'Since late-life'=1, 'Since mid-life'=2, 'Since early-life'=3; all other risks are scored 0 to 3 where 0 means that the risk is never present and 3 means that the risk is present at three timepoints. For data protection reasons, some categories have been collapsed. For diabetes, no participants had diabetes in early-life. A) Mean scores for males and females. B) Percentage of males and females with different levels of risk factor exposures. TBI=traumatic brain injury \* t-test p<0.05; \*\* t-test p<0.01

4.3.4. Associations of lifetime CRS with later-life cognitive performance and brain health measures

#### 4.3.4.1. Cognitive performance measures age 69

Table 15 shows the estimated associations of lifetime CRS with each outcome measure in males and females. There was no evidence for effect modifications by *APOE*-ε4 status (Table 16).

With minimal adjustments, each unit increase in CRS associated with a 1.92 SD reduction in male verbal memory performance (95% CI -2.44, -1.40) and a 2.54 SD reduction in female verbal memory performance (95% CI -3.01, -2.07). Associations remained with adjustments for early-life predictors of CRS, although there was some attenuation, mainly driven by childhood cognition; adjustment for childhood cognition (M1) attenuated the estimated effect of CRS on verbal memory performance by 39% in males and 33% in females. Fully adjusted model effects are demonstrated in Figure 12; there was a trend-level difference in male and female effect estimates (F=3.25, p=0.07).

In males, increased CRS significantly associated with poorer search speed performance in males (M0  $\beta$ =-1.79[95% CI -22.34, -1.25]) but not in females (M0  $\beta$ =-0.50[-1.02, 0.04]). The association in males remained with adjustment for early-life predictors, although there was a 9% attenuation when accounting for childhood cognition. Fully adjusted effect estimates significantly differed between males and females (F=10.32, p<0.001), shown in Figure 12.

Increased CRS associated with poorer ACE-III performance in males (M0  $\beta$ =-2.24[bias corrected 95% CI -2.84, -1.64]) and females (M0  $\beta$ =-2.42[bias corrected 95% CI -3.16, -1.9]). There was some effect attenuation across adjusted models, with effects attenuating from M0 to M1 (childhood cognition) by 31% in males and 48% in females, but associations remained independently of early-life predictors of CRS. Fully adjusted model predictions, which did not substantially differ between males and females<sup>a</sup>, are shown in Figure 12.

<sup>&</sup>lt;sup>a</sup> Post-estimation test was not possible with bootstrapping applied. Running the post-estimation on multiple imputation estimates, without bootstrapping, showed no significant difference in estimates between males and females (F=0.08, p=0.78).

#### 4.3.4.2. Insight 46 cognitive performance and neuroimaging outcomes aged ~70

Model estimations for the effect of lifetime CRS on outcomes are presented in Table 15. There was no evidence for effect modifications by *APOE*-ɛ4 status for any cognitive or neuroimaging outcomes (Table 16).

No association of CRS with PACC performance was found in males or females.

In males, each unit increase in CRS significantly associated with a 75.82cm<sup>3</sup> reduction in TBV (M0, 95% CI -138.44, -13.19), while females showed a trend-level association of increased CRS with reduced TBV (M0  $\beta$ =-59.74[-118.96, -0.51]). The association in males remained independently of early-life predictors of CRS. Fully adjusted model predictions, which did not significantly differ between males and females (F=0.20, p=0.66) are shown in Figure 13.

Males also showed a significant association of increased CRS with reduced hippocampal volume (M0  $\beta$ =0.69cm<sup>3</sup>[-1.13, -0.24]), while no association was found in females. The male association remained significant with early-life covariable adjustments. Fully adjusted model predictions, which did not significantly differ between males and females (F=1.68, p=0.20), are demonstrated in Figure 13. To determine whether brain volume reductions associated with CRS were generalised across the brain, models of hippocampal volume were adjusted for TBV as a sensitivity analysis. This slightly attenuated the estimated effects of CRS on hippocampal volume (male M3+TBV  $\beta$ =-0.520[-0.959, -0.098], p=0.02; female M3+TBV  $\beta$ =-0.175[-0.536, 0.185], p=0.34).

No associations of CRS with WMHV or amyloid SUVR or  $A\beta$  status were detected in males or females.

**Table 15.** Model estimates and 95% confidence intervals for the association of cumulative risks score with standardised cognitive performance outcomes and neuroimaging outcomes, cumulatively adjusting for confounding variables.

			м	ales			Fen	nales	
Outcome		M0	M1	M2	M3	M0	M1	M2	M3
Verbal	N		10	011			10	)63	
memory z-	β	-1.921	-1.167	-1.052	-1.051	-2.539	-1.703	-1.617	-1.674
score	95%	-2.439,	-1.664,	-1.553,	-1.553,	-3.013,	-2.153,	-2.069,	-2.121,
	CI	-1.403	-0.671	-0.551	-0.549	-2.065	-1.253	-1.166	-1.226
	р	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**
Search speed	N		10	038			10	)73	
z-score	β	-1.794	-1.634	-1.584	-1.571	-0.495	-0.303	-0.256	-0.248
	95%	-2.338,	-2.205,	-2.165,	-2.154,	-1.024,	-0.854,	-0.813,	-0.807,
	CI	-1.251	-1.064	-1.003	-0.988	0.035	0.248	0.301	0.310
	р	<0.001**	<0.001**	<0.001**	<0.001**	0.07	0.28	0.37	0.38
ACE-III total z-	N	595	585	578	525	644	641	630	575
score <sup>a</sup>	β	-2.237	-1.539	-1.407	-1.634	-2.420	-1.258	-1.231	-1.349
	95%	-2.835,	-2.11,	-1.992,	-2.299,	-3.161,	-1.786,	-1.836,	-1.925,
	CI	-1.647	-0.979	-0.827	-1.084	-1.786	-0.746	-0.718	-0.763
	р	*	*	*	*	*	*	*	*
PACC z-score	N		2	56			2	46	
	β	-0.067	-0.188	-0.129	-0.145	-0.404	-0.247	-0.197	-0.222
	95%	-1.075,	-1.128,	-1.053,	-1.065,	-1.324,	-1.084,	-1.039,	-1.063,
	CI	0.941	0.753	0.794	0.775	0.516	0.591	0.645	0.619
	р	0.90	0.69	0.78	0.76	0.39	0.56	0.65	0.60
TBV	N		2	39			2	29	
	β	-75.815	-75.352	-75.612	-76.941	-59.737	-59.869	-58.843	-58.362
	95%	-138.436,	-138.156,	-138.658,	-140.010,	-118.964,	-119.376,	-118.324,	-118.120,
	CI	-13.194	-12.548	-12.565	-13.872	-0.511	-0.362	0.638	1.397
	р	0.02*	0.02*	0.02*	0.02*	0.05	0.05	0.05	0.06
Hippocampal	N		2	.39			2	29	
volume	β	-0.686	-0.699	-0.701	-0.704	-0.292	-0.287	-0.282	-0.292
	95%	-1.132,	-1.147,	-1.150,	-1.154,	-0.656,	-0.652,	-0.646,	-0.658,
	CI	-0.241	-0.251	-0.251	-0.255	0.072	0.078	0.083	0.074
	р	<0.001**	<0.001**	<0.001**	<0.001**	0.12	0.12	0.13	0.12
WMHV	N		2	.34			2	21	
	β	1.226	1.121	1.121	1.138	1.929	1.929	1.890	1.891
	95%	0.261,	0.239.	0.238,	0.245,	0.471,	0.468,	0.451,	0.450,
	CI	5.750	5.255	5.275	5.287	7.910	7.954	7.915	7.951
<u></u>	р	0.80	0.89	0.89	0.87	0.36	0.36	0.38	0.38
SUVR	N	0.000	0.000	.35	0.001	0.001	2	27	0.014
	p orv	-0.083	-0.086	-0.089	-0.091	0.001	-0.001	-0.001	0.014
	95%	-0.194,	-0.197,	-0.201,	-0.200,	-0.094,	-0.095,	-0.096,	-0.075,
	CI	0.028	0.025	0.023	0.018	0.096	0.094	0.094	0.103
AQ atatush	P	0.14	0.13	0.12	0.10	0.98	0.99	0.99	0.75
Ap status"		0.261	0.246	.35	0.190	2 072	2 001	2011	6 5 4 7
		0.261	0.240	0.241	0.180	3.072	3.091	3.011	0.547
	95%	0.007,	0.006,	0.006,	0.003,	0.101,	0.100,	0.097,	0.151,
	5	9.780	9.318	9.307	9.485	93.451	92.122	93.235	284.099
	р	0.47	0.45	0.44	0.40	0.52	0.52	0.53	0.33

<sup>a</sup>Bootstrapping applied: confidence intervals are bias-corrected, p values are not applicable, models are not imputed.

<sup>b</sup>Logistic regression, coefficient presented is an OR for being amyloid positive compared with amyloid negative.

\*Evidence of association (bias-corrected confidence intervals do not cross zero); \*\* p<.01

M0: unadjusted (PACC adjusted for age at cognitive testing; neuroimaging outcomes adjusted for age at scan and TIV[TBV, hippocampal volume, WMHV only]); M1: M0+childhood cognition; M2: M1+childhood SEP; M3: M2+APOE- $\epsilon$ 4 status

**Table 16.** Interactive effects of cumulative risks score with APOE- $\varepsilon$ 4 status on cognitive performance and neuroimaging outcome measures, in males and females, in minimally adjusted models.

Outcome		Males	Females
Verbal memory z-score	Ν	654	686
	β	0.314	1.031
	95% CI	-1.012, 1.640	-0.257, 2.319
	р	0.64	0.12
Search speed z-score	Ν	669	693
	β	0.257	-0.036
	95% CI	-1.006, 1.521	-1.356, 1.284
	р	0.69	0.96
ACE-III total <sup>a</sup>	Ν	540	587
	β	0.465	1.102
	95% CI	-1.319, 1.865	-0.458, 2.674
	р	-	-
PACC z-score	Ν	199	188
	β	0.064	0.025
	95% CI	-2.167, 2.296	-2.411, 2.461
	р	0.95	0.98
TBV	N	184	174
	β	41.969	85.751
	95% CI	-111.015, 194.952	-69.659, 241.162
	р	0.59	0.28
Hippocampal volume	Ν	184	174
	β	-0.225	-0.210
	95% CI	-1.219, 0.769	-1.170, 0.749
	р	0.66	0.67
WMHV	Ν	181	169
	β	0.470	2.889
	95% CI	0.012, 17.776	0.062, 134.618
	р	0.68	0.59
SUVR	Ν	181	172
	β	-0.128	0.106
	95% CI	-0.386, 0.130	-0.143, 0.354
	р	0.33	0.40
Aβ status <sup>ь</sup>	Ν	181	172
	OR	0.076	0.023
	95% CI	0.000, 423.196	0.000, 96.940
	a	0.56	0.38

M0, minimally adjusted: PACC adjusted for age at cognitive testing; neuroimaging outcomes adjusted for age at scan and TIV (TBV, hippocampal volume, WMHV only); models assessing APOE interactions for other outcomes are unadjusted.

<sup>a</sup>Bootstrapping applied: confidence intervals are bias-corrected, p values are not applicable, models are not imputed.

<sup>b</sup>Logistic regression, coefficient presented is an OR for being amyloid positive compared with amyloid negative.



**Figure 12.** Predicted associations of cumulative risks score with standardised cognitive test performance at age 69 in males and females, in models fully adjusting for confounders (childhood cognition, childhood SEP, APOE- $\varepsilon$ 4 status). A) ACE-III: male n=525, female n=575. Bootstrapping is applied, confidence intervals are bias-corrected, and the model is not imputed. B) Word list recall: male n=1011, female n=1063. C) Timed letter search: male n=1038, female n=1073.



**Figure 13.** Predicted associations of cumulative risks score with A) total brain volume (TBV) and B) hippocampal volume at age ~70, in Insight 46 males (n=239) and females (n=229), adjusted for age at scan, total intracranial volume (TIV), childhood cognition, childhood socioeconomic position (SEP), and APOE-ε4 status.

# 4.3.5. Use of risk-modifying treatments

When additionally accounting for ever use of risk-modifying treatments (hearing aid use, antihypertensive medication, anti-depressant medication, diabetes medications), all associations previously detected with adjustment for early-life predictors of lifetime CRS were maintained (Table 17).

When accounting for all risk-modifying treatments, the association of CRS with verbal memory was attenuated by 4% in males and 10% in females, compared with models fully adjusted for early-life factors. This was primarily driven by an attenuating effect of anti-depressant medication use.

In males, the significant association of CRS with search speed performance was attenuated by 2% when accounting for all risk-modifying treatments, but adjustment for hearing aid use increased effect estimates by 6%.

The association of CRS with ACE-III performance was attenuated by 3% in males and 10% in females when accounting for all risk-modifying treatments, primarily driven by an attenuating effect of anti-depressant medication use.

Accounting for all risk-modifying treatments strengthened the significant associations of CRS with TBV and hippocampal volume in males by 6% and 27%, respectively. For TBV this was primarily driven by a strengthening effect of anti-depressant medication, while hearing aid use primarily explained the strengthening effect for hippocampal volume.

**Table 17.** Model estimates and 95% confidence intervals showing the estimated association of cumulative risks score with standardised cognitive performance measures at age 69 and volumetric neuroimaging measures aged ~70 in males and females, with full adjustments for confounding variables and additional adjustments for the use of risk-modifying treatments.

					Males						Females		
Outcome		M3	1	2	3	4	5	M3	1	2	3	4	5
Verbal memory z-	Ν			10	)11					1	.063		
score	β	-1.051	-1.065	-1.055	-1.000	-1.035	-1.008	-1.674	-1.706	-1.621	-1.562	-1.607	-1.514
	95%	-1.553 <i>,</i>	-1.576,	-1.561,	-1.527,	-1.537,	-1.545 <i>,</i>	-2.121,	-2.159,	-2.070,	-2.019,	-2.054,	-1.978,
	CI	-0.549	-0.555	-0.549	-0.474	-0.532	-0.471	-1.226	-1.253	-1.172	-1.105	-1.160	-1.051
	р	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**
Search speed z-	Ν				1038					1	.073		
score	β	-1.571	-1.673	-1.539	-1.492	-1.542	-1.542	-0.248	-0.205	-0.200	-0.225	-0.217	-0.126
	95%	-2.154,	-2.261,	-2.125,	-2.102,	-2.125,	-2.159 <i>,</i>	-0.807, 0.310	-0.771, 0.362	-0.760, 0.361	-0.797, 0.347	-0.780 <i>,</i>	-0.711, 0.460
	CI	-0.988	-1.084	-0.953	-0.881	-0.959	-0.926					0.346	
	р	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	0.38	0.48	0.48	0.44	0.45	0.67
ACE-III total z-score <sup>a</sup>	N	525			575								
	β	-1.634	-1.658	-1.644	-1.553	-1.631	-1.588	-1.349	-1.345	-1.317	-1.287	-1.296	-1.208
	95%	-2.299,	-2.378,	-2.311,	-2.255,	-2.302,	-2.313,	-1.925,	-1.895,	-1.874,	-1.854,	-1.874,	-1.762,
	CI	-1.084	-1.087	-1.074	-0.997	-1.093	-0.988	-0.763	-0.746	-0.730	-0.694	-0.698	-0.604
	р	*	*	*	*	*	*	*	*	*	*	*	*
TBV	N			2	39					2	229		
	β	-76.941	-78.214	-74.951	-79.910	-77.496	-81.182	-58.362	-58.541	-56.243	-60.392	-46.831	-48.922
	95%	-	-	-	-	-	-	-118.120,	-118.902,	-116.434,	-120.982,	-104.005,	-107.772,
	CI	140.010,	143.881,	139.476,	144.436,	140.290,	149.861,	1.397	1.821	3.948	0.197	10.342	9.927
		-13.872	-12.546	-10.425	-15.384	-14.702	-12.502						
	р	0.02*	0.02*	0.02*	0.02*	0.02*	0.02*	0.06	0.06	0.07	0.05	0.11	0.10
Hippocampal	Ν			2	39					2	229		
volume	β	-0.704	-0.815	-0.742	-0.741	-0.703	-0.894	-0.292	-0.318	-0.284	-0.300	-0.265	-0.301
	95%	-1.154,	-1.282,	-1.202,	-1.199,	-1.154,	-1.384,	-0.658. 0.074	-0.691, 0.055	-0.654, 0.085	-0.671, 0.071	-0.635,	-0.685 <i>,</i> 0.084
	CI	-0.255	-0.348	-0.282	-0.284	-0.252	-0.405					0.105	
	р	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	<0.001**	0.12	0.09	0.13	0.11	0.16	0.12

M3: (Neuroimaging outcomes adjusted for age at scan and TIV), childhood cognition, childhood SEP, APOE-ε4 status; 1=M3+hearing aid use; 2=M3+anti-hypertensive medication use; 3=M3+anti-depressant medication use; 4=M3+diabetes medication use; 5=M3+all risk-modifying treatments.

<sup>*a</sup>Bootstrapping applied: confidence intervals are bias-corrected, p values are not applicable, models are not imputed.*</sup>

\*Evidence of association (bias-corrected confidence intervals do not cross zero); \*\* p<.01
# 4.3.6. Secondary analyses

# 4.3.6.1. Cumulative risk exposures to midlife

Midlife CRS was calculated for 1,604 (52% female) participants who had available data at both the early- (up to 36 years) and mid-life (ages 43 and 53) timepoints, for all risk factors. In the whole cohort, males (n=774) had a mean midlife CRS of 0.23 (SD 0.13, range 0-0.71), which did not significantly differ from that of females (n=830, mean 0.22, SD 0.13, range 0-0.69; p=0.105; Appendix B Figure 1). T-tests showed that mean scores per risk factor in the whole-cohort (Appendix B Figure 2) were higher in males than females for hearing impairment, TBI, hypertension, high alcohol consumption and smoking, while females had higher scores than males for obesity, depression, and physical inactivity.

Summary statistics showing mean total midlife CRS, mean scores per risk factor, and t-tests of sex differences in the Insight 46 sample are shown in Appendix B Table 1. Midlife CRS was calculated for 420 (49% female) Insight 46 participants. Males (n=215) had a mean midlife CRS of 0.19 (SD 0.10, range 0-0.48) which did not significantly differ from that of females (n=205, mean 0.17, SD 0.11, range 0-0.54; p=0.07). T-tests demonstrated that mean scores per risk were higher in males than females for TBI, hypertension and high alcohol consumption, while scores were higher in females for depression.

Estimated associations of midlife CRS with cognitive performance outcomes at age 69 and Insight 46 volumetric neuroimaging outcomes are presented in Table 18. As in the main analyses, independently of early-life predictors of lifetime CRS, increased midlife CRS associated with poorer verbal memory and ACE-III performance in males and females, poorer search speed performance in males only, and reduced hippocampal volume in males but not females. The negative association of lifetime CRS with TBV in males was replicated with midlife CRS, and a significant negative association of midlife CRS with TBV was also detected in females. **Table 18.** Effect of midlife cumulative risks score on cognitive performance outcomes at age 69, and volumetric neuroimaging outcomes at age ~70 in males and females, with full model adjustments.

Outcome		Males	Females
Verbal memory z-score	N	1011	1063
	β	-0.927	-1.763
	95% CI	-1.412, -0.441	-2.227, -1.299
	р	<0.001**	<0.001**
Search speed z-score	N	1038	1073
	β	-1.145	-0.358
	95% CI	-1.715, -0.576	-0.920, 0.205
	р	<0.001**	0.21
ACE-III total <sup>a</sup>	N	572	625
	β	-1.456	-1.259
	95% CI	-1.996, -0.881	-1.762, -0.621
	р	*	*
TBV	Ν	239	229
	β	-71.163	-65.050
	95% CI	-130.417, -11.909	-122.135, -7.966
	р	0.02*	0.03*
Hippocampal volume	N	239	229
	β	-0.571	-0.110
	95% CI	-0.991, -0.152	-0.471, 0.252
	р	0.01*	0.55

\*<0.05 \*\* p<.01

<sup>a</sup>Bootstrapping applied: confidence intervals are bias-corrected, p values are not applicable, models are not imputed.

Models are fully adjusted for: childhood cognition, childhood SEP, APOE- $\varepsilon$ 4 status [TBV and hippocampal volume also adjusted for age at scan and TIV]

# 4.3.6.2. Non-weighted lifetime cumulative risks score

The sex-stratified distributions of non-weighted risk scores in the whole-cohort and Insight 46 samples are shown in Appendix B Figures 3 and 4. In the whole cohort, males (n=740) had a mean count of 9.11 (SD 3.72, range 0-23) risk factors across early-, mid- and later-life while females (n=769) had a mean count of 8.35 (SD 3.59, range 1-20). This translated to non-weighted risk scores of 0.25 (SD 0.10, range 0-0.64) in males and 0.23 (SD 0.10, range 0.03-0.56) in females, where the sex difference was significant (t-test p=0.001). A significant sex difference was also found in Insight 46 (p<0.001) where males had higher mean scores (0.24, SD 0.09, range 0.06-0.56, n=201) than females (0.19, SD 0.09, range 0.03-0.44, n=188).

The fully adjusted estimated effects of non-weighted lifetime CRS on each cognitive and neuroimaging outcome are presented in Table 19. As in the main analyses, increased risk exposures associated with poorer verbal memory and ACE-III performance in males and females, and with poorer search speed performance in males only, while there were no associations of CRS with PACC performance in either sex. Similarly to the main analyses, a negative association of CRS with hippocampal volume was found in males only. The negative association of weighted CRS with TBV in males was replicated, and a significant negative association of non-weighted CRS on TBV in females was also detected. No associations of non-weighted CRS with WMHV were found, but in contrast to the main analyses a significant association of non-weighted CRS with amyloid SUVR was detected in males; increased risk exposures associated with reduced A $\beta$  levels, although not sufficiently for an effect to be seen on the likelihood of having a positive A $\beta$  load.

**Table 19.** Model estimates and 95% confidence intervals for the association of non-weighted lifetime cumulative risks score with standardised cognitive performance outcomes and neuroimaging outcomes, fully adjusted for early-life confounding variables.

Outcome	Values	Males	Females
Verbal memory z-	Ν	1011	1063
score	β	-0.827	-1.923
	95% CI	-1.432, -0.221	-2.481, -1.366
	р	0.01*	<0.001**
Search speed z-score	N	1038	1073
	β	-2.085	-0.585
	95% CI	-2.772, -1.398	-1.272, 0.102
	р	<0.001**	0.10
ACE-III total z-score <sup>a</sup>	Ν	525	575
	β	-1.276	-1.700
	95% CI	-2.034, -0.569	-2.404, -0.991
	р	*	*
PACC total z-score	Ν	256	246
	β	0.090	-0.388
	95% CI	-0.899, 1.080	-1.344, 0.568
	р	0.86	0.42
TBV	N	239	229
	β	-100.408	-90.950
	95% CI	-168.156, -32.659	-158.224, -23.675
	р	<0.001**	0.01*
Hippocampal volume	Ν	239	229
	β	-0.585	-0.273
	95% CI	-1.068, -0.102	-0.694, 0.147
	р	0.02*	0.20
WMHV	Ν	234	221
	β	2.934	2.306
	95% CI	0.512, 16.804	0.440. 12.095
	р	0.23	0.32
SUVR	Ν	235	227
	β	-0.137	2.845
	95% CI	-0.253, -0.022	0.040, 202.861
	р	0.02*	0.64
Aβ status <sup>b</sup>	Ν	235	227
	β	0.096	2.845
	OR	0.001, 7.047	0.040, 202.861
	р	0.28	0.63

<sup>a</sup>Bootstrapping applied: confidence intervals are bias-corrected, p values are not applicable, models are not imputed.

<sup>b</sup>Logistic regression, coefficient presented is an OR for being amyloid positive compared with amyloid negative.

\*p<0.05 or bias-corrected confidence intervals do not cross zero; \*\* p<.01

M0: unadjusted (PACC adjusted for age at cognitive testing; neuroimaging outcomes adjusted for age at scan and TIV[TBV, hippocampal volume, WMHV only]); M1: M0+childhood cognition; M2: M1+childhood SEP; M3: M2+APOE- $\epsilon$ 4 status

## 4.4. Discussion

## 4.4.1. Key findings

A lifetime cumulative dementia risks score (CRS) was derived from life course data, quantifying exposures to modifiable dementia risk factors up to age 69. Sex differences in the frequency of some individual risk factors were found and, although males in the neuroimaging sub-study had greater CRS than females, there were no sex differences in CRS in the whole-cohort sample.

Early-life (childhood cognitive performance, childhood SEP) and genetic (*APOE*- $\epsilon$ 4 status) factors were predictive of CRS in the whole-cohort sample. Independently of these predictors, males and females showed adverse associations of increased risk exposures with poorer cognitive performance at age 69, with equal effects on cognitive state (ACE-III) but stronger associations with verbal memory in females and processing speed in males. Greater cumulative risk exposures associated with reduced brain volumes at age ~70, particularly in males, but there were no associations with AD (A $\beta$ ) or cSVD (WMHV) pathology measures in either sex. There was no evidence that associations between CRS and outcomes were modified by *APOE*- $\epsilon$ 4 status. Use of risk-modifying treatments did not have a substantial impact on the estimated effects of cumulative risk exposures on later-life outcomes.

Associations of lifetime cumulative risk exposures with later-life cognitive performance and brain volumes were replicated when risk exposures were measured only up until midlife, and when lifetime CRS was not weighted by PAFs.

## 4.4.2. Interpretation of findings

While males had a significantly higher count of lifetime risk exposures than females, the lack of sex difference in weighted lifetime CRS in the whole cohort sample reflects sex differences across individual risk factors assigned different weightings. That non-weighted lifetime CRS was significantly greater in males than females indicates that some of the risk-factor weightings could be sex-biased. Demonstrating sex differential patterns of risk, greater frequency of hearing impairment, TBI, hypertension and high alcohol consumption among males than females, and greater depression and physical inactivity in females, aligned with previous reports of sex differences in these risk factors.<sup>120,293-297</sup> However, significantly greater

weighted CRS in Insight 46 males than females seemed to be driven by greater levels of physical activity among Insight 46 females than in the whole cohort, in agreement with previous analyses demonstrating that greater physical activity predicted agreement to take part in the Insight 46 sub-study.<sup>194</sup>

Indeed, better general health and socioeconomic circumstances within Insight 46<sup>194</sup> likely explains reduced risk exposures in Insight 46 than in the whole cohort. Minimal variation in childhood cognitive performance and SEP, which are higher in Insight 46,<sup>194</sup> might also explain the lack of associations between early-life factors and lifetime CRS among Insight 46 participants. In the whole cohort, higher childhood cognition and SEP predicted lower CRS, reflecting how such early-life advantages can contribute to maintained health and social advantages throughout life,<sup>205,298</sup> in agreement with the accumulative model of life course epidemiology (Section 1.1.). While greater cognitive performance in childhood could have downstream effects on socioeconomic circumstances throughout life (e.g. improved education and career opportunities), greater early cognitive performance could also be indicative of developmental advantages prenatally or during infancy (e.g. diet in infancy), which may have lasting impacts on susceptibility to some health-related dementia risk factors such as obesity or hypertension, for example. However, puberty timing was not found to be predictive of CRS, implying that early-life sex hormone exposure did not influence the occurrence of modifiable risk factors throughout life; social advantage may be more predictive of lifetime dementia risk factor exposures than biological development.

Interestingly though, higher cumulative risk exposures were found in whole-cohort *APOE*- $\varepsilon$ 4 non-carriers than carriers, emphasising the strength of the *APOE*- $\varepsilon$ 4 association with dementia; while  $\varepsilon$ 4 carriers might have fewer lifetime exposures to modifiable dementia risks, dementia rates are higher among carriers.<sup>288</sup> Nonetheless, associations between CRS and outcome measures were not modified by *APOE*- $\varepsilon$ 4 status, indicating that adverse associations of greater risk exposures with cognitive performance and brain health at age ~70 were not exacerbated in  $\varepsilon$ 4 carriers. Conversely, the CAIDE population-based study found that a composite of midlife (mean age 50.6(6.0), range 39-64) lifestyle risks (physical inactivity, alcohol consumption, smoking, high dietary fat) associated with increased dementia risk ~21 years later (mean age 71.6(4.1), range 65-79) to a greater extent in  $\varepsilon$ 4 carriers than non-carriers.<sup>299</sup> The extent to which  $\varepsilon$ 4 status modifies the associations of lifestyle risks with later-

life dementia-related outcomes could depend on the risks examined, when these risks occur, and the age at which outcomes are assessed. In the current analyses, participants were younger than in the CAIDE study; it is possible that  $\varepsilon 4$  modifying effects are not detected in a preclinical cohort. Reasons for why NSHD ɛ4 carriers had fewer risk exposures than noncarriers are unclear. Pleiotropic effects of ɛ4 whereby some ɛ4 benefits are recognised at younger ages (e.g. better cognitive performance in young carriers than non-carriers)<sup>262</sup> could provide some explanation. There may also be a survivor bias in which only the healthiest  $\varepsilon 4$ carriers are retained in the cohort at age ~70, meaning that ε4 carriers with high risk factor exposures are not observed in the data collected at older ages. However, APOE genotype was not associated with cumulative risk exposures in the Insight 46 sub-sample, and lifestyle risks in the CAIDE study did not significantly differ between ε4 carriers and non-carriers. Additionally, in the population-based Rotterdam study, no APOE group differences (low-[  $\epsilon^2/\epsilon^3$ ,  $\epsilon^2/\epsilon^2$ ], mid-[ $\epsilon^3/\epsilon^3$ ], high-dementia-risk[ $\epsilon^4/\epsilon^4$ ,  $\epsilon^4/\epsilon^2$ ,  $\epsilon^4/\epsilon^3$ ]) were found in a count of protective health and lifestyle factors (smoking abstinence, no depression, no diabetes, regular physical activity, avoiding social isolation, adhering to a healthy diet including limited alcohol consumption), although BMI was lower and cholesterol levels were higher among  $\epsilon 4$ carriers.<sup>300</sup> To understand the association of *APOE* genotype with lifetime cumulative risk exposures, further work is required to determine whether the association is robust across varied cohorts.

Although early-life predictors of CRS were found, adverse associations of greater risk exposures with later-life cognitive performance and brain volumes remained independently of these early-life predictors, reinforcing the value of interventions in adulthood to limit the adverse effects of risk exposures on later-life outcomes. Accounting for the use of risk-modifying treatments did not, however, substantially alter the significant associations of CRS with later-life outcomes, implying that treatments to control existing risk factors do not effectively mitigate against the adverse effects of CRS on memory performance and cognitive state, in agreement with previous findings of cognitive performance improvements with anti-depressant use.<sup>301</sup> Conversely, hearing aid use slightly strengthened the negative effects of CRS on processing speed performance and hippocampal volume. Previous evidence has found benefits of hearing aid use, including increased social interaction,<sup>302,303</sup> but adherence to

hearing aid use is generally poor <sup>304</sup> which potentially contributed to the negative confounding observed. While further work is needed to elucidate potential mediating effects of risk factor treatments, it is possible that interventions to prevent the occurrence of risk factors could be beneficial for later-life outcomes. Indeed, multidomain lifestyle intervention trials have shown promise in reducing risk factor incidence and improving cognitive performance. For example, the 2-year FINGER trial of combined dietary, exercise, cognitive training and vascular risk monitoring interventions showed improved cognitive performance on a neuropsychological test battery assessing executive function, processing speed and memory.<sup>104,105</sup>

Associations of increased cumulative modifiable risk exposures with reduced cognitive state at age 69 could reflect diminished cognitive resilience (Section 1.6.) to ageing, resulting from reduced opportunities for the development and maintenance of cognitive reserve throughout the life course. However, equal effects of CRS on cognitive state between sexes while there were sex variations in CRS effects on individual cognitive domains demonstrates that, although composite outcomes may not differ by sex, pathways to these outcomes may be different; the adverse effect of CRS on cognitive state could be driven by effects on memory in females and processing speed in males. As shown in Chapter 3, both these domains show female advantages throughout adulthood, with a larger female advantage in verbal memory performance. Although there was a slightly greater adverse effect of CRS on verbal memory performance in females than males, possibly representing greater CRS-associated reductions in female verbal memory cognitive resources given a higher starting point, this sex difference was not statistically significant. Conversely, the adverse effect of CRS on processing speed performance was only detected in males, demonstrating a male-specific adverse effect of greater lifetime risk exposures on a cognitive domain shown to be particularly vulnerable to the effects of ageing.<sup>17</sup> Previous NSHD analyses have, however, found adverse effects of increased BMI (at age 60-64, and between ages 36 and 43) on processing speed performance at age 60-64 only in females;<sup>206</sup> an individual risk factor association could be masked when measuring cumulative risk exposures, but there may also be differences in how risk factors associate with outcomes measured at different ages. Interestingly, longitudinal analyses of verbal memory and processing speed performance within NSHD showed faster age-related decline in processing speed performance in males than females, while there were no sex differences in rates of verbal memory decline between ages 43 and 69.5 Taken together,

evidence suggests that increased exposure to modifiable dementia risk factors could contribute to accelerated cognitive ageing observed in males, although further work is required to determine if CRS associates with longitudinal cognitive trajectories.

Within the Insight 46 sub-sample, CRS was not found to associate with performance on a cognitive composite designed to detect subtle pre-clinical cognitive changes indicative of AD (PACC), possibly reflecting a lack of association for cumulative risk exposures with AD-specific cognitive changes. However, only PACC total was examined in these analyses; it remains possible that CRS associates with subtle cognitive changes assessed in the PACC sub-tests (Table 2), particularly given sex-specific findings for verbal memory and processing speed in the whole-cohort. Additionally, given better general health, socioeconomic circumstances, and cognitive performance within the Insight 46 cohort, <sup>194</sup> it is possible that Insight 46 participants had greater cognitive resilience than the whole cohort, buffering against the adverse effects of greater lifetime risk exposures. Indeed, a sensitivity analysis to examine the associations of CRS with ACE-III performance in the Insight 46 sub-sample (male n=163, female n=155; Appendix B Table 2) did not detect an association in males, and the significant effect in females was weaker than that found in the whole cohort.

However, it is also notable that CRS was not associated with any of the neuroimaging measures of specific dementia-related pathology (A $\beta$ , WMHV) examined, further indicating that the estimated effects of lifetime risk exposures are not specific to indicators of possible dementia at age ~70. The lack of association between CRS, which includes several cardiovascular risks, and WMHV (a marker of cSVD) was surprising; the effects of cardiovascular risks on WMHV were possibly diluted by the inclusion of other non-cardiovascular risks which are not independently associated with WHMV. There was evidence for CRS associations with reduced brain volumes (stronger in males) which, alongside the male-specific association with processing speed, suggests that greater lifetime risk factor exposures exacerbate non-dementia-specific cognitive and brain ageing, particularly in males. Interestingly, males, but not females, showed an association of CRS with reduced hippocampal volume which remained independently of total brain volume, indicating that CRS-associated volume reductions are not evenly distributed throughout the male brain. Given a slightly stronger effect of CRS on verbal memory performance in females, it is surprising that CRS was not associated with female hippocampal volume, as a brain region important for memory

153

function.<sup>305</sup> However, a sensitivity analysis did not detect associations of CRS with verbal memory for either sex in the Insight 46 sub-sample (Appendix B Table 3), possibly reflecting improved cognitive resilience to hippocampal atrophy within the Insight 46 cohort. As the cohort ages, continued follow-up assessments<sup>306</sup> will be valuable in identifying whether and how cognitive resilience changes over time.

The potential effects of CRS on brain ageing to a greater extent in males than females is consistent with previous Insight 46 work which found older predicted brain age differences (PAD; the difference between chronological age and predicted brain age derived from T1-weighted MRI measures of grey matter, white matter, CSF and intracranial volume) in males.<sup>236</sup> In analyses adjusted for sex, greater PAD associated with increased WMHV and cardiovascular risk (measured using the Framingham risk score) at ages 36 and 69,<sup>236</sup> indicating a link between cerebrovascular disease and accelerated brain ageing in the Insight 46 cohort; future research examining CRS-PAD associations could inform whether lifetime risks, including but not exclusive to cardiovascular risk, differentially associate with brain ageing in males and females.

While the primary focus of the current analyses was on cumulative modifiable risk exposures throughout the life course, midlife is a proposed period of vulnerability,<sup>8,307</sup> and many of the Lancet risks (e.g. TBI, hypertension, obesity) are attributed to midlife.<sup>56</sup> Indeed, associations of lifetime CRS with later-life cognitive and volumetric neuroimaging outcomes were replicated when only modifiable risk exposures accumulated up to midlife were quantified, demonstrating that the adverse effects of risk factor exposures on later-life outcomes were already present by midlife. While this does not prove that midlife is a sensitive period, the finding does add to research demonstrating that risks up to and including midlife more strongly associate with brain health measures than risks in later-life. Cardiovascular risk measured in early adulthood (age 36) has been particularly associated with later-life neuroimaging outcomes (older PAD, reduced TBV, increased WMHV) in Insight 46, while cardiovascular risk at ages 53 and 69 showed progressively weaker associations with such outcomes.<sup>236,308</sup> Further work is needed to investigate whether risk exposures within each of the three timepoints included in the CRS (early-, mid-, later-life) differentially associate with later-life outcomes. This is also important given that the directionality of some health and lifestyle factors associated with dementia risk can reverse in later-life; for example,

154

hypotension and weight loss (lower BMI) at older ages could be part of the dementia prodrome.<sup>309,310</sup>

Associations of lifetime CRS with later-life cognitive and volumetric neuroimaging outcomes were also replicated when weightings per risk factor were removed, in agreement with previous studies which did not find differences in weighted and non-weighted Lancet risk exposure associations with cognitive performance.<sup>111,273</sup> An additional association detected between non-weighted CRS and cerebral amyloid levels in males, that was not found for weighted CRS, was in the unexpected direction, whereby greater risk exposure associated with lower levels of cerebral amyloid. Reasons for this are unclear, requiring independent validation.

## 4.4.3. Strengths and limitations

The strength of these analyses lies in the wealth of prospective life course data facilitating the quantification of lifetime exposures to each of the twelve modifiable dementia risks identified in the Lancet commission, where other studies have been cross-sectional and have not had available data for all twelve risks.<sup>111,311</sup> Importantly, the availability of prospective early-life data also enabled analyses accounting for early-life predictors of cumulative risk exposures, demonstrating that although early-life factors are important, there are opportunities throughout life to mitigate the adverse effects of cumulative risks on later-life outcomes.

While the inclusion of variables relating to lifetime use of risk-modifying treatments was informative, these variables were not independent from cumulative risk score. For example, diabetes medication use is indicative of having diabetes, and even if diabetes is controlled with medication, the diabetes diagnosis remains. Additionally, treatment use could reflect greater risk factor severity; anti-depressant medications are more likely to be prescribed for more severe depression,<sup>312</sup> for example. Adjustments for risk-modifying treatments were intended to inform whether the use of such treatments mitigated CRS-outcome associations, but such adjustments can be problematic. As described in Section 3.4.3., adjusting for a mediator can induce collider bias, particularly if there is unmeasured confounding. Further, Tobin et al. (2005)<sup>313</sup> demonstrated that, when examining determinants of blood pressure, adjusting for anti-hypertensive use as a covariate induced estimation bias whereby effect sizes

were reduced; individuals with higher underlying blood pressure (an unobserved measure) are more likely to be those taking anti-hypertensive medications which reduce observed blood pressure. In individuals who receive risk-modifying treatment, there are two components to the observed risk factor: i) the extent to which treatment reduces the risk factor, negatively associated with the risk (e.g. anti-hypertensive use reduces blood pressure), and ii) the underlying presence of the risk as a prerequisite for treatment, positively associated with the risk (e.g. high blood pressure increases likelihood of taking anti-hypertensive medication). Adjusting for treatment can therefore have minimal impact on effect estimates, or there may even be counterintuitive results. For example, if underlying blood pressure (ii) is higher than the blood pressure-lowering effect of anti-hypertensive medication (i), the overall association of treatment with blood pressure would be positive, implying that anti-hypertensive use increases blood pressure. One solution suggested by Tobin et al. is to apply a sensible constant to the observed risk factor measurement in treated individuals, based on prior knowledge of the treatment effect size (e.g. anti-hypertensives reduce blood pressure by 15mmHg). An avenue for future research addressing treatment effects could be to incorporate treatment effect weightings into the CRS, instead of adjusting for treatments as covariates.

Further work is also warranted to include more detailed investigation of medication types, adherence, severity of the risk requiring treatment, and the potential impact of treatment on other risk factors. The current analyses of risk-modifying treatments take a broad approach whereby multiple risk factors are combined in the CRS and adjustments are made for treatments which modify only some of these risk factors. This inadvertently leads to some illogical research questions, asking 'does the effect of depression on cognition go through anti-hypertensive use?', for example. In further work, risk-modifying treatment effects should be examined for individual risk factors.

While the cumulative risks score was derived to maximise the use of available life course data pertaining to each individual risk factor, some limitations in the categorisation of individual risks must be acknowledged. For instance, with no consistent measure of social isolation established,<sup>56</sup> proxies for social isolation were derived from structural social health measures<sup>314</sup> (marital and cohabitation status, social network size, social contact frequency); isolation indicated by any one of these measures was taken to represent social isolation during the relevant time-period. It is possible that, if someone is unmarried and living alone, they

might have a large social network outside of the home and are therefore not socially isolated, although evidence has shown that married people tend to have more social contact than single people, particularly among older generations.<sup>56</sup> Additionally, smoking was counted as a risk if participants indicated current smoking during each timepoint, but this categorisation does not indicate how heavily someone smoked, which has previously been negatively associated with cognitive performance.<sup>248</sup> There is, however, still value in quantifying the number of timepoints throughout life that individuals were smokers, and a level of smoking heaviness associated with dementia risk is not specified in the Lancet commission.<sup>56</sup>

It could be argued that collapsing dementia risk data into a single measure of cumulative risk exposures is disadvantageous given that the fine-grained detail of individual risks and their associations with later-life outcomes are diluted. However, there is value in quantifying multiple risk exposures throughout life given that, in reality, risk factors are interrelated. For instance, physical inactivity increases the risk of obesity, which is also associated with hypertension and diabetes.<sup>315</sup>

The Lancet commission acknowledges the interrelationships between risk factors and recognises that the contribution of each factor may not be equal.<sup>56</sup> While the risk-specific PAFs calculated in the Lancet commission were used to produce the weighted lifetime CRS in these analyses, these PAFs have not been validated and a particular limitation for these analyses is that PAFs were not sex-specific. Nonetheless, results from the secondary analyses of non-weighted CRS did not substantially differ from weighted CRS findings in males or females.

Another limitation is that, while the Lancet risks were identified as those with the most convincing evidence during an extensive review of dementia risks literature, this body of evidence is likely to be male-centric given a general lack of research explicitly testing for sex differences and that females have been traditionally underrepresented in research. Additionally, these twelve risks are unlikely to be exhaustive (see Section 1.10.2.). In fact, with the emergence of new evidence between the 2017<sup>316</sup> and 2020 Lancet reviews, three risks were added (heavy alcohol consumption, high air pollution exposure, TBI).<sup>56</sup> Even so, comparison of dementia risk scores within the Dunedin, New Zealand, population cohort did not find substantially improved predictions of midlife brain health (predicted brain age, WMHV, hippocampal volume) or objective or subjective cognitive decline when an extensive

list of 48 risk factors were measured,<sup>271</sup> indicating that measures of fewer, more selective dementia risk factors can still be informative.

Finally, there may be reverse directionality for some risk factors whereby there is uncertainty whether a factor contributes to dementia risk or if it forms part of the dementia prodrome. For example, depression in earlier-life is considered a risk for later dementia while depression in later-life could be a non-cognitive dementia feature.<sup>292</sup> However, the results from secondary analyses, which reduced the potential effects of bidirectionality by measuring risk factors only up until midlife, did not substantially differ from the main analyses.

# 4.4.4. Summary

Independently of early-life predictors of lifetime dementia risk factor exposures, there were adverse associations of accumulated risks with later-life cognitive performance and brain volumes in both sexes, which were already established by midlife. There were nuances in which cognitive domains more strongly associated with cumulative risk exposures between the sexes, and associations with reduced brain volumes were stronger in males. These findings demonstrate evidence for greater risk factor exposures with more advanced cognitive and brain ageing not specific to dementia-related pathology, particularly in males, reflecting reduced male resilience to life course health and lifestyle risks.

# 5.0. Empirical section C: Menopause and female later-life cognition and brain health

This section aims to examine how variations in menopause, as a female-specific hormonal transition, associate with cognition and brain health in later-life, to build an understanding of menopause as a possible contributor to increased female dementia risk, and to determine how risk for adverse cognitive and brain ageing might differ between women.

The primary research question addressed in this section is:

 Does age at menopause associate with measures of cognitive performance and brain health in later-life?

The section is split into two empirical chapters:

- 5i. Menopause and later-life cognition (published in a peer reviewed journal)<sup>317</sup>
- 5ii. Menopause and later-life brain health

# 5i. Menopause and later-life cognition

In this section, associations between age at menopause and cognitive performance at age ~70 were examined to determine whether relationships were stronger for certain cognitive domains. NSHD and Insight 46 women with available data for menopause age and cognitive state (NSHD, ACE-III age 69) or PACC performance (Insight 46, age 69-71) were included in multivariable linear regression analyses assessing the relationships between menopause age and cognitive performance, and the extent to which associations could be explained by relevant early cognitive, sociodemographic, reproductive, and health-related covariables. Older age at menopause associated with better performance across all cognitive outcomes, most strongly for assessments of visual processing, and associative learning and memory. These associations were not modified by menopause type (natural or surgical) or *APOE*- $\epsilon$ 4 status, but effect estimates were attenuated by life course covariables, most notably by childhood cognition. Additionally adjusting for HT use did not have a meaningful impact on effect estimates, implying that HT use did not mediate the relationship of menopause timing with post-menopausal cognitive performance.

#### 5i.1. Introduction

As described in Section 1.11., endogenous oestrogen levels gradually decline during natural menopause, while surgeries to remove the uterus or one or both of the ovaries can cause an earlier and sometimes more dramatic decline in oestrogen levels.<sup>142</sup> Oestrogen has pleiotropic effects influencing both the reproductive axis and higher mental function, <sup>305</sup> and menopause is often accompanied by neurological symptoms including cognitive difficulties, particularly with memory and attention.<sup>144</sup> However, an understanding of the longer-term association between menopause and cognitive function in later-life is not yet established. There is conflicting evidence around how age at menopause, or taking menopausal HT, is associated with dementia risk, cognitive impairment or later-life cognitive function.<sup>144,154</sup> It is important to understand the association between menopause and well characterised cognitive function in later-life, prior to overt dementia symptoms, to help develop a better understanding of female cognitive ageing.

Evidence shows increased dementia risk among female carriers of the *APOE*- $\varepsilon$ 4 risk allele for AD, compared with male  $\varepsilon$ 4 carriers, especially at later, post-menopausal ages.<sup>95,318</sup> However, evidence linking menopause age with *APOE* genotype is mixed.<sup>319,320</sup> Research examining interactions of *APOE* and menopause age on dementia risk and later-life cognitive performance is also lacking.

While meta-analysis does not support an overall link between menopause age and risk for developing dementia, most studies investigating the association between menopause and later-life outcomes are unable to account for childhood cognition, a key confound given higher childhood cognition predicts both later menopause age<sup>156,157</sup> and better cognitive performance in later-life.<sup>158,159</sup> Studies also typically lack prospectively recorded data for premenopausal covariables such as BMI and smoking. In addition, most studies have wide age ranges at cognitive testing and short follow-up periods. The NSHD and Insight 46 cohorts provide a unique opportunity to overcome some of these issues, with prospectively recorded data obtained over 70 years of follow-up in an age-homogenous cohort.

Previous NSHD work has investigated associations between menopause age and cognitive performance. Among women who were post-menopause by age 57 years, positive associations of menopause age with NART and verbal memory, but not processing speed, performance at age 54 have been detected. Associations attenuated with adjustments for

childhood cognition, previous task performance, and additional socioeconomic factors.<sup>321</sup> Small positive associations between age at natural menopause and better verbal memory performance, but not processing speed, from ages 43 to 69 were also found after accounting for lifetime factors, although the effect estimates attenuated with adjustment for childhood cognition.<sup>155</sup> A broader examination of which cognitive domains most strongly associate with menopause age has not yet been conducted, nor have clinically relevant cognitive assessments been examined in relation to menopause age.

# 5i.1.1. Objectives and research questions

This work expands on previous NSHD analyses, by addressing the relationship between age at menopause and performance on clinically relevant cognitive assessments completed in laterlife; a test of cognitive state at age 69 (ACE-III) and a composite measure, mainly used in clinical trials (PACC),<sup>322</sup> completed by Insight 46 participants. Overall task performance and subdomain performance is assessed, to examine which cognitive domains associate with menopause age.

The following research questions are addressed:

- 1. Does age at menopause associate with cognitive performance on the ACE-III and PACC in later-life, and which cognitive domains show the strongest associations?
- 2. Are associations independent of a range of relevant confounders including early cognitive and sociodemographic factors, reproductive, and health-related factors?
- 3. Are associations modified by menopause type (natural or surgical)?
- 4. Are associations modified by APOE-ε4 status?
- 5. Does HT use contribute to associations of menopause age with later-life cognitive performance?

## 5i.1.2. Hypotheses

Given previous evidence associating later menopause age with better cognitive performance, later menopause age was expected to positively associate with cognitive performance at age ~70, most notably for assessments of verbal abilities and memory. Early cognitive, sociodemographic, reproductive, and health-related factors were expected to attenuate

associations between menopause age and later-life cognition, in agreement with previous studies testing menopause age-cognition relationships within NSHD.<sup>155</sup> Given previous evidence for poorer cognitive and dementia-related outcomes in women who had surgically-induced menopause at earlier compared with later ages,<sup>151</sup> an effect modification by menopause type was anticipated whereby women who had surgical menopause at earlier ages would show poorer cognitive performance than women who had natural menopause at earlier ages. Although associations of *APOE* genotype with menopause age are unclear, an effect modification by *APOE*-ɛ4 status was hypothesised, particularly for domains linked with AD (i.e. memory). Since the ɛ4 dementia risk is greater in females than males, menopause age-cognition associations were expected to be stronger in carriers than non-carriers. The relationship between HT use and later-life cognitive performance is not yet well understood, hence a hypothesis for whether, and in what direction, adjustments for HT use might alter effect estimates cannot be drawn.

# 5i.2. Analytic method

## 5i.2.1. Analytic sample

Between ages 43 and 54, 1,572 female NSHD study members completed annual postal questionnaires for the Women's Health in the Middle Years survey<sup>192</sup> (Section 2.2.1.1.), providing information on their age at menopause and whether they underwent a surgical or natural menopause. As outlined in Section 2.3.2., NSHD participants completed the ACE-III at age 69 (Table 1), and at wave I of Insight 46 data collection (age 69-71), participants completed a modified version of the PACC (Table 2). Women were included in these analyses if they had known age at menopause and available ACE-III data at age 69 (whole-cohort) or available PACC data at age 69-71 (Insight 46 wave 1) (Figure 14).



Figure 14. Flow chart demonstrating the sample selection for analyses.

# 5i.2.2. Menopause age and type

Age at menopause was ascertained for all menopause types as age at final menstrual period, indicated on self-reported questionnaires<sup>192</sup> as months since birth and later converted into years. Menopause type was recorded as natural if no hysterectomy or oophorectomy surgery was reported prior to the final menstrual period. Women who reported having a hysterectomy or unilateral or bilateral oophorectomy before reaching a natural menopause were categorised as having had a surgical menopause. Information on the type of surgeries women reported is presented in Table 20.

**Table 20.** Characteristics for women with available data on age at menopause and cognitive home assessment at age 69 (ACE-III), and Insight 46 sub-study neuropsychology assessment (PACC). The descriptive data included in this table have not been imputed.

	ACE-III completed (NSHD)		PACC completed (Insight 46)	
Variable	Ν	Mean(SD);range/%	Ν	Mean(SD);range/%
Age of period cessation (years since birth) (mean(SD); range)	746	49.81(5.95);28.75-62.50	197	49.89(5.71);30.25-60.50
Natural menopause (mean(SD); range)		51.95(4.06);34.50-61.92	134	52.47(3.22);40.50-59.50
Surgical menopause (mean(SD); range)	205	44.18(6.44);28.75-62.50	63	44.41(5.98);30.25-60.50
ACE-III total raw score (mean(SD); range)	746	91.79(6.01);62-100	167	93.28(5.27);70-100
ACE-III attention & orientation raw score	746	16.61(1.95);5-18	167	16.74(1.90);8-18
(mean(SD); range)				
ACE-III language raw score (mean(SD); range)	746	25.28(1.13);16-26	167	25.49(0.99);19-26
ACE-III memory raw score (mean(SD); range)	746	23.79(2.66);12-26	167	24.50(1.97);15-26
ACE-III verbal fluency raw score (mean(SD); range)	746	11.09(22.04);2-14	192	11.5(1.87);2-14
ACE-III visuospatial function raw score (mean(SD); range)		15.01(1.30);8-16	167	15.13(1.34);8-16
PACC total raw score <sup>a</sup> (mean(SD); range)	167	39.62(6.59);13.50-52	197	39.8(6.62);13.50-52.75
PACC DSST raw score (mean(SD); range)	167	48.80(10.24);24-76	197	49.15(10.11);24-76
PACC FNAME-12A raw score (mean(SD); range)	165	68.78(18.36);3-95	195	69.18(18.16);3-95
PACC logical memory delayed raw score (mean(SD); range)		12.17(3.27);0-20	197	12.39(3.41);0-23
PACC MMSE raw score (mean(SD); range)		29.23(1.10);23-30	197	29.28(1.04);23-30
Childhood cognition z-score age 8 <sup>b</sup> (mean(SD); range)	693	0.16(0.80);-2.11-2.39	197	0.40(0.77);-1.59-2.47
Childhood social class	702		192	
Manual (%)	377	53.70	94	48.96
Non-manual (%)		46.30	98	51.04
Education (to age 26)			192	
Ordinary (GCSE-level or below) (%)	462	65.16	95	49.48
Advanced (A-level or higher) (%)	247	34.84	97	50.52
Age at menarche (years since birth) (mean(SD); range)		13.02(1.19);9-18.50	172	12.88(1.20);9.92-17.50
Number of natural-born children			177	
0-2 children (%)	420	65.42	115	64.70
3 or more children (%)	222	34.58	62	35.03
Menopause type			197	
Natural (%)	541	72.50	134	68.02
Surgical (%)	205	27.50	63	31.98
Type of surgery	205		63	
Hysterectomy only (%)	101	49.27	34	53.97
Unilateral oophorectomy (with/without		10.24	5	7.94
hysterectomy) (%)				
Bilateral oophorectomy (with/without		40.49	24	38.09
hysterectomy) (%)				
BMI at age 36 years (kg/m <sup>2</sup> ) (mean(SD); range)		23.18(3.31);16.23-40.39	183	23.10(3.27);17.16-39.16
Smoking pack years at age 36 years (mean(SD); range)		1.12(2.01);0-10	183	0.84(1.72);0-7.50

APOE-ɛ4 status	665		197	
ε4 present (%)	195	29.32	54	27.41
ε4 absent (%)	470	70.68	143	72.59
Affective symptoms age 69 (GHQ caseness)			192	
Yes (%)	140	18.82	24	12.50
No (%)	604	81.18	168	87.50
Age at Insight 46 cognitive testing (mean(SD);		N/A	184	70.68(0.68);69.27-71.86
range)				
Ever use of HT	667		191	
No (%)	282	42.28	58	30.37
Yes (%)	385	57.72	133	69.63
For HT users, age at first use	384		132	
≤45 years	83	21.61%	23	17.42%
46-51 years	231	60.16%	89	67.43%
≥52 years	70	18.23%	220	15.15%
For HT users, length of HT use	377		130	
<5 years (%)	183	48.54	61	46.92
5-10 years (%)	139	36.87	48	36.92
>10 years (%)	55	14.59	21	16.15

ACE-III = Addenbrooke's Cognitive Examination; PACC = Preclinical Alzheimer's Cognitive Composite; NSHD=National Survey of Health and Development; SD=standard deviation; DSST=Digit-Symbol Substitution Test; FNAME=Face-Name Associative Memory Examination; MMSE=Mini-Mental State Examination; GHQ= General Health Questionnaire; HT= Hormone Therapy

a: PACC total raw score Is the mean of scores across the four PACC sub-tests (DSST, FNAME-12A, logical memory delayed recall, MMSE) calculated for each participant. Where FNAME-12A data was missing, PACC total was calculated as the mean of scores across DSST, logical memory delayed recall and MMSE.

*b: Z*-score standardised to the sample at the time. If data are missing for cognition at age 8, values from age 11 (n=11) or age 15 (n=10) years were used instead.

#### *5i.2.3. Cognitive outcome measures*

As outlined in Section 2.3.2. (Table 1), NSHD participants completed the ACE-III, a measure of cognitive state, during home visits at age 69. Total ACE-III score (maximum 100) is the sum of scores across five cognitive sub-domains: attention and orientation (scored 0-18), verbal fluency (0-14), memory (0-26), language (0-26), and visuospatial function (0-16). Raw ACE-III total and sub-domain scores were standardised to the analytical sample.

Also described in Section 2.3.2., Insight 46 participants completed a comprehensive neuropsychology test battery<sup>158,193</sup> when aged between 69 and 71 (Table 2). This included a modified version of the PACC,<sup>158</sup> comprising of four sub-tests: the digit-symbol substitution test (DSST; Wechsler Adult Intelligence Scale-Revised)<sup>213</sup> assessing processing speed, associative learning, attention, and executive function; the 12-item face-name associative memory examination (FNAME-12A)<sup>215</sup> assessing associative, episodic memory; logical memory IIa (Wechsler Memory Scale-Revised)<sup>212</sup> assessing episodic memory; and the minimental state examination (MMSE),<sup>211</sup> a 30-point test of overall cognitive state. Raw scores on each sub-test were standardised to the analytical sample and averaged to generate a total PACC score for each participant.

#### 5i.2.4. Covariables

Covariables were selected based on previous analyses and evidence linking variables with menopause age or cognition.<sup>155,157,158</sup> Childhood cognition at age 8 years was the sum of four tests of verbal and non-verbal ability (Section 2.3.4.4.), standardised to the sample at the time of testing. If data were missing for cognition at age 8, available data for cognition at age 11 (n=11), or at age 15 years (n=10) were used instead. Childhood SEP up to age 15 (Section 2.3.4.3.) was categorised as manual or non-manual. Highest educational attainment up to age 26 (Section 2.3.4.2.) was categorised as ordinary (GCSE-level or equivalent) or below, or advanced (A-level or equivalent, or above). Age at menarche was recorded as years since birth, reported by a school doctor at age 14-15 years or self-reported at age 48.<sup>323</sup> Parity was indicated by self-reported number of natural-born children, excluding still births and miscarriages. Due to small proportions of nulliparous or single parity women in the whole-NSHD sample, this variable was categorised as 0-2 children, or 3 or more children. Menopause type was categorised as natural or surgical, as defined above (Section 5i.2.2.). Given some

evidence for a relationship between BMI and menopause timing (higher BMI, later menopause) and that obesity associates with an increased likelihood of surgical menopause,<sup>324</sup> BMI at age 36, negatively associated with cognitive performance at age 60-64 in this cohort,<sup>206</sup> was included as a covariable, recorded as a continuous value (kg/m<sup>2</sup>). Smoking pack years was self-reported at age 36 years. *APOE*- $\epsilon$ 4, linked with an earlier menopause age,<sup>319</sup> was categorised as  $\epsilon$ 4-present or  $\epsilon$ 4-absent (Section 2.3.4.1.). Affective symptom caseness at age 69 was determined using a cut-off of 5 or more on the 28-item GHQ.<sup>282</sup> Age at cognitive testing for Insight 46 participants was derived from the recorded age, in years, at which participants underwent neuroimaging. Ever or never use of any type of prescribed menopausal HT by age 69 was self-reported by questionnaire.

## 5i.2.5. Statistical analyses

Analyses were conducted using Stata version 17.0.

Associations between menopause age and z-score standardised cognitive outcomes (ACE-III total and sub-domains, and PACC total and sub-tests) were assessed using multivariable linear regression analyses, cumulatively adjusting for covariables. Unadjusted models (model 0/M0) were followed by adjustments for early cognitive and sociodemographic factors (M1:childhood cognition, childhood SEP, education), reproductive factors (M2:M1 plus age at menarche, parity, menopause type), and health-related factors (M3:M2 plus BMI, smoking, affective symptoms, *APOE*-ɛ4 status, and age at cognitive testing [Insight 46 only]).

The potential moderating role of menopause type was examined by testing for menopause age-by-type interactions on standardised ACE-III total and PACC total scores in fully adjusted models (M3). Similarly, menopause age-by-*APOE*-ε4 interactions were added to fully adjusted models to examine whether associations were modified by *APOE*-ε4 status.

The potential role of HT use in associations of menopause age with cognitive outcomes was assessed by further adjusting for HT (M3 plus HT). Whether HT use was associated with ACE-III total and PACC total scores was also examined without adjustments, and after accounting for menopause age.

In sensitivity analyses, ran for ACE-III total and PACC total, participants who scored less than 82 on the ACE-III, a threshold indicative of possible cognitive impairment,<sup>325</sup> were excluded (whole-cohort n=50; Insight 46 n=5). Additionally, since the surgical removal of both ovaries results in the cessation of all ovarian oestrogen production, in contrast to surgeries in which at least one ovary is conserved,<sup>142</sup> analyses excluding women who had bilateral oophorectomy (whole-cohort n=83; Insight 46 n=24) were conducted.

Multiple imputation (Section 2.4.) with 50 imputations (unless otherwise specified) was used to account for missing data in the covariables and, where applicable, in outcomes (PACC FNAME-12A only; missing n=2). Separate imputation models were run on the maximal analytical samples for the ACE-III (whole-cohort; n=746) and PACC (Insight 46; n=197) outcomes. The imputation models included the outcome measures available for each sample. For the whole cohort, these were ACE-III total and all ACE-III sub-domains. For Insight 46, these were PACC DSST, PACC FNAME-12A, PACC logical memory delayed recall, PACC MMSE, and ACE-III total. PACC total was passively generated as an average of the four PACC sub-tests, because of missing data for FNAME-12A (n=2), which was imputed using Gaussian normal regression in the Insight 46 sample. For both samples, the imputation models additionally included menopause age and additional covariables included in the analytical models (childhood cognition, childhood SEP, education, age at menarche, parity, menopause type, BMI, smoking, APOE-E4 status, affective symptoms, age at cognitive testing [Insight 46 only], and HT use). No auxiliary variables were included due to the need for simplified imputation models to allow multiple imputation to be combined with bootstrapping (see below). For analyses testing menopause age-by-type and menopause age-by-APOE-E4 interactions, these interactions were added to the imputation.

Where outcome measures were skewed (ACE-III total, ACE-III sub-domains, and MMSE), bootstrapping was applied (Section 2.4.). Bootstrapping was combined with multiple imputation by running the multiple imputation and analysis model on the imputed data within each bootstrap sample. To reduce processing demands 8 imputations were used, as this was deemed sufficient according to Monte Carlo Error estimations (Appendix C Table 1).

## 5i.3. Results

#### 5i.3.1. Participant characteristics

Data for menopause age was available for 1,378 women. Of these, 746 were still in the cohort and completed the ACE-III at age 69, and 197 Insight 46 participants completed the PACC (Figure 14). Two participants did not complete the FNAME-12A assessment due to technical problems and lack of time; multiple imputation was applied to account for these missing data (described above).

Table 20 displays participant characteristics. Mean age at menopause was comparable between the whole-cohort (mean=49.81 years, range 28.75-62.50) and the Insight 46 samples (mean=49.89 years, range 30.25-60.50), and between *APOE*-ɛ4 carriers and non-carriers in the whole-cohort (carriers mean menopause age=49.99 (SD=6.10), non-carriers=49.66 (5.89); t-test p=0.52) and in Insight 46 (carriers mean=49.50 (5.71), non-carriers=50.04 (5.66); t-test p=0.55). Women who had a surgical rather than natural menopause experienced menopause at a younger age (7.70 and 8.06 years younger, on average, for the whole-cohort and Insight 46 participants than in the whole-cohort. Consistent with a previous report,<sup>194</sup> cognitive scores were also generally higher in Insight 46 participants, as was childhood SEP and education.

Most women included in these analyses had used HT (57.72% and 69.63% in the whole-cohort and Insight 46, respectively; Table 20). Of the women who had used HT, most started taking HT between age 46 and 51 years (whole-cohort=60.16%; Insight 46=67.42%; Table 20), and less than 5 years was the most common length of HT use (whole-cohort=48.54%; Insight 46=46.92%; Table 20).

## 5i.3.2. Associations of menopause age with later-life cognitive performance

Among women from NSHD who completed the ACE-III at age 69, positive associations were detected for later menopause age with better task performance on the ACE-III total and across all ACE-III sub-domains (n=746; Figure 15; Appendix C Table 2). In the unadjusted model, each 1-year increase in age at menopause associated with a 0.024 SD increase in standardised ACE-III total score (95% CI 0.012, 0.036), equating to 0.02 additional points for the ACE-III total raw

score (maximum 100). Attention and orientation was the only ACE-III outcome measure for which the unadjusted effect estimate was not significant.

As shown in Figure 15, the effect estimates for all ACE-III outcomes were attenuated after adjustments for early cognition and sociodemographic factors in model 1. Adjustments for reproductive factors in model 2 increased the effect estimates for ACE-III total and the language, verbal fluency, and visuospatial sub-domains. The association of menopause age with ACE-III memory performance was further attenuated and no longer significant after model 2 adjustments. With further adjustment for health-related factors in model 3, no significant associations of menopause age with any ACE-III outcomes remained, but the largest effect estimates were for ACE-III total ( $\beta$ =0.010[-0.004, 0.024]) and the memory ( $\beta$ =0.009[-0.006, 0.023]) and visuospatial function ( $\beta$ =0.013[-0.004, 0.026]) sub-domains.

For women in the Insight 46 sub-sample, later menopause age associated with better task performance on the PACC total and across all PACC sub-tests (n=197; Figure 16; Appendix C Table 3). In the unadjusted model, each 1-year increase in menopause age associated with a 0.029 SD increase in PACC total z-score (95% CI 0.011, 0.048), equating to 0.01 additional points on the PACC total raw score. MMSE was the only PACC sub-test for which the unadjusted effect estimate was not significant.

As shown in Figure 16, adjusting for early cognition and sociodemographic factors in model 1 attenuated the associations with PACC outcomes; no associations remained significant. Additionally adjusting for reproductive factors in model 2 increased the effect estimates for PACC total, DSST, and FNAME-12A. With full adjustments (M3) the largest effects were for FNAME-12A performance, remaining significant ( $\beta$ =0.037[0.005, 0.069]), and for DSST, although non-significant ( $\beta$ =0.031[-0.001, 0.062]). No significant associations remained for PACC total, logical memory delayed, nor MMSE performance.

171



**Figure 15.** Model estimates and bootstrap bias-corrected 95% confidence intervals for the effect of 1-year increase in age at menopause on standardised z-scores for the Addenbrooke's Cognitive Examination (ACE-III; total score and sub-domains) at age 69 in the National Survey of Health and Development (NSHD) whole-cohort. N = 746.



**Figure 16.** Model estimates and 95% confidence intervals for the effect of 1-year increase in age at menopause on standardised z-scores for the Preclinical Alzheimer's Cognitive Composite (PACC; total score and sub-tests) at age 69 to 71 in the Insight 46 sample. N = 197.

DSST = Digit-Symbol Substitution Test; FNAME = Face-Name Associative Memory Examination; MMSE = Mini-Mental State Examination

\*Bootstrap bias-corrected confidence interval.

## 5i.3.3. Effect modification by menopause type and APOE-ε4 status

No interactive effects of menopause age-by-menopause type on standardised ACE-III total (95% CI -0.03, 0.02) nor PACC total (95% CI -0.073, 0.019; p=0.243) performance were detected (Appendix C Table 4). Additionally, no interactive effects of menopause age-by-*APOE*- $\epsilon$ 4 status on ACE-III total (95% CI -0.009, 0.006) nor PACC total (95% CI -0.039, 0.037; p=0.949) were detected (Appendix C Table 5).

# *5i.3.4.* The role of menopausal hormone therapy

Compared with fully adjusted models (M3), further adjusting for HT use (M4) had little impact on the effect estimates for any outcomes (Figures 15 and 16; Appendix C Tables 2 and 3). There was no evidence of associations between HT use and ACE-III total nor PACC total performance (Appendix C Table 6).

# *5i.3.5.* Supplementary and sensitivity analyses

Individually adjusting for each model 1 covariable (childhood cognition, childhood SEP, education) revealed that the attenuation of effect estimates was driven by childhood cognition (Appendix C Table 7).

Where negative confounding in model 2 was observed (ACE-III: total, language, verbal fluency, visuospatial function; PACC: total, DSST, FNAME-12A), individually adjusting for reproductive covariables (age at menarche, parity, menopause type) showed that the negative confounding was driven by menopause type (Appendix C Table 8). No menopause age-by-menopause type interactions were detected on these outcomes (Appendix C Table 9). Unadjusted regression models of menopause type on cognitive outcomes showed poorer cognitive performance in women with surgical compared with natural menopause, and these associations were negatively confounded when adjusting for menopause age (Appendix C Table 10).

Excluding women with possible cognitive impairment (total ACE-III score <82) did not substantially change the effect estimates for the association of menopause age with ACE-III total, although the estimates for PACC total were slightly attenuated (Appendix C Tables 11-14). Similarly, excluding women with bilateral oophorectomy did not substantially change the

effect estimates for the ACE-III total outcome, while the estimates for PACC total were slightly attenuated (Appendix C Tables 15-17).

# 5i.4. Discussion

## 5i.4.1. Key findings

Later menopause age associated with better cognitive performance in later-life across all outcomes. Associations were strongest for the ACE-III memory and visuospatial function sub-domains, and the PACC DSST and FNAME sub-tests, demonstrating relationships of menopause age with visual processing, and associative learning and memory domains. Adjusting for early-life factors attenuated all effect estimates, driven by childhood cognition, and accounting for menopause type revealed negative confounding for some outcomes. No significant interactions with menopause type or *APOE*-ɛ4 status were detected, and further adjustment for HT use did not meaningfully alter the estimated effects.

## *5i.4.2.* Interpretation of findings

These findings support previous evidence for positive associations between age at menopause and cognitive outcomes,<sup>154,157,321</sup> with prolonged exposure to the neuroprotective benefits of endogenous oestrogen a hypothesised mechanism.<sup>305</sup> Additionally, these findings are consistent with previous evidence of small positive associations between menopause age and verbal memory performance in NSHD.<sup>157</sup> During the menopause transition, women often report memory problems,<sup>144</sup> and oestrogen receptors are found in high concentrations within the hippocampus, a brain region important for learning and memory.<sup>305</sup> However, the associations of menopause age with memory performance were not consistent across different memory assessments; the association with delayed episodic memory (logical memory delayed recall) in the Insight 46 sample was not particularly strong compared with other outcome measures. There are several memory types (e.g. episodic, associations between menopause age and memory assessed in the sub-tests included in these analyses. Associations between menopause age and memory assessed.

The association with a measure of processing speed in the Insight 46 cohort contrasts with previous evidence from NSHD where associations with processing speed were not detected.<sup>157,321</sup> The relationships could differ in the Insight 46 sub-sample compared with the whole-cohort, particularly given demographic differences.<sup>194</sup> However, the processing speed measure completed by Insight 46 participants (DSST) differs from the letter cancellation task completed between age 43 and 69 in the whole-cohort; <sup>157,321</sup> the DSST includes an associative learning component. The relationship with a measure of associative memory (FNAME-12A) in the Insight 46 cohort might also suggest that menopause age links with associative learning in later-life. Additionally, both the DSST and FNAME-12A are reliant on visual processing, in agreement with the whole-cohort association with visuospatial function.

As previously shown,<sup>157,321</sup> most associations were not independent of life course covariables and childhood cognition was a particularly important factor. Adjustment for childhood cognition, which predicts both menopause age<sup>156,157</sup> and later-life cognitive performance,<sup>158,159</sup> most strongly attenuated the associations compared with other covariables such as SEP and education. Upstream, developmental factors (e.g. genetic factors, pre-natal exposures, early-life experiences)<sup>157,326</sup> giving rise to childhood cognition could link the timing of menopause with later-life cognitive outcomes, and childhood cognition could be a proxy indicator of lifetime oestrogen exposure.<sup>156</sup>

The negative confounding by menopause type likely reflects negative associations of surgical, compared with natural, menopause and cognitive outcomes,<sup>151</sup> which contrasts the positive associations between menopause age and cognitive performance. Interestingly, the association between menopause age and memory performance at age 69 was explained by other reproductive factors, including menopause type; menopause type rather than timing could be important for memory in later-life, although surgical menopause is synonymous with earlier menopause. Differences in menopause age-cognition associations by menopause type were not detected, but there are differences between natural and surgical menopause beyond menopause timing. Surgeries inducing menopause lead to more rapid declines in oestrogen levels than during natural menopause; removal of both ovaries results in the most acute cessation of ovarian oestrogen production.<sup>142</sup> Nonetheless, excluding women who had a bilateral oophorectomy did not substantially change the results. Women with surgical menopause are also more likely to use HT, and to have poorer overall health and lower SEP

than women who have a natural menopause.<sup>327,328</sup> Whether and how the associations of menopause age with later-life cognitive outcomes might differ by menopause type is a complex topic which requires further investigation utilising different and larger cohorts. Understanding the potential impacts of surgical menopause on later-life outcomes, including cognitive abilities, will be important for informing clinical guidance encompassing the risks and benefits of such surgical interventions.

Additionally, the relationships between menopause age and later-life cognitive performance were not modified by *APOE*- $\varepsilon$ 4 status. Although there are some reports of earlier<sup>319</sup> and later<sup>320</sup> menopause timing in carriers than non-carriers, menopause age did not differ by *APOE* genotype in NSHD and Insight 46 women. Since adverse effects of *APOE*- $\varepsilon$ 4 tend to be more evident at older ages, continued follow-up of the NSHD and Insight 46 cohorts will be valuable to further examine the potential for interactions between menopause timing and *APOE* genotype. However, Chapter 3 analyses did reveal some sex-by-*APOE*- $\varepsilon$ 4 interactions on cognitive performance at older ages (53 onwards). It is possible that menopause timing is not an underlying mechanism exacerbating  $\varepsilon$ 4-associated dementia risk in females.

# *5i.4.3. Strengths and limitations*

The main strength of this work is the use of longitudinal, prospective life course data which provides a unique opportunity to account for early-life confounds such as childhood cognition. The age homogenous cohort is particularly beneficial given that the exposure variable (menopause age) was age dependent. However, the generalisability of these results to other generations could be limited given secular changes in women's access to education and in HT use, for example. While the potential contribution of HT use in relation to associations between menopause age and later-life cognition has been considered, with little evidence for potential mediating effects, differentiating between the different types of HT used was not possible. There is a need for more in-depth analyses of HT and cognition beyond the scope of these analysis, which might benefit from the inclusion of additional data sets, given potential variations according to dosage, formulation, duration of use, and age when HT is initiated.<sup>329</sup> While the effects reported here are small, these are consistent with other research, and some residual associations were still detected after adjustments. In future research, follow-up cognitive assessments in this cohort will facilitate further examination of associations

between menopause and cognitive decline and, as explored in Section 5ii, the availability of neuroimaging data within Insight 46 provides an opportunity to examine the potential neural mechanisms underlying these associations.

# 5i.4.4. Summary

These analyses provide further evidence that later age at menopause associates with better cognitive performance in later-life and identify that the associations are most notable for visual processing, and associative learning and memory domains. However, life course covariables, particularly childhood cognition, contribute to associations. Such factors are therefore important to consider when examining the potential mechanisms underlying relationships between menopause and female cognitive ageing.

# 5ii. Menopause and later-life brain health

The purpose of this section was to test relationships between age at natural menopause and brain health at age ~70, across a range of multi-modal neuroimaging measures. Women from Insight 46 who had a natural menopause, had available data for their age at menopause, and who completed PET-MRI imaging at Insight 46 wave I were included. Multivariable regression analyses tested non-linear associations, effect modifications by APOE-E4 status, and accounted for relevant early cognitive, sociodemographic, reproductive, and health-related covariables. There was evidence for an inverted-U non-linear relationship of menopause age with AD pathology (A $\beta$ ), and later menopause age positively (and linearly) associated with larger TBV (indicating reduced non-specific brain ageing), but menopause age was not associated with hippocampal volume or measures of NAWM microstructural integrity. Some associations were modified by APOE genotype; in ɛ4 carriers but not in non-carriers, later menopause age associated with increased cortical thickness and reduced markers of cSVD (WMHV). Associations were mostly independent of life course covariables, including HT use, but blood pressure and smoking at age 36 partially explained associations with TBV and WMHV, respectively. Overall, these findings indicate that later age at natural menopause could be beneficial for later-life brain health, particularly in APOE-E4 carriers. An alternative, but complementary, interpretation is that earlier age at natural menopause is detrimental for later-life brain health.

#### 5ii.1. Introduction

Given that menopause age is associated with later-life cognitive performance measures (Section 5i.), it is plausible that brain structures and functions which subserve cognitive abilities will also be associated with menopause age. Indeed, menopause can be considered a neurological transition, given the neurological nature of some menopausal symptoms (e.g. hot flushes, sleep disturbance, brain fog).<sup>8</sup> Oestrogen also supports neural functioning and has neuroprotective properties (e.g. neuronal maintenance, Aβ production inhibition),<sup>84</sup> hence declining oestrogen levels during menopause are hypothesised to increase the vulnerability of the female brain to neuropathology and dementia processes. The female-specific process of menopause could therefore offer some explanation for greater dementia risk in females. However, the relationship between menopause and dementia-related brain health in later-life is not yet fully understood.

Some studies have demonstrated variations in brain health across menopause phases. Perimenopausal women have showed smaller grey and white matter volumes, and lower energy metabolism than pre- and post-menopausal women, while post-menopausal women showed similar or greater brain volumes and energy metabolism levels than pre-menopausal women.<sup>163</sup> Longitudinally though, evidence shows faster brain ageing in post-menopausal women, with faster declines in TBV and glucose metabolism, and faster increases in WMHV and Aβ deposition than age-matched men and pre-menopausal women.<sup>162,330</sup> This supports the hypothesis that menopause accelerates neural ageing processes.

Although meta-analysis does not support an overall association of menopause age with dementia risk, several individual studies associated later menopause age with lower dementia risk,<sup>154</sup> and consistent positive associations with cognitive performance are reported,<sup>154,156,157,317,331</sup> including in this cohort (Section 5i.). Only a few studies have examined menopause age and brain health, reporting mixed results. In the UK Biobank, later age at natural menopause associated with smaller TBV and hippocampal volumes,<sup>164</sup> and with younger predicted brain ages.<sup>332</sup> In a small study (n=35), TBV was not linked with menopause age but poorer cerebrovascular function and greater WMHV were detected in women with late ( $\geq$ 53 years) compared with early ( $\leq$ 49 years) natural menopause.<sup>333</sup> For some, but not all, women, menopause could be a 'tipping point' for accelerated brain ageing and the development of pathology.<sup>318</sup> Indeed, peri- and post-menopausal *APOE*- $\epsilon$ 4 carriers are shown
to, independently of age, have higher A $\beta$  levels than pre-menopausal and male  $\epsilon$ 4 carriers.<sup>163</sup> While evidence linking *APOE*- $\epsilon$ 4 status and menopause age is mixed,<sup>319,320</sup> ApoE is shown to have a role in ovarian sex hormone production and in the neural effects of oestrone.<sup>334</sup> Interactions of menopausal hormone changes and *APOE*- $\epsilon$ 4 status on brain health are yet to be examined and, although Section 5i analyses did not find evidence for menopause age-by-*APOE*- $\epsilon$ 4 interactions on later-life cognitive performance, such investigations are warranted.

Whether menopausal HT confers benefits or risks to later-life dementia-related outcomes is also unclear<sup>170</sup> and neuroimaging outcomes from RCTs are inconsistent<sup>179,335</sup> (Section 1.11.5.). Discrepant HT findings can be attributed to heterogeneity in dosages, formulations, duration of use, and the timing of HT initiation in relation to menopause. Given uncertainty over how HT might influence brain health outcomes, most research assessing menopause and brain health excludes women who have used HT, meaning that the potential role of HT in menopause-brain health associations is unclear.

# *5ii.1.1. Objectives and research questions*

This work aims to examine the associations of menopause age with a range of multimodal neuroimaging markers of brain health reflecting dementia- (A $\beta$ , hippocampal volume), non-specific ageing- (total brain volumes, cortical thickness), and vascular-related (NAWM microstructural integrity, WMHV) pathways in Insight 46 participants at age ~70 who underwent a natural menopause.

Learning from the previous analyses in Section 5i, there are some variations in the methodological approach taken here, namely that women who underwent surgical menopause are now excluded. This is in recognition of the distinctions between natural and surgical menopause; women with surgical menopause have underlying etiologies indicating a need for surgery, are more likely to use HT, and by definition have earlier ages at menopause than women who go through natural menopause.<sup>336</sup> Additionally, the possibility of non-linear associations is recognised given mixed findings in the relatively little research which has examined associations of menopause age with neuroimaging outcomes;<sup>332,333</sup> non-linear associations of menopause age with brain-health outcomes are now examined prior to subsequent analyses.

The following research questions are addressed:

- 1. Are there linear or non-linear associations of age at natural menopause with neuroimaging markers of brain health at age ~70?
- 2. Are associations moderated by APOE-E4 status?
- 3. Are associations independent of relevant early-life, reproductive, and health-related covariables?
- Does HT use contribute to associations of menopause age with neuroimaging markers of brain health at age ~70?

# 5ii.1.2. Hypotheses

Previous examinations of menopause age and brain health are scarce, but given positive associations with cognitive performance, it follows that later menopause age will also positively associate with brain health, particularly given evidence for faster brain ageing in post-menopausal women. As observed in Section 5i, life course covariables were expected to attenuate effect estimates, with little mediating effect of HT use. While effect modifications by *APOE*- $\epsilon$ 4 status were not detected in Section 5i, some evidence linking menopause status and *APOE* genotype with A $\beta$  levels leads to the hypothesis that relationships between menopause age and brain health, particularly AD-related measures, may be stronger in  $\epsilon$ 4 carriers than in non-carriers.

## 5ii.2. Analytic method

## 5ii.2.1. Analytic sample

Insight 46 females who also completed Women's Health in the Middle Years Survey questionnaires (Section 2.2.1.1.) were included in these analyses if they reported a natural menopause, had available data on their age at menopause, and if they underwent neuroimaging at wave I of Insight 46 data collection, generating a maximal analytical sample of 126 females (Figure 17).



*Figure 17.* Flow chart demonstrating sample selection for analyses. Scan refers to combined amyloid PET-MRI.

# 5ii.2.2. Menopause age

As in Section 5i., age at menopause was prospectively self-reported as age (years) since birth that menstrual periods stopped. Women indicated, via questionnaire, whether their periods had stopped naturally or if there was another reason for their cessation (e.g. surgery to remove the ovaries, Section 5i.2.2.).

## *5ii.2.3. Neuroimaging outcome measures*

As outlined in Section 2.3.3., Insight 46 participants underwent combined PET-MRI neuroimaging aged ~70. For these analyses, the following outcomes were of interest: TBV, hippocampal volume, WMHV, continuous A $\beta$  SUVR, A $\beta$  status, NAWM microstructural integrity measures (FA and MD), and cortical thickness across ROI (frontal, occipital, parietal, temporal lobes, and the Harvard AD signature region).<sup>220</sup>

For analyses assessing WMHV as an outcome, participants who failed BaMoS QC were excluded (n=3). Participants whose scans failed NAWM QC (n=16) were excluded from NAWM analyses.

## 5ii.2.4. Covariables

Lifetime covariables were selected based on previous analyses linking menopause age with later-life outcomes in this cohort<sup>317</sup> and on associations of covariables with menopause age and brain health measures.<sup>158,326</sup> Childhood cognition at age 8 years (Section 2.3.4.4.) was standardised to the sample at the time of testing. If data were missing for cognition at age 8, available data for cognition at age 11 (n=7), or at age 15 years (n=1) were instead used. Educational attainment up to age 26 (Section 2.3.4.2.) was categorised as ordinary (GCSE-level or equivalent) or below, or advanced (A-level or equivalent, or above). Adult SEP up to age 53 (Section 2.3.4.3.) was dichotomised to manual and non-manual. In contrast to Section 5i analyses which included a measure of childhood SEP, adulthood SEP is instead used here given evidence linking mid- but not early-life SEP with volumetric measures of brain health.<sup>337</sup> Age at menarche was recorded as years since birth, according to school doctor- or self-report (Section 5i.2.4.). Parity was indicated by self-reported number of biological children, excluding still births and miscarriages, coded as an ordinal variable. As in 5i. (Section 5i.2.4.), BMI and smoking pack years at age 36 were included as covariables, APOE-E4 status was categorised as  $\epsilon$ 4 present or absent, and menopausal HT use by age 69 was categorised as ever or never used. Blood pressure at age 36 was included as an additional covariable in these analyses, given reported relationships between vascular health and brain health (specifically cSVD) which are particularly notable in females.<sup>40</sup> Seated blood pressure at age 36 was measured in the upper arm twice after 5 minutes of rest, using a Hawksley Random Zero sphygmomanometer.

# 5ii.2.5. Statistical analyses

Analyses were conducted using Stata version 17.0.

Multivariable regression analyses were used to assess the associations of menopause age with brain health measures (TBV, hippocampal volume, continuous SUVR, NAWM MD and FA measures, and all cortical thickness ROI). Multivariable logistic regression was used to

184

examine the odds of being amyloid positive given each 1-year increase in menopause age. The association of menopause age with WMHV, which has a skewed distribution, was examined using a generalised linear model with gamma distribution log link. All minimally adjusted models (M0) were adjusted for age at scan. Models were also adjusted for TIV for volumetric measures (TBV, hippocampal volume, and WMHV).

Firstly, non-linear associations were examined by including a quadratic term for menopause age in fully adjusted models (M0+childhood cognition, education, SEP, puberty, parity, BMI, smoking and blood pressure at age 36, and *APOE*- $\epsilon$ 4 status). Where non-linear associations were detected (p<0.1), subsequent analyses included a quadratic term for menopause age, otherwise subsequent analyses used linear models.

The potential moderating role of *APOE*- $\epsilon$ 4 status was examined by testing for menopause ageby-*APOE*- $\epsilon$ 4 interactions in fully adjusted models. Where interactions were significant (p<0.1), *post-hoc* analyses were conducted, stratifying by *APOE*- $\epsilon$ 4 status.

The contribution of life course covariables was examined by cumulatively adjusting for earlylife (M1:M0+childhood cognition, education, SEP), reproductive (M2:M1+puberty, parity), and health-related (M3:M2+smoking, BMI, and blood pressure at age 36, *APOE*-ɛ4 status) factors.

To examine whether HT contributed to associations of menopause age with brain health, HT use was further adjusted for (M3+HT). The relationship of HT use with neuroimaging outcomes was also examined in minimally adjusted models, and in models adjusting for menopause age, including HT use as the predictor variable where never use was the reference category.

Given evidence for sex differences in WMHV where women typically have greater WMHV levels than men (Chapter 3), and evidence that WMHV predicts cortical thickness in this cohort,<sup>219</sup> sensitivity analyses were ran whereby all models for non-WMHV outcomes were adjusted for WMHV. Further sensitivity analyses examined whether the main findings could be replicated when excluding the twelve women with clinically diagnosed neurological conditions (Table 21), since neurological changes associated with such conditions (e.g. greater WMHV in multiple sclerosis patients) could influence the estimated effects of menopause age on neuroimaging outcomes. Additionally, since only two women in the sample had an early menopause (aged <45 years), analyses were re-ran excluding these participants.

185

Multiple imputation with 50 imputations was used to impute missing data in covariables (Section 2.4.2.). Separate imputation models were run on the maximal analytic sample for the main analyses (n=126) and for the sensitivity analyses excluding 12 women with clinically diagnosed neurological conditions (n=114). The imputation models included all variables used in the analytic models (menopause age, all outcome variables, all covariables), quadratic terms for menopause age, interaction terms for menopause age and *APOE*- $\epsilon$ 4 status, and auxiliary variables selected to improve predictions of imputed values (diastolic and systolic blood pressure at age 63, diastolic and systolic blood pressure at lnsight 46 wave I, childhood social class, and smoking pack years at age 63). For each sample, missing values were imputed using Gaussian normal regression for TIV, age at scan, childhood cognition, puberty age, smoking pack years at age 36, BMI at age 36, and diastolic and systolic blood pressure at age 36. Missing values were imputed using logistic regression for adult social class, education, *APOE*- $\epsilon$ 4 status, and HT use. Ordinal logistic regression was used to impute missing values for number of natural-born children.

#### 5ii.3. Results

#### *5ii.3.1. Participant characteristics*

The maximal sample, summarised in Figure 17, was 126 women with known age at natural menopause and who had undergone neuroimaging at Insight 46 wave I. Participant characteristics for the analytical sample are presented in Table 21. Mean age at menopause was not statistically different between *APOE*-ε4 carriers (mean 51.93 years, SD=2.84, n=36) and non-carriers (mean 52.67 years, SD=3.41, n=90) (t=1.14, p=0.256).

Most women in the sample had ever used HT (60.3%; Table 21). Of those who had used HT, most started HT between age 46 and 51 (69.5%; Table 21) and most took HT for less than 5 years (60.6%; Table 21).

Variable	Ν	Mean(SD)/%	Range
Menopause age (years)	126	52.5(3.3)	40.5, 59.5
TBV (cm <sup>3</sup> )	126	1043.6(82.2)	818.6,
			1265.2
Hippocampal volume (cm <sup>3</sup> )	126	3.0(0.3) 2.3, 3.7	
WMHV (cm <sup>3</sup> )	123	5.1(5.4)	0.4, 32.8
SUVR	126	0.6(0.1)	0.5, 0.9
Aβ positive	20	15.9%	-
Aβ negative	106	84.1%	-
NAWM MD (z-score)	115	0.2(0.3)	-0.6, 1.2
NAWM FA (z-score)	115	-0.1(0.2)	-0.8, 0.5
CT: ADsig Harvard(mm)	126	2.7(0.1)	2.4, 2.9
CT: frontal(mm)	126	2.8(0.1)	2.4, 2.9
CT: occipital(mm)	126	2.2(0.1)	1.9, 2.4
CT: parietal(mm)	126	2.5(0.1)	2.3, 2.7
CT: temporal(mm)	126	2.9(0.1)	2.6, 3.1
TIV (cm <sup>3</sup> )	126	1340.0(92.9)	1129.8,
			1558.1
Age at scan (years)	126	70.7(0.7)	69.5, 71.8
Childhood cognition (z-score)	126	0.5(0.8)	-1.6, 2.5
SEP age 53			
Manual	11	8.7%	-
Non-manual	115	91.3%	-
Educational attainment to age 26			
Ordinary (GCSE-level/equivalent or below)	61	48.8%	-
Advanced (A-level/equivalent or above)	64	51.2%	-
Age at menarche (years)	107	13.0(1.3)	9.9, 17.5
Number of natural-born children			
0	15	12.0%	-
1	11	8.8%	-
2	64	51.2%	-
3	28	22.4%	-
4	7	5.6%	-
Smoking pack years age 36	117	0.9(1.7)	0, 6.9
BMI age 36 (kg/m <sup>2</sup> )	117	22.8(2.9)	17.2, 36.1
DBP age 36 (mmHg)	116	74.1(11.6)	43.6, 113.6
SBP age 36 (mmHg)	116	116.7(12.8)	87.6, 150.3
APOE-e4 status			
ε4 non-carrier	90	71.4%	-
ε4 carrier	36	28.6%	-

**Table 21.** Participant characteristics for women with available age at natural menopause and neuroimaging outcome data, based on complete case data.

HT use			
Never	48	39.7%	-
Ever	73	60.3%	-
For HT users, age at first use			
≤45	6	8.3%	-
46-51	50	69.5%	-
≥52	16	22.2%	-
For HT users, HT duration			
<5 years	43	60.6%	-
5+ years	28	39.4%	-
Clinical neurological condition			
None	114	90.5%	-
Depression	3	2.4%	-
Epilepsy	2	1.6%	-
MS	2	1.6%	-
PD	2	1.6%	-
Stroke	2	1.6%	-
MCI	1	0.8%	-

Descriptive statistics for continuous variables are presented as mean, standard deviation, and range. For categorical variables, the percentage of participants in each category are presented.

TBV=total brain volume; WMHV=white matter hyperintensity volume; SUVR=standarised uptake value ratio; Aβ=beta-amyloid; NAWM=normal appearing white matter; MD=mean diffusivity; FA=fractional anisotropy; CT=cortical thickness; ADsig=Alzheimer's Disease signature; TIV=total intracranial volume; SEP=socioeconomic position; BMI=body mass index; DBP=diastolic blood pressure; SBP=systolic blood pressure; APOE=apolipoprotein e; HT=hormone therapy; MS=multiple sclerosis; PD=Parkinson's Disease; MCI=mild cognitive impairment

# 5ii.3.2. Amyloid

There was evidence for a non-linear association of menopause age with SUVR ( $\beta$ =-0.001[95% CI -0.002,0.000], p=0.02; Appendix C Table 18) and with amyloid status (OR=0.91[0.83,1.00], p=0.06; Appendix C Table 18), but only marginal evidence for *APOE* interactions (SUVR p=0.10; amyloid status p=0.78; Appendix C Table 19). Women who had menopause earlier or later had lower SUVR (Figure 18) and less likelihood of being amyloid positive than women who had menopause in the mid-range. Cumulatively adjusting for covariables and HT use did not alter the estimated associations (Appendix C Table 21), and there were no associations of HT use with SUVR or A $\beta$  status (Appendix C Table 24).

## 5ii.3.3. WMHV

There was no evidence for a non-linear association of menopause age on WMHV (p=0.79; Appendix C Table 18), but there was a significant modifying effect of *APOE*- $\varepsilon$ 4 status (p=0.03; Figure 19; Appendix C Table 20). *APOE*- $\varepsilon$ 4 carriers (n=35) showed a 12% decrease in WMHV per 1-year increase in menopause age (M0 relative WMHV change=0.88[0.79,0.98], p=0.03) while non-carriers (n=88) showed a non-significant 4% WMHV increase (M0 relative WMHV change=1.04[0.98 1.10], p=0.21). Further adjusting for covariables (Figure 19; Appendix C Table 22) did not substantially alter the effect estimates for  $\varepsilon$ 4 non-carriers, but in carriers there was an effect attenuation when adjusting for health-related factors (M3). Individually adjusting for M3 covariables demonstrated that this effect attenuation was mostly driven by adjustment for smoking (Appendix C Table 25). Additional adjustment for HT use did not alter the model estimates for  $\varepsilon$ 4 carriers or non-carriers, and there was no association of HT use with WMHV (Appendix C Table 24).



**Figure 18.** Predicted values and 95% confidence intervals for the non-linear association of age at natural menopause with amyloid SUVR, based on the fully adjusted model assuming mean values for continuous variables and reference categories for categorical variables (n=126).

Model adjusted for: age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure, APOE- $\epsilon$ 4 status

SUVR=standardised uptake value ratio; SEP=socioeconomic position; BMI=body mass index; APOE=apolipoprotein-e

#### 5ii.3.4. TBV

There was no evidence for a non-linear association of menopause age with TBV (p=0.57; Appendix C Table 18) or for effect modification by *APOE*- $\varepsilon$ 4 status (p=0.19; Appendix C Table 20). There was an overall positive association for increasing menopause age with greater TBV (Figure 20; M0: $\beta$ =3.17[0.72,5.62], p=0.01), attenuated by adjustment for health-related covariables (Appendix C Table 23). Individually adjusting for M3 covariables revealed that this attenuation was mostly driven by blood pressure adjustments (Appendix C Table 26). Further adjustment for HT use did not alter the estimated association of menopause age with TBV (Appendix C Table 23), and there was no association of HT use with TBV (Appendix C Table 24).



**Figure 19.** Model estimates and 95% confidence intervals demonstrating the predicted relative change in white matter hyperintensity volume (WMHV) per 1-year increase in menopause age, for APOE- $\varepsilon$ 4 carriers (n=35) and non-carriers (n=88), across models cumulatively adjusted for life course covariables.

M0:age at scan, TIV; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

WMHV=white matter hyperintensity volume; HT=hormone therapy; TIV=total intracranial volume; SEP=socioeconomic position



**Figure 20.** Model estimates and 95% confidence intervals demonstrating the predicted change in total brain volume (TBV) per 1-year increase in menopause age, for the pooled analytical sample (n=126) across models cumulatively adjusted for life course covariables.

M0:age at scan, TIV; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ɛ4 status

TBV=total brain volume; HT=hormone therapy; TIV=total intracranial volume; SEP=socioeconomic position

#### 5ii.3.5. Hippocampal volume

There was no evidence for non-linear associations (Appendix C Table 18), *APOE*-ε4 interactions (Appendix C Table 20), or for any meaningful association of menopause age with hippocampal volume (Appendix C Table 23). HT use was not associated with hippocampal volume (Appendix C Table 24).

# 5ii.3.6. NAWM

There was no evidence for non-linear associations (Appendix C Table 18), *APOE*-ε4 interactions (Appendix C Table 20), or associations of menopause age with NAWM FA or MD (Appendix C Table 23). HT use was not associated with NAWM measures (Appendix C Table 24).

## 5ii.3.7. Cortical thickness

There was no evidence for non-linear associations of menopause age with cortical thickness in any ROI (all p>0.1; Appendix C Table 18), but linear associations with all ROI were modified by *APOE*- $\epsilon$ 4 status (p<0.1; Appendix C Table 20). Across all ROI, *APOE*- $\epsilon$ 4 carriers (n=36) showed greater cortical thickness with increasing age at menopause (Table 22; Figure 21) while in non-carriers (n=90) cortical thickness was reduced with increasing menopause age (Table 22; Figure 21). For the frontal, occipital and temporal ROI, model estimates did not substantially change with cumulative adjustments for life course covariables (Figure 21, Appendix C Table 22). However, in  $\epsilon$ 4 carriers the positive association of menopause age with cortical thickness in the Harvard ADsig ROI was attenuated with adjustment for reproductive factors, driven by adjustment for parity, and the positive association with parietal cortical thickness was attenuated with adjustment for health-related factors (Figure 21, Appendix C Tables 27 and 28). Additionally adjusting for HT use did not substantially alter the effect estimates for  $\epsilon$ 4 carriers or non-carriers across all ROI (Figure 21), and HT use did not associate with cortical thickness in any ROI (Appendix C Table 24). **Table 22.** Model estimates and 95% confidence intervals demonstrating the predicted change in cortical thickness regions of interest (ROI), based on minimally adjusted models (M0), for APOE-ε4 carriers (n=36) and non-carriers (n=90).

	ROI (mm)	β	Lower 95% Cl	Upper 95% Cl	P-value
	Harvard ADsig	0.009	0.002	0.017	0.02
	Frontal	0.007	-0.003	0.016	0.13
ε4 carriers	Occipital	0.006	-0.004	0.017	0.24
	Parietal	0.011	0.004	0.019	<0.01
	Temporal	0.010	-0.001	0.020	0.06
	Harvard ADsig	-0.004	-0.009	0.001	0.09
ε4 non-	Frontal	-0.003	-0.009	0.002	0.27
carriers					
	Occipital	-0.003	-0.009	0.002	0.20
	Parietal	-0.003	-0.008	0.001	0.16
	Temporal	-0.004	-0.010	0.002	0.17

M0:age at scan, TIV

ROI=regions of interest; CI=confidence interval; ADsig=Alzheimer's Disease signature region



**Figure 21.** Model estimates and 95% confidence intervals demonstrating the predicted change in cortical thickness regions of interest (ROI; A=Harvard ADsig, B=Frontal, C=Occipital, D=Parietal, E=Temporal) per 1-year increase in menopause age, for APOE- $\varepsilon$ 4 carriers (n=36) and non-carriers (n=90), across models cumulatively adjusted for life course covariables.

M0:age at scan; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ɛ4 status[pooled analyses only]

ADsig=Alzheimer's Disease signature region; HT=hormone therapy; SEP=socioeconomic position

#### *5ii.3.8. Sensitivity analyses*

When adjusting non-WMHV outcome models for WMHV, the results did not differ from the main analyses (Appendix C Tables 29-31).

In analyses excluding the twelve women with neurological conditions (Appendix C Tables 32-37), results for amyloid SUVR, WMHV, hippocampal volume, and most cortical thickness ROI did not substantially differ from the main analyses. For amyloid status there was no longer evidence for a non-linear association with menopause age (n=114; p=0.23). For TBV, a modifying effect of *APOE*- $\epsilon$ 4 status was detected (n=114; p=0.08) whereby the positive association of later menopause age with greater TBV was stronger in  $\epsilon$ 4 carriers (n=35) than in non-carriers (n=79). There was also evidence of an *APOE*- $\epsilon$ 4 interaction for NAWM measures (n=102; FA p=0.08, MD p=0.02). In  $\epsilon$ 4 carriers (n=30), no associations of menopause age with FA or MD were detected, but in non-carriers (n=72) later menopause age associated with reduced FA and increased MD. Therefore, in  $\epsilon$ 4 non-carriers increasing menopause age associated with poorer NAWM microstructural integrity. There was no longer a significant *APOE*- $\epsilon$ 4 interaction for frontal cortical thickness (n=114; p=0.10).

When excluding the two women who had early menopause (aged <45 years; Appendix C Tables 38-43), non-linear associations of menopause age with SUVR and amyloid status were replicated, as were menopause age-by-*APOE*- $\epsilon$ 4 interactive effects on WMHV and cortical thickness measures, except for frontal cortical thickness (n=124; p=0.111). As in the main analyses, no associations of menopause age with hippocampal volume or NAWM FA were detected. An interactive effect of menopause age and *APOE*- $\epsilon$ 4 on TBV was detected (n=124; p=0.080) whereby  $\epsilon$ 4 carriers (n=36) showed a stronger positive relationship between later menopause age and greater TBV than non-carriers (n=88). An effect modification by *APOE*- $\epsilon$ 4 was also detected for NAWM MD (n=109; p=0.081); in non-carriers (n=78) but not in carriers (n=31), later menopause age associated with greater MD.

#### 5ii.4. Discussion

## 5ii.4.1. Key findings

As expected, based on previously reported associations of later menopause with better cognitive performance (Section 5i.),<sup>154-156,317</sup> overall positive associations of natural

menopause age with better brain health at age ~70 were found, such as greater total brain volumes with increasing menopause age. Some associations were modified by *APOE*- $\epsilon$ 4 genotype. Cortical thickness, for example, was positively associated with menopause age in  $\epsilon$ 4 carriers while non-carriers showed non-significant negative associations. Overall, associations were independent of life course covariables, including HT use. However, midlife blood pressure and smoking partially explained some associations.

# 5ii.4.2. Interpretation of findings

These findings generally support the hypothesis that prolonged exposure to endogenous oestrogen with increasing menopause age is beneficial for later-life brain health. However, HT use - which indicates exogenous oestrogen exposure - was not found to contribute to menopause age associations with brain health, nor was HT use associated with the neuroimaging outcomes assessed. Variations in HT dosages, administration routes, formulations, durations, and timing of initiation in relation to menopause are sources of discrepancy within HT literature<sup>170,329</sup> and require further examination before conclusions can be drawn on the effects of HT on later-life brain health.

However, it is surprising that no association with hippocampal volume - a key region implicated in AD<sup>338</sup> - was detected despite previous evidence demonstrating the hippocampi as oestrogen-sensitive regions<sup>339,340</sup> and that, in this cohort, a cognitive domain most strongly linked with menopause age is memory (Section 5i.),<sup>317</sup> which is subserved by the hippocampi.<sup>338</sup> Although replication and further investigation is needed, this finding could indicate that menopause age is not associated with AD-specific regional pathology, rather with more global measures of brain health.

Later menopause age was linked with greater TBV, generally a positive indicator of brain health since TBV reduces with advancing age and in dementia,<sup>341</sup> contrasting previous reports of null or negative associations.<sup>164,333</sup> However, previous studies differed from the current analyses as they: excluded HT users, did not include women with menopause age younger than 45, and had larger age ranges at neuroimaging. The association detected with TBV here was slightly attenuated by adjustments for health-related factors, primarily driven by blood pressure at age 36. While evidence shows increases in blood pressure during and post-menopause,<sup>342</sup>

197

examination of pre-menopause blood pressure in relation to menopause age is scarce. Findings elsewhere do, however, demonstrate that higher blood pressure typically associates with smaller TBV,<sup>98,343,344</sup> which could explain the effect attenuation observed here.

In the existing literature, menopause age has not been examined in relation to  $A\beta$ , but these analyses demonstrate evidence for non-linear associations whereby continuous  $A\beta$  levels and the likelihood of being  $A\beta$  positive were lowest in women with the earliest and latest ages at menopause. Reasons for this finding are unclear and could be driven by methodological factors such as using a white matter reference region; these results require replication and further examination.

Although no moderating effects of *APOE*- $\varepsilon$ 4 status for amyloid measures were detected, ApoE - a lipid transporter protein closely linked with cholesterol<sup>92</sup> - is known to have a role in Aβ deposition and metabolism,<sup>92,345</sup> and *APOE*- $\varepsilon$ 4 genotype is linked with higher Aβ levels.<sup>346</sup> The evidence linking *APOE*- $\varepsilon$ 4 genotype with menopause age is mixed<sup>319,320</sup> and, in this sample, no significant differences in menopause age between  $\varepsilon$ 4 carriers and non-carriers were detected. Nonetheless, some findings report links between ApoE, female reproductive factors, and brain health, indicating potential mechanisms through which *APOE* genotype might interact with menopause-brain associations. For instance, cholesterol levels are shown to increase across the menopause transition and post-menopause.<sup>348</sup> ApoE is also shown to facilitate neuroprotective effects of oestrogen while oestrogen can also stimulate ApoE production in the brain.<sup>349</sup>

For cortical thickness and WMHV, stronger positive associations of menopause age with brain health were detected in  $\varepsilon$ 4 carriers than in non-carriers. Given that  $\varepsilon$ 4 carriers are more prone to brain changes leading to dementia-related pathology,<sup>350,351</sup> having prolonged exposure to neuroprotective oestrogen (through having menopause at a later age) could be particularly beneficial to  $\varepsilon$ 4 carriers in limiting the build-up of brain pathologies. Alternatively, an earlier age at menopause could be particularly detrimental for  $\varepsilon$ 4 carriers.

Although research linking *APOE* genotype with menopause and cortical thickness is lacking, some research does indicate sex-specific associations of *APOE* genotype and cortical thickness; healthy female ɛ4 carriers showed greater cortical thinning than female non-

carriers, while healthy male carriers and non-carriers showed no differences in cortical thinning.<sup>352</sup> For white matter hyperintensities, indicative of cerebrovascular disease (specifically cSVD),<sup>353</sup> higher cholesterol levels associated with greater WMHV<sup>354</sup> provides some plausibility for cholesterol serving some mechanistic role in links between menopause, *APOE* genotype, and WMHV. However, the association of menopause age and WMHV in  $\varepsilon$ 4 carriers was partially explained by smoking at age 36. Previous research has associated smoking with earlier age at menopause, possibly due to anti-oestrogen effects of tobacco,<sup>355</sup> and with increased WMHV which reflects adverse effects of smoking on vascular health.<sup>356,357</sup> Such associations could be reflective of smoking as a marker for poorer general health and social disadvantage, although adjustments for health-related factors such as hypertension did not attenuate the association of smoking with WMHV progression over a 10 year follow-up.<sup>356</sup>

When excluding women with neurological conditions, some *APOE* interactions were unmasked. *APOE*- $\epsilon$ 4 carriers showed stronger positive associations of menopause age with TBV, and in non-carriers increasing menopause age associated with poorer NAWM microstructural integrity. It is worth noting that excluding women with neurological conditions did exclude one woman in the main analytic sample who had an early menopause (<45 years); changes in the results could reflect potential effects of early menopause, rather than the effects of neurological conditions. Indeed, a sensitivity analysis excluding two women who had menopause younger than age 45 demonstrated similar effect modifications by *APOE*- $\epsilon$ 4 status. Nonetheless, previous analyses of Insight 46 data showed sex differences in NAWM; females had poorer FA and, specifically in females, higher systolic blood pressure and cardiovascular risk scores associated with poorer FA.<sup>237</sup> Further research is required to determine the mechanisms underlying menopause associations with NAWM microstructural integrity and how menopause, or other female reproductive variables, might contribute to poorer cerebrovascular health observed in females compared with males.

# *5ii.4.3. Strengths and limitations*

A major strength of this research is the range of multimodal neuroimaging measures examined, facilitating an examination of which indicators of brain health do and do not associate with menopause age. This study also uses life course data prospectively recorded over 70 years, providing a valuable opportunity to examine prospectively recorded age at menopause in relation to later-life brain health, where other studies are reliant on retrospectively recorded menopause age which does not offer the same level of accuracy and could also be prone to error.

However, as a sub-study of the NSHD cohort, Insight 46 has a limited sample size which is further reduced when confining analyses to females. This may limit the potential power to detect statistically significant relationships, particularly when interrogating potential differences in estimates stratified by APOE- $\varepsilon$ 4 genotype. Nonetheless, that some effect modifications by APOE- $\varepsilon$ 4 were detected is encouraging and provides justification for further research, including using larger cohorts. Additionally, as will be discussed in Section 6.3., there is a selection bias within Insight 46 whereby participants tend to have better general health and higher socioeconomic backgrounds than the whole-NSHD sample.<sup>194</sup> There is a need for the current findings to be replicated in more diverse cohorts, particularly given that Insight 46 (and the whole-NSHD cohort) is representative only of white British individuals born in 1946. Finally, while these analyses included women who took HT, with adjustments for HT use and examination of associations between HT use and neuroimaging outcomes, there is recognition that further research is needed to examine how variations in HT, such as dosages and timing of HT initiation in relation to menopause, might associate with later-life brain health.

#### 5ii.4.4. Summary

Later age at natural menopause generally associated with better brain health at age ~70 (particularly with reduced non-specific brain ageing and fewer markers of cSVD) and some associations were stronger in *APOE*- $\varepsilon$ 4 carriers than in non-carriers. Most associations were independent from life course covariables, including HT use, although midlife smoking and blood pressure partially explained some relationships. These findings support the notion that later menopause age is beneficial for brain health given prolonged exposure to endogenous oestrogen. These findings can also be interpreted such that earlier menopause age, indicating reduced oestrogen exposure, increases risk for adverse brain health outcomes. Further research is needed to replicate these findings in other cohorts and to determine the underlying mechanisms.

# 5.1. Summary

Together, the analyses conducted in this empirical section demonstrate that, overall, later age at menopause is beneficial for later-life cognitive performance and brain health, aligning with previous research indicating that later age at menopause is protective. Alternatively, the same results may indicate that an earlier age at menopause is detrimental for later-life cognition and brain health. Not all associations were independent of life course covariables; notably, childhood cognition mostly explained associations of menopause age with later-life cognition, indicating some upstream, developmental factor might explain the associations, or perhaps higher childhood cognition is a proxy indicator for increased lifetime oestrogen exposure. While evidence was found for an effect modification by APOE-E4 status in associations between menopause age and brain health measures, APOE genotype did not modify associations of menopause age with cognitive performance. A possible explanation for this discrepancy could be that dementia-related brain pathology develops and can be detected prior to the onset of cognitive symptoms. Since the NSHD and Insight 46 cohorts are mainly healthy and dementia free at the time of cognitive testing and neuroimaging, any differences in cognitive performance between  $\varepsilon 4$  carriers and non-carriers in relation to menopause age could be subtle and not detected. Further examination of associations between menopause age, APOE genotype, and cognitive performance assessed in subsequent data collection waves is warranted.

# 6.0. Discussion

# 6.1. Summary of key findings

This thesis examined sex differences in cognitive performance across the life course and in measures of brain health in later-life. There was consideration of how various developmental, sociodemographic, lifestyle, genetic, and sex-specific reproductive factors might contribute to later-life cognition and brain health, within a mostly cognitively unimpaired birth cohort. The findings of each analysis are discussed in the relevant chapters, with the key findings also summarised below.

- A) A descriptive narrative of sex differences and similarities in cognitive performance across domains, throughout life, and in neuroimaging indicators of brain health in later-life was produced (Chapter 3). The role of socioeconomic, educational, and lifestyle (smoking, physical activity/PA) covariables in sex differences were examined, alongside the potential for APOE-ε4 genotype to modify sex differences.
  - i. **Cognitive performance:** Differences in mean performance between males and females were detected across a range of cognitive assessments, at several timepoints throughout the life course. Females showed better performance on more cognitive tests than males, but most notably on tasks assessing verbal and memory domains. Males, on the other hand, tended to show non-verbal and visuospatial task advantages. When accounting for socioeconomic and educational covariables, which were higher in males, female cognitive performance advantages were strengthened while male advantages were weakened, reflecting negative and positive confounding, respectively. Lifestyle covariables (smoking, PA) did not substantially contribute to cognitive performance sex differences. Generally, sex differences in cognitive performance in younger adulthood were not modified by APOE- $\varepsilon$ 4 status, but some APOE effect modifications were detected at older ages, demonstrating stronger female memory performance advantages in  $\varepsilon 4$  non-carriers than in carriers.
  - Brain health: At age ~70 years, females showed larger total brain volumes (relative to head size) than males. Females also showed poorer cerebrovascular health (specifically, greater indicators of cerebral small vessel

disease/cSVD) than males, indicated by greater white matter hyperintensity volumes and poorer microstructural integrity of normal-appearing white matter (NAWM), but no sex differences were detected in measures associated with AD (hippocampal volume and A $\beta$  levels). Socioeconomic, educational, and lifestyle covariables did not substantially contribute to sex differences in neuroimaging measures and effect modification by *APOE*- $\epsilon$ 4 status was detected only for NAWM measures, whereby only female non-carriers showed poorer NAWM integrity.

- B) To quantify lifetime exposures to twelve identified modifiable dementia risk factors, a cumulative risks score (CRS) was derived from life course data (Chapter 4). The extent to which early cognitive, reproductive, socioeconomic, and genetic factors associated with cumulative risk exposures, and whether CRS associated with later-life cognitive performance and brain health were compared in males and females. This was to determine whether the potential associations with non-sex-specific dementia risk factor exposures were stronger in one sex than the other. Interactive effects of *APOE*-ε4 status and CRS on later-life outcomes were also examined in both sexes.
  - i. **Cumulative risk exposures:** In the whole cohort, males had more cumulative exposure to modifiable dementia risk factors across the life course than females. In both sexes, higher childhood cognitive performance, childhood SEP, and *APOE*-ε4 non-carrier status associated with greater CRS. These associations were not replicated in the neuroimaging sub-study sample.
  - ii. Cognitive performance: Independently of genetic and early-life cognitive and socioeconomic predictors of CRS, greater cumulative risk exposures associated with poorer cognitive performance in males and females at age 69. Associations of CRS with cognitive state were equal between males and females, while the association with verbal memory performance was slightly (non-significantly) stronger in females, and the association with a measure of processing speed was only detected in males. No effect modifications by APOE genotype were found.
  - iii. **Brain health:** In males and females, greater risk exposures associated with smaller TBV although this was stronger in males. Only males showed an

association of greater CRS with smaller hippocampal volume, and neither sex showed associations of CRS with measures of cSVD (WMHV) or AD pathology (Aβ). Associations were not modified by *APOE*-ε4 status.

- C) Prospectively reported age at menopause, as a female-specific hormonal transition, was examined for its associations with cognitive performance across a range of domains and with neuroimaging measures of brain health (Chapter 5). The extent to which these associations could be explained by relevant early cognitive, sociodemographic, reproductive, genetic, and health-related covariables was also examined, alongside tests for whether APOE-ε4 genotype modified associations of menopause age with later-life outcomes.
  - i. **Cognitive performance:** Later age at menopause associated with better performance across a range of cognitive domains, but the strongest associations were found for the domains of visual processing, associative learning, and memory. These associations were largely explained by covariables, most notably by cognitive performance in childhood, which is positively associated with menopause age and later-life cognitive performance. There was no evidence for interactive effects of *APOE*-ε4 genotype and menopause age.
  - ii. Brain health: Later menopause age generally associated with better indicators of brain health (greater TBV, for example), although these were partly explained by covariables, particularly by health (blood pressure) and lifestyle (smoking) factors. Some evidence for interactive effects of APOE genotype was found whereby ε4 carriers showed beneficial associations of later menopause age with reduced WMHV and greater cortical thickness.

Figure 22, which will be referenced in the following sections of this discussion, summarises the interrelationships between each of the three empirical chapters and provides an illustrative schema of causal directions through which the findings can be interpreted.



*Figure 22.*Illustrative schema of causal directions between life course variables, later-life brain health, and later-life cognitive performance in males and females, as described and assessed in each of the empirical chapters in this thesis.

- a) Known early-life predictors of later-life cognitive performance; higher childhood cognition, higher educational attainment, and APOE-ε4 non-carrier (vs. carrier) status associated with better later-life cognitive performance on the Addenbrooke's Cognitive Examination-3<sup>rd</sup> edition (ACE-III).<sup>159</sup>
- b) Greater childhood cognition and socioeconomic position (SEP), and APOE-ε4 carrier (vs. non-carrier) status associated with fewer lifetime dementia risk factor exposures (Chapter 4). Note that education was included in the cumulative risks score (CRS), so education was not further adjusted for in Chapter 4 analyses.
- c) Greater dementia risk exposures (increased CRS) exacerbated brain ageing in males; males showed negative associations of increased CRS with smaller brain volumes (total brain volume and hippocampal volume), while females showed a trend-level association with total brain volume but no associations with hippocampal volume (Chapter 4).
- d) Greater dementia risk exposures adversely associated with later-life cognitive performance, with more associations detected for males than females. Both sexes showed an equal negative association of CRS with cognitive state, females showed a slightly stronger association with verbal memory, while only males showed an association with processing speed performance (Chapter 4).
- e) Higher childhood cognition associated with later menopause age<sup>156,157</sup> (Chapter 5).
- f) Later menopause age associated with reduced brain ageing (greater brain volume) and with better cerebrovascular health (Chapter 5); in APOE-ε4 carriers, later menopause age associated with reduced white matter hyperintensity volume (WMHV).
- g) Later menopause age associated with better later-life cognitive performance, although this is largely explained by childhood cognition<sup>155</sup> (Chapter 5).
- h) Some existing evidence from Insight 46 has demonstrated associations of poorer brain health measures (amyloid positivity, greater WMHV, lower brain volumes) with poorer later-life cognitive performance, although sex differences in these associations are not explicitly examined.<sup>40,158,217,218,236</sup> Females showed poorer cerebrovascular health, but better cognitive performance than males (Chapter 3), indicating a female cognitive resilience buffer against cerebrovascular pathology. As depicted in the illustrative line graph, reduced cognitive resilience buffer in males suggests a steeper brain health-cognition associations could be non-linear and may differ once pathology reaches a certain threshold (see Section 6.2.6.).

# 6.2. Interpretation of key findings

#### 6.2.1. Sex differences in reserve and resilience

As outlined in Section 1.6., resilience is the ability to maintain cognitive function despite brain ageing and pathology. This is achieved through mechanisms including cognitive and brain reserve, which refer to cognitive adaptability and structural brain characteristics, respectively.<sup>45</sup> Depicted in Figure 22, the findings from this thesis indicate differences in how well males and females can maintain cognitive function with brain ageing and pathology, and maintain cognitive function and brain health with exposures to life course health and lifestyle risks.

In Chapter 3, females were found to maintain cognitive performance advantages into later-life, despite showing poorer cerebrovascular health (increased cSVD markers) than males. Insight 46 females therefore demonstrated cognitive resilience - cognitive abilities were maintained despite cSVD pathology. Given that females outperformed males on most cognitive assessments throughout the life course, it is plausible that females have increased cognitive reserve, with greater cognitive resources and functional networks which can be drawn on and adapted to minimise the functional impacts of brain pathologies. Although the direct associations of brain health and cognitive performance have not been assessed in this thesis, colleagues have assessed these associations in the Insight 46 sample, generally indicating poorer cognitive performance with more advanced brain ageing and pathologies. Excluding participants with cognitive impairment, poorer performance on the PACC at age ~70 has been associated with amyloid positivity, increased WMHV,<sup>158</sup> and older predicted brain age,<sup>236</sup> but not with NAWM<sup>40</sup> or total brain volume.<sup>158</sup> While sex differences in these associations have not been explicitly tested in this cohort, the relationship between brain health and cognitive performance is hypothesised to be stronger and steeper in males, as outlined in Figure 22h. In a non-dementia sample, at a given level of brain pathology, males are expected to show poorer cognitive ability than females, given less cognitive reserve to protect against the adverse effects of pathology on cognitive function.

Similarly, findings from Chapter 4 suggest that males are less able to resist the adverse effects of accumulated modifiable dementia risk factor exposures than females on cognitive performance (Figure 22c) and brain volumes (Figure 22d). Given that these associations are stronger in men,

and men were more likely to have greater exposure to modifiable dementia risks, cognitive and brain volume sex differences favouring women observed in Chapter 3 could reflect reduced cognitive and brain reserve in males, diminished by modifiable lifestyle risk exposures. Indeed, more male cognitive advantages were observed during childhood and young adulthood, when fewer risk exposures have been accumulated, than in later-life (Chapter 3). It also follows that fewer cumulative risk exposures in females, and reduced effects of cumulative risks on brain ageing (brain volumes) in females, contributes to the preservation of female structural brain resources (brain reserve), thereby facilitating female cognitive resilience to cerebrovascular pathology.

## 6.2.2. Brain ageing vs. cerebrovascular health

Findings from Chapters 3 and 4 demonstrate that more advanced brain ageing in males (smaller relative brain volumes) is linked with accumulated dementia risk exposures, while poorer brain health in females, specific to cSVD risk, is not. Although a measure of cumulative risk exposures was not associated with cSVD markers (WMHV), other Insight 46 analyses associated individual risk factors with later-life brain health. These relationships are most evident in females and particularly demonstrate links of midlife cardiovascular risks with later-life cerebrovascular health; glycemia (an indicator of diabetes) measured from age 53 to 69 associated with reduced TBV only in females,<sup>129</sup> and only in females did greater cardiovascular risk and blood pressure in midlife associate with poorer NAWM integrity.<sup>40</sup> Taken together, evidence indicates potential variations in how accumulated or individual risk factor exposures associate with age- and vascular-related brain health between the sexes; males show greater brain ageing vulnerability to accumulated risks, while females are more vulnerable to cerebrovascular impacts of individual risks, particularly cardiovascular risks present in midlife.

Nonetheless, both sexes did show adverse associations of accumulated risk exposures with laterlife cognitive performance, albeit across more domains in males than females (Chapter 4). As shown in Figure 22(c,d,h), CRS-cognition associations could be mediated through CRS-associated brain ageing in males, but less so in females, in whom the potential pathways between accumulated risks and cognitive performance are less clear. Aside from brain health, societal factors linked with cumulative health and lifestyle risks also warrant consideration. For instance, the traditional caregiving role of women could mean that men with greater risk factor exposures receive more support than women, who might manage their own health to a greater extent than men do. Women with greater cumulative risk exposures could take on additional burdens, such as managing health appointments and medications, which could provide stimulation to help maintain cognitive abilities, but could also divert cognitive resources away from the functional systems required to complete cognitive assessments. Additionally, female underrepresentation in clinical research means that the symptoms for some clinical conditions might not accurately reflect female clinical presentations. Females can therefore experience delayed diagnoses and subsequent treatments or support when they present with 'atypical' symptoms; the health worries associated with such delays could adversely impact cognitive functioning.

It is also possible that subtle effects of individual risks on brain health are diluted when these risks are included in a cumulative measure; the absence of significant CRS-brain health associations does not necessarily mean that mediating effects of brain health are not present in females. If there are such mediating effects, fewer CRS-cognition associations in females are assumed to reflect female cognitive resilience to cSVD. Further, female-specific exposures not included in the CRS could provide pathways through which accumulated risks may associate with brain health in females. For example, the menopause transition has been linked with several risks included in the CRS (e.g. hypertension, diabetes, weight gain, and obesity)<sup>167</sup> and Chapter 5 demonstrates associations of menopause age with TBV and cSVD. It is possible that menopause-related variables (e.g. timing, symptoms) interact with accumulated risks, providing a mechanism through which accumulated risks may health.

#### 6.2.3. Midlife and menopause as a sensitive period for females

As mentioned above, menopause, which typically occurs during midlife, has been associated with several cardiovascular risks, with peri- and post-menopausal women showing increased blood pressure and BMI, for example, than pre-menopausal women,<sup>167</sup> even when groups are age-

matched.<sup>358</sup> Additionally, cardiovascular risk prevalence is greater in males up until midlife, when female prevalence increases to match that of males.<sup>167</sup> Given that oestrogen has both vascular and neuroprotective properties,<sup>83,84,359</sup> declining oestrogen levels during menopause could contribute to increased cardiovascular risks and vulnerability of the brain to those risks. Menopause could therefore act as a trigger for cardiovascular risks and further risk factor accumulation by inducing a chain of risks. Female risk factor accumulation may be more concentrated at midlife and beyond, potentially with more rapid risk accumulation, while male risk exposures are more evenly dispersed across the life course, possibly exerting more prolonged influence on brain ageing (Chapter 4, Figure 22c). Indeed, Chapter 4 sensitivity analyses did demonstrate that risks accumulated up to and including midlife already associated with later-life brain volumes.

There were also parallels in the examination of the interrelationships between cumulative risk exposures and menopause timing. It was previously reported that later menopause age associated with better verbal memory performance but not with performance on an assessment of processing speed.<sup>155</sup> Similarly, cumulative risk exposures in females associated with poorer verbal memory performance but not with processing speed. While not tested in this thesis, it is possible that earlier menopause timing increases cumulative risk exposures by shifting the female sensitive period to cardiovascular risks and subsequent risk accumulation earlier. Interactive effects of cumulative risk exposures and menopause timing are therefore expected whereby women with earlier menopause are more likely to show stronger CRS associations with cognition (particularly verbal memory) and brain ageing (given positive associations of menopause age with TBV; Figure 22f) than women with later menopause timing.

#### 6.2.4. The role of *APOE* genotype

Chapter 3 analyses showing reduced later-life memory performance in female APOE- $\varepsilon$ 4 carriers than non-carriers align with reports of greater APOE- $\varepsilon$ 4-associated dementia risk in females than males.<sup>95</sup> However, the mechanisms underlying this sex difference in  $\varepsilon$ 4 dementia risk are not yet well understood.

Interestingly, Chapter 5 demonstrated menopause associations with cerebrovascular health only in  $\varepsilon$ 4 carriers (later menopause age, reduced cSVD markers), indicating a greater sensitivity to menopause timing among carriers than non-carriers. Given the role of ApoE protein as a lipid transporter, including in the delivery of the cholesterol precursor needed for oestrogen synthesis,<sup>92,360</sup> there are biological mechanisms through which *APOE* genotype and menopause can plausibly interact. For example, adipose tissue (which increases at menopause) becomes the primary source of endogenous oestrogen post-menopause,<sup>361</sup> when ovarian oestrogen production declines. It is possible that earlier menopause timing results in an earlier switch to reliance on adipose-derived oestrogen, which could be disadvantageous for *APOE*- $\varepsilon$ 4 carriers, given that  $\varepsilon$ 4 disrupts ApoE lipid transporter function.<sup>92</sup> In partial support of this hypothesis, there is some evidence for interactive effects of BMI and *APOE* genotype on female, but not male, predicted brain age.<sup>362</sup> However, higher BMI associated with younger predicted brain age in  $\varepsilon$ 4 carriers, in the opposite direction to expected if  $\varepsilon$ 4 disrupts adipose tissue oestrogen synthesis.

Alternatively, ε4 carriers metabolise glucose less efficiently than non-carriers and are more reliant on ketone bodies (produced from fatty acids)<sup>363</sup> as an energy source.<sup>318</sup> Glucose metabolism, supported by oestrogen, also declines during menopause,<sup>163</sup> hence greater reliance on ketone metabolism post-menopause could be exacerbated in ε4 carriers. Some literature suggests that over-reliance on ketone metabolism could leave brain white matter, as a ketone source, vulnerable to degeneration.<sup>318,364</sup> Therefore, ε4 carriers who undergo menopause at earlier ages could have prolonged over-reliance on ketone metabolism, adversely impacting white matter integrity.

While interactive effects of *APOE* with menopause age were found, there was no evidence for *APOE* interactions with cumulative risk exposures in either sex (Chapter 4). Taken together, these findings could indicate that  $\varepsilon$ 4 contributes to female dementia risk via female-specific reproductive processes, such as menopause, rather than by increasing the vulnerability of the female brain to accumulated health and lifestyle risks which are not sex-specific. However, more work is clearly needed to understand the mechanisms through which *APOE*- $\varepsilon$ 4 genotype increases dementia risk more greatly in females.

#### 6.2.5. Early-life and sociocultural influences

A particular strength of the NSHD cohort is the wealth of prospective data from across the life course, including early-life which many other studies of ageing are lacking. With previous NSHD analyses demonstrating associations of early-life variables, including childhood cognition and parental SEP, with later-life cognitive performance,<sup>159</sup> such early-life covariables were considered throughout the analyses in this thesis.

There were some variations in the extent to which early-life covariables explained associations of predictor variables with later-life cognition and brain health outcomes. Menopause-cognition associations were largely explained by early-life factors (primarily childhood cognition; Chapter 5) while cumulative risk exposure associations with cognitive and brain health measures were not (Chapter 4). As described in Chapter 5, childhood cognition could be a proxy measure of some upstream developmental process during infancy or *in utero*, possibly reflecting the role of sex hormones on brain development.<sup>156</sup> This could represent a sensitive period during which the pathways linking female reproductive and cognitive ageing are established. However, this thesis has not explicitly tested the sensitive period model, which would be methodologically challenging (see Section 1.1.). For example, repeated measures of endogenous and exogenous sex hormone exposures across the life course would be needed, and the effects of these exposures at different timepoints on later-life cognition compared.<sup>6,365</sup> This would establish whether early-life sex hormone exposures have stronger effects than the same exposure at other timepoints, but only under the assumption that the level of sex hormone exposure is equal at all timepoints, which given the known variations in sex hormone levels across the life course, <sup>366</sup> is not realistic.

Conversely, although early-life factors associated with cumulative risk exposures, there was an additional effect of lifetime risk exposures on later-life outcomes. Less advantageous early-life exposures (e.g. lower SEP) could induce a chain of risks (e.g. fewer educational and occupational opportunities, poorer diet), which may explain the associations of early-life factors with cumulative risk scores. The accumulative model proposes that it is the built-up effect of these chained risks which influence later-life outcomes rather than the initial (early-life) exposures alone, a notion which the Chapter 4 findings support.

It is important to note that many sociocultural variables included in these analyses are traditionally gendered in this cohort (e.g. lower educational and occupational attainment in women) and there could be variations between sexes in how early-life factors cascade to other sociocultural exposures throughout life. For example, although females had greater cognitive performance in childhood than males, educational attainment and occupational SEP were greater in males (Chapter 3). There is a controversial argument that educational attainment reflects baseline cognitive ability,<sup>9</sup> which may be at least partially true given positive associations of childhood cognition and educational attainment,<sup>159</sup> but higher educational attainment cannot be achieved if access to education is limited, which has traditionally been the case for women and for those with lower socioeconomic backgrounds.

Given that societal gender roles vary over time, the generational context of the NSHD cohort must be considered. People born in 1946 would have been entering the labour market in the 1960s, when the culture in Britain tended toward women taking on more household and childcare duties than men and, although female representation in the work force was increasing, men were the main breadwinners. When women did work, fewer hours were worked, and the job roles available tended to be lower skilled and lower paid than those available to men, hence there was less incentive for women to pursue higher education.<sup>367,368</sup>

Evidence from Chapter 3 demonstrates how gendered inequalities in education and SEP could suppress female cognitive advantages. Indeed, secular improvements in educational access across generations (as traditional gender roles become less prevalent within society)<sup>369</sup> have been associated with increased female memory and verbal memory advantages.<sup>54</sup> Traditional gendered socioeconomic inequalities, as observed within NSHD, could therefore limit opportunities to develop and maintain cognitive and brain reserve resources. With fewer opportunities for activities which support brain maintenance (e.g. through occupational complexity), females could show more rapid reductions than males in their existing reserve resources as they advance into older age, as will be discussed in the following Section (6.2.6.).

213

#### 6.2.6. Pathways to greater female dementia risk

While this thesis has not explicitly examined sex differences in dementia diagnoses, variations in male and female pathways to cognitive and brain health measures up until an age at which dementia diagnoses remain relatively rare (~70 years) were tested in a mostly dementia-free, cognitively unimpaired cohort. At this age, there was evidence that females showed better cognitive and brain reserve than males, providing structural and functional brain resources through which females can cope with (exert cognitive resilience to) increased levels of cerebrovascular pathology. Although this does appear paradoxical to evidence showing greater dementia risk in females, sex differences in resilience pathways were demonstrated and it remains possible that changes in resilience with continued ageing, brain pathology progression and expression, and cognitive and functional impairment diagnoses, may also differ by sex.

Indeed, among MCI participants in the ADNI cohort (mean age at baseline 74 years), females showed greater decline than males on the AD Assessment Scale-Cognition score (ADAS-Cog) over 10 years.<sup>370</sup> Wider evidence has also shown that male AD patients outperform female AD patients across several cognitive domains including visuospatial abilities, language, and episodic memory.<sup>371</sup> Interestingly, a quantification of cognitive reserve among Aβ positive ADNI participants (baseline age 73.9 years) demonstrated differences in cognitive reserve-related cognitive decline across AD dementia stages (mean follow-up 2 years). Greater cognitive reserve (reflecting the extent to which ADAS-Cog performance was higher or lower than expected given indicators of brain health, e.g. GMV) associated with slower decline in memory and executive functioning in participants with unimpaired cognition and MCI, but with faster decline in AD participants.<sup>372</sup> Although sex differences were not examined, such findings do support the theory that greater baseline cognitive resilience slows cognitive decline in the preclinical phases of AD dementia, but once dementia becomes clinically manifest, cognitive resilience advantages are lost. Together, findings imply that, in older age when cognitive impairment and brain pathology becomes more prevalent, females lose their cognitive resilience advantage, leaving them vulnerable to more rapid functional decline. The relationship between brain health and cognitive function (e.g. Figure 22h) may, therefore, not be linear. If females show cognitive resilience advantages until a threshold of pathology is reached, then the female slope is expected to be gentler than the male slope at lower levels of pathology, but to become steeper than the male slope at high pathology levels.

Alternatively, lifetime cognitive advantages in females, particularly in memory domains which are first affected in AD dementia, could result in delayed detection of dementia-related cognitive decline. Indeed, some evidence indicates that females receive dementia diagnoses at more advanced disease stages than males, with females showing lower (poorer) MMSE scores than males when initially diagnosed with AD.<sup>373</sup> This difference might also reflect a greater privilege for male clinical assessments; for example, female caregivers might be more likely to raise concerns about functional impairments, particularly since males are less likely to be widowed than females.<sup>374</sup> Delayed diagnosed. Cognitive assessments widely used in clinic and research often have ceiling effects (e.g. MMSE) and lack the sensitivity to detect early signs of dementia, but the range of cognitive outcomes examined throughout this thesis include more sensitive measures with greater performance variability (e.g. PACC), with demonstrable sex differences in task performance. Further utilisation of such assessments could be valuable for detecting dementia at earlier stages, which would be beneficial for both sexes.

# 6.3. Methodological limitations

## 6.3.1. Multiple testing

Several cognitive and neuroimaging outcome measures have been investigated across multiple statistical models throughout this thesis, increasing the likelihood of detecting associations where there are none (type I error). Some methods commonly used to address this issue of multiple testing include adopting family-wise error or false discovery rates which control for the type I error rate across all comparisons or for a fraction of type I errors among the significant results, respectively.

However, such adjustments can be too conservative and increase the risk of type II errors, when null associations are incorrectly accepted. Additionally, corrections for multiple testing are not required when the variables examined are correlated because the overall number of hypotheses tested is reduced. In this thesis, the main hypotheses focused on the themes of cognitive performance and neuroimaging indicators of brain health; the cognitive variables are correlated (e.g. better ACE-III performance associates with better performance on other cognitive assessments), as are the neuroimaging variables (e.g. larger TBV associates with greater WMHV). For these reasons, it was deemed unnecessary to correct for multiple testing in this thesis.

### 6.3.2. Non-dementia outcomes

Although the overall purpose of this thesis was to develop an understanding of how various life course factors might contribute to increased dementia risk in females, use of a mostly cognitively unimpaired cohort in their 70's precluded the opportunity for dementia diagnosis as an outcome measure. Nonetheless, Insight 46 neuroimaging measures of AD-related pathology (principally PET amyloid imaging) provide a valuable indication for the likelihood of later AD diagnosis; Aβ positive individuals are thought to be in the preclinical phases of AD dementia.<sup>193,375</sup> Throughout this thesis, there was a general lack of significant findings for Aβ outcomes, aligning with the norm across other Insight 46 analyses to date, whereby *APOE* genotype is the variable most consistently associated with Aβ: 60.8% of Aβ positive individuals are  $\varepsilon$ 4 carriers, compared with 22.7% of Aβ negative individuals.<sup>158</sup> Such null findings could reflect the preclinical nature of the Insight 46 cohort at age ~70; Aβ levels may not have yet reached a threshold to noticeably impact functional abilities or neurodegenerative processes. Alternatively, risk factors (such as those identified in the Lancet commission)<sup>56</sup> could be linked with dementia via non-amyloid-related pathways including ageing- (e.g. brain volumes) or vascular-related (e.g. WMHV, NAWM microstructural integrity) pathways.

Further follow-up of the cohort is ongoing, including repeated cognitive testing, amyloid PET imaging and additional tau PET imaging.<sup>306</sup> Subsequent analyses of additional pathology data, including examination of cognitive and brain health trajectories alongside dementia diagnoses, which are expected to increase as the cohort advances into older age,<sup>94</sup> will be important to develop a full picture of life course sex differences in resilience pathways and to elucidate what this could mean for dementia risk in males and females.
It is also important to note that, alongside neuroimaging outcomes, there has been a focus on cognitive performance assessments to infer levels of functional abilities related to atypical ageing or dementia. While cognitive assessments are useful tools, with some validated measures (e.g. Memory Impairment Screen, 10-point Cognitive Screener) used in the clinical diagnosis of dementia,<sup>376</sup> performance on such assessments are not necessarily true to real life, everyday functioning.<sup>377</sup> For instance, some tasks are unusual and would not typically be encountered in daily life, and test anxiety could also influence performance. Further research could benefit from incorporating additional functional measures intrinsic to the clinical definition of dementia, such as ability to undertake activities of daily living which can be assessed by questionnaire administered to individuals and close informants such as family members.<sup>378</sup>

#### 6.3.3. Data availability and biases

The vast amount of prospective data available spanning the life course is a strong advantage of NSHD and Insight 46, including the rarely available measures of cognition in childhood; but there are some limitations of birth cohort and secondary data analyses which should be acknowledged. First, the variables available at each data collection wave reflect the topics and research questions of interest at that time, which within NSHD have substantially evolved since the survey's conception as a maternity study. Therefore, variables of interest are not always available across all timepoints (e.g. physical activity was not measured before age 36). Some variables can also lack sufficient detail to address present day research questions; for example, age at HT initiation is only available in age group categories which hinders analyses assessing a sensitive period for HT use in relation to age at menopause.

Second, given that NSHD data spans over seven decades, where variables have been repeatedly measured, the measurement technique can vary over time and may not reflect current day best practices. For example, several different questionnaires were used across data collections to assess depressive affect, transitioning from the Present State Examination (PSE) to the Psychiatric Symptom Frequency (PSF) scale, and then to the 28-item General Health Questionnaire (GHQ-28) which is more widely used in clinical and research settings today.<sup>379</sup>

Third, NSHD study members were invited to participate in Insight 46 if they had a clinic visit at age 60-64, indicated they were willing to attend a clinic visit in London, and if they had available relevant data available across childhood and adulthood, as previously outlined elsewhere.<sup>193</sup> This was to maximise the life course data available for analyses alongside the additional Insight 46 data collected, but the approach has induced a sample selection bias. Compared with the wider NSHD cohort, Insight 46 participants are found to be generally healthier, to have better cognitive function, higher educational attainment, and to be from higher socioeconomic backgrounds.<sup>194</sup> Individuals who may be at the greatest risk for adverse outcomes, including dementia, are therefore underrepresented within Insight 46, limiting the generalisability of findings to the wider NSHD cohort and the general population, whereby associations of education and SEP, for example, with health-related outcomes may be underestimated. Indeed, that early-life variables associated with cumulative risk exposures in the whole-NSHD but not Insight 46 sample (Chapter 4) could be explained by less socioeconomic variability within Insight 46, as well as lower statistical power.

Fourth, attrition bias can induce similar problems. As is the case with most longitudinal studies, there is inevitable sample attrition over time due to various factors including loss of contact, emigration, and no longer wanting or being able to participate. Recent NSHD retention rates are, however, high; at age 69, 94% of the 2816 remaining study members took part in data collection.<sup>191</sup> High retention rates may limit the level of attrition bias, but there is still evidence for greater likelihood of participation among study members with higher SEP, educational attainment, childhood cognition, and better health.<sup>191</sup> The study members retained at older data collections could, therefore, be under representative of individuals who may be at increased risk of adverse outcomes.

Finally, and related to attrition bias, is the issue of survivor bias, whereby participants remaining in the study, particularly at older ages, are those with the best health since those with poorer health may withdraw due to illness or die. This is of particular concern for sex difference analyses given that females have a greater life expectancy than males. Indeed, within NSHD 20% of males were not contacted for the age 69 data collection because they had died, compared with 15% of women.<sup>191</sup> Other evidence also shows that males have greater cardiovascular disease mortality 218 rates than females at younger ages.<sup>380,381</sup> Therefore, the males remining in the study in later-life are likely to be those with the best cardiovascular health, while females with poorer cardiovascular health can still participate; surviving males could be healthier than surviving females, which could influence sex differences in later-life outcomes such as cognitive ageing, brain health, and dementia.

#### 6.3.4. Generalisability

Some generalisability limitations linked with sample selection and attrition biases have been outlined in the previous Section (6.3.3.). This section will discuss some of the generalisability issues intrinsic to a birth cohort representative of the post-war population in Britain.

A key strength of NSHD is its very narrow age range, with all participants born in the same week of 1946, which minimises the influence of age variability on the reported associations with laterlife outcomes. This also means that wider societal, political, environmental, and economic events (e.g. the 1970s financial crash, introduction of the internet) were experienced at the same age across study members. However, it follows that people of other generations experience different societal exposures, hence the relationships of life course variables and later-life outcomes could vary between generations that are not represented in NSHD data. As mentioned in Section 6.2.5., changing societal gender roles over time and secular improvements in educational access are particularly relevant when considering how sex differences in cognitive ageing and dementia risk might change across generations.<sup>54</sup> With continued reductions in gendered educational and occupational disparities,<sup>382</sup> it is hypothesised that females' heightened dementia risk could diminish when more recent generations reach older age, given greater opportunities for the development and maintenance of cognitive and brain reserve resources than females born in 1946 or earlier.

Additionally, NSHD is a White British cohort, hence it is not representative of non-British populations or other ethnic groups. This is a particularly relevant limitation for the generalisability of dementia sex differences research, given reported geographical variations in dementia incidence sex differences; evidence indicates that AD dementia incidence does not differ by sex

in the US, but in European countries females have higher incidence than males.<sup>383</sup> Cultural differences in health, lifestyle, and socioeconomic circumstances, for example, could contribute to such variations. Indeed, evidence has shown female cognitive disadvantages (including on attention and orientation, memory, and fluency tasks) in lower income counties, contrasting with findings of female cognitive advantages in high income countries.<sup>384</sup> Such sex differences are largely explained by gendered educational inequalities whereby women have reduced access to education, which are greatest in low-income countries.<sup>384</sup>

Further, ethnic diversity in Britain has vastly increased since 1946, but ethnic minorities are frequently underrepresented in research. Relatively little is therefore known about cognitive ageing and dementia risk in these groups, and much less about sex differences across ethnicities. When adjusting for sex, analyses of UK Biobank data have indicated increased dementia risk in Black compared with White participants, and equal dementia risk in South Asian and White participants.<sup>385</sup> Similarly, English medical record and mortality data shows increased dementia incidence in Black compared with White people, accounting for sex, and South Asian and Black people with dementia are shown to die at younger ages than White people with dementia.<sup>386</sup> In non-British samples, there is some evidence for cognitive and dementia sex differences which vary between ethnic groups. A US study of non-Hispanic White, Black, and Hispanic males and females found female memory advantages over males within the Black and Hispanic groups, while non-Hispanic White females outperformed all sex and ethnic groups.<sup>387</sup> Over an average 15 years of follow-up, decline in memory and visuospatial performance was also found to be steeper in Black females compared with non-Hispanic White females.<sup>387</sup> Elsewhere, AD risk is reported to be greater among Black American than non-Hispanic White adults, <sup>388</sup> and Black American females are reported to have the highest prevalence of cardiovascular risks linked with dementia (including hypertension and type 2 diabetes) than other ethnic or sex groups.<sup>389</sup> The intersectionality of sex and ethnicity is therefore an important issue which needs to be examined to understand cognitive and brain ageing, and dementia mechanisms within all groups of society, but this cannot be addressed in an ethnically homogenous cohort such as NSHD.

#### 6.3.5. Sex and gender interrelationships

This thesis has focused on sex differences, referring to the biological characteristics distinguishing males from females.<sup>13</sup> While gender (self-identified and based on societal constructs of masculinity and femininity)<sup>13</sup> need not match biological sex, interrelationships between sex and gender are likely. Given the gendered nature of some variables (e.g. education, occupation, physical activity levels, smoking habits) linked with dementia and cognitive resilience, it will be important to consider how gender identity might modify sex differences in life course cognitive and brain ageing pathways.

In 2016, the Sex and Gender Equity in Research (SAGER) guidelines<sup>14</sup> were published to streamline and encourage the reporting of sex and gender in scientific research and outputs. While the guidelines recommend that sex should distinguish males and females based on biological characteristics, no specific recommendations are made for gender reporting. It is acknowledged that, although often misconceptualised as binary (masculine/feminine), gender identity lies on a spectrum.<sup>14</sup> The term transgender describes someone whose gender identity (internal sense of gender) or gender expression (public presentation of gender) does not conform to their sex assigned at birth; some people who identify as transgender might have a binary gender identity, while others might not. Individuals who identify as non-binary or gender fluid might identify as both, neither, or somewhere in-between the traditional masculine and feminine dichotomy.<sup>390</sup>

Although there is no current consensus for how best to measure gender in scientific research, this is an active topic of discussion. In 2007, the Canadian Women's Health Research Network outlined a gender framework suggesting four domains of gender-related variables through which researchers might be better able to conceptualise gender: gender identification (inner sense of self), gender role (behavioural norms), gender relations (interpersonal interactions), and institutionalised gender (distribution of power within society, including discrimination).<sup>391,392</sup> More recently, the GOING-FWD (Gender Outcomes International Group – to Further Well-being Development) consortium incorporated this framework into a five-step process developed to facilitate retrospective analyses of gender in relation to non-communicable disease outcomes.<sup>12</sup> With the development of gender-based frameworks and increased awareness of sex and gender

distinctions, researchers will be better equipped to address questions of sex and gender interactions in the coming years.

Regarding the findings presented in this thesis, gender identify could modify some of the sex differences observed. For example, according to the minority stress model, people who identify as gender diverse could experience stigmatisation and discrimination, leading to an accumulation of psychological stress which can induce negative impacts on mental and physical wellbeing.<sup>390,393</sup> Individuals who identify as gender diverse could, therefore, experience increased life course exposures to modifiable dementia risk factors and could lack the necessary support to manage multiple health and lifestyle adversities, given a lack of acceptance or understanding of gender diversity within society, including in clinical settings.

A recent examination of survey data from the United States has reported sex and gender differences in dementia risk scores, across age-matched groups.<sup>394</sup> Individuals assigned female at birth (female sex) showed lower midlife dementia risk (measured using CAIDE [ages 40-65] and LIBRA [age  $\geq$ 50] scores) than those assigned male at birth (male sex), but females showed greater later-life AD risk (indicated by greater Australian National University-Alzheimer's Disease Risk Index/ANU-ADRI scores [age  $\geq$  55] scores) than males. While midlife dementia risk scores did not differ between gender identity groups, gender diverse cohorts (transgender women, transgender men, and non-binary adults) did show increased later-life AD risk (ANU-ADRI scores) compared with cisgender men and women. The authors postulate that social stigma and discrimination experienced by individuals expressing gender identities deviating from the social norms of binary masculinity and femininity matching biological sex, could be an important contributor to increased AD risk. Indeed, levels of depression were greatest among non-binary adults and transgender males, providing some support for the minority stress model.

Additionally, generational variations in the impacts of gender identity on health and lifestyle throughout the life course are anticipated. Only in more recent years has gender fluidity been increasingly recognised and accepted within society.<sup>390</sup> In an older cohort such as the NSHD, there could have been increased societal pressure for binary expression of gender identify, with gender

222

fluid individuals experiencing discrimination, or else suppressing their true gender identity, with implications for mental well-being and general health.

# 6.4. Policy and research implications

#### 6.4.1. Sex- and gender-based analysis (SGBA) research policies

The findings from this thesis highlight that there are complicated sex differences in the life course pathways and resilience mechanisms to later-life cognitive function and brain health, reinforcing the importance of sex-aware research and therefore providing support for the implementation of sex- and gender-based analysis (SGBA) policies. Such policies encourage careful consideration of sex and gender throughout the research process including design (e.g. inclusion of both sexes), analyses (e.g. sex-disaggregated data), and reporting (e.g. clarity on how sex and gender were considered). The SAGER guidelines<sup>14</sup> are an example of an international framework to encourage better sex and gender reporting in research publications, but individual countries have also introduced policies to embed SGBA in their health-related research.

In 1993, the US National Institutes of Health (NIH) implemented the Women and Minorities as Subjects in Clinical Research policy<sup>395</sup> to ensure that women and minority groups were included in all clinical research, with success in that at least half of participants in recent NIH-funded clinical research are women.<sup>396</sup> The Canadian Institutes for Health Research followed in 2009, providing clear definitions for sex and gender, requiring grant applicants to report whether they had integrated sex or gender into their proposals, and later asking applicants to justify any omissions to do so.<sup>397,398</sup> The US NIH additionally introduced a policy in 2016 to embed sex as a biological variable (SABV) throughout the design, analysis, and reporting phases of all vertebrate animal and human studies, with researchers needing to provide strong justification for single-sex studies.<sup>396,399</sup> The UK has lagged behind, with the first medical research funder (the MRC) introducing a sex and gender policy only in July 2023.<sup>400</sup> However, there is increasing awareness across the UK research sector, with several organisations advocating for improved SGBA in health research. A collaboration of various stakeholders (researchers, regulators, funders, participants, publishers, clinicians, the Department for Health and Social Care) are currently designing a sex

and gender policy framework, due to be launched in 2024, as part of the Medical Science Sex and Gender Equity (MESSAGE) initiative.<sup>1</sup> Work is clearly ongoing to develop effective SGBA frameworks which can be readily adopted by the medical research community. Research specifically aiming to examine and understand sex differences in health outcomes, such as in this thesis, can contribute to the success of SGBA policies, including by helping to encourage a culture of change in the way that researchers approach sex and gender.

## 6.4.2. Women's health

This thesis reinforces the notion that female-specific reproductive processes can relate to female health-related outcomes in later-life, thereby advocating for an emphasis on women's health, which has traditionally been overlooked in research and health care. Although menopause was the primary female-specific reproductive factor examined in this thesis, there are several other female-specific health issues which were not examined, including pre-menstrual syndrome, endometriosis, and complications of pregnancy to name a few. Throughout the duration of this doctoral work, there has been an increase in public awareness and discussion of women's health issues, including the implementation of public health policies pertaining to women's health and wellbeing. For instance, the first Women's Health Strategy for England was launched in July 2022,<sup>401</sup> setting out plans for improvements in how the health and care system listens to women, aiming to boost health outcomes for women and girls. Priority areas include, amongst others, menstrual health and gynaecological conditions, fertility and pregnancy, menopause, mental health and wellbeing, and healthy ageing. New and planned initiatives under the strategy include the creation of women's health hubs, dedicated women's health areas on the NHS website, and increased accessibility to menopausal HT.<sup>401,402</sup> Media attention and advocacy from public figures, including TV presenters Davina McCall and Naga Munchetty, has also helped to reduce taboo surrounding women's health, widening the discussion around topics such as menstruation and menopause, including among men. Societal shifts in the understanding of women's health could improve support, for example within family networks and in the workplace, and thereby reduce the individual and societal burden of women's health issues.

224

The apparent wave of public and governmental support for women's health improvements should be taken as an opportunity to re-evaluate how best to record women's health data, improving the resources available to examine the determinants of women's health outcomes and treatment effectiveness. For instance, as highlighted in Chapter 5, our current understanding of HT effects on later-life cognition and brain health is poor, often hindered by inadequate data regarding HT timing, dosages, and administration methods. There should also be continued research examining women's health in relation to later-life health outcomes, to determine how recent policy implementations and societal attitude changes might alter associations as women across different generations age.

# 6.4.3. Reducing dementia risk exposures

The demonstration of adverse associations between increased lifetime exposures to the Lancet commission's twelve modifiable dementia risk factors and later-life cognitive and brain health outcomes adds to existing literature reporting dose-response effects of health and lifestyle risk factors.<sup>108,111,271,273</sup> Together, this associative evidence implies that reducing lifetime risk exposures could have benefits for later-life outcomes, and multi-domain lifestyle intervention trials have shown promise in this respect.<sup>104,105</sup> However, the development of public health policies to reduce accumulated risk exposures across the life course will need to consider how modifiable the identified risk factors truly are for individuals and governments. For instance, reducing air pollution exposure at an individual level would be difficult, with air pollution improvements requiring substantial worldwide efforts which could take several years to be those at the highest risk of dementia but also the least likely to engage with prevention strategies, given greater health and social barriers.<sup>403</sup>

There also needs to be consideration for when in the life course interventions might be most effective, including how this might differ between sexes and other societal groups. For example, if midlife, or more specifically the menopause transition, is indeed a sensitive period for females, then public health policies aimed at improving women's health outcomes might target women of this age. Conversely, if males are vulnerable to more prolonged accumulation of risks across the life course, then policies targeting males earlier in life could be beneficial. In fact, the associations of early-life variables with accumulated risks (Chapter 4) do suggest that early-life could be a key target for policies to reduce lifetime accumulation of risks, particularly in males, thereby promoting healthier lifestyles throughout the entire life course.

This thesis also highlights how gendered socioeconomic and educational inequalities may limit female opportunities to develop and maintain the structural and functional brain resources needed to protect against abnormal brain ageing- and dementia pathology-related functional impairments. While this thesis demonstrates greater female cognitive resilience than males in a cohort at age ~70, how cognitive resilience might change in older age and as pathology levels increase remains to be determined. Nonetheless, policies aimed at reducing gendered disparities, including improving educational access especially in lower-income countries, could enhance female cognitive abilities and presumably also the resources supporting cognitive reserve. Other avenues for reducing gendered socioeconomic disparities might include supporting and encouraging women to return to work after having children in tandem with strategies to facilitate and encourage men to take on greater responsibilities for child rearing, and supporting women in work during the menopause transition. In addition to health benefits, such strategies could also have wider economic advantages associated with a stronger workforce.

# 6.5. Future directions

## 6.5.1. Cognitive resilience and risk factor exposure interrelationships

Further analyses should examine the extent to which potential sex differences in cognitive resilience might contribute to the observed sex differences in how cumulative dementia risk exposures associate with later-life cognitive and brain health outcomes. Three additional research questions to be addressed are: 1) Do males show reduced cognitive resilience than females at age ~70? 2) Are the associations of cumulative dementia risk factor exposures with cognitive performance mediated by brain health, and are mediating effects strongest in males? 3) Is

cognitive resilience modified by cumulative dementia risk factor exposures, and are such interactive effects strongest in males?

Question 1 can be addressed by testing for sex differences in the associations of neuroimaging measures of brain health and cognitive performance in later-life. The strongest brain-cognition associations are expected in males (Figure 22h) since greater cognitive resilience in females suggests a disconnect between brain health and cognitive function which would be less apparent in males.

Question 2 can be tested using sex-stratified mediation analyses. The mediating effects of brain health on CRS-cognition associations are expected to be strongest in males, given greater female resistance to brain ageing and pathology when exposed to dementia risk factors (Figure 22c) and presumed greater female cognitive resilience that buffers against reduced brain health (Figure 22h).

Question 3 would be assessed by testing for interactive effects of CRS and later-life brain health on cognitive performance. Those with greater risk factor exposures would be expected to show reduced resilience (i.e. stronger brain-cognition associations), with greater modifying effects in males than females since CRS more strongly associated with later-life outcomes in males (Figure 22c, d).

#### 6.5.2. Addressing risk-modifying treatment use

As described in Section 4.4.3., there are limitations in adjusting for risk-modifying treatments, such as anti-hypertensive medications, when examining the associations of CRS with later-life outcomes. It is still important to consider how such treatments might contribute to these associations. One way to address this in further work could be to apply a weighting to each risk factor included in the CRS for which risk-modifying treatments exist. Weightings should be determined using existing literature and knowledge of the treatment effect size; for example, the extent to which blood pressure is reduced by anti-hypertensive medication.<sup>313</sup>

#### 6.5.3. Integrating female reproductive health

Further analyses should examine how menopause timing and additional reproductive health variables might contribute to female cognitive resilience and associations of dementia risk factor exposures with later-life outcomes.

Findings from this thesis, and wider literature, imply that later menopause (reflecting prolonged endogenous oestrogen exposure) extends brain structure and function maintenance (promoting greater cognitive resilience), and possibly delays the accumulation of dementia risk factor exposures (Section 6.2.3.). The following questions can address these assumptions: 1) Does menopause timing modify cognitive resilience to brain ageing and pathology in later-life? 2) Is there an association between menopause age and cumulative risk exposures, and are there interactive effects of menopause age and risk factor exposures on later-life cognitive and brain health outcomes?

To test question 1, interactive effects of menopause age and brain health measures on cognitive performance should be examined. It is hypothesised that women with later menopause would show greater cognitive resilience, with weaker brain-cognition associations than those who experienced menopause at younger ages.

Addressing question 2, later menopause age is expected to associate with fewer dementia risk exposures. Menopause age-by-CRS interactions are hypothesised to reveal stronger associations of CRS with later-life cognition and brain health in women with earlier menopause ages.

More broadly, future research should also examine how additional female-specific reproductive exposures might influence female resilience mechanisms, as potential explanatory factors in observed sex differences. Indeed, there is evidence of GMV reductions associated with pregnancy, which are maintained for at least 2-years post-partum.<sup>404</sup> Hypertensive disorders of pregnancy are also of great interest, with long-lasting cerebrovascular changes (increased white matter hyperintensities and reduced NAWM microstructural integrity) and reduced cognitive functioning observed in women who had pre-eclampsia compared with normotensive pregnancies.<sup>405-407</sup> Additionally, while menopause age is informative, there will also be value in examining how menopausal symptoms relate to dementia risk exposures and later-life cognition

and brain health. For example, fewer or less severe symptoms could reflect a greater ability to maintain functional abilities with hormonal changes. Social relationships, work-related performance, and physical activities, for example, may therefore be better maintained, potentially minimising additional accumulation of health and lifestyle dementia risks.

Given the suggestion that menopause and perhaps other female reproductive transitions (e.g. menarche, pregnancy) could be sensitive periods for female brain health, future work could also examine whether women who have greater dementia risk factor exposures during these periods have poorer cognitive and brain health outcomes in later-life than those with fewer risk exposures during sensitive periods. For example, risk exposures in the five years either side of menopause could be quantified and examined for associations with later-life outcomes. The expectation would be that women with greater risk exposures would have poorer outcomes. Such analyses could be further extended by additionally quantifying risk exposures in males at ages matched to female menopause for comparison; adverse associations of increased risk exposures would be hypothesised to be most evident in females.

## 6.5.4. Longitudinal follow-up

This thesis has examined cross-sectional sex differences in cognitive performance and brain health measures, but a great strength of NSHD is that there are longitudinal measures (Table 1) which facilitate the examination of cognitive trajectories, necessary to understand variations in cognitive ageing and the underlying mechanisms. Given the inference that midlife could be a sensitive period for females (Section 6.2.3.), a further question to address is whether dementia risk factor exposures accumulated up to and including midlife (up to age 53) differentially associate with rates of verbal memory and processing speed decline in males and females from age 53 to 69, with careful consideration for how accumulated risks to midlife might interact with additional risk exposures during the follow-up period. Accumulated risks to midlife would be expected to associate with steeper cognitive performance decline in females, reflecting how midlife risk exposures might increase female vulnerability to accelerated cognitive ageing in the years preceding any dementia diagnosis.

Now aged 78, study members are continuing to participate in further waves of data collection for Insight 46.<sup>306</sup> As the cohort continues to advance into older age, when brain pathology levels and rates of dementia diagnoses are anticipated to increase, there is a valuable opportunity to further examine the life course pathways to dementia and how these differ by sex. To test the hypothesis that females show a rapid decline in their level of cognitive resilience once a threshold of brain pathology is reached, thereby contributing to greater dementia rates in females, follow-up analyses should re-examine the associations observed at age ~70 (Figure 22) at subsequent ages, in addition to testing for sex differences in cognitive and brain ageing trajectories from age 69 onwards, and how life course health, lifestyle, and reproductive variables might contribute to such trajectories in males and females. A challenge for such analyses is that the statistical power to detect dementia incidence could be low due to small sex-stratified sample sizes in future Insight 46 data collections. The Insight 46 selection bias, whereby Insight 46 participants have better general health than the wider NSHD cohort, might also mean that the proportion of dementia cases are lower than in the whole cohort or general population.

#### 6.5.5. Examining survivor bias

Described in Section 6.3.3., survivor bias is a limitation of these analyses given attrition of the NSHD cohort over time. This is a particular concern for sex difference analyses since more female than male study members are retained at older ages, given higher male mortality rates at younger ages. In further research, the analyses included in this thesis could be re-run using inverse probability weighting. This method aims to recreate a representative sample of the initial cohort by assigning weights to individuals according to their likelihood of remaining in the cohort, based on their characteristics. For example, participants with higher SEP are more likely to have available ACE-III data at age 69 than those with lower SEP; participants with higher SEP would be given a lower weighting in the analysis, thereby increasing the proportional representation of participants with lower SEP who were more likely to be lost to follow-up. Comparing such weighted analyses to the current findings in this thesis can inform the extent to which survivor bias may have driven the findings, which is important to understand given the hypothesis that increased female life expectancy could contribute to females' increased dementia risk.

#### 6.5.6. Varied cohorts

As discussed in Section 6.3.4., there are limitations in the generalisability of the findings presented in this thesis, primarily to non-British and non-White populations, and to other generations. Further research should therefore utilise more diverse cohorts to assess whether the associations observed in NSHD and Insight 46 (Figure 22) are replicated in different groups. Importantly, the intersectionality of sex and ethnicity should be tested, acknowledging that the presence or absence of sex differences (e.g. in level of cognitive resilience, or in the associations of cumulative risk exposures with later-life brain health) might not be equal across all ethnic groups.

Additionally, the sex differences in relationships between life course variables as presented in Figure 22 should be examined and compared between cohorts spanning generations (e.g. subsequent British birth cohorts; National Child Development Study/NCDS (1958), 1970 British Cohort Study/BCS70, Millennium Cohort Study/MCS (2000)). This will allow for trends to be observed over time, to determine whether societal changes in gender roles and gendered socioeconomic disparities translate into reduced sex differences in cognitive resilience, cognitive and brain ageing, and ultimately dementia diagnoses, although dementia outcomes will only be possible several decades in the future (particularly for MCS).

## 6.6. Conclusions

The findings of this thesis highlight that males and females differ in their abilities to maintain functional abilities with poorer brain health related to ageing and cerebrovascular disease at age ~70. Females showed cognitive resilience to increased levels of cerebral small vessel disease pathology, while males demonstrated reduced cognitive resilience to greater brain ageing (smaller MRI-measured brain volumes). The use of a life course approach incorporating early-life exposures, health and lifestyle variables identified elsewhere as dementia risk factors, and female-specific reproductive processes (menopause timing) revealed that exposures at different life stages differentially associate with later-life cognitive and brain health outcomes between males and females, thereby showing sex differences in the life course mechanisms underlying

cognitive resilience. There was also evidence to support previous reports that gendered socioeconomic and educational disparities may suppress female cognitive abilities. This thesis therefore advocates for the importance of sex- and gender-based analyses, which will be integral to building a deeper understanding of the sociocultural and biological processes related to the life course development and risk of dementia in males and females, ultimately helping to inform public health prevention strategies to minimise the future individual and societal burden of dementia in an ageing population.

# References

- Witt, A., Politis, M. & Womersley, K. A whole sector approach to policy change will accelerate integration of sex and gender in research. *BMJ* 383 (2023). https://doi.org/10.1136/bmj.p2913
- Gaugler, J., James, B., Johnson, T., Scholz, K. & Weuve, J. 2016 Alzheimer's disease facts and figures. *Alzheimer's & Dementia* 12, 459–509 (2016). https://doi.org/10.1016/j.jalz.2016.03.001
- Mazure, C. M. & Swendsen, J. Sex differences in Alzheimer's disease and other dementias. *Lancet Neurol* 15, 451 (2016). https://doi.org/10.1016%2FS1474-4422(16)00067-3
- Jack, C. R. *et al.* Tracking pathophysiological processes in Alzheimer's disease: An updated hypothetical model of dynamic biomarkers. *The Lancet Neurology* **12**, 207–216 (2013). https://doi.org/10.1016/s1474-4422(12)70291-0
- Davis, D. *et al.* Decline in Search Speed and Verbal Memory Over 26 Years of Midlife in a British Birth Cohort. *Neuroepidemiology* 49, 121–128 (2017). https://doi.org/10.1159/000481136
- Ben-Shlomo, Y. & Kuh, D. A life course approach to chronic disease epidemiology: conceptual models, empirical challenges and interdisciplinary perspectives. *Int J Epidemiol* **31**, 285–293 (2002). https://doi.org/10.1093/ije/31.2.285
- Roza, S. J. *et al.* Effects of maternal smoking in pregnancy on prenatal brain development. The Generation R Study. *Eur J Neurosci* 25, 611–617 (2007). https://doi.org/10.1111/j.1460-9568.2007.05393.x
- Brinton, R. D., Yao, J., Yin, F., Mack, W. J. & Cadenas, E. Perimenopause as a neurological transition state. *Nature Reviews Endocrinology* **11**, 393–405 (2015). https://doi.org/10.1038/nrendo.2015.82
- Walhovd, K. B., Lövden, M. & Fjell, A. M. Timing of lifespan influences on brain and cognition. *Trends Cogn Sci* 27, 901–915 (2023). https://doi.org/10.1016/j.tics.2023.07.001
- Coylewright, M., Reckelhoff, J. F. & Ouyang, P. Menopause and Hypertension. *Hypertension* 51, 952–959 (2008). https://doi.org/10.1161/hypertensionaha.107.105742
- 11. Rioux, C. *et al.* Sex and gender terminology: a glossary for gender-inclusive epidemiology. *J Epidemiol Community Health* **76**, 764–768 (2022).

- Raparelli, V. *et al.* Identification and inclusion of gender factors in retrospective cohort studies: the GOING-FWD framework. *BMJ Glob Health* 6, (2021). https://doi.org/10.1136/bmjgh-2021-005413
- Office for National Statistics. Sex and gender within the context of data collected for the Sustainable Development Goals (SDGs). https://www.ons.gov.uk/economy/environmentalaccounts/articles/whatisthedifferenceb etweensexandgender/2019-02-21. Date accessed: 01/12/2021
- Heidari, S., Babor, T. F., De Castro, P., Tort, S. & Curno, M. Sex and Gender Equity in Research: rationale for the SAGER guidelines and recommended use. *Res Integr Peer Rev* 1, 1–9 (2016). https://doi.org/10.1186/s41073-016-0007-6
- 15. Zucchella, C. *et al.* Neuropsychological testing. *Pract Neurol* **18**, 227–237 (2018). https://doi.org/10.1136/practneurol-2017-001743
- 16. Horn, J. L. & Cattell, R. B. Age differences in fluid and crystallized intelligence. *Acta Psychol (Amst)* **26**, 107–129 (1967). https://doi.org/10.1016/0001-6918(67)90011-X
- 17. Deary, I. J. *et al.* Age-associated cognitive decline. *Br Med Bull* **92**, 135–152 (2009). https://doi.org/10.1093/bmb/ldp033
- Hedden, T. & Gabrieli, J. D. E. Insights into the ageing mind: a view from cognitive neuroscience. *Nature Reviews Neuroscience* 5, 87–96 (2004). https://doi.org/10.1038/nrn1323
- 19. World Health Organization. Dementia fact sheet. https://www.who.int/news-room/fact-sheets/detail/dementia Date accessed: 03/12/2021
- Ewers, M., Sperling, R. A., Klunk, W. E., Weiner, M. W. & Hampel, H. Neuroimaging markers for the prediction and early diagnosis of Alzheimer's disease dementia. *Trends Neurosci* 34, 430–442 (2011). https://doi.org/10.1016/j.tins.2011.05.005
- 21. Smith, G. E. & Bondi, M. W. Mild Cognitive Impairment and Dementia: Definitions, Diagnosis, and Treatment. (American Academy of Clinical Neuropsychology, 2013). https://books.google.co.uk/books?hl=en&lr=&id=M6JSJmdc\_m0C&oi=fnd&pg=PP1&dq= defining+dementia+and+mild+cognitive+impairment+MCI&ots=sID4u1JygU&sig=bV49YrS 2vJfMXRyDscRarWcWJG0&redir\_esc=y#v=onepage&q=defining%20dementia%20and%20 mild%20cognitive%20impairment%20MCI&f=false Date accessed: 10/02/2022
- 22. National Institute of Aging. Alzheimer's Disease Fact Sheet. https://www.nia.nih.gov/health/alzheimers-disease-fact-sheet Date accessed: 03/12/2021
- 23. Rabinovici, G. D. & Miller, B. L. Frontotemporal lobar degeneration. *CNS Drugs* **24**, 375–398 (2010). https://doi.org/10.2165/11533100-00000000-00000

- Podcasy, J. L. & Epperson, C. N. Considering sex and gender in Alzheimer disease and other dementias. *Dialogues Clin Neurosci* 18, 437 (2016). https://doi.org/10.31887%2FDCNS.2016.18.4%2Fcepperson
- 25. Walker, Z., Possin, K. L., Boeve, B. F. & Aarsland, D. Lewy body dementias. *The Lancet* **386**, 1683–1697 (2015). https://doi.org/10.1016/s0140-6736(15)00462-6
- 26. Gorelick, P. B. *et al.* Vascular Contributions to Cognitive Impairment and Dementia. *Stroke*42, 2672–2713 (2011). https://doi.org/10.1161/str.0b013e3182299496
- Yang, T., Sun, Y., Lu, Z., Leak, R. K. & Zhang, F. The impact of cerebrovascular aging on vascular cognitive impairment and dementia. *Ageing Res Rev* 34, 15 (2017). https://doi.org/10.1016%2Fj.arr.2016.09.007
- 28. Psych Scene Hub. White Matter Hyperintensities on MRI Artefact or Something Sinister? https://psychscenehub.com/psychinsights/white-matter-hyperintensities-mri/ Date accessed: 03/12/202129.
- 29. Li, Q. *et al.* Cerebral Small Vessel Disease. *Cell Transplant* **27**, 1711 (2018). https://doi.org/10.1177/0963689718795148
- Snyder, H. M. *et al.* Vascular contributions to cognitive impairment and dementia including Alzheimer's disease. *Alzheimers Dement* **11**, 710–717 (2015). https://doi.org/10.1016/j.jalz.2014.10.008
- 31. Eisenmenger, L. B. *et al.* Vascular contributions to Alzheimer's disease. *Transl Res* **254**, 41–53 (2023). https://doi.org/10.1016/j.trsl.2022.12.003
- Hugo, J. & Ganguli, M. Dementia and cognitive impairment: epidemiology, diagnosis, and treatment. *Clin Geriatr Med* 30, 421–442 (2014). https://doi.org/10.1016/j.cger.2014.04.001
- Tsoi, K. K. F., Chan, J. Y. C., Hirai, H. W., Wong, S. Y. S. & Kwok, T. C. Y. Cognitive tests to detect dementia a systematic review and meta-analysis. *JAMA Intern Med* 175, 1450– 1458 (2015). https://doi.org/10.1001/jamainternmed.2015.2152
- Knott, L. Screening for cognitive impairment: Cognitive function tests.
  https://patient.info/doctor/screening-for-cognitive-impairment (2022). Date accessed: 16/04/2024
- NHS. Tests for diagnosing dementia. https://www.nhs.uk/conditions/dementia/symptoms-and-diagnosis/tests/ Date accessed: 16/04/2024

- Pleen, J., Camerucci, E., Al-Sabbagh, M. & Cunningham, K. Blood-Based Biomarkers in Alzheimer Disease: Clinical Implementation and Limitations. *Practical Neurology* 23, 27-39 (2024).
- 37. Gonzalez-Ortiz, F. *et al.* Plasma phospho-tau in Alzheimer's disease: towards diagnostic and therapeutic trial applications. *Mol Neurodegener* **18**, 1–12 (2023). https://doi.org/10.1186/s13024-023-00605-8
- Driscoll, I. *et al.* Longitudinal pattern of regional brain volume change differentiates normal aging from MCI. *Neurology* **72**, 1906–1913 (2009). https://doi.org/10.1212%2FWNL.0b013e3181a82634
- 39. Sudre, C. H. *et al.* Bayesian model selection for pathological neuroimaging data applied to white matter lesion segmentation. *IEEE Trans Med Imaging* **34**, 2079–2102 (2015). https://doi.org/10.1109/tmi.2015.2419072
- 40. James, S.-N. *et al.* Neuroimaging, clinical and life course correlates of normal-appearing white matter integrity in 70-year-olds. *Brain Commun* **18**, fcad225 (2023). https://doi.org/10.1093/braincomms/fcad225
- Maniega, S. M. *et al.* White matter hyperintensities and normal-appearing white matter integrity in the aging brain. *Neurobiol Aging* 36, 909–918 (2015). https://doi.org/10.1016/j.neurobiolaging.2014.07.048
- 42. Van Norden, A. G. W. *et al.* Diffusion tensor imaging and cognition in cerebral small vessel disease: The RUN DMC study. *Biochimica et Biophysica Acta* **1822**, 401–407 (2012). https://doi.org/10.1016/j.bbadis.2011.04.008
- Matsuda, H., Shigemoto, Y. & Sato, N. Neuroimaging of Alzheimer's disease: focus on amyloid and tau PET. Jpn J Radiol **37**, 735–749 (2019). https://doi.org/10.1007/s11604-019-00867-7
- Kim, J., Jeong, M., Stiles, W. R. & Choi, H. S. Neuroimaging Modalities in Alzheimer's Disease: Diagnosis and Clinical Features. *Int J Mol Sci* 23, (2022). https://doi.org/10.3390%2Fijms23116079
- 45. Stern, Y. *et al.* Whitepaper: Defining and investigating cognitive reserve, brain reserve, and brain maintenance. *Alzheimers Dement* **16**, 1305–1311 (2020). https://doi.org/10.1016/j.jalz.2018.07.219
- 46. Arenaza-Urquijo, E. M. & Vemuri, P. Improving the resistance and resilience framework for aging and dementia studies. *Alzheimers Res Ther* **12**, 1–4 (2020).
- 47. Hyde, J. S. Sex and cognition: gender and cognitive functions. *Curr Opin Neurobiol* **38**, 53– 56 (2016). https://doi.org/10.1016/j.conb.2016.02.007

- 48. Jäncke, L. Sex/gender differences in cognition, neurophysiology, and neuroanatomy. *F1000Res* **7**, (2018). https://doi.org/10.12688%2Ff1000research.13917.1
- Linn, M. C. & Petersen, A. C. Emergence and Characterization of Sex Differences in Spatial Ability: A Meta-Analysis. *Child Dev* 56, 1479 (1985). https://psycnet.apa.org/doi/10.2307/1130467
- Sharps, M. J., Price, J. L. & Williams, J. K. SPATIAL COGNITION AND GENDER Instructional and Stimulus Influences on Mental Image Rotation Performance. *Psychol Women Q* 18, 413–425 (1994).
- 51. Levine, D. A. *et al.* Sex Differences in Cognitive Decline Among US Adults. *JAMA Netw Open* **4**, e210169 (2021). https://doi.org/10.1001/jamanetworkopen.2021.0169
- Zaninotto, P., Batty, G. D., Allerhand, M. & Deary, I. J. Cognitive function trajectories and their determinants in older people: 8 years of follow-up in the English Longitudinal Study of Ageing. *J Epidemiol Community Health* 72, 685–694 (2018). https://doi.org/10.1136%2Fjech-2017-210116
- 53. Ferreira, L., Ferreira Santos-Galduróz, R., Ferri, C. P. & Fernandes Galduróz, J. C. Rate of cognitive decline in relation to sex after 60 years-of-age: A systematic review. *Geriatr Gerontol Int* 14, 23–31 (2014). https://doi.org/10.1111/ggi.12093
- 54. Bloomberg, M. *et al.* Sex differences and the role of education in cognitive ageing: analysis of two UK-based prospective cohort studies. *Lancet Public Health* 6, e106–e115 (2021). https://doi.org/10.1016/S2468-2667(20)30258-9
- 55. Hyde, J. S. The gender similarities hypothesis. *American Psychologist* **60**, 581–592 (2005). https://doi.org/10.1037/0003-066x.60.6.581
- 56. Livingston, G. *et al.* Dementia prevention, intervention, and care: 2020 report of the Lancet Commission. *The Lancet* **396**, 413–446 (2020). https://doi.org/10.1016/S0140-6736(20)30367-6
- 57. Eliot, L., Ahmed, A., Khan, H. & Patel, J. Dump the "dimorphism": Comprehensive synthesis of human brain studies reveals few male-female differences beyond size. *Neurosci Biobehav Rev* 125, 667–697 (2021). https://doi.org/10.1016/j.neubiorev.2021.02.026
- Cosgrove, K. P., Mazure, C. M. & Staley, J. K. Evolving Knowledge of Sex Differences in Brain Structure, Function, and Chemistry. *Biological Psychiatry* 62, 847–855 (2007). https://doi.org/10.1016/j.biopsych.2007.03.001
- 59. Esposito, G., Van Horn, J. D., Weinberger, D. R. & Berman, K. F. Gender differences in cerebral blood flow as a function of cognitive state with PET. *J Nucl Med* **37**, 559-564 (1996).

- Reiman, E. M., Armstrong, S. M., Matt, K. S. & Mattox, J. H. The application of positron emission tomography to the study of the normal menstrual cycle. *Human Reproduction* 11, 2799–2805 (1996). https://doi.org/10.1093/oxfordjournals.humrep.a019214
- Ferretti, M. T. *et al.* Sex differences in Alzheimer disease The gateway to precision medicine. *Nature Reviews Neurology* 14, 457–469 (2018). https://doi.org/10.1038/s41582-018-0032-9
- 62. Kanaan, R. A. *et al.* Gender Differences in White Matter Microstructure. *PLoS One* **7**, e38272 (2012). https://doi.org/10.1371%2Fjournal.pone.0038272
- Van Hemmen, J. *et al.* Sex Differences in White Matter Microstructure in the Human Brain Predominantly Reflect Differences in Sex Hormone Exposure. *Cerebral Cortex* 27, 2994– 3001 (2017). https://doi.org/10.1093/cercor/bhw156
- 64. Cox, S. R. *et al.* Ageing and brain white matter structure in 3,513 UK Biobank participants. *Nat Commun* **7**, (2016). https://doi.org/10.1038/ncomms13629
- 65. Sachdev, P., Chen, X. & Wen, W. White matter hyperintensities in mid-adult life. *Curr Opin Psychiatry* **21**, 268–274 (2008). https://doi.org/10.1097/yco.0b013e3282f945d5
- Lohner, V. *et al.* The Relation Between Sex, Menopause, and White Matter Hyperintensities: The Rhineland Study. *Neurology* 99, 935-942 (2022). https://doi.org/10.1212/wnl.000000000200782
- 67. Vinke, E. J. *et al.* Trajectories of imaging markers in brain aging: the Rotterdam Study. *Neurobiol Aging* **71**, 32–40 (2018). https://doi.org/10.1016/j.neurobiolaging.2018.07.001
- Zyriax, B. C. & Windler, E. Lifestyle changes to prevent cardio- and cerebrovascular disease at midlife: A systematic review. *Maturitas* 167, 60–65 (2023). https://doi.org/10.1016/j.maturitas.2022.09.003
- Armstrong, N. M. *et al.* Sex differences in brain aging and predictors of neurodegeneration in cognitively healthy older adults. *Neurobiol Aging* 81, 146–156 (2019). https://doi.org/10.1016/j.neurobiolaging.2019.05.020
- 70. Salwierz, P. *et al.* Sex and gender differences in dementia. *Int Rev Neurobiol* **164**, 179–233 (2022). https://doi.org/10.1016/bs.irn.2022.07.002
- 71. Andrew, M. K. & Tierney, M. C. The puzzle of sex, gender and Alzheimer's disease: Why are women more often affected than men? *Women's Health* **14**, (2018). https://doi.org/10.1177%2F1745506518817995
- 72. Lapane, K. L. *et al.* Gender differences in predictors of mortality in nursing home residents with AD. *Neurology* **56**, 650–654 (2001). https://doi.org/10.1212/wnl.56.5.650

- Edland, S. D., Rocca, W. A., Petersen, R. C., Cha, R. H. & Kokmen, E. Dementia and Alzheimer Disease Incidence Rates Do Not Vary by Sex in Rochester, Minn. *Arch Neurol* 59, 1589–1593 (2002). https://doi.org/10.1001/archneur.59.10.1589
- Bachman, D. L. *et al.* Incidence of dementia and probable Alzheimer's disease in a general population: the Framingham Study. *Neurology* 43, 515–519 (1993). https://doi.org/10.1212/wnl.43.3\_part\_1.515
- 75. Gao, S., Hendrie, H. C., Hall, K. S. & Hui, S. The relationships between age, sex, and the incidence of dementia and Alzheimer disease: A meta-analysis. *Arch Gen Psychiatry* **55**, 809–815 (1998). https://doi.org/10.1001/archpsyc.55.9.809
- 76. Xirocostas, Z. A., Everingham, S. E. & Moles, A. T. The sex with the reduced sex chromosome dies earlier: a comparison across the tree of life. *Biol Lett* 16, (2020). https://doi.org/10.1098/rsbl.2019.0867
- Davis, E. J. *et al.* A second X chromosome contributes to resilience in a mouse model of Alzheimer's disease. *Sci Transl Med* 12, (2020). https://doi.org/10.1126/scitranslmed.aaz5677
- 78. Brooks, W. H. X chromosome inactivation and autoimmunity. *Clin Rev Allergy Immunol* **39**, 20–29 (2010). https://doi.org/10.1007/s12016-009-8167-5
- Youness, A., Miquel, C. H. & Guéry, J. C. Escape from X Chromosome Inactivation and the Female Predominance in Autoimmune Diseases. *Int J Mol Sci* 22, 1–12 (2021). https://doi.org/10.3390%2Fijms22031114
- Kinney, J. W. *et al.* Inflammation as a central mechanism in Alzheimer's disease.
  *Alzheimer's & Dementia: Translational Research & Clinical Interventions* 4, 575 (2018).
  https://doi.org/10.1016/j.trci.2018.06.014
- 81. Klein, S. L. & Flanagan, K. L. Sex differences in immune responses. *Nature Reviews Immunology 2016 16:10* **16**, 626–638 (2016). https://doi.org/10.1038/nri.2016.90
- Casaletto, K. B. *et al.* Sex-specific effects of microglial activation on Alzheimer's disease proteinopathy in older adults. *Brain* 145, 3536 (2022). https://doi.org/10.1093/brain/awac257
- Barth, C., Villringer, A. & Sacher, J. Sex hormones affect neurotransmitters and shape the adult female brain during hormonal transition periods. *Frontiers in Neuroscience* 9, 37 (2015). https://doi.org/10.3389%2Ffnins.2015.00037
- Pertesi, S., Coughlan, G., Puthusseryppady, V., Morris, E. & Hornberger, M. Menopause, cognition and dementia – A review. *Post Reproductive Health* 25, (2019). https://doi.org/10.1177/2053369119883485

- Currie, H. & Moger, S. J. Menopause Understanding the impact on women and their partners. *Post Reprod Health* 25, 183–190 (2019). https://doi.org/10.1177/2053369119895413
- Subramaniapillai, S., Almey, A., Natasha Rajah, M. & Einstein, G. Sex and gender differences in cognitive and brain reserve: Implications for Alzheimer's disease in women. *Front Neuroendocrinol* 60, 100879 (2021). https://doi.org/10.1016/j.yfrne.2020.100879
- Barnes, L. L. *et al.* Sex differences in the clinical manifestations of Alzheimer disease pathology. *Arch Gen Psychiatry* 62, 685–691 (2005).
  https://doi.org/10.1001/archpsyc.62.6.685
- Koran, M. E. I., Wagener, M. & Hohman, T. J. Sex Differences in the Association between AD Biomarkers and Cognitive Decline. *Brain Imaging Behav* 11, 205 (2017). https://doi.org/10.1007/s11682-016-9523-8
- Liu, C., Liu, C., Kanekiyo, T., Xu, H. & Bu, G. Apolipoprotein E and Alzheimer disease: risk, mechanisms and therapy. *Nat Rev Neurol* 9, 106–118 (2013). https://doi.org/10.1038/nrneurol.2012.263
- Farrer, L. A. *et al.* Effects of Age, Sex, and Ethnicity on the Association Between Apolipoprotein E Genotype and Alzheimer Disease: A Meta-analysis. *JAMA* 278, 1349– 1356 (1997).
- 91. Corder, E. *et al.* Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. *Science* **261**, 921–923 (1993). https://doi.org/10.1126/science.8346443
- 92. Husain, M. A., Laurent, B. & Plourde, M. APOE and Alzheimer's Disease: From Lipid Transport to Physiopathology and Therapeutics. *Front Neurosci* 15, 85 (2021). https://doi.org/10.3389%2Ffnins.2021.630502
- 93. Alzforum. *Therapeutics: LX1001.* (2023). https://www.alzforum.org/therapeutics/lx1001 Date accessed: 24/08/2024
- 94. Cao, Q. *et al.* The Prevalence of Dementia: A Systematic Review and Meta-Analysis. *J Alzheimers Dis* **73**, 1157–1166 (2020). https://doi.org/10.3233/jad-191092
- 95. Neu, S. C. *et al.* Apolipoprotein E genotype and sex risk factors for Alzheimer disease: A meta-analysis. *JAMA Neurol* 74, 1178–1189 (2017). https://doi.org/10.1001/jamaneurol.2017.2188
- 96. Stern, Y. et al. Whitepaper: Defining and investigating cognitive reserve, brain reserve, and brain maintenance. Alzheimers Dement 16, 1305–1311 (2020). https://doi.org/10.1016/j.jalz.2018.07.219

- 97. Richards, M. & Sacker, A. Lifetime antecedents of cognitive reserve. *J Clin Exp Neuropsychol* **25**, 614–624 (2003). https://doi.org/10.1076/jcen.25.5.614.14581
- 98. Lane, C. A. *et al.* Associations between blood pressure across adulthood and late-life brain structure and pathology in the neuroscience substudy of the 1946 British birth cohort (Insight 46): an epidemiological study. *Lancet Neurol* 18, 942–952 (2019). https://doi.org/10.1016/s1474-4422(19)30228-5
- 99. Kivimäki, M. *et al.* Physical inactivity, cardiometabolic disease, and risk of dementia: an individual-participant meta-analysis. *BMJ* **365**, (2019). https://doi.org/10.1136/bmj.l1495
- Chang, R. C. C., Ho, Y. S., Wong, S., Gentleman, S. M. & Ng, H. K. Neuropathology of cigarette smoking. *Acta Neuropathol* 127, 53–69 (2014). https://doi.org/10.1007/s00401-013-1210-x
- 101. Wennberg, A. M. V., Wu, M. N., Rosenberg, P. B. & Spira, A. P. Sleep Disturbance, Cognitive Decline, and Dementia: A Review. *Semin Neurol* 37, 395 (2017). https://doi.org/10.1055/s-0037-1604351
- 102. Ohara, T. *et al.* Association Between Daily Sleep Duration and Risk of Dementia and Mortality in a Japanese Community. *J Am Geriatr Soc* 66, 1911–1918 (2018). https://doi.org/10.1111/jgs.15446
- Kivipelto, M. *et al.* The Finnish Geriatric Intervention Study to Prevent Cognitive Impairment and Disability (FINGER): study design and progress. *Alzheimers Dement* 9, 657–665 (2013). https://doi.org/10.1016/j.jalz.2012.09.012
- 104. Ngandu, T. *et al.* A 2 year multidomain intervention of diet, exercise, cognitive training, and vascular risk monitoring versus control to prevent cognitive decline in at-risk elderly people (FINGER): a randomised controlled trial. *Lancet* **385**, 2255–2263 (2015). https://doi.org/10.1016/s0140-6736(15)60461-5
- Rosenberg, A. *et al.* Multidomain lifestyle intervention benefits a large elderly population at risk for cognitive decline and dementia regardless of baseline characteristics: The FINGER trial. *Alzheimers Dement* 14, 263–270 (2018). https://doi.org/10.1016/j.jalz.2017.09.006
- Vos, S. J. B. *et al.* Modifiable Risk Factors for Prevention of Dementia in Midlife, Late Life and the Oldest-Old: Validation of the LIBRA Index. *Journal of Alzheimer's Disease* 58, 537– 547 (2017). https://doi.org/10.3233/jad-161208
- Kivipelto, M. *et al.* Risk score for the prediction of dementia risk in 20 years among middle aged people: a longitudinal, population-based study. *Lancet Neurology* 5, 735–741 (2006). https://doi.org/10.1016/s1474-4422(06)70537-3

- 108. Stephen, R. et al. Associations of CAIDE Dementia Risk Score with MRI, PIB-PET measures, and cognition. Journal of Alzheimer's Disease 59, 695 (2017). https://doi.org/10.3233/jad-170092
- 109. Kivimäki, M. *et al.* Estimating Dementia Risk Using Multifactorial Prediction Models.
  JAMA Netw Open 6, e2318132–e2318132 (2023).
  https://doi.org/10.1001/jamanetworkopen.2023.18132
- 110. Anstey, K. J. *et al.* Dementia Risk Scores and Their Role in the Implementation of Risk Reduction Guidelines. *Front Neurol* **12**, (2022). https://doi.org/10.3389/fneur.2021.765454
- 111. LaPlume, A. A., McKetton, L., Anderson, N. D. & Troyer, A. K. Sex differences and modifiable dementia risk factors synergistically influence memory over the adult lifespan. *Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring* 14, e12301 (2022). https://doi.org/10.1002%2Fdad2.12301
- 112. Labaka, A., Goñi-Balentziaga, O., Lebeña, A. & Pérez-Tejada, J. Biological Sex Differences in Depression: A Systematic Review. *Biol Res Nurs* 20, 383–392 (2018). https://doi.org/10.1177/1099800418776082
- 113. Altemus, M., Sarvaiya, N. & Neill Epperson, C. Sex differences in anxiety and depression clinical perspectives. *Front Neuroendocrinol* **35**, 320–330 (2014). https://doi.org/10.1016/j.yfrne.2014.05.004
- 114. National Institute of Mental Health. Major depression.
  https://www.nimh.nih.gov/health/statistics/major-depression Date accessed: 30/11/2021
- 115. Lovejoy, J. C. & Sainsbury, A. Sex differences in obesity and the regulation of energy homeostasis. *Obes Rev* 10, 154–167 (2009). https://doi.org/10.1111/j.1467-789x.2008.00529.x
- 116. Kelly, T., Yang, W., Chen, C. S., Reynolds, K. & He, J. Global burden of obesity in 2005 and projections to 2030. *Int J Obes (Lond)* 32, 1431–1437 (2008). https://doi.org/10.1038/ijo.2008.102
- 117. World Health Organization. 10 facts on gender and tobacco. https://www.slideshare.net/drtonythomas/10-facts-on-gender-and-tobacco Date accessed: 30/11/2021
- 118. Wade, J. M. Is it Race, Sex, Gender or All Three? Predicting Risk for Alcohol Consumption in Emerging Adulthood. J Child Fam Stud 29, 3481–3492 (2020). https://doi.org/10.1007/s10826-020-01780-8

- 119. White, A. M. Gender Differences in the Epidemiology of Alcohol Use and Related Harms in the United States. *Alcohol Res* 40, 1–13 (2020). https://doi.org/10.35946%2Farcr.v40.2.01
- 120. Ramirez, L. A. & Sullivan, J. C. Sex Differences in Hypertension: Where We Have Been and Where We Are Going. Am J Hypertens **31**, 1247–1254 (2018). https://doi.org/10.1093/ajh/hpy148
- Santosa, A. *et al.* Gender differences and determinants of prevalence, awareness, treatment and control of hypertension among adults in China and Sweden. *BMC Public Health* 20, 1–13 (2020). https://doi.org/10.1186/s12889-020-09862-4
- Leening, M. J. G. *et al.* Sex differences in lifetime risk and first manifestation of cardiovascular disease: prospective population based cohort study. *BMJ* 349, (2014). https://doi.org/10.1136/bmj.g5992
- 123. Merz, A. A. & Cheng, S. Sex differences in cardiovascular ageing. *Heart* **102**, 825–831 (2016). https://doi.org/10.1136/heartjnl-2015-308769
- 124. Gilsanz, P. *et al.* Female sex, early-onset hypertension, and risk of dementia. *Neurology* **89**, 1886–1893 (2017). https://doi.org/10.1212%2FWNL.000000000004602
- 125. Kimm, H. et al. Mid-life and late-life vascular risk factors and dementia in Korean men and women. Arch Gerontol Geriatr 52, e117–e122 (2011). https://doi.org/10.1016/j.archger.2010.09.004
- 126. Chatterjee, S. *et al.* Type 2diabetes as a risk factor for dementia in women compared with men: A pooled analysis of 2.3 million people comprising more than 100,000 cases of dementia. *Diabetes Care* **39**, 300–307 (2016). https://doi.org/10.2337%2Fdc15-1588
- 127. Sundermann, E. E. *et al.* Prediabetes Is Associated With Brain Hypometabolism and Cognitive Decline in a Sex-Dependent Manner: A Longitudinal Study of Nondemented Older Adults. *Front Neurol* **12**, 551975 (2021). https://doi.org/10.3389/fneur.2021.551975
- Moran, C., Gilsanz, P., Beeri, M. S., Whitmer, R. A. & Lacy, M. E. Sex, diabetes status and cognition: findings from the study of longevity in diabetes. *BMJ Open Diabetes Res Care* 9, e001646 (2021). https://doi.org/10.1136/bmjdrc-2020-001646
- 129. Fatih, N. et al. Sex-related differences in whole brain volumes at age 70 in association with hyperglycemia during adult life. *Neurobiol Aging* (2021). https://doi.org/10.1016/j.neurobiolaging.2021.09.008
- Noale, M. *et al.* Incidence of dementia: Evidence for an effect modification by gender. The ILSA Study. *Int Psychogeriatr* 25, 1867–1876 (2013). https://doi.org/10.1017/s1041610213001300

- Artero, S. *et al.* Risk profiles for mild cognitive impairment and progression to dementia are gender specific. *J Neurol Neurosurg Psychiatry* **79**, 979–984 (2008). https://doi.org/10.1136/jnnp.2007.136903
- 132. Gong, J., Harris, K., Peters, S. A. E. & Woodward, M. Sex differences in the association between major cardiovascular risk factors in midlife and dementia: a cohort study using data from the UK Biobank. *BMC Med* 19, 110 (2021). https://doi.org/10.1186/s12916-021-01980-z
- 133. Hulka, B. S. & Meirik, O. Research on the menopause. *Maturitas* **23**, 109–112 (1996). https://doi.org/10.1016/0378-5122(95)00967-1
- 134. Burger, G. G., Dudley, E. C., Robertson, D. M. & Dennerstein, L. Hormonal changes in the menopause transition. *Recent Prog Horm Res* 57, 257–275 (2002). https://doi.org/10.1210/rp.57.1.257
- 135. Hoek, A., Schoemaker, J. & Drexhage, H. A. Premature Ovarian Failure and Ovarian Autoimmunity. *Endocr Rev* **18**, 107–134 (1997). https://doi.org/10.1210/edrv.18.1.0291
- 136. Eleazu, I. C., Jones-O'Connor, M. & Honigberg, M. C. The Impact of Premature Menopause on Future Risk of Cardiovascular Disease. *Current Treatment Options in Cardiovascular Medicine 2020 22:12* 22, 1–11 (2020). https://link.springer.com/article/10.1007/s11936-020-00854-6
- 137. Faris, S. Late-onset menopause: What is causing your delay?
  https://www.healthline.com/health/menopause/late-onset Date accessed: 17/9/2021
- Harlow, S. D. *et al.* Executive summary of the Stages of Reproductive Aging Workshop +
  10: addressing the unfinished agenda of staging reproductive aging. *Menopause* 19, 387 (2012). https://doi.org/10.1097%2Fgme.0b013e31824d8f40
- 139. Kase, N. G. Impact of hormone therapy for women aged 35 to 65 years, from contraception to hormone replacement. *Gend Med* 6, 37–59 (2009). https://doi.org/10.1016/j.genm.2009.02.001
- 140. Khaw, K. T. Epidemiology of the menopause. **48**, 249–261 (1992). https://doi.org/10.1093/oxfordjournals.bmb.a072546
- 141. Nwadike, V. R. Surgical Menopause. https://www.healthline.com/health/surgicalmenopause#causes Date accessed: 08/2/2022
- 142. Utian, W. H. Menopause-related definitions. Int Congr Ser 1266, 133–138 (2004).
- 143. Greendale, G. A., Karlamangla, A. S. & Maki, P. M. The Menopause Transition and Cognition. *JAMA* **323**, 1495–1496 (2020). https://doi.org/10.1001/jama.2020.1757

- 144. Maki, P. M. & Henderson, V. W. Cognition and the menopause transition. *Menopause* **23**, 803–805 (2016). https://doi.org/10.1097/gme.000000000000681
- 145. Greendale, G. A. *et al.* Effects of the menopause transition and hormone use on cognitive performance in midlife women. *Neurology* **72**, 1850–1857 (2009). https://doi.org/10.1212/wnl.0b013e3181a71193
- Schaafsma, M., Homewood, J. & Taylor, A. Subjective cognitive complaints at menopause associated with declines in performance of verbal memory and attentional processes. *Climacteric* 13, 84-98 (2010). https://doi.org/10.3109/13697130903009187
- Epperson, C. N., Sammel, M. D. & Freeman, E. W. Menopause effects on verbal memory: Findings from a longitudinal community cohort. *Journal of Clinical Endocrinology and Metabolism* 98, 3829–3838 (2013). https://doi.org/10.1210/jc.2013-1808
- 148. Mitchell, E. S. & Woods, N. F. Cognitive symptoms during the menopausal transition and early postmenopause. *Climacteric* 14, 252-261 (2011). https://doi.org/10.3109/13697137.2010.516848
- 149. Williams, D. R. E. *et al.* Frequency and severity of vasomotor symptoms among peri- and postmenopausal women in the United States. *Climacteric* 11, 32-43 (2008). https://doi.org/10.1080/13697130701744696
- Edwards, H., Duchesne, A., Au, A. S. & Einstein, G. The many menopauses: searching the cognitive research literature for menopause types. *Menopause* 26, 45 (2019). https://doi.org/10.1097%2FGME.00000000001171
- 151. Georgakis, M. K., Beskou-Kontou, T., Theodoridis, I., Skalkidou, A. & Petridou, E. T. Surgical menopause in association with cognitive function and risk of dementia: A systematic review and meta-analysis. *Psychoneuroendocrinology* **106**, 9–19 (2019). https://doi.org/10.1016/j.psyneuen.2019.03.013
- 152. Grodstein, F. Cardiovascular risk factors and cognitive function. *Alzheimer's & Dementia* **3**, S16–S22 (2007). https://doi.org/10.1016/j.jalz.2007.01.001
- Appiah, D. *et al.* Is Surgical Menopause Associated With Future Levels of Cardiovascular Risk Factor Independent of Antecedent Levels? The CARDIA Study. *Am J Epidemiol* 182, 991–999 (2015). https://doi.org/10.1093/aje/kwv162
- 154. Georgakis, M. K. *et al.* Age at menopause and duration of reproductive period in association with dementia and cognitive function: A systematic review and meta-analysis. *Psychoneuroendocrinology* **73**, 224–243 (2016). https://doi.org/10.1016/j.psyneuen.2016.08.003

- 155. Kuh, Di., Cooper, R., Moore, A., Richards, M. & Hardy, R. Age at menopause and lifetime cognition: Findings from a British birth cohort study. *Neurology* **90**, E1673–E1681 (2018). https://doi.org/10.1212/wnl.000000000005486
- Richards, M., Kuh, D., Hardy, R. & Wadsworth, M. Lifetime cognitive function and timing of the natural menopause. *Neurology* 53, 308–314 (1999). https://doi.org/10.1212/wnl.53.2.308
- 157. Kuh, Di., Cooper, R., Moore, A., Richards, M. & Hardy, R. Age at menopause and lifetime cognition: Findings from a British birth cohort study. *Neurology* **90**, E1673–E1681 (2018). https://doi.org/10.1212/wnl.000000000005486
- 158. Lu, K. *et al.* Cognition at age 70: Life course predictors and associations with brain pathologies. *Neurology* **93**, E2144–E2156 (2019). https://doi.org/10.1212/wnl.00000000008534
- Richards, M. *et al.* Identifying the lifetime cognitive and socioeconomic antecedents of cognitive state: Seven decades of follow-up in a British birth cohort study. *BMJ Open* 9, (2019). https://doi.org/10.1136/bmjopen-2018-024404
- 160. Udeh-Momoh, C. & Watermeyer, T. Female specific risk factors for the development of Alzheimer's disease neuropathology and cognitive impairment: Call for a precision medicine approach. *Ageing Res Rev* **71**, 101459 (2021). https://doi.org/10.1016/j.arr.2021.101459
- 161. Kim, G. W., Park, K. & Jeong, G. W. Effects of Sex Hormones and Age on Brain Volume in Post-Menopausal Women. *Journal of Sexual Medicine* **15**, 662–670 (2018). https://doi.org/10.1016/j.jsxm.2018.03.006
- 162. Mosconi, L. *et al.* Increased Alzheimer's risk during the menopause transition: A 3-year longitudinal brain imaging study. *PLoS One* **13**, (2018). https://doi.org/10.1371/journal.pone.0207885
- 163. Mosconi, L. *et al.* Menopause impacts human brain structure, connectivity, energy metabolism, and amyloid-beta deposition. *Scientific Reports 2021 11:1* **11**, 1–16 (2021). https://doi.org/10.1038/s41598-021-90084-y
- 164. Ambikairajah, A., Tabatabaei-Jafari, H., Hornberger, M. & Cherbuin, N. Age, menstruation history, and the brain. *Menopause* 28, 167–174 (2020). https://doi.org/10.1097/gme.00000000001688
- 165. Vélez, M. P., Alvarado, B. E., Lord, C. & Zunzunegui, M. V. Life course socioeconomic adversity and age at natural menopause in women from Latin America and the Caribbean. *Menopause* **17**, 552–559 (2010).

- 166. Lawlor, D. A., Ebrahim, S. & Smith, G. D. The association of socio-economic position across the life course and age at menopause: the British Women's Heart and Health Study. *BJOG* **110**, 1078–1087 (2003).
- 167. Rosano, D. G. M. C., Vitale, C., Marazzi, G. & Volterrani, M. Menopause and cardiovascular disease: the evidence. *Climactertic* 10, 19-24 (2007). https://doi.org/10.1080/13697130601114917
- 168. Zilberman, J. M. *et al.* Association between hypertension, menopause, and cognition in women. *J Clin Hypertens* **17**, 970–976 (2015). https://doi.org/10.1111/jch.12643
- 169. Jones, R. *et al.* Novel coronary heart disease risk factors at 60–64 years and life course socioeconomic position: The 1946 British birth cohort. *Atherosclerosis* 238, 70–76 (2015). https://doi.org/10.1016%2Fj.atherosclerosis.2014.11.011
- Maki, P. & Hogervorst, E. The menopause and HRT. HRT and cognitive decline. *Best Pract Res Clin Endocrinol Metab* 17, 105–122 (2003). https://doi.org/10.1016/s1521-690x(02)00082-9
- Shao, H. *et al.* Hormone therapy and Alzheimer disease dementia: New findings from the Cache County Study. *Neurology* 79, 1846–1852 (2012). https://doi.org/10.1212/wnl.0b013e318271f823
- 172. Kuh, D. & Hardy, R. *A Life Course Approach to Women's Health*. (Oxford University Press, 2002). https://doi.org/10.1093/acprof:oso/9780192632890.001.0001
- 173. Rossouw, J. E. *et al.* Risks and benefits of estrogen plus progestin in healthy postmenopausal women: Principal results from the women's health initiative randomized controlled trial. *J Am Med Assoc* 288, 321–333 (2002). https://doi.org/10.1001/jama.288.3.321
- 174. Anderson, G. L. & Limacher, M. Effects of Conjugated Equine Estrogen in Postmenopausal Women with Hysterectomy: The Women's Health Initiative Randomized Controlled Trial. J Am Med Assoc 291, 1701–1712 (2004). https://doi.org/10.1001/jama.291.14.1701
- 175. Resnick, S. m. *et al.* The Women's Health Initiative Study of Cognitive Aging (WHISCA): A randomized clinical trial of the effects of hormone therapy on age-associated cognitive decline. *Clinical Trials* **1**, 440–450 (2004). https://doi.org/10.1191/1740774504cn040oa
- 176. Espeland, M. A. *et al.* Long-term effects of conjugated equine estrogen therapies on domain-specific cognitive function: results from the Women's Health Initiative study of cognitive aging extension. *J Am Geriatr Soc* 58, 1263–1271 (2010). https://doi.org/10.1111%2Fj.1532-5415.2010.02953.x

- 177. Craig, M. C., Maki, P. M. & Murphy, D. G. M. The Women's Health Initiative Memory Study: findings and implications for treatment. *Lancet Neurol* 4, 190–194 (2005). https://doi.org/10.1016/s1474-4422(05)01016-1
- 178. Shumaker, S. A. *et al.* Conjugated Equine Estrogens and Incidence of Probable Dementia and Mild Cognitive Impairment in Postmenopausal Women: Women's Health Initiative Memory Study. *JAMA* 291, 2947–2958 (2004). https://doi.org/10.1001/jama.291.24.2947
- 179. Resnick, S. M. *et al.* Postmenopausal hormone therapy and regional brain volumes: The WHIMS-MRI Study. *Neurology* 72, 135–142 (2009). https://doi.org/10.1212/01.wnl.0000339037.76336.cf
- Coker, L. H. *et al.* Postmenopausal hormone therapy and subclinical cerebrovascular disease: The WHIMS-MRI Study. *Neurology* 72, 125–134 (2009). https://doi.org/10.1212%2F01.wnl.0000339036.88842.9e
- Espeland, M. A. *et al.* Long-term effects on cognitive function of postmenopausal hormone therapy prescribed to women aged 50 to 55 years. *JAMA Intern Med* **173**, 1429–1436 (2013). https://doi.org/10.1001/jamainternmed.2013.7727
- 182. Gleason, C. E. *et al.* Effects of Hormone Therapy on Cognition and Mood in Recently Postmenopausal Women: Findings from the Randomized, Controlled KEEPS–Cognitive and Affective Study. *PLoS Med* **12**, e1001833 (2015). https://doi.org/10.1371/journal.pmed.1001833
- 183. Resnick, S. M. & Henderson, V. W. Hormone therapy and risk of Alzheimer disease: A critical time. *Journal of the American Medical Association* 288, 2170–2172 (2002). https://doi.org/10.1001/jama.288.17.2170
- 184. Maki, P. M. The Critical Window Hypothesis of Hormone Therapy and Cognition: A Scientific Update on Clinical Studies. *Menopause* 20, 695 (2013). https://doi.org/10.1097/gme.0b013e3182960cf8
- 185. Brinton, R. D. The healthy cell bias of estrogen action: mitochondrial bioenergetics and neurological implications. *Trends in Neurosciences* **31**, 529–537 (2008). https://doi.org/10.1016/j.tins.2008.07.003
- Brinton, R. D. Investigative Models for Determining Hormone Therapy-Induced Outcomes in Brain: Evidence in Support of a Healthy Cell Bias of Estrogen Action. *Ann N Y Acad Sci* **1052**, 57–74 (2005). https://doi.org/10.1196/annals.1347.005
- 187. Daniel, J. M. Estrogens, estrogen receptors, and female cognitive aging: the impact of timing. *Horm Behav* **63**, 231–237 (2013). https://doi.org/10.1016/j.yhbeh.2012.05.003

- 188. Petra, S. et al. Cognitive health after menopause: does menopausal hormone therapy affect it? Best Pract Res Clin Endocrinol Metab 36 101565 (2021). https://doi.org/10.1016/j.beem.2021.101565
- 189. Barth, C. et al. Menopausal hormone therapy and the female brain: leveraging neuroimaging and prescription registry data from the UK Biobank cohort. medRxiv (preprint). https://doi.org/10.1101/2024.04.08.24305450
- 190. Wadsworth, M. The origins and innovatory nature of the 1946 British national birth cohort study. *Longit Life Course Stud* **1**, 121–136 (2010).
- Kuh, D. *et al.* The MRC National Survey of Health and Development reaches age 70: maintaining participation at older ages in a birth cohort study. *Eur J Epidemiol* **31**, 1135 (2016). https://doi.org/10.1007%2Fs10654-016-0217-8
- 192. Kuh, D. & Hardy, R. Women's health in midlife: Findings from a British birth cohort study. Journal of the British Menopause Society 9, 55–60 (2003). https://doi.org/10.1258/136218003100322206
- 193. Lane, C. A. *et al.* Study protocol: Insight 46 a neuroscience sub-study of the MRC National Survey of Health and Development. *BMC Neurol* 17, 1–25 (2017). https://doi.org/10.1186/s12883-017-0846-x
- 194. James, S. N. *et al.* Using a birth cohort to study brain health and preclinical dementia: Recruitment and participation rates in Insight 46. *BMC Res Notes* 11, 1–9 (2018). https://doi.org/10.1186/s13104-018-3995-0
- 195. Pigeon, D. A. Tests used in the 1954 and 1957 surveys. in *The home and the school*. (ed. Douglas, J. W. B.) 129–132 (MacGibbon & Kee, London, 1964).
- 196. Pigeon, D. A., Douglas, J. W. B., Ross, J. M. & Simpson, H. R. Details of the fifteen years tests. in *All our future*. (Davies, London, 1968).
- 197. Heim, A. W. *Manual for the Group Test of General Intelligence AH4*. (National Foundation for Educational Research, London, 1955).
- 198. Ministry of Education. *Reading Ability*. (HMSO, London, 1950).
- 199. Richards, M. & Wadsworth, M. E. J. Long term effects of early adversity on cognitive function. *Arch Dis Child* **89**, 922–927 (2004). https://doi.org/10.1136/adc.2003.032490
- Richards, M., Hardy, R., Kuh, D. & Wadsworth, M. E. J. Birth weight and cognitive function in the British 1946 birth cohort: longitudinal population based study. *BMJ* 322, 199–203 (2001). https://doi.org/10.1136%2Fbmj.322.7280.199

- 201. Richards, M., Shipley, B., Fuhrer, R. & Wadsworth, M. E. J. Cognitive ability in childhood and cognitive decline in mid-life: longitudinal birth cohort study. *BMJ* **328**, 552 (2004). https://doi.org/10.1136/bmj.37972.513819.ee
- 202. Nelson, H. E. & Willison, J. *National Adult Reading Test (NART)*. (Nfer-Nelson, Windsor, 1991).
- 203. Roth, M. *et al.* CAMDEX: A Standardised Instrument for the Diagnosis of Mental Disorder in the Elderly with Special Reference to the Early Detection of Dementia. *The British Journal of Psychiatry* **149**, 698–709 (1986). https://doi.org/10.1192/bjp.149.6.698
- 204. James, S. N. *et al.* Lifetime affective problems and later-life cognitive state: Over 50 years of follow-up in a British birth cohort study. *J Affect Disord* 241, 348–355 (2018). https://doi.org/10.1016%2Fj.jad.2018.07.078
- 205. Hurst, L. *et al.* Lifetime socioeconomic inequalities in physical and cognitive aging. *Am J Public Health* **103**, 1641–1648 (2013). https://doi.org/10.2105%2FAJPH.2013.301240
- 206. Masi, S. *et al.* Patterns of adiposity, vascular phenotypes and cognitive function in the 1946 British Birth Cohort. *BMC Med* **16**, 75 (2018). https://doi.org/10.1186/s12916-018-1059-x
- 207. Proitsi, P. *et al.* Lifetime cognition and late midlife blood metabolites: findings from a British birth cohort. *Transl Psychiatry* 8, 203 (2018). https://doi.org/10.1038%2Fs41398-018-0253-0
- Mathuranath, P. S., Nestor, P. J., Berrios, G. E., Rakowicz, W. & Hodges, J. R. A brief cognitive test battery to differentiate Alzheimer's disease and frontotemporal dementia. *Neurology* 55, 1613–1620 (2000). https://doi.org/10.1212/01.wnl.0000434309.85312.19
- 209. Noone, P. Addenbrooke's Cognitive Examination-III. *Occup Med (Chic III)* **65**, 418–420 (2015). https://doi.org/10.1093/occmed/kqv041
- 210. Reitan, R. M. & Wolfson, D. The Halstead-Reitan neuropsychological test battery: Theory and clinical interpretation. *Reitan Neuropsychology* **4**, (1985).
- Folstein, M. F., Folstein, S. E. & Mchugh, P. R. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. *J Psychiatr Res.* 12, 189-198 (1975). https://doi.org/10.1016/0022-3956(75)90026-6
- 212. Wechsler, D. Wechsler memory scale revised edition. (1987).
- 213. Wechsler, D. Wechsler adult intelligence scale revised edition. (1981).
- 214. Jaeger, J. Digit Symbol Substitution Test: The Case for Sensitivity Over Specificity in Neuropsychological Testing. J Clin Psychopharmacol 38, 513 (2018). https://doi.org/10.1097/jcp.00000000000941

- Papp, K. V. *et al.* Development of a psychometrically equivalent short form of the facename associative memory exam for use along the early Alzheimers disease trajectory. *Clinical Neuropsychologist* 28, 771–785 (2014). https://doi.org/10.1080/13854046.2014.911351
- 216. Wechsler, D. The Wechsler abbreviated scale of intelligence. (1999).
- 217. Lu, K. *et al.* Increased variability in reaction time is associated with amyloid beta pathology at age 70. *Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring* **12**, e12076 (2020). https://doi.org/10.1002/dad2.12076
- 218. Lu, K. *et al.* Dissociable effects of APOE ε4 and β-amyloid pathology on visual working memory. *Nature Aging 2021 1:11* **1**, 1002–1009 (2021). https://doi.org/10.1038/s43587-021-00117-4
- Parker, T. D. *et al.* Amyloid β influences the relationship between cortical thickness and vascular load. *Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring* 12, (2020). https://doi.org/10.1002%2Fdad2.12022
- 220. Dickerson, B. C. *et al.* Alzheimer-signature MRI biomarker predicts AD dementia in cognitively normal adults. *Neurology* **76**, 1395–1402 (2011). https://doi.org/10.1212/wnl.0b013e3182166e96
- 221. Science, D. of E. and. Burnham Further Education Committee Grading Courses. (1972).
- 222. Bann, D., Cooper, R., Wills, A. K., Adams, J. & Kuh, D. Socioeconomic position across life and body composition in early old age: Findings from a British birth cohort study. J Epidemiol Community Health (1978) 68, 516–523 (2014). https://doi.org/10.1136/jech-2013-203373
- 223. Bland, R. Measuring 'Social Class' A discussion of the Registrar-General's Classification. *Sociology* **13**, (1979). https://doi.org/10.1177/003803857901300209
- 224. Hausmann, M., Schoofs, D., Rosenthal, H. E. S. & Jordan, K. Interactive effects of sex hormones and gender stereotypes on cognitive sex differences—A psychobiosocial approach. *Psychoneuroendocrinology* **34**, 389–401 (2009). https://doi.org/10.1016/j.psyneuen.2008.09.019
- 225. Van Tuyckom, C., Van De Velde, S. & Bracke, P. Does country-context matter? A crossnational analysis of gender and leisure time physical inactivity in Europe. *Eur J Public Health* **23**, 452–457 (2013). https://doi.org/10.1093/eurpub/cks009
- 226. Peters, S. A. E., Huxley, R. R. & Woodward, M. Do smoking habits differ between women and men in contemporary Western populations? Evidence from half a million people in the UK Biobank study. *BMJ Open* **4**, e005663 (2014). https://doi.org/10.1136/bmjopen-2014-005663

- 227. James, S.-N. *et al.* Timing of physical activity across adulthood on later-life cognition: 30 years follow-up in the 1946 British birth cohort. *J Neurol Neurosurg Psychiatry* **94**, 349–356 (2023). https://doi.org/10.1136/jnnp-2022-329955
- 228. Erickson, K. I., Hillman, C. H. & Kramer, A. F. Physical activity, brain, and cognition. *Curr Opin Behav Sci* **4**, 27–32 (2015). https://doi.org/10.1249%2FMSS.00000000001936
- Anstey, K. J., Von Sanden, C., Salim, A. & O'Kearney, R. Smoking as a Risk Factor for Dementia and Cognitive Decline: A Meta-Analysis of Prospective Studies. *Am J Epidemiol* 166, 367–378 (2007). https://doi.org/10.1093/aje/kwm116
- 230. Nebel, R. A. *et al.* Understanding the impact of sex and gender in Alzheimer's disease: A call to action. *Alzheimer's & Dementia* 14, 1171–1183 (2018). https://doi.org/10.1016/j.jalz.2018.04.008
- 231. Macintyre, S. & Hunt, K. Socio-economic Position, Gender and Health: How Do They Interact? J Health Psychol. 2, 315-334 (1997). https://doi.org/10.1177/135910539700200304
- Lenehan, M. E., Summers, M. J., Saunders, N. L., Summers, J. J. & Vickers, J. C. Relationship between education and age-related cognitive decline: a review of recent research. *Psychogeriatrics* 15, 154–162 (2015). https://doi.org/10.1111/psyg.12083
- 233. Wolfova, K., Csajbok, Z., Kagstrom, A., Kåreholt, I. & Cermakova, P. Role of sex in the association between childhood socioeconomic position and cognitive ageing in later life. *Scientific Reports* **11**, 4647 (2021). https://doi.org/10.1038/s41598-021-84022-1
- 234. Ghebremedhin, E. *et al.* Gender and age modify the association between APOE and ADrelated neuropathology. *Neurology* **56**, 1696–1701 (2001). https://doi.org/10.1212/wnl.56.12.1696
- 235. Vernon, P. E. All Our Future. A Longitudinal Study of Secondary Education. By J. W. B. Douglas, J. M. Ross and H. R. Simpson. London: Peter Davies. 1968. Pp. 241. Price 42s. *The British Journal of Psychiatry* 115, 1081–1082 (1969).
- 236. Wagen, A. Z. *et al.* Life course, genetic, and neuropathological associations with brain age in the 1946 British Birth Cohort: a population-based study. *Lancet Healthy Longev* 3, e607–e616 (2022). https://doi.org/10.1016/s2666-7568(22)00167-2
- James, S.-N. *et al.* Neuroimaging, clinical and life course correlates of normal-appearing white matter integrity in 70-year-olds. *Brain Commun* 18, (2023). https://doi.org/10.1093/braincomms/fcad225
- 238. Hyde, J. S. Sex and cognition: gender and cognitive functions. *Curr Opin Neurobiol* **38**, 53–56 (2016). https://doi.org/10.1016/j.conb.2016.02.007
- Mielke, G. I., da Silva, I. C. M., Kolbe-Alexander, T. L. & Brown, W. J. Shifting the Physical Inactivity Curve Worldwide by Closing the Gender Gap. *Sports Medicine* 48, 481–489 (2018). https://doi.org/10.1007/s40279-017-0754-7
- 240. Dumith, S. C., Hallal, P. C., Reis, R. S. & Kohl, H. W. Worldwide prevalence of physical inactivity and its association with human development index in 76 countries. *Prev Med (Baltim)* 53, 24–28 (2011). https://doi.org/10.1016/j.ypmed.2011.02.017
- 241. Pampel, F. C. Global patterns and determinants of sex differences in smoking. *Int J Comp Sociol* **47**, 466–487 (2006). https://doi.org/10.1177%2F0020715206070267
- 242. Wareham, N. J. *et al.* Validity and repeatability of the EPIC-Norfolk Physical Activity Questionnaire. *Int J Epidemiol* **31**, 168–174 (2002). https://doi.org/10.1093/ije/31.1.168
- 243. Logan, S. & Johnston, R. Investigating gender differences in reading. *Educational Review* 62, (2010). https://doi.org/10.1080/00131911003637006
- 244. Spencer, S. Girls at risk. Early school-leaving and early marriage in the 1950s. *Journal of Educational Administration and History* **41**, (2009). https://doi.org/10.1080/00220620902808251
- 245. Gerstorf, D., Herlitz, A. & Smith, J. Stability of sex differences in cognition in advanced old age: the role of education and attrition. *J Gerontol B Psychol Sci Soc Sci* **61**, (2006). https://doi.org/10.1093/geronb/61.4.p245
- 246. Karlsson, P., Thorvaldsson, V., Skoog, I., Gudmundsson, P. & Johansson, B. Birth cohort differences in fluid cognition in old age: comparisons of trends in levels and change trajectories over 30 years in three population-based samples. *Psychol Aging* **30**, 83–94 (2015). https://doi.org/10.1037/a0038643
- 247. Mielke, M. M., Vemuri, P. & Rocca, W. A. Clinical epidemiology of Alzheimer's disease: Assessing sex and gender differences. *Clinical Epidemiology* 6, 37–48 (2014). https://doi.org/10.2147%2FCLEP.S37929
- 248. Richards, M., Jarvis, M. J., Thompson, N. & Wadsworth, M. E. J. Cigarette Smoking and Cognitive Decline in Midlife: Evidence from a Prospective Birth Cohort Study. *Am J Public Health* **93**, 994–998 (2003). https://doi.org/10.2105/ajph.93.6.994
- 249. NHS. Early or delayed puberty. https://www.nhs.uk/conditions/early-or-delayed-puberty/ Date accessed: 20/6/2023
- 250. Peper, J. S. & Dahl, R. E. Surging Hormones: Brain-Behavior Interactions During Puberty. *Curr Dir Psychol Sci* **22**, 134 (2013). https://doi.org/10.1177/0963721412473755
- 251. Rehman, A. & Khalili, Y. Al. Neuroanatomy, Occipital Lobe. StatPearls (2021).
- 252. Bui, T. & Das, J. M. Neuroanatomy, Cerebral hemisphere. StatPearls (2023).

- 253. Sisk, C. L. & Zehr, J. L. Pubertal hormones organize the adolescent brain and behavior. *Front Neuroendocrinol* **26**, 163–174 (2005). https://doi.org/10.1016/j.yfrne.2005.10.003
- 254. Pirau, L. & Lui, F. Frontal Lobe Syndrome. *StatPearls* (2021).
- 255. Patel, A., Biso, G. M. N. R. & Fowler, J. B. Neuroanatomy, Temporal Lobe. *StatPearls* (2021).
- Giedd, J. N., Castellanos, F. X., Rajapakse, J. C., Vaituzis, A. C. & Rapoport, J. L. Sexual dimorphism of the developing human brain. *Prog Neuropsychopharmacol Biol Psychiatry* 21, 1185–1201 (1997). https://doi.org/10.1016/s0278-5846(97)00158-9
- 257. Wisdom, N. M., Callahan, J. L. & Hawkins, K. A. The effects of apolipoprotein E on nonimpaired cognitive functioning: A meta-analysis. *Neurobiol Aging* **32**, 63–74 (2011). https://doi.org/10.1016/j.neurobiolaging.2009.02.003
- 258. Ihle, A., Bunce, D. & Kliegel, M. APOE ε4 and Cognitive Function in Early Life: A metaanalysis. *Neuropsychology* **26**, 267–277 (2012). https://doi.org/10.1037/a0026769
- 259. Rusted, J. M. *et al.* APOE e4 polymorphism in young adults is associated with improved attention and indexed by distinct neural signatures. *Neuroimage* **65**, 364–373 (2013). https://doi.org/10.1016/j.neuroimage.2012.10.010
- 260. Mondadori, C. R. A. *et al.* Better Memory and Neural Efficiency in Young Apolipoprotein E ε4 Carriers. *Cerebral Cortex* **17**, 1934–1947 (2007). https://doi.org/10.1093/cercor/bhl103
- 261. Smith, C. J. & Ashford, J. W. Apolipoprotein ε4-Associated Protection Against Pediatric Enteric Infections Is a Survival Advantage in Pre-Industrial Populations. *Journal of Alzheimer's Disease* **93**, 907–918 (2023). https://doi.org/10.3233/jad-221218
- 262. Tuminello, E. R. & Han, S. D. The apolipoprotein e antagonistic pleiotropy hypothesis: Review and recommendations. *Int J Alzheimers Dis*, (2011). https://doi.org/10.4061/2011/726197
- 263. Promjunyakul, N. O. *et al.* Baseline NAWM structural integrity and CBF predict periventricular WMH expansion over time. *Neurology* **90**, e2119 (2018). https://doi.org/10.1212%2FWNL.00000000005684
- 264. Tuladhar, A. M. *et al.* White matter integrity in small vessel disease is related to cognition. *Neuroimage Clin* **7**, 518–524 (2015). https://doi.org/10.1016%2Fj.nicl.2015.02.003
- 265. Jack, C. R. *et al.* Age, Sex, and APOE ε4 Effects on Memory, Brain Structure, and β-Amyloid Across the Adult Life Span. *JAMA Neurol* 72, 511–519 (2015). https://doi.org/10.1001/jamaneurol.2014.4821

- 266. Janota, C., Lemere, C. A. & Brito, M. A. Dissecting the Contribution of Vascular Alterations and Aging to Alzheimer's Disease. *Mol Neurobiol* 53, 3793–3811 (2016). https://doi.org/10.1007/s12035-015-9319-7
- 267. Ruigrok, A. N. V. *et al.* A meta-analysis of sex differences in human brain structure. *Neuroscience and Biobehavioral Reviews* **39**, 34–50 (2014). https://doi.org/10.1016%2Fj.neubiorev.2013.12.004
- Pegueroles, J. *et al.* Longitudinal brain structural changes in preclinical Alzheimer's disease. *Alzheimers Dement* 13, 499–509 (2017).
  https://doi.org/10.1016/j.jalz.2016.08.010
- 269. Oschwald, J. *et al.* Brain structure and cognitive ability in healthy aging: A review on longitudinal correlated change. *Rev Neurosci* **31**, 1–57 (2019). https://doi.org/10.1515/revneuro-2018-0096
- 270. Villemagne, V. L. *et al.* Amyloid β deposition, neurodegeneration, and cognitive decline in sporadic Alzheimer's disease: a prospective cohort study. *Lancet Neurol* 12, 357–367 (2013). https://doi.org/10.1016/s1474-4422(13)70044-9
- 271. Reuben, A. *et al.* Improving risk indexes for Alzheimer's disease and related dementias for use in midlife. *Brain Commun* **4**, (2022). https://doi.org/10.1093/braincomms/fcac223
- Seux, M. *et al.* Correlates of cognitive status of old patients with isolated systolic hypertension: The Syst-Eur vascular dementia project. *Journal of Hypertension* 16, 963-969 (1998). https://doi.org/10.1097/00004872-199816070-00009
- 273. LaPlume, A. A., McKetton, L., Levine, B., Troyer, A. K. & Anderson, N. D. The adverse effect of modifiable dementia risk factors on cognition amplifies across the adult lifespan. *Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring* 14, e12337 (2022). https://doi.org/10.1002%2Fdad2.12337
- 274. Wang, X. J. *et al.* Early-Life Risk Factors for Dementia and Cognitive Impairment in Later Life: A Systematic Review and Meta-Analysis. *Journal of Alzheimer's Disease* 67, 221–229 (2019). https://doi.org/10.3233/jad-180856
- Puolakka, E. *et al.* Childhood socioeconomic status and lifetime health behaviors: The Young Finns Study. *Int J Cardiol* 258, 289–294 (2018). https://doi.org/10.1016/j.ijcard.2018.01.088
- 276. Dundas, R., Leyland, A. H. & MacIntyre, S. Early-Life School, Neighborhood, and Family Influences on Adult Health: A Multilevel Cross-Classified Analysis of the Aberdeen Children of the 1950s Study. Am J Epidemiol 180, 197 (2014). https://doi.org/10.1093%2Faje%2Fkwu110

- 277. Rodgers, B. & Mann, S. A. The reliability and validity of PSE assessments by lay interviewers: a national population survey. *Psychol Med* **16**, 689–700 (1986). https://doi.org/10.1017/s0033291700010436
- 278. MD Calc. CAGE Questions for Alcohol Use: Screens for excessive drinking and alcoholism. https://www.mdcalc.com/calc/1729/cage-questions-alcohol-use Date accessed: 03/10/2023
- 279. World Health Organization. AUDIT: the Alcohol use disorders identification test: guidelines for use in primary health care. https://www.who.int/publications/i/item/WHO-MSD-MSB-01.6a Date accessed: 03/10/2023
- 280. Byford, M., Abbott, R. A., Maughan, B., Richards, M. & Kuh, D. Adolescent mental health and subsequent parenting: a longitudinal birth cohort study. *J Epidemiol Community Health* **68**, 396 (2014). https://doi.org/10.1136/jech-2013-202997
- 281. Lindelow, M., Hardy, R. & Rodgers, B. Development of a scale to measure symptoms of anxiety and depression in the general UK population: the psychiatric symptom frequency scale. *J Epidemiol Community Health (1978)* **51**, 549–557 (1997). https://doi.org/10.1136/jech.51.5.549
- 282. Sterling, M. General Health Questionnaire-28 (GHQ-28). *Journal of Physiotherapy* **57**, (2011). https://doi.org/10.1016/s1836-9553(11)70060-1
- 283. Douglas, J. W. B. & Waller, R. E. Air Pollution and Respiratory Infection in Children. *Br J Prev Soc Med* **20**, 1 (1966).
- 284. Allinson, J. P. *et al.* Early childhood lower respiratory tract infection and premature adult death from respiratory disease in Great Britain: a national birth cohort study. *The Lancet* 401, 1183–1193 (2023). https://doi.org/10.1016/S0140-6736(23)00131-9
- 285. Elander, J. & Rutter, M. Use and development of the Rutter parents' and teachers' scales. International Journal of Methods in Psychiatric Research 6, 63-78 (1996). https://psycnet.apa.org/doi/10.1002/(SICI)1234-988X(199607)6:2%3C63::AID-MPR151%3E3.3.CO;2-M
- 286. Colman, I., Wadsworth, M. E. J., Croudace, T. J. & Jones, P. B. Forty-year psychiatric outcomes following assessment for internalizing disorder in adolescence. *American Journal of Psychiatry* 164, 126–133 (2007). https://doi.org/10.1176/ajp.2007.164.1.126
- 287. Cai, Y. et al. Cross-sectional associations between air pollution and chronic bronchitis: an ESCAPE meta-analysis across five cohorts. *Thorax* 69, 1005–1014 (2014). https://doi.org/10.1136/thoraxjnl-2013-204352

- 288. Liu, C.-C., Kanekiyo, T., Xu, H. & Bu, G. Apolipoprotein E and Alzheimer disease: risk, mechanisms, and therapy. *Nat Rev Neurol* 9, 106 (2013). https://doi.org/10.1038/nrneurol.2012.263
- Kang, J. H., Logroscino, G., De Vivo, I., Hunter, D. & Grodstein, F. Apolipoprotein E, cardiovascular disease and cognitive function in aging women. *Neurobiol Aging* 26, 475– 484 (2005). https://doi.org/10.1016/j.neurobiolaging.2004.05.003
- 290. Cooper, R., Hardy, R. & Kuh, D. Timing of menarche, childbearing and hysterectomy risk. *Maturitas* **61**, 317 (2008). https://doi.org/10.1016%2Fj.maturitas.2008.09.025
- 291. Kivimäki, M. *et al.* Body mass index and risk of dementia: Analysis of individual-level data from 1.3 million individuals. *Alzheimer's & Dementia* **14**, 601 (2018). https://doi.org/10.1016/j.jalz.2017.09.016
- 292. Bennett, S. & Thomas, A. J. Depression and dementia: cause, consequence or coincidence? *Maturitas* **79**, 184–190 (2014). https://doi.org/10.1016/j.maturitas.2014.05.009
- 293. Homans, N. C. *et al.* Prevalence of age-related hearing loss, including sex differences, in older adults in a large cohort study. *Laryngoscope* **127**, 725–730 (2017). https://doi.org/10.1002/lary.26150
- 294. Faul, M. & Coronado, V. Epidemiology of traumatic brain injury. *Handb Clin Neurol* **127**, 3– 13 (2015). https://doi.org/10.1016/b978-0-444-52892-6.00001-5
- 295. Grant, B. F. *et al.* Epidemiology of DSM-5 Alcohol Use Disorder: Results From the National Epidemiologic Survey on Alcohol and Related Conditions III. *JAMA Psychiatry* **72**, 757–766 (2015). https://doi.org/10.1001/jamapsychiatry.2015.0584
- 296. Eid, R. S., Gobinath, A. R. & Galea, L. A. M. Sex differences in depression: Insights from clinical and preclinical studies. *Prog Neurobiol* **176**, 86–102 (2019). https://doi.org/10.1016/j.pneurobio.2019.01.006
- 297. Rosenfeld, C. S. Sex-dependent differences in voluntary physical activity. *J Neurosci Res* **95**, 279–290 (2017). https://doi.org/10.1002/jnr.23896
- 298. McElroy, E. *et al.* Influence of childhood socioeconomic position and ability on mid-life cognitive function: evidence from three British birth cohorts. *J Epidemiol Community Health* **75**, 643–650 (2021). https://doi.org/10.1136%2Fjech-2020-215637
- 299. Kivipelto, M. *et al.* Apolipoprotein E ε4 magnifies lifestyle risks for dementia: a population-based study. *J Cell Mol Med* 12, 2762 (2008). https://doi.org/10.1111/j.1582-4934.2008.00296.x

- 300. Licher, S. *et al.* Genetic predisposition, modifiable-risk-factor profile and long-term dementia risk in the general population. *Nature Medicine 2019* 25, 1364–1369 (2019). https://doi.org/10.1038/s41591-019-0547-7
- 301. Lee, J.-O., Kim, J.-W., Kang, H.-J., Hong, J.-P. & Kim, J.-M. Predictors of Cognitive Improvement during 12 Weeks of Antidepressant Treatment in Patients with Major Depressive Disorder. *Clinical Psychopharmacology and Neuroscience* 16, 461 (2018). https://doi.org/10.9758%2Fcpn.2018.16.4.461
- 302. Dawes, P. et al. Hearing-aid use and long-term health outcomes: Hearing handicap, mental health, social engagement, cognitive function, physical health, and mortality. Int J Audiol 54, 838–844 (2015). https://doi.org/10.3109/14992027.2015.1059503
- 303. Weinstein, B. E., Sirow, L. W. & Moser, S. Relating Hearing Aid Use to Social and Emotional Loneliness in Older Adults. Am J Audiol 25, 54–61 (2016). https://doi.org/10.1044/2015\_aja-15-0055
- 304. Aazh, H., Prasher, D., Nanchahal, K. & Moore, B. C. J. Hearing-aid use and its determinants in the UK National Health Service: A cross-sectional study at the Royal Surrey County Hospital. Int J Audiol 54, 152–161 (2015). https://doi.org/10.3109/14992027.2014.967367
- 305. Zárate, S., Stevnsner, T. & Gredilla, R. Role of estrogen and other sex hormones in brain aging. Neuroprotection and DNA repair. *Frontiers in Aging Neuroscience* 9, 430 (2017). https://doi.org/10.3389%2Ffnagi.2017.00430
- Murray-Smith, H. *et al.* Updating the study protocol: Insight 46 a longitudinal neuroscience sub-study of the MRC National Survey of Health and Development - phases 2 and 3. *BMC Neurol* 24, 40 (2024). https://doi.org/10.1186/s12883-023-03465-3
- 307. Irwin, K., Sexton, C., Daniel, T., Lawlor, B. & Naci, L. Healthy aging and dementia: Two roads diverging in midlife? *Front Aging Neurosci* 10, 412313 (2018). https://doi.org/10.3389/fnagi.2018.00275
- 308. Lane, C. A. *et al.* Associations Between Vascular Risk Across Adulthood and Brain Pathology in Late Life: Evidence From a British Birth Cohort. *JAMA Neurol* 77, 175 (2020). https://doi.org/10.1001/jamaneurol.2019.3774
- 309. Singh-Manoux, A. et al. Obesity trajectories and risk of dementia: 28 years of follow-up in the Whitehall II Study. Alzheimer's & Dementia 14, 178–186 (2018). https://doi.org/10.1016/j.jalz.2017.06.2637
- 310. McGrath, E. R. *et al.* Blood pressure from mid-to late life and risk of incident dementia. *Neurology* **89**, 2447–2454 (2017). https://doi.org/10.1212/wnl.00000000004741

- 311. LaPlume, A. A., McKetton, L., Anderson, N. D. & Troyer, A. K. Sex differences and modifiable dementia risk factors synergistically influence memory over the adult lifespan. *Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring* 14, e12301 (2022). https://doi.org/10.1002%2Fdad2.12301
- 312. Sirey, J. A. *et al.* Predictors of antidepressant prescription and early use among depressed outpatients. *Am J Psychiatry* **156**, 690–696 (1999). https://doi.org/10.1176/ajp.156.5.690
- 313. Tobin, M. D., Sheehan, N. A., Scurrah, K. J. & Burton, P. R. Adjusting for treatment effects in studies of quantitative traits: antihypertensive therapy and systolic blood pressure. *Stat Med* 24, 2911–2935 (2005). https://doi.org/10.1002/sim.2165
- 314. Maddock, J. *et al.* Social Health and Change in Cognitive Capability among Older Adults: Findings from Four European Longitudinal Studies. *Gerontology* **69**, 1330–1346 (2023). https://doi.org/10.1159/000531969
- 315. Warburton, D. E. R., Nicol, C. W. & Bredin, S. S. D. Health benefits of physical activity: the evidence. *CMAJ: Canadian Medical Association Journal* **174**, 801 (2006). https://doi.org/10.1503/cmaj.051351
- 316. Livingston, G. *et al.* Dementia prevention, intervention, and care. *The Lancet* **390**, 2673–2734 (2017). https://doi.org/10.1016/s0140-6736(17)31363-6
- Needham, L. P. *et al.* A comprehensive assessment of age at menopause with wellcharacterized cognition at 70 years: A population-based British birth cohort. *Maturitas* 170, 31–38 (2023). https://doi.org/10.1016/j.maturitas.2023.01.009
- Riedel, B. C., Thompson, P. M. & Brinton, R. D. Age, APOE and sex: Triad of risk of Alzheimer's disease. *Journal of Steroid Biochemistry and Molecular Biology* 160, 134–147 (2016). https://doi.org/10.1016/j.jsbmb.2016.03.012
- 319. Koochmeshgi, J., Hosseini-Mazinani, S. M., Seifati, S. M., Hosein-Pur-Nobari, N. & Teimoori-Toolabi, L. Apolipoprotein E genotype and age at menopause. in *Annals of the New York Academy of Sciences* vol. 1019 564–567 (New York Academy of Sciences, 2004).
- 320. Meng, F.-T. et al. ApoE genotypes are associated with age at natural menopause in Chinese females. AGE 2011 34:4 34, 1023–1032 (2011). https://doi.org/10.1007%2Fs11357-011-9287-4
- 321. Kok, H. S. *et al.* Cognitive function across the life course and the menopausal transition in a British birth cohort. *Menopause* 13, 19–27 (2006). https://doi.org/10.1097/01.gme.0000196592.36711.a0
- 322. Donohue, M. C. *et al.* The preclinical Alzheimer cognitive composite: measuring amyloidrelated decline. *JAMA Neurol* **71**, 961–970 (2014). https://doi.org/10.1001/jamaneurol.2014.803

- 323. Cooper, R. *et al.* Validity of age at menarche self-reported in adulthood. *J Epidemiol Community Health (1978)* **60**, 993 (2006). https://doi.org/10.1136/jech.2005.043182
- 324. Al-Safi, Z. A. & Polotsky, A. J. Obesity and Menopause. *Best Pract Res Clin Obstet Gynaecol* **29**, 548–553 (2015). https://doi.org/10.1016/j.bpobgyn.2014.12.002
- 325. Hsieh, S., Schubert, S., Hoon, C., Mioshi, E. & Hodges, J. R. Validation of the Addenbrooke's Cognitive Examination III in Frontotemporal Dementia and Alzheimer's Disease. *Dement Geriatr Cogn Disord* 36, 242–250 (2013). https://doi.org/10.1159/000351671
- 326. Hardy, R. & Kuh, D. Social and environmental conditions across the life course and age at menopause in a British birth cohort study. *BJOG* **112**, 346–354 (2005). https://doi.org/10.1111/j.1471-0528.2004.00348.x
- 327. Sievert, L. L., Murphy, L., Morrison, L. A., Reza, A. M. & Brown, D. E. Age at menopause and determinants of hysterectomy and menopause in a multi-ethnic community: The Hilo Women's Health Study. *Maturitas* 76, 334–341 (2013). https://doi.org/10.1016/j.maturitas.2013.08.007
- 328. Secoșan, C. *et al.* Surgically Induced Menopause-A Practical Review of Literature. *Medicina (Kaunas)* **55**, (2019). https://doi.org/10.3390%2Fmedicina55080482
- 329. McCarrey, A. C. & Resnick, S. M. Postmenopausal hormone therapy and cognition. *Horm Behav* **74**, 167–172 (2015). https://doi.org/10.1016%2Fj.yhbeh.2015.04.018
- Than, S. *et al.* Interactions Between Age, Sex, Menopause, and Brain Structure at Midlife: A UK Biobank Study. *J Clin Endocrinol Metab* 106, 410–420 (2021). https://doi.org/10.1210/clinem/dgaa847
- Richards, M., Kuh, D., Hardy, R. & Wadsworth, M. Lifetime cognitive function and timing of the natural menopause. *Neurology* 53, 308–314 (1999). https://doi.org/10.1212/wnl.53.2.308
- 332. Subramaniapillai, S. *et al.* Sex- and age-specific associations between cardiometabolic risk and white matter brain age in the UK Biobank cohort. *Hum Brain Mapp* 43, 3759–3774 (2022). https://doi.org/10.1002/hbm.25882
- 333. Moir, M. E. *et al.* Age at natural menopause impacts cerebrovascular reactivity and brain structure. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology* **324**, R207–R215 (2023). https://doi.org/10.1152/ajpregu.00228.2022
- 334. Valencia-Olvera, A. C. et al. Role of estrogen in women's Alzheimer's disease risk as modified by APOE. J Neuroendocrinol 35, e13209 (2022). https://doi.org/10.1111/jne.13209

- 335. Kantarci, K. *et al.* Brain structure and cognition 3 years after the end of an early menopausal hormone therapy trial. *Neurology* **90**, E1404–E1412 (2018). https://doi.org/10.1212/wnl.00000000005325
- 336. Rodriguez, M. & Shoupe, D. Surgical Menopause. *Endocrinol Metab Clin North Am* **44**, 531–542 (2015). https://doi.org/10.1016/j.ecl.2015.05.003
- Elbejjani, M. *et al.* Life-Course Socioeconomic Position and Hippocampal Atrophy in a Prospective Cohort of Older Adults. *Psychosom Med* **79**, 14–23 (2017). https://doi.org/10.1097/psy.00000000000365
- 338. Rao, Y. L. *et al.* Hippocampus and its involvement in Alzheimer's disease: a review. *3 Biotech 2022 12:2* **12**, 1–10 (2022). https://doi.org/10.1007%2Fs13205-022-03123-4
- 339. Bean, L. A., Ianov, L. & Foster, T. C. Estrogen Receptors, the Hippocampus, and Memory. *Neuroscientist* **20**, 534 (2014). https://doi.org/10.1177/1073858413519865
- 340. Spencer, J. L. *et al.* Uncovering the mechanisms of estrogen effects on hippocampal function. *Front Neuroendocrinol* **29**, 219–237 (2008). https://doi.org/10.1016/j.yfrne.2007.08.006
- 341. Fotenos, A. F., Snyder, A. Z., Girton, L. E., Morris, J. C. & Buckner, R. L. Normative estimates of cross-sectional and longitudinal brain volume decline in aging and AD. *Neurology* 64, 1032–1039 (2005). https://doi.org/10.1212/01.wnl.0000154530.72969.11
- 342. Maric-Bilkan, C., Gilbert, E. L. & Ryan, M. J. Impact of ovarian function on cardiovascular health in women: Focus on hypertension. *Int J Womens Health* 6, 131–139 (2014). https://doi.org/10.2147%2FIJWH.S38084
- 343. Beauchet, O. *et al.* Blood pressure levels and brain volume reduction: A systematic review and meta-analysis. *J Hypertens* **31**, 1502–1516 (2013). https://doi.org/10.1097/hjh.0b013e32836184b5
- 344. Power, M. C. *et al.* Life-course blood pressure in relation to brain volumes. *Alzheimer's & Dementia* **12**, 890–899 (2016). https://doi.org/10.1016%2Fj.jalz.2016.03.012
- 345. Kanekiyo, T., Xu, H. & Bu, G. ApoE and Aβ in Alzheimer's Disease: Accidental Encounters or Partners? *Neuron* **81**, 740–754 (2014). https://doi.org/10.1016/j.neuron.2014.01.045
- 346. Kok, E. *et al.* Apolipoprotein E–dependent accumulation of Alzheimer disease–related lesions begins in middle age. *Ann Neurol* **65**, 650–657 (2009). https://doi.org/10.1002/ana.21696
- 347. Van Beresteijn, E. C. H. *et al.* Perimenopausal increase in serum cholesterol: a 10-year longitudinal study. *Am J Epidemiol* 137, 383–392 (1993).
  https://doi.org/10.1093/oxfordjournals.aje.a116686

- 348. Kok, H. S. *et al.* Heart Disease Risk Determines Menopausal Age Rather Than the Reverse. *J Am Coll Cardiol* **47**, 1976–1983 (2006). https://doi.org/10.1016/j.jacc.2005.12.066
- 349. Gamache, J., Yun, Y. & Chiba-Falek, O. Sex-dependent effect of APOE on Alzheimer's disease and other age-related neurodegenerative disorders. *Dis Model Mech* **13**, (2020). https://doi.org/10.1242/dmm.045211
- Reinvang, I., Espeseth, T. & Westlye, L. T. APOE-related biomarker profiles in non-pathological aging and early phases of Alzheimer's disease. *Neurosci Biobehav Rev* 37, 1322–1335 (2013). https://doi.org/10.1016/j.neubiorev.2013.05.006
- 351. Flowers, S. A. & Rebeck, G. W. APOE in the normal brain. *Neurobiol Dis* **136**, 104724 (2020). https://doi.org/10.1016/j.nbd.2019.104724
- 352. Liu, Y. *et al.* Effect of APOE ε4 Allele on Cortical Thicknesses and Volumes: The AddNeuroMed Study. *Journal of Alzheimer's Disease* **21**, 947–966 (2010). https://doi.org/10.3233/jad-2010-100201
- 353. Chutinet, A. & Rost, N. S. White matter disease as a biomarker for long-term cerebrovascular disease and dementia topical collection on cerebrovascular disease and stroke. *Curr Treat Options Cardiovasc Med* 16, 1–12 (2014). https://doi.org/10.1007/s11936-013-0292-z
- 354. Willey, J. Z. *et al.* Lipid Profile Components and Subclinical Cerebrovascular Disease in the Northern Manhattan Study. *Cerebrovascular Diseases* **37**, 423–430 (2014). https://doi.org/10.1159%2F000362920
- 355. Parente, R. C., Faerstein, E., Celeste, R. K. & Werneck, G. L. The relationship between smoking and age at the menopause: A systematic review. *Maturitas* **61**, 287–298 (2008). https://doi.org/10.1016/j.maturitas.2008.09.021
- 356. Power, M. C. *et al.* Smoking and white matter hyperintensity progression. *Neurology* **84**, 841–848 (2015). https://doi.org/10.1212%2FWNL.00000000001283
- 357. Debette, S. et al. Midlife vascular risk factor exposure accelerates structural brain aging and cognitive decline. Neurology 77, 461–468 (2011). https://doi.org/10.1212%2FWNL.0b013e318227b227
- 358. Gambacciani, M. *et al.* Climacteric modifications in body weight and fat tissue distribution. *Climacteric* **2**, 37–44 (1999). https://doi.org/10.3109/13697139909025561
- 359. Moolman, J. A. Unravelling the cardioprotective mechanism of action of estrogens. *Cardiovasc Res* **69**, 777–780 (2006). https://doi.org/10.1016/j.cardiores.2006.01.001
- 360. Martínez-Martínez, A. B. *et al.* Beyond the CNS: The many peripheral roles of APOE. *Neurobiol Dis* **138**, (2020). https://doi.org/10.1016/j.nbd.2020.104809

- 361. Schindler, L. S. *et al.* Associations between abdominal adipose tissue, reproductive span, and brain characteristics in post-menopausal women. *Neuroimage Clin* 36, (2022). https://doi.org/10.1016%2Fj.nicl.2022.103239
- 362. Subramaniapillai, S. *et al.* Sex differences in brain aging among adults with family history of Alzheimer's disease and APOE4 genetic risk. *Neuroimage Clin* **30**, 102620 (2021). https://doi.org/10.1016/j.nicl.2021.102620
- 363. Kolb, H. *et al.* Ketone bodies: from enemy to friend and guardian angel. *BMC Medicine* 2021 19:1 **19**, 1–15 (2021). https://doi.org/10.1186/s12916-021-02185-0
- 364. Klosinski, L. P. *et al.* White Matter Lipids as a Ketogenic Fuel Supply in Aging Female Brain: Implications for Alzheimer's Disease. *EBioMedicine* 2, 1888 (2015). https://doi.org/10.1016%2Fj.ebiom.2015.11.002
- 365. Lynch, J. & Smith, G. D. A life course approach to chronic disease epidemiology. *Annu Rev Public Health* 26, 1-25 (2005).
  https://doi.org/10.1146/annurev.publhealth.26.021304.144505
- 366. Rehbein, E., Hornung, J., Sundström Poromaa, I. & Derntl, B. Shaping of the Female Human Brain by Sex Hormones: A Review. *Neuroendocrinology* **111**, 183–206 (2021). https://doi.org/10.1159/000507083
- 367. Davis, A. & King, L. Gendered Perspectives on Men's Changing Familial Roles in Postwar England, c.1950–1990. Gend Hist **30**, 70–92 (2018). https://doi.org/10.1111%2F1468-0424.12333
- 368. Brooke, S. Gender and Working Class Identity in Britain during the 1950s. *Journal of Social History* **34**, 773-795 (2001). https://www.jstor.org/stable/3789418
- 369. National Centre for Social Research. British social attitudes 30: How and why Britain's attitudes and values are changing. https://natcen.ac.uk/publications/british-social-attitudes-30 Date accessed: 08/2/2024
- 370. Sohn, D. *et al.* Sex Differences in Cognitive Decline in Subjects with High Likelihood of Mild Cognitive Impairment due to Alzheimer's disease. *Sci Rep* 8, (2018). https://doi.org/10.3389%2Ffnagi.2022.959394
- 371. Laws, K. R., Irvine, K. & Gale, T. M. Sex differences in cognitive impairment in Alzheimer's disease. *World J Psychiatry* **6**, 54 (2016). https://doi.org/10.5498%2Fwjp.v6.i1.54
- 372. Van Loenhoud, A. C. *et al.* Cognitive reserve and clinical progression in Alzheimer disease: A paradoxical relationship. *Neurology* 93, e334 (2019). https://doi.org/10.1212/wnl.000000000007821

- 373. Pradier, C. *et al.* The Mini Mental State Examination at the Time of Alzheimer's Disease and Related Disorders Diagnosis, According to Age, Education, Gender and Place of Residence: A Cross-Sectional Study among the French National Alzheimer Database. *PLoS One* **9**, (2014). https://doi.org/10.1371%2Fjournal.pone.0103630
- 374. Bradford, A., Kunik, M. E., Schulz, P., Williams, S. P. & Singh, H. Missed and delayed diagnosis of dementia in primary care: Prevalence and contributing factors. *Alzheimer Dis Assoc Disord* 23, 306–314 (2009). https://doi.org/10.1097%2FWAD.0b013e3181a6bebc
- 375. Sperling, R. A. *et al.* The A4 study: stopping AD before symptoms begin? *Sci Transl Med* **6**, (2014). https://doi.org/10.1126/scitranslmed.3007941
- 376. National Institute for Health and Care Excellence. How should I assess a person with suspected dementia? https://cks.nice.org.uk/topics/dementia/diagnosis/assessment/ Date accessed: 09/2/2024
- 377. Bielak, A. A. M., Hatt, C. R. & Diehl, M. Cognitive Performance in Adults' Daily Lives: Is There a Lab-Life Gap? *Res Hum Dev* 14, 219–233 (2017). https://doi.org/10.1080/15427609.2017.1340050
- 378. Pashmdarfard, M. & Azad, A. Assessment tools to evaluate Activities of Daily Living (ADL) and Instrumental Activities of Daily Living (IADL) in older adults: A systematic review. *Med J Islam Repub Iran* 34, 33 (2020). https://doi.org/10.34171/mjiri.34.33
- 379. van Zyl, C. J. J. A network analysis of the General Health Questionnaire. J Health Psychol 26, 249–259 (2021). https://doi.org/10.1177/1359105318810113
- 380. Mikkola, T. S., Gissler, M., Merikukka, M., Tuomikoski, P. & Ylikorkala, O. Sex Differences in Age-Related Cardiovascular Mortality. *PLoS One* 8, (2013). https://doi.org/10.1371/journal.pone.0063347
- 381. Office for National Statistics. Leading causes of death, UK: 2001 to 2018. Office for National Statistics. https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/causesof death/articles/leadingcausesofdeathuk/2001to2018 Date accessed: 10/2/2024
- 382. Paterson, L. Education and high-status occupations in the UK since the middle of the twentieth century. Br J Sociol Educ 43, 375–396 (2022). http://dx.doi.org/10.1080/01425692.2022.2026763
- 383. Mielke, M. M. Sex and gender differences in Alzheimer disease dementia. *Psychiatric Times* **35**, 14–15 (2018).
- 384. Bloomberg, M. *et al.* Comparison of sex differences in cognitive function in older adults between high- and middle-income countries and the role of education: a population-based multicohort study. *Age Ageing* **52**, (2023). https://doi.org/10.1093/ageing/afad019

- 385. Mukadam, N., Marston, L., Lewis, G. & Livingston, G. Risk factors, ethnicity and dementia: A UK Biobank prospective cohort study of White, South Asian and Black participants. *PLoS One* 17, (2022). https://doi.org/10.1371/journal.pone.0275309
- 386. Mukadam, N. *et al.* Incidence, age at diagnosis and survival with dementia across ethnic groups in England: A longitudinal study using electronic health records. *Alzheimers Dement* 19, 1300–1307 (2023). https://doi.org/10.1002/alz.12774
- Avila, J. F. *et al.* Sex/Gender Differences in Cognitive Trajectories Vary as a Function of Race/Ethnicity. *Alzheimers Dement* 15, 1516 (2019). https://doi.org/10.1016%2Fj.jalz.2019.04.006
- 388. 2021 Alzheimer's disease facts and figures. *Alzheimers Dement* **17**, 327–406 (2021). https://doi.org/10.1002/alz.12328
- 389. Misiura, M. B. *et al.* Intersectionality in Alzheimer's Disease: The Role of Female Sex and Black American Race in the Development and Prevalence of Alzheimer's Disease. *Neurotherapeutics* 20, 1019–1036 (2023). https://doi.org/10.1007/s13311-023-01408-x
- 390. Diamond, L. M. Gender Fluidity and Nonbinary Gender Identities Among Children and Adolescents. *Child Dev Perspect* 14, 110–115 (2020). http://dx.doi.org/10.1111/cdep.12366
- 391. Johnson, J. L., Greaves. & Repta, R. Better science with sex and gender: A primer for health research. *Vancouver: Women's Health Research Network* (2007).
- 392. Johnson, J. L., Greaves, L. & Repta, R. Better science with sex and gender: Facilitating the use of a sex and gender-based analysis in health research. *Int J Equity Health* 8, 1–11 (2009). http://dx.doi.org/10.1186/1475-9276-8-14
- 393. Meyer, I. H. Prejudice, Social Stress, and Mental Health in Lesbian, Gay, and Bisexual Populations: Conceptual Issues and Research Evidence. *Psychol Bull* **129**, 674–697 (2003). https://doi.org/10.1037%2F0033-2909.129.5.674
- 394. Brady, B., Zheng, L., Kootar, S. & Anstey, K. J. Sex and gender differences in risk scores for dementia and Alzheimer's disease among cisgender, transgender, and non-binary adults. *Alzheimer's & Dementia* 20, 5–15 (2024). https://doi.org/10.1002/alz.13317
- 395. Institute of Medicine (US) Committee on Ethical and Legal Issues Relating to the Inclusion of Women in Clinical Studies. Reports on women's participation in clinical studies, 1977-1993. In Women and Health Research: Ethical and Legal Issues of Including Women in Clinical Studies. (Eds. Mastronianni, A. C., Faden, R. & Federman, D.) (Washington: National Academies Press, 1994).

- 396. Arnegard, M. E., Whitten, L. A., Hunter, C. & Clayton, J. A. Sex as a Biological Variable: A 5-Year Progress Report and Call to Action. J Womens Health 29, 858–864 (2020). https://doi.org/10.1089/jwh.2019.8247
- 397. Haverfield, J. & Tannenbaum, C. A 10-year longitudinal evaluation of science policy interventions to promote sex and gender in health research. *Health Res Policy Syst* 19, 1–12 (2021). https://doi.org/10.1186/s12961-021-00741-x
- 398. Government of Canada. Health Portfolio Sex and Gender-Based Analysis Plus Policy: Advancing equity, diversity and inclusion. https://www.canada.ca/en/healthcanada/corporate/transparency/heath-portfolio-sex-gender-based-analysis-policy.html Date accessed: 13/2/2024
- 399. National Institutes of Health. Consideration of Sex as a Biological Variable in NIH-funded Research. https://grants.nih.gov/grants/guide/notice-files/not-od-15-102.html Date accessed: 13/2/2024
- 400. UK Research and Innovation. Sex in experimental design. https://www.ukri.org/councils/mrc/guidance-for-applicants/policies-and-guidance-forresearchers/sex-in-experimental-design/ Date accessed: 13/2/2024
- 401. Department of Health and Social Care. Policy paper: Women's Health Strategy for England. https://www.gov.uk/government/publications/womens-health-strategy-forengland Date accessed: 13/2/2024
- 402. Wellbeing of women. Women's Health Strategy: One Year On. https://www.wellbeingofwomen.org.uk/news/womens-health-strategy-one-year-on/ Date accessed: 13/2/2024
- 403. Schott, J. M. How preventable is dementia? *Pract Neurol* **22**, 446–447 (2022). https://doi.org/10.1136/pn-2022-003418
- 404. Hoekzema, E. *et al.* Pregnancy leads to long-lasting changes in human brain structure. *Nature Neuroscience 2016 20:2* **20**, 287–296 (2016). https://doi.org/10.1038/nn.4458
- 405. Siepmann, T. *et al.* Long-term cerebral white and gray matter changes after preeclampsia. *Neurology* **88**, 1256–1264 (2017). https://doi.org/10.1212/wnl.00000000003765
- 406. Postma, I. R., De Groot, J. C., Aukes, A. M., Aarnoudse, J. G. & Zeeman, G. G. Cerebral white matter lesions and perceived cognitive dysfunction: the role of pregnancy. *Am J Obstet Gynecol* **211**, 257.e1-257.e5 (2014). https://doi.org/10.1016/j.ajog.2014.02.031
- 407. Mielke, M. M. *et al.* Impaired Cognition and Brain Atrophy Decades After Hypertensive Pregnancy Disorders. *Circ Cardiovasc Qual Outcomes* 9, S70–S76 (2016). https://doi.org/10.1161/circoutcomes.115.002461

# Appendix A



**Supplementary Figure 1.** Histogram displaying the distribution of continuous beta-amyloid standardised uptake value ratio (SUVR) in Insight 46 males (n=235) and females (n=227), measured using amyloid PET-MRI at age 69-71 during wave I of Insight 46.



# Appendix B (Chapter 4 supplementary)

**Supplementary Figure 1.** Distribution of cumulative dementia risk scores up to midlife in males (N=774) and females (N=830), in the whole-cohort sample.



**Supplementary Figure 2.** Distribution of scores to midlife per risk factor, in males (N=774) and females (N=830) in the whole cohort analytical sample. The maximum score per risk is 2. Low education 'No'=0, 'Yes'=2; TBI 'Never'=0, 'Ever'=2; diabetes 'Never'=0, 'Since mid-life'=1, 'Since early-life'=2 (for data protection reasons, diabetes categories have been collapsed); all other risks are scored 0 to 2 where 0 means that the risk is never present and 2 means that the risk is present at two timepoints. **A)** Mean scores for males and females. **B)** Percentage of males and females with different levels of risk factor exposures. TBI=traumatic brain injury. \* t-test p<0.05; \*\* t-test p<0.01



**Supplementary Figure 3.** Distribution of non-weighted lifetime cumulative dementia risk scores in males (N=740) and females (N=769), in the whole-NSHD analytical sample.



**Supplementary Figure 4.** Distribution of non-weighted lifetime cumulative dementia risk scores in males (N=201) and females (N=188), in the Insight 46 analytical sample.

Supplementary Table 1. Distributions of total midlife cumulative risks score (CRS) and individual risk factor scores in Insight 46 males and females.

	Male		Female		
	2	215 (51.2%)		05 (48.8%)	
Variable	Ν	Mean(SD)	Ν	Mean(SD)	Linear regression t-test p-value <sup>a</sup>
Midlife cumulative risks score; range		0.19(0.10);		0.17(0.11);	0.073
		0.00, 0.48		0.00, 0.54	
Low education		0.23(0.64)		0.26(0.67)	0.566
Hearing impairment		0.24(0.47)		0.21(0.43)	0.386
ТВІ		0.44(0.83)		0.21(0.61)	<0.001
Hypertension		0.74(0.72)		0.50(0.62)	<0.001
High alcohol consumption		0.65(0.76)		0.15(0.43)	<0.001
Obesity		0.22(0.50)		0.25(0.50)	0.573
Smoking		0.73(0.76)		0.65(0.75)	0.235
Depression		0.25(0.49)		0.43(0.59)	<0.001
Social isolation		0.26(0.49)		0.19(0.48)	0.103
Physical inactivity		0.72(0.72)		0.81(0.75)	0.206
Air pollution		0.37(0.56)		0.34(0.57)	0.607
Diabetes		0.02(0.15)		0.01(0.09)	0.502

Bold text indicates a significant (p<0.05) sex difference.

SD=standard deviation; TBI=traumatic brain injury

**Supplementary Table 2.** Model estimates and 95% bias-corrected confidence intervals demonstrating the estimated fully adjusted effects of cumulative risks score (CRS) on ACE-III performance at age 69, standardised to the Insight 46 sample, in Insight 46 males and females. Models are fully adjusted for childhood cognition, childhood SEP, and APOE-ε4 status.

	Males	Females
Ν	163	155
β	-0.875	-1.494
95% CI	-2.383, 0.641	-3.120, -0.033
р	-	*

\*Significant effect, confidence interval does not cross zero

Bootstrapping applied: confidence intervals are bias-corrected, p values are not applicable.

**Supplementary Table 3.** Model estimates and 95% confidence intervals demonstrating the estimated fully adjusted effects of cumulative risks score (CRS) on verbal memory performance at age 69, standardised to the Insight 46 sample, in Insight 46 males and females. Models are fully adjusted for childhood cognition, childhood SEP, and APOE- $\varepsilon$ 4 status.

	Males	Females
Ν	197	180
β	-0.783	-0.684
95% CI	-2.136, 0.571	-2.043, 0.675
р	0.26	0.32

# Appendix C (Chapter 5 supplementary)

### 5i. Supplementary material

**Supplementary Table 1.** Monte Carlo Error (MCE) estimations generated from fully adjusted multivariable regression models for menopause age on standardised outcome measures, with multiple imputation by chained equations (MICE) applied using 8 imputations. MCE of the coefficient should be <10% of its standard error.

Outcome		Beta	Standard error
	Model estimation	0.010	0.006
	MCE	0.0002	
ACE III attention and orientation	Model estimation	0.001	0.008
ACE-III attention and orientation	MCE	0.0002	
	Model estimation	0.005	0.007
ACE-III language	MCE	0.0003	
	Model estimation	0.009	0.007
ACE-III Memory	MCE	0.0002	
	Model estimation	0.006	0.007
ACE-III verbar idency	MCE	0.0002	
	Model estimation	0012	0.007
ACE-III VISUOSPatiai Tunction	MCE	0.0002	
MAASE	Model estimation	0.002	0.016
IVIIVISE	MCE	0.0004	

**Supplementary Table 2.** Model estimates and bootstrap 95% confidence intervals, corresponding with Figure 2, for the effect of 1-year increase in age at menopause on standardised z-scores for the Addenbrooke's Cognitive Examination (ACE-III; total score and sub-domains) at age 69 in the National Survey of Health and Development (NSHD) whole-cohort. N=746.

		M0			M1			M2			M3			M4	
Outcome	β	Lower Cl	Upper Cl												
ACE-III total ACE-III	0.024	0.012	0.036	0.009	-0.002	0.019	0.011	-0.002	0.025	0.010	-0.004	0.024	0.010	-0.004	0.024
attention and orientation	0.006	-0.007	0.022	0.001	-0.013	0.015	0.001	-0.015	0.016	0.001	-0.015	0.016	0.001	-0.015	0.016
ACE-III language	0.012	0.001	0.026	0.000	-0.009	0.013	0.005	-0.008	0.022	0.005	-0.008	0.021	0.005	-0.008	0.021
ACE-III memory	0.023	0.009	0.036	0.012	0.000	0.025	0.011	-0.004	0.026	0.009	-0.006	0.023	0.009	-0.006	0.024
ACE-III verbal fluency	0.016	0.005	0.027	0.005	-0.007	0.016	0.006	-0.010	0.021	0.006	-0.009	0.022	0.006	-0.009	0.022
ACE-III visuospatial function	0.020	0.007	0.032	0.010	-0.001	0.022	0.014	-0.002	0.028	0.013	-0.004	0.026	0.012	-0.004	0.026

 $\beta$ = z-score standardised coefficient beta; CI=confidence interval (bootstrap bias corrected). Note that, because bootstrapping is applied, p-values are not applicable.

*M0: unadjusted; M1: childhood cognition, childhood socioeconomic position, education; M2: M1+ age at menarche, parity, menopause type; M3: M2+ BMI, smoking, APOE-ε4 status, affective symptoms; M4: M3+ hormone therapy (HT)*  **Supplementary Table 3.** Model estimates and 95% confidence intervals, corresponding with Figure 3, for the effect of 1-year increase in age at menopause on standardised z-scores for the Preclinical Alzheimer's Cognitive Composite (PACC; total score and sub-tests) at age 69 to 71 in the Insight 46 sample. N=197.

		M0			M1			M2	M3		M4				
Outcome	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
PACC total	0.029	0.011, 0.048	0.002	0.015	-0.002, 0.032	0.089	0.018	-0.004, 0.040	0.112	0.019	-0.004, 0.042	0.099	0.019	-0.004, 0.042	0.099
DSST	0.033	0.008, 0.057	0.008	0.018	-0.006, 0.041	0.147	0.030	-0.001, 0.0622	0.058	0.031	-0.001, 0.062	0.058	0.031	-0.001, 0.062	0.058
FNAME- 12A	0.033	0.008, 0.058	0.010	0.017	-0.007, 0.041	0.164	0.035	0.003, 0.067	0.030	0.037	0.005, 0.069	0.023	0.037	0.005, 0.069	0.023
Logical															
memory delayed	0.031	0.007, 0.056	0.012	0.015	-0.009, 0.039	0.222	0.003	-0.028, 0.034	0.852	0.007	-0.025, 0.038	0.684	0.006	-0.025, 0.038	0.684
recall MMSE*	0.021	-0.003, 0.052	-	0.009	-0.014, 0.039	-	0.004	-0.026, 0.037	-	0.001	-0.033, 0.032	-	0.003	-0.035, 0.031	-

β=z-score standardised coefficient beta; CI=confidence interval; DSST=Digit-Symbol Substitution Test; FNAME=Face-Name Associative Memory Examination; MMSE=Mini-Mental State Examination

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable. Some bootstrap replications failed (MO: no failures. M1: 12 replication failures. M2: 10 replication failures. M3: 8 replication failures. M4: 16 replication failures).

*M0: unadjusted; M1: childhood cognition, childhood socioeconomic position, education; M2: M1+ age at menarche, parity, menopause type; M3: M2+ BMI, smoking, APOE-ε4 status, affective symptoms, age at cognitive testing; M4: M3+ hormone therapy (HT)*  **Supplementary Table 4.** Model estimations for menopause age-by-menopause type (natural or surgical) interaction terms in fully adjusted models (M3) regressing menopause age (in years) on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

Outcome	N	Menopause Age x Menopause Type β	Lower Cl	Upper Cl	Р
ACE-III total*	746	-0.005	-0.033	0.021	-
PACC total	197	-0.027	-0.073	0.019	0.243

 $\beta$ = z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: natural menopause

M3: adjusted for childhood cognition, childhood socioeconomic position, education, age at menarche, parity, BMI, smoking, APOE- $\epsilon$ 4 status, affective symptoms, age at cognitive testing [PACC total only]

**Supplementary Table 5.** Model estimations for menopause age-by-APOE-ε4 status (ε4 carrier or non-carrier) interaction terms in fully adjusted models (M3) regressing menopause age (in years) on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

Outcome	Ν	Menopause Age x APOE β	Lower Cl	Upper Cl	Р
ACE-III total*	746	-0.001	-0.002	0.002	-
PACC total	197	-0.001	-0.039	0.037	0.949

 $\beta$ = z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: APOE-ɛ4 non-carrier

M3: adjusted for childhood cognition, childhood socioeconomic position, education, age at menarche, parity, menopause type, BMI, smoking, affective symptoms, age at cognitive testing [PACC total only]

**Supplementary Table 6.** Model estimations and 95% confidence intervals for the effect of ever using menopausal hormone therapy (HT), compared with never using HT, on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

			Unadj	usted		Ac	ljusted for	menopause	age
Outcome	Ν	β	Lower Cl	Upper Cl	Р	β	Lower Cl	Upper Cl	p-value
ACE-III total*	746	-0.041	-0.157	0.147	-	0.022	-0.151	0.191	-
PACC total	197	-0.024	-0.265	0.216	0.841	0.059	-0.183	0.301	0.630

 $\beta$ = z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: HT never used

Unadjusted: only the exposure (HT use) and the outcome are included in the model.

Adjusted for menopause age: Age at menopause is added to the model as a covariable. There are no additional covariables in the model.

**Supplementary Table 7.** Model estimates and 95% confidence intervals, individually adjusting for early cognitive and sociodemographic covariables, for the effect of 1-year increase in menopause age on standardised z-scores for the Addenbrooke's Cognitive Examination (ACE-III; total score and subdomains; n=746) the Preclinical Alzheimer's Cognitive Composite (PACC; n=197). Multiple imputation by chained equations (MICE) was applied with 50 imputations.

To assess whether effect attenuations from M0 to M1 in the main analyses were driven by a particular variable included in M1, childhood cognition, childhood socioeconomic position, and education covariables were individually adjusted for. The model p-values and confidence intervals are not presented here, the interest is in the change in effect estimates when adjusting for each covariable.

	M0 (unadjusted)	M0 + childhood cognition	M0 + childhood SEP	M0 + education
Outcome	β	β	β	β
ACE-III total	0.024	0.011	0.019	0.018
ACE-III attention and orientation	0.006	0.001	0.005	0.003
ACE-III language	0.012	0.001	0.008	0.008
ACE-III memory	0.023	0.013	0.019	0.019
ACE-III verbal fluency	0.016	0.006	0.012	0.012
ACE-III visuospatial function	0.020	0.011	0.017	0.016
PACC total	0.029	0.014	0.025	0.025
PACC DSST	0.033	0.017	0.029	0.028
PACC FNAME-12A	0.033	0.016	0.028	0.027
PACC logical memory delayed recall	0.031	0.014	0.028	0.027
PACC MMSE	0.021	0.008	0.016	0.015

β= z-score standardised coefficient beta; CI=confidence interval; DSST=Digit-Symbol Substitution Test; FNAME=Face-Name Associative Memory Examination; MMSE=Mini-Mental State Examination; SEP=socioeconomic position

**Supplementary Table 8.** Model estimates and 95% confidence intervals, individually adjusting for reproductive covariables, for the effect of 1-year increase in menopause age on standardised z-scores for the Addenbrooke's Cognitive Examination (ACE-III) total and ACE-III language, verbal fluency, and visuospatial function sub-domains (n=746), and on standardised z-scores for the Preclinical Alzheimer's Cognitive Composite (PACC) total and digit-symbol substitution test (DSST) and face-name associative memory examination (FNAME) sub-tests (n=197). Multiple imputation by chained equations (MICE) was applied with 50 imputations.

To assess whether negative confounding observed for ACE-III total, ACE-III language, ACE-III verbal fluency, ACE-III visuospatial function, PACC total, PACC DSST, and PACC FNAME-12A between M1 and M2 in the main analyses was driven by a particular variable included in M2, age at menarche, parity, and menopause type were individually adjusted for. The model p-values and confidence intervals are not presented here, as the interest is in the change in effect estimates when adjusting for each covariable.

	M1	M1 + age at menarche	M1 + parity	M1 + menopause type
Outcome	β	β	β	β
ACE-III total	0.009	0.009	0.010	0.012
ACE-III language	0.000	0.001	0.000	0.005
ACE-III verbal fluency	0.005	0.004	0.005	0.007
ACE-III visuospatial function	0.010	0.009	0.010	0.014
PACC total	0.015	0.016	0.014	0.018
PACC DSST	0.018	0.017	0.018	0.031
PACC FNAME-12A	0.017	0.019	0.016	0.034

β= z-score standardised coefficient beta; CI=confidence interval; DSST=Digit-Symbol Substitution Test; FNAME=Face-Name Associative Memory Examination; MMSE=Mini-Mental State Examination

M1: adjusted for childhood cognition, childhood socioeconomic position, and education

**Supplementary Table 9.** Menopause age-by-menopause type interactions, for outcomes which show negative confounding from menopause type in the main analyses. Model estimations for menopause age-by-menopause type (natural or surgical) interaction terms in fully adjusted models (M3) regressing menopause age (in years) on standardised Addenbrooke's Cognitive Examination (ACE-III) language, verbal fluency, and visuospatial fluency (N=746) and on standardised Preclinical Alzheimer's Cognitive Composite (PACC) digit-symbol substitution test (DSST) and face-name associative memory examination (FNAME-12A) (N=197).

Outcome	Menopause Age x Menopause Type β	Lower Cl	Upper Cl	Р
ACE-III language*	0.005	-0.029	0.031	-
ACE-III verbal fluency*	-0.014	-0.047	0.014	-
ACE-III visuospatial function*	0.023	-0.005	0.051	-
PACC DSST	-0.045	-0.110	0.020	0.171
PACC FNAME-12A	-0.028	-0.094	0.038	0.399

β=z-score standardised coefficient beta; CI=confidence interval. Reference category: natural menopause

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

M3: adjusted for childhood cognition, childhood socioeconomic position, education, age at menarche, parity, BMI, smoking, APOE- $\epsilon$ 4 status, affective symptoms, age at cognitive testing (PACC measures only)

**Supplementary Table 10.** Model estimations for type of menopause on standardised Addenbrooke's Cognitive Examination (ACE-III) total and subdomain scores and on standardised Preclinical Alzheimer's Cognitive Composite (PACC) total and sub-test scores. Multiple imputation is not applied because there is no missing data for the menopause age covariable.

			Unadj	usted		Adjusted for menopause age			
Outcome	N	β	Lower Cl	Upper Cl	Р	β	Lower Cl	Upper Cl	Р
ACE-III total*	746	-0.131	-0.300	0.044	-	0.084	-0.149	0.296	-
ACE-III attention and orientation*	746	-0.031	-0.223	0.135	-	0.023	-0.174	0.208	-
ACE-III language*	746	-0.013	-0.182	0.139	-	0.122	-0.085	0.335	-
ACE-III memory*	746	-0.164	-0.349	0.008	-	0.019	-0.201	0.240	-
ACE-III verbal fluency*	746	-0.096	-0.253	0.056	-	0.051	-0.178	0.244	-
ACE-III visuospatial function*	746	-0.088	-0.249	0.084	-	0.103	-0.098	0.296	-
PACC total	197	-0.216	-0.444	0.013	0.064	0.039	-0.262	0.339	0.799
PACC DSST	197	-0.133	-0.434	0.169	0.386	0.234	-0.161	0.629	0.245
PACC FNAME-12A	195	-0.104	-0.406	0.199	0.500	0.249	-0.153	0.652	0.223
PACC logical memory delayed recall	197	-0.413	-0.710	-0.117	0.007	-0.286	-0.681	0.109	0.154
PACC MMSE*	197	-0.223	-0.606	0.082	-	-0.098	-0.485	0.312	-

β= z-score standardised coefficient beta; CI=confidence interval; DSST=digit-symbol substitution task FNAME-12A=face-name associative memory examination; MMSE=mini-mental state examination

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

*Reference category: natural menopause* 

### Model outputs from section 5i sensitivity analyses excluding women with ACE-III total <82

**Supplementary Table 11.** Model estimates and bootstrap 95% confidence intervals for the effect of 1-year increase in age at menopause on standardised z-scores for the Addenbrooke's Cognitive Examination (ACE-III) total score at age 69 in the National Survey of Health and Development (NSHD) whole-cohort, and the Preclinical Alzheimer's Cognitive Composite (PACC) at age 69 to 71 in the Insight 46 sample.

			M0			M1			M2			M3			M4	
Outcome	N	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
ACE-III	606	0.015	0.005,		0.000	-0.003,		0.011	0.001,		0.010	0.000,		0.010	-0.001,	
total*	696	0.015	0.024	-	0.006	0.014	-	0.011	0.021	-	0.010	0.021	-	0.010	0.020	<u>-</u> נ
DACC total	102	0.026	0.010,	0 002	0.014	-0.001,	0.072	0.015	-0.006,	0 1 5 0	0.016	-0.005,	0 1 2 2	0.016	-0.005,	0 1 2 2
PACC TOTAL 192	0.026 0.043 0.00	0.002	0.014	0.030	0.072	0.015	0.036	0.150	0.016	0.036	0.133	0.016	0.036	0.133		

 $\beta$ = z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable.

M0: unadjusted; M1: childhood cognition, childhood socioeconomic position, education; M2: M1+ age at menarche, parity, menopause type; M3: M2+ BMI, smoking, APOE-ε4 status, affective symptoms (+age at cognitive testing for PACC outcome only); M4: M3+ hormone therapy (HT)

**Supplementary Table 12.** Model estimations for menopause age-by-menopause type (natural or surgical) interaction terms in fully adjusted models (M3) regressing menopause age (in years) on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

Outcome	N	Menopause Age x Menopause Type β	Lower Cl	Upper Cl	Ρ
ACE-III total*	696	0.007	-0.012	0.029	-
PACC total	192	0.001	-0.042	0.044	0.997

 $\beta$ =z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: natural menopause

M3: adjusted for childhood cognition, childhood socioeconomic position, education, age at menarche, parity, BMI, smoking, APOE- $\epsilon$ 4 status, affective symptoms, age at cognitive testing [PACC total only]

**Supplementary Table 13.** Model estimations for menopause age-by-APOE-ε4 status (ε4 carrier or non-carrier) interaction terms in fully adjusted models (M3) regressing menopause age (in years) on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

Outcome	N	Menopause Age x <i>APOE</i> β	Lower Cl	Upper Cl	Ρ
ACE-III total*	696	-0.001	-0.002	0.003	-
PACC total	192	0.004	-0.030	0.038	0.813

 $\beta$ = z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: APOE-ɛ4 non-carrier

M3: adjusted for childhood cognition, childhood socioeconomic position, education, age at menarche, parity, menopause type, BMI, smoking, affective symptoms, age at cognitive testing [PACC total only]

**Supplementary Table 14.** Model estimations and 95% confidence intervals for the effect of ever using menopausal hormone therapy (HT), compared with never using HT, on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

_				Unadj	usted		Adjusted for menopause age				
Outcom	ne	Ν	β	Lower Cl	Upper Cl	Р	β	Lower Cl	Upper Cl	Р	
ACE-III to	tal*	696	-0.063	-0.177	0.056	-	-0.023	-0.158	0.097	-	
PACC to	tal	192	-0.042	-0.259	0.174	0.700	0.029	-0.188	0.246	0.793	

 $\beta$ =z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: HT never used

Unadjusted: only the exposure (HT use) and the outcome are included in the model.

Adjusted for menopause age: Age at menopause is added to the model as a covariable. There are no additional covariables in the model.

### Model outputs from section 5i sensitivity analyses excluding women who had surgical menopause via bilateral oophorectomy

**Supplementary Table 15.** Model estimates and bootstrap 95% confidence intervals for the effect of 1-year increase in age at menopause on standardised z-scores for the Addenbrooke's Cognitive Examination (ACE-III) total score at age 69 in the National Survey of Health and Development (NSHD) whole-cohort, and the Preclinical Alzheimer's Cognitive Composite (PACC) at age 69 to 71 in the Insight 46 sample.

		M0		M1		M2		M3		M3						
Outcome	Ν	β	95% CI	Р	β	Lower Cl	Ρ	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
ACE-III	663	0.023	0.011,	_	0.012	-0.001,	_	0.015	-0.001,	_	0.015	-0.003,	_	0.014	-0.001,	_
total*	005	0.025	0.037		0.012	0.021	-	0.015	0.031		0.015	0.030		0.014	0.030	-
DACC total	172	0 020	0.010,	0 002	0.014	-0.004,	0 1 2 6	0.011	-0.014,	0 202	0.012	-0.014,	0 242	0.012	-0.014,	0 2 4 2
FACCIOID	1/5	0.050	0.050	0.005	0.014	0.032	0.130	0.011	0.037	0.362	0.015	0.039	0.545	0.015	0.039	0.545

 $\beta$ = *z*-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable.

*M0: unadjusted; M1: childhood cognition, childhood socioeconomic position, education; M2: M1+ age at menarche, parity, menopause type; M3: M2+ BMI, smoking, APOE-ε4 status, affective symptoms (+age at cognitive testing for PACC outcome only); M4: M3+ hormone therapy (HT)* 

**Supplementary Table 16.** Model estimations for menopause age-by-APOE-ε4 status (ε4 carrier or non-carrier) interaction terms in fully adjusted models (M3) regressing menopause age (in years) on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

Outcome	Ν	Menopause Age x <i>APOE</i> β	Lower Cl	Upper Cl	Р
ACE-III total*	663	-0.001	-0.003	0.003	-
PACC total	173	0.008	-0.032	0.047	0.710

 $\beta$ = *z*-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: APOE-ɛ4 non-carrier

M3: adjusted for childhood cognition, childhood socioeconomic position, education, age at menarche, parity, menopause type, BMI, smoking, affective symptoms, age at cognitive testing [PACC total only]

**Supplementary Table 17.** Model estimations and 95% confidence intervals for the effect of ever using menopausal hormone therapy (HT), compared with never using HT, on standardised Addenbrooke's Cognitive Examination (ACE-III) total and Preclinical Alzheimer's Cognitive Composite (PACC) total performance, in the National Survey for Health and Development (NSHD) whole-cohort and Insight 46 samples, respectively.

			Unadj	usted		Adj	usted for m	enopause a	age
Outcome	Ν	β	Lower Cl	Upper Cl	Р	β	Lower Cl	Upper Cl	Р
ACE-III total*	663	-0.068	-0.230	0.108	-	-0.009	-0.222	0.140	-
PACC total	173	-0.041	-0.297	0.215	0.752	0.031	-0.226	0.287	0.814

 $\beta$ =z-score standardised coefficient beta; CI=confidence interval

\*Bootstrapping applied; bias-corrected confidence intervals; p-values not applicable

Reference category: HT never used

Unadjusted: only the exposure (HT use) and the outcome are included in the model.

Adjusted for menopause age: Age at menopause is added to the model as a covariable. There are no additional covariables in the model.

### 5ii. Supplementary material

**Supplementary table 18.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in neuroimaging outcomes with each 1-year increase in menopause age. Models are fully adjusted and include a quadratic term for menopause age.

Outcome variable	Ν	Menopause age <sup>2</sup> β	Lower Cl	Upper Cl	Р
SUVR	126	-0.001	-0.002	0.000	0.015
Amyloid status [OR;	126	0.914	0.833	1.002	0.055
reference=negative]					
WMHV*	123	0.999	0.988	1.009	0.786
TBV*	126	-0.141	-0.628	0.345	0.566
Hippocampal volume*	126	0.001	-0.002	0.004	0.690
NAWM FA	110	-0.002	-0.005	0.001	0.132
NAWM MD	110	0.002	-0.003	0.006	0.456
CT: Harvard ADsig	126	0.000	-0.001	0.001	0.573
CT: Frontal	126	0.000	-0.001	0.001	0.557
CT: Occipital	126	0.000	-0.001	0.001	0.664
CT: Parietal	126	0.000	0.000	0.001	0.462
CT: Temporal	126	0.001	0.000	0.002	0.300

Models are adjusted for age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure, APOE-ε4 status [\*and TIV]

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; TIV=total intracranial volume; SUVR=standardised uptake value ratio; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions

**Supplementary table 19.** Model estimates and 95% confidence intervals demonstrating the predicted interaction of APOE- $\varepsilon$ 4 status (reference= $\varepsilon$ 4 non-carrier) with non-linear change in amyloid SUVR and likelihood of being amyloid positive with each 1-year increase in menopause age. Models are fully adjusted and include a quadratic term for menopause age and APOE- $\varepsilon$ 4-by-menopause age interaction terms. N=126

Outcome variable	APOE-by-menopause age <sup>2</sup> β/OR	Lower Cl	Upper Cl	Ρ
SUVR	-0.002	-0.004	0.000	0.103
Amyloid status*	0.966	0.755	1.235	0.782

\*Reference category=amyloid negative; OR

Models are adjusted for age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, and blood pressure.

SUVR=standardised uptake value ratio; CI=confidence interval; OR=odds ratio; SEP=socioeconomic position
**Supplementary table 20.** Model estimates and 95% confidence intervals demonstrating the predicted interaction of APOE- $\epsilon$ 4 status (reference= $\epsilon$ 4 non-carrier) with linear change in neuroimaging outcomes with each 1-year increase in menopause age. The models are fully adjusted and include an APOE- $\epsilon$ 4-by-menopause age interaction term. N=123

Outcome variable	Ν	APOE-by-menopause	Lower Cl	Upper Cl	Р
		age β			
WMHV*	123	0.868	0.767	0.982	0.025
TBV*	126	4.105	-2.110	10.319	0.193
Hippocampal volume*	126	-0.019	-0.057	0.019	0.332
NAWM FA	110	0.020	-0.021	0.060	0.341
NAWM MD	110	-0.043	-0.101	0.014	0.140
CT: Harvard ADsig	126	0.012	0.002	0.022	0.016
CT: Frontal	126	0.010	-0.002	0.023	0.091
CT: Occipital	126	0.010	-0.002	0.022	0.093
CT: Parietal	126	0.013	0.003	0.022	0.012
CT: Temporal	126	0.014	0.001	0.027	0.037

Model is adjusted for age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure [\*and TIV]

CI=confidence interval; SEP=socioeconomic position; TIV=total intracranial volume; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions

**Supplementary table 21.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in amyloid SUVR and likelihood of being amyloid positive with each 1-year increase in menopause age, with cumulative adjustments for life course covariables. Models include a quadratic term for menopause age. N=126

	MO			M1		M2				M3		M3+HT			
Outcome	β/OR	95% CI	Р												
variable															
SUVR	-0.001	-0.002,	0.015	-0.001	-0.002,	0.012	-0.001	-0.002,	0.008	-0.001	-0.002,	0.015	-0.001	-0.002,	0.015
		0.000			0.000			0.000			0.000			0.000	
Amyloid	0.930	0.864,	0.052	0.933	0.868,	0.060	0.924	0.855,	0.045	0.914	0.833,	0.055	0.913	0.833,	0.053
status*		1.001			1.003			0.998			1.002			1.001	
* ~ (	· .			0.0			•								

\*Reference category=amyloid negative; OR

M0:age at scan; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ε4 status

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; HT=hormone therapy; SUVR=standardised uptake value ratio

**Supplementary table 22.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age among APOE-ε4 carriers and non-carriers), with cumulative adjustments for life course covariables.

				M0		M1		M2		M3			M3+HT				
ΑΡΟΕ-ε4	Outcome	Ν	β	95% CI	Р												
	WMHV*	35	0.885	0.792,	0.032	0.879	0.783,	0.029	0.877	0.766,	0.057	0.923	0.783,	0.341	0.926	0.783,	0.374
				0.989			0.987			1.004			1.088			1.096	
	CT: Harvard	36	0.009	0.002,	0.018	0.009	0.001,	0.037	0.007	-0.002,	0.113	0.007	-0.004,	0.187	0.007	-0.004,	0.209
	ADsig			0.017			0.017			0.016			0.018			0.018	
	CT: Frontal	36	0.007	-0.002,	0.130	0.006	-0.003,	0.193	0.006	-0.005,	0.237	0.006	-0.007,	0.318	0.006	-0.007,	0.348
Carriers				0.016			0.016			0.017			0.019			0.019	
	CT: Occipital	36	0.006	-0.004,	0.236	0.007	-0.004,	0.207	0.006	-0.006,	0.296	0.005	-0.010,	0.469	0.005	-0.010,	0.497
				0.017			0.019			0.019			0.020			0.021	
	CT: Parietal	36	0.011	0.004,	0.003	0.011	0.003,	0.009	0.010	0.001,	0.024	0.009	-0.001,	0.074	0.009	-0.001,	0.086
				0.019			0.018			0.019			0.020			0.019	
	CT:	36	0.010	-0.001,	0.064	0.011	0.000,	0.049	0.010	-0.003,	0.113	0.008	-0.006,	0.232	0.008	-0.007,	0.256
	Temporal			0.020			0.022			0.022			0.023			0.023	
	WMHV*	88	1.038	0.979,	0.209	1.040	0.979,	0.204	1.032	0.969,	0.326	1.023	0.963,	0.459	1.024	0.964,	0.442
				1.100			1.105			1.098			1.086			1.088	
	CT: Harvard	90	-0.004	-0.009,	0.092	-0.004	-0.009,	0.080	-0.003	-0.008,	0.166	-0.004	-0.009,	0.134	-0.005	-0.009,	0.067
	ADsig			0.001			0.001			0.001			0.001			0.000	
	CT: Frontal	90	-0.003	-0.009,	0.267	-0.004	-0.010,	0.195	-0.003	-0.009,	0.299	-0.003	-0.009,	0.255	-0.004	-0.010,	0.167
Non-				0.002			0.002			0.003			0.003			0.002	
carriers	CT: Occipital	90	-0.003	-0.009,	0.195	-0.004	-0.009,	0.170	-0.003	-0.009,	0.253	-0.004	-0.010,	0.157	-0.005	-0.010,	0.066
				0.002			0.002			0.002			0.002			0.000	
	CT: Parietal	90	-0.003	-0.008,	0.161	-0.003	-0.008,	0.186	-0.002	-0.007,	0.358	-0.002	-0.007,	0.340	-0.003	-0.008,	0.228
				0.001			0.002			0.003			0.003			0.002	
	CT:	90	-0.004	-0.010,	0.173	-0.004	-0.010,	0.268	-0.004	-0.010,	0.291	-0.004	-0.011,	0.198	-0.005	-0.012,	0.100
	Temporal			0.002			0.003			0.003			0.002			0.001	

M0:age at scan, [\*TIV]; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

CI=confidence interval; SEP=socioeconomic position; TIV=total intracranial volume; HT=hormone therapy; WMHV=white matter hyperintensity volume; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions

**Supplementary table 23.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age, with cumulative adjustments for life course covariables, with APOE pooled.

			M0			M1			M2			M3			M3+HT	
Outcome	Ν	β	95% CI	Р												
TBV*	126	3.167	0.717,	0.012	3.224	0.732,	0.012	2.861	0.309,	0.028	2.417	-0.152,	0.065	2.419	-0.159,	0.066
			5.617			5.716			5.414			4.986			4.998	
Hippocampal	126	0.008	-0.007,	0.275	0.009	-0.006,	0.260	0.010	-0.005,	0.196	0.009	-0.007,	0.262	0.009	-0.007,	0.258
volume*			0.023			0.024			0.026			0.025			0.024	
NAWM FA	110	-0.008	-0.022,	0.292	-0.007	-0.022,	0.357	-0.007	-0.023,	0.352	-0.006	-0.022,	0.447	-0.006	-0.023,	0.439
			0.007			0.008			0.008			0.010			0.010	
NAWM MD	110	0.012	-0.009,	0.261	0.009	-0.012,	0.405	0.008	-0.015,	0.493	0.007	-0.017,	0.575	0007	-0.016,	0.559
			0.032			0.030			0.030			0.030			0.030	

M0:age at scan, [\*TIV]; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ε4 status

CI=confidence interval; SEP=socioeconomic position; TIV=total intracranial volume; HT=hormone therapy; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity

**Supplementary table 24.** Model estimates and 95% confidence intervals demonstrating the predicted difference in neuroimaging outcomes measures in hormone therapy (HT) ever users compared with never users.

			N	10		M0 + menopause age				
Outcome	Ν	β	Lower	Upper	Р	β	Lower	Upper	Р	
			CI	CI			CI	CI		
SUVR	121	0.002	-0.025	0.028	0.894	0.001	-0.025	0.028	0.926	
Aβ status (OR;	121	0.359	0.080	1.613	0.182	0.354	0.078	1.599	0.177	
reference=negative)										
WMHV*	118	0.097	-0.282	0.475	0.617	0.111	-0.268	0.489	0.567	
TBV*	121	-4.492	-21.559	12.575	0.603	-5.445	-22.200	11.309	0.521	
Hippocampal	121	-0.078	-0.174	0.019	0.115	-0.081	-0,177	0.016	0.101	
volume*										
NAWM FA	105	0.029	-0.067	0.126	0.546	0.033	-0.064	0.130	0.498	
NAWM MD	105	-0.028	-0.162	0.106	-0.679	-0.033	-0.168	0.101	0.623	
CT: Harvard ADsig	121	-0.015	-0.042	0.011	0.261	-0.015	-0.041	0.012	0.282	
CT: Frontal	121	-0.014	-0.046	0.017	0.367	-0.014	-0.046	0.018	0.384	
CT: Occipital	121	-0.029	-0.060	0.002	0.062	-0.029	-0.059	0.002	0.069	
CT: Parietal	121	-0.012	-0.038	0.014	0.361	-0.012	-0.038	0.015	0.375	
CT: Temporal	121	-0.028	-0.062	0.006	0.101	-0.027	-0.061	0.006	0.111	

Reference category: Never use MO=scan age, [\*TIV]

CI=confidence interval; TIV=total intracranial volume; HT=hormone therapy; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions

**Supplementary table 25.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in WMHV with each 1-year increase in menopause age among APOE-ε4 carriers, cumulatively adjusting for health-related (M3) covariables. N=35

		M2+smokin	g	M2+BMI			M2+SBP+DBP				M2+SBP		M2+DBP			
Outcome	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	
WMHV	0.934	0.807,	0.356	0.880	0.772,	0.056	0.877	0.755,	0.088	0.877	0.773,	0.041	0.892	0.780,	0.097	
		1.080			1.003			1.020			0.995			1.021		

M0:age at scan, TIV; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

WMHV=white matter hyperintensity volume; CI=confidence interval; SEP=socioeconomic position; TIV=total intracranial volume; SBP=systolic blood pressure; DBP=diastolic blood pressure

**Supplementary table 26.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in TBV(cm<sup>3</sup>) with each 1-year increase in menopause age, cumulatively adjusting for health-related (M3) covariables. N=126

	M2+smoking		g		M2+BMI		M2+SBP+DBP		M2+SBP			M2+DBP			M2+APOE-E4 status			
Outcome	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
TBV	2.785	0.229,	0.033	2.860	0.298,	0.029	2.544	0.004,	0.050	2.746	0.191,	0.035	2.853	0.292,	0.029	2.796	0.227,	0.033
		5.342			5.422			5.084			5.302			5.415			5.365	

M0:age at scan, TIV; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ε4 status

TBV=total brain volume; CI=confidence interval; SEP=socioeconomic position; TIV=total intracranial volume; SBP=systolic blood pressure; DBP=diastolic blood pressure

Supplementary table 27. Model estimates and 95% confidence intervals demonstrating the predicted linear change in cortical thickness(mm) in the Harvard ADsig ROI with each 1-year increase in menopause age among APOE- $\varepsilon$ 4 carriers, cumulatively adjusting for reproductive (M2) covariables. N=36

		M1+p	uberty			M1+	parity	
Outcome variable	β	Lower Cl	Upper Cl	Р	β	Lower Cl	Upper Cl	Р
Harvard ADsig	0.009	0.000	0.017	0.043	0.008	-0.001	0.017	0.098

M0:age at scan; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

Harvard ADsig=Harvard Alzheimer's Disease signature regions; ROI=regions of interest; CI=confidence interval; SEP=socioeconomic position

**Supplementary table 28.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in cortical thickness(mm) in the parietal ROI with each 1-year increase in menopause age among APOE-ε4 carriers, cumulatively adjusting for health-related (M3) covariables. N=36

	M2+smoking			M2+BMI			M2+SBP+DBP				M2+SBP		M2+DBP		
Outcome	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
variable														<u> </u>	
Parietal	0.010	0.000,	0.045	0.010	0.001,	0.034	0.010	0.001,	0.039	0.010	0.001,	0.027	0.010	0.001,	0.033
		0.019			0.019			0.019			0.019			0.019	

M0:age at scan; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

Harvard ADsig=Harvard Alzheimer's Disease signature regions; ROI=regions of interest; CI=confidence interval; SEP=socioeconomic position; SBP=systolic blood pressure; DBP=diastolic blood pressure

## Model outputs from section 5ii sensitivity analyses adjusting for WMHV

**Supplementary table 29.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in amyloid SUVR and in the likelihood of being amyloid positive with each 1-year increase in menopause age, in fully-adjusted models (M3). Models include a quadratic term for menopause age. N=126

Outcome variable	Menopause age <sup>2</sup> β/OR	Lower Cl	Upper Cl	Р
SUVR	-0.001	-0.002	0.000	0.015
Amyloid status*	0.915	0.835	1.002	0.056

\*OR; Reference category=amyloid negative

Models are adjusted for: age at scan, WMHV, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure, APOE-E4 status

CI=confidence interval; OR=odds ratio; HT=hormone therapy; SEP=socioeconomic position; WMHV=white matter hyperintensity volume; SUVR=standardised uptake value ratio

**Supplementary table 30.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age, in fully-adjusted models (M3).

Ν	β	Lower Cl	Upper Cl	Ρ
126	2.404	-0.179	4.986	0.068
126	0.009	-0.007	0.024	0.278
110	-0.005	-0.021	0.010	0.493
110	0.005	-0.017	0.028	0.634
	N 126 126 110 110	Νβ1262.4041260.009110-0.0051100.005	NβLower Cl1262.404-0.1791260.009-0.007110-0.005-0.0211100.005-0.017	Ν β Lower Cl Upper Cl   126 2.404 -0.179 4.986   126 0.009 -0.007 0.024   110 -0.005 -0.021 0.010   110 0.005 -0.017 0.028

Models are adjusted for: age at scan, [\*TIV], WMHV, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure, APOE-ε4 status

CI=confidence interval; SEP=socioeconomic position; TIV=total intracranial volume; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity

**Supplementary table 31.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in cortical thickness(mm) ROI with each 1-year increase in menopause age among APOE-ε4 carriers (n=36) and non-carriers (n=90), in fully-adjusted models (M3).

APOE-ε4 status	Outcome variable	β	Lower Cl	Upper Cl	Р
	Harvard ADsig	0.009	-0.003	0.022	0.129
	Frontal	0.006	-0.009	0.021	0.415
Carriers	Occipital	0.005	-0.012	0.022	0.535
	Parietal	0.011	-0.001	0.022	0.068
	Temporal	0.012	-0.004	0.028	0.137
	Harvard ADsig	-0.003	-0.008	0.002	0.178
	Frontal	-0.003	-0.009	0.003	0.309
Non-carriers	Occipital	-0.003	-0.009	0.002	0.236
	Parietal	-0.002	-0.007	0.003	0.405
	Temporal	-0.004	-0.011	0.003	0.259

Models are adjusted for: age at scan, WMHV, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure

Harvard ADsig=Harvard Alzheimer's Disease signature regions; ROI=regions of interest; CI=confidence interval; SEP=socioeconomic position; WMHV=white matter hyperintensity volume

## Model outputs from section 5ii sensitivity analyses excluding women with clinically diagnosed neurological conditions

**Supplementary table 32.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in neuroimaging outcomes with each 1-year increase in menopause age. Models are fully adjusted and include a quadratic term for menopause age.

Outcome variable	Ν	Menopause age <sup>2</sup> β	Lower Cl	Upper Cl	Р
SUVR	114	-0.001	-0.002	0.000	0.006
Amyloid status {OR;	114	0.980	0.948	1.013	0.232
reference=negative]					

Models are adjusted for age at scan, [\*TIV], childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure, and APOE-ε4 status.

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; TIV-total intracranial volume; SUVR=standardised uptake value ratio; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions

**Supplementary table 33.** Model estimates and 95% confidence intervals demonstrating the predicted interaction of APOE- $\epsilon$ 4 status (reference= $\epsilon$ 4 non-carrier) with non-linear change in amyloid SUVR with each 1-year increase in menopause age. The model is fully adjusted and includes a quadratic term for menopause age and APOE- $\epsilon$ 4-by-menopause age interaction terms. N=114

Outcome variable	APOE-by-menopause $age^2 \beta$	Lower Cl	Upper Cl	р
SUVR	-0.001	-0.004	0.001	0.303

Models are adjusted for age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, and blood pressure.

CI=confidence interval; SEP=socioeconomic position; SUVR=standardised uptake value ratio

**Supplementary table 34.** Model estimates and 95% confidence intervals demonstrating the predicted interaction of APOE- $\epsilon$ 4 status (reference= $\epsilon$ 4 non-carrier) with linear change in neuroimaging outcomes with each 1-year increase in menopause age. The models are fully adjusted and include an APOE- $\epsilon$ 4-by-menopause age interaction term.

Outcome variable	Ν	APOE-by-menopause	Lower Cl	Upper Cl	Р
		age β			
Amyloid status [OR;	114	1.126	0.813	1.559	0.476
reference=negative]					
WMHV*	113	0.858	0.752	0.979	0.023
TBV*	114	5.551	-0.7122	11.813	0.082
Hippocampal volume*	114	-0.026	-0.065	0.013	0.187
NAWM FA	102	0.037	-0.005	0.078	0.082
NAWM MD	102	-0.069	-0.126	-0.012	0.017
CT: Harvard ADsig	114	0.014	0.004	0.024	0.007
CT: Frontal	114	0.011	-0.002	0.023	0.104
CT: Occipital	114	0.011	-0.002	0.023	0.091
CT: Parietal	114	0.015	0.005	0.025	0.003
CT: Temporal	114	0.014	0.000	0.027	0.050

Reference category=amyloid negative

Models are adjusted for age at scan, [\*TIV], childhood cognition, education, SEP, puberty, parity, smoking, BMI, and blood pressure.

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; TIV=total intracranial volume; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions **Supplementary table 35.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in amyloid SUVR with each 1-year increase in menopause age, with cumulative adjustments for life course covariables. The model includes a quadratic term for menopause age. *N*=114

		M0 M1 M2 M3								M3+HT					
Outcome	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
SUVR	-0.001	-0.002,	0.013	-0.001	-0.002,	0.015	-0.001	-0.002,	0.009	-0.001	-0.002,	0.006	-0.001	-0.002,	0.007
	0.000				0.000		0.000				0.000		0.000		

M0:age at scan; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ε4 status

SUVR=standardised uptake value ratio; CI=confidence interval; SEP=socioeconomic position; HT=hormone therapy

**Supplementary table 36.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age, with cumulative adjustments for life course covariables.

						M1		M2			M3			M3+HT		
Outcome	Ν	β	95% CI	Р												
Amyloid status [OR;	114	1.018	0.903,	0.769	1.032	0.912,	0.617	1.023	0.900,	0.727	1047	0.911,	0.516	1.047	0.911,	0.515
reference=negative]			1.148			1.167			1.164			1.204			1.204	
Hippocampal volume*	114	0.008	-0.008,	0.310	0.009	-0.007,	0.259	0.009	-0.008,	0.299	0.007	-0.010,	0.431	0.008	-0.010,	0.389
			0.024			0.026			0.026			0.024			0.025	
CT: Frontal	114	-	-0.006,	0.656	-	-0.006,	0.634	-	-0.006,	0.788	-	-0.007,	0.739	-	-0.007,	0.759
		0.001	0.004		0.001	0.004		0.001	0.005		0.001	0.005		0.001	0.005	

M0:age at scan, [\*TIV]; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ε4 status

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; HT=hormone therapy; TIV=total intracranial volume; CT=cortical thickness

**Supplementary table 37.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age among APOE-ε4 carriers and non-carriers, with cumulative adjustments for life course covariables.

				M0			M1			M2			M3			M3+HT	
ΑΡΟΕ-ε4	Outcome	Ν	β	95% CI	р												
	WMHV*	34	0.881	0.787,	0.028	0.876	0.778,	0.029	0.860	0.753,	0.026	0.910	0.779,	0.232	0.915	0.778,	0.280
				0.987			0.986			0.982			1.062			1.075	
	TBV*	35	5.717	-0.204,	0.058	5.881	-0.515,	0.070	5.258	-1.664,	0.129	3.711	-4.301,	0.341	3.774	-4.534,	0.349
				11.637			12.277			12.181			11.723			12.082	
	NAWM FA	30	0.010	-0.020,	0.485	0.007	-0.027,	0.689	0.004	-0.032,	0.829	0.003	-0.037,	0.857	0.004	-0.038,	0.838
				0.040			0.041			0.03			0.044			0.046	
	NAWM MD	30	-	-0.062,	0.112	-	-0.069,	0.145	-	-0.077,	0.216	-	-0.086,	0.228	-	-0.088,	0.226
Carriers			0.028	0.007		0.029	0.011		0.029	0.019		0.032	0.023		0.032	0.023	
currers	CT: Harvard	35	0.009	0.001,	0.025	0.008	-0.001,	0.069	0.007	-0.002,	0.136	0.007	-0.003,	0.173	0.007	-0.004,	0.204
	ADsig			0.017			0.016			0.016			0.017			0.018	
	CT: Occipital	35	0.006	-0.005,	0.305	0.006	-0.006,	0.287	0.005	-0.007,	0.393	0.005	-0.010,	0.497	0.005	-0.011,	0.531
				0.017			0.018			0.018			0.020			0.021	
	CT: Parietal	35	0.011	0.004,	0.004	0.010	0.002,	0.016	0.010	0.001,	0.029	0.009	-0.001,	0.072	0.009	-0.002,	0.092
				0.019			0.018			0.018			0.019			0.020	
	CT: Temporal	35	0.008	-0.002,	0.100	0.009	-0.002,	0.098	0.009	-0.004,	0.155	0.009	-0.005,	0.211	0.009	-0.006,	0.223
				0.019			0.020			0.021			0.023			0.023	
	WMHV*	79	1.053	0.985,	0.131	1.045	0.975,	0.210	1.034	0.961,	0.371	1.016	0.947,	0.656	1.018	0.949,	0.619
				1.125			1.121			1.113			1.091			1.091	
	TBV*	79	0.861	-2.135,	0.569	1.244	-1.826,	0.422	0.411	-2.861,	0.803	0.388	-2.878,	0.813	0.347	-2.941,	0.834
				3.856			4.314			3.682			3.653			3.634	
	NAWM FA	72	-	-0.043, -	0.017	-	-0.046, -	0.018	-	-0.052, -	0.014	-	-0.050, -	0.019	-	-0.051, -	0.020
			0.024	0.004		0.025	0.004		0.029	0.006		0.028	0.005		0.028	0.005	
	NAWM MD	72	0.035	0.008,	0.011	0.037	0.008,	0.012	0.039	0.007,	0.017	0.038	0.005,	0.023	0.038	0.006,	0.022
Non-				0.0622			0.065			0.070			0.070			0.070	
carriers	CT: Harvard	79	-	-0.011, -	0.011	-	-0.012, -	0.008	-	-0.012, -	0.027	-	-0.012,	0.038	-	-0.012, -	0.025
	ADsig		0.006	0.001		0.007	0.002		0.006	0.001		0.006	0.000		0.006	0.001	
	CT: Occipital	79	-	-0.011,	0.079	-	-0.012,	0074	-	-0.011,	0.155	-	-0.011,	0.179	-	-0.011,	0.112
			0.005	0.001		0.006	0.001		0.005	0.002		0.004	0.002		0.005	0.001	
	CT: Parietal	79	-	-0.011, -	0.018	-	-0.011, -	0.016	-	-0.011,	0.064	-	-0.010,	0.101	-	-0.010,	-
			0.006	0.001		0.006	0.001		0.005	0.000		0.005	0.001		0.005	0.001	0.080
	CT: Temporal	79	-	-0.012,	0.148	-	-0.011,	0.219	-	-0.012,	0.238	-	-0.012,	0.226	-	-0.013,	0.169
			0.005	0.002		0.004	0.003		0.005	0.003		0.005	0.003		0.005	0.002	

M0:age at scan, [\*TIV]; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

CI=confidence interval; SEP=socioeconomic position; HT=hormone therapy; TIV=total intracranial volume; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions

## Model outputs from section 5ii sensitivity analyses excluding women with early menopause (age <45 years)

**Supplementary table 38.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in amyloid standardised uptake value ratio (SUVR) and odds ratio for being amyloid positive, with each 1-year increase in menopause age. Models are fully adjusted and include a quadratic term for menopause age.

Outcome variable	N	Menopause age <sup>2</sup> β	Lower CI	Upper Cl	Р
SUVR	124	-0.002	-0.003	-0.000	0.007
Amyloid status {OR;	124	0.915	0.834	1.003	0.058
reference=negative]					

reference=negative] Models are adjusted for age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, blood pressure, and APOE-ε4 status.

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; SUVR=standardised uptake value ratio

Supplementary table 39. Model estimates and 95% confidence intervals demonstrating the predicted interaction of APOE- $\varepsilon$ 4 status (reference= $\varepsilon$ 4 non-carrier) with non-linear change in amyloid SUVR and A6 odds ratio, with each 1-year increase in menopause age. Models are fully adjusted and include a quadratic term for menopause age and APOE- $\varepsilon$ 4-by-menopause age interaction terms.

Outcome variable	Ν	APOE-by-menopause $age^2 \beta$	Lower Cl	Upper Cl	р
SUVR	124	-0.002	-0.004	0.001	0.249
Amyloid status {OR; reference=negative]	124	0.969	0.759	1.237	0.800

Models are adjusted for age at scan, childhood cognition, education, SEP, puberty, parity, smoking, BMI, and blood pressure.

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; SUVR=standardised uptake value ratio

**Supplementary table 40.** Model estimates and 95% confidence intervals demonstrating the predicted interaction of APOE- $\epsilon$ 4 status (reference= $\epsilon$ 4 non-carrier) with linear change in neuroimaging outcomes with each 1-year increase in menopause age. The models are fully adjusted and include an APOE- $\epsilon$ 4-by-menopause age interaction term.

Outcome variable	N	APOE-by-menopause	Lower Cl	Upper Cl	Р
		age β			
WMHV*	121	-0.140	-0.270	-0.010	0.035
TBV*	124	5.739	-0.691	12.168	0.080
Hippocampal volume*	124	-0.014	-0.054	0.258	0.486
NAWM FA	109	0.028	-0.014	0.070	0.188
NAWM MD	109	-0.053	-0.112	0.007	0.081
CT: Harvard ADsig	124	0.012	0.002	0.022	0.023
CT: Frontal	124	0.010	-0.002	0.023	0.111
CT: Occipital	124	0.011	-0.001	0.024	0.076
CT: Parietal	124	0.012	0.002	0.022	0.019
CT: Temporal	124	0.012	-0.002	0.026	0.089

Reference category=amyloid negative

Models are adjusted for age at scan, [\*TIV], childhood cognition, education, SEP, puberty, parity, smoking, BMI, and blood pressure.

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; TIV=total intracranial volume; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; FA=fractional anisotropy; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions **Supplementary table 41.** Model estimates and 95% confidence intervals demonstrating the predicted non-linear change in amyloid SUVR and odds ratio for being amyloid positive, with each 1-year increase in menopause age, with cumulative adjustments for life course covariables. The models include a quadratic term for menopause age. N=124.

		M0			M1			M2	M2		M3			M3+HT		
Outcome	β	95% CI	Р													
SUVR	-0.001	-0.003,	0.012	-0.001	-0.003,	0.015	-0.002	-0.003,	0.014	-0.002	-0.003,	0.007	-0.002	-0.003,	0.007	
		-0.000			-0.000			-0.000			-0.000			-0.000		
Amyloid status {OR;	0.930	0.864,	0.052	0.933	0.868,	0.061	0.924	0.855,	0.045	0.915	0.834,	0.058	0.913	0.833,	0.056	
reference=negative]		1.001			1.003			0.998			1.003			1.002		

M0:age at scan; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-e4 status

SUVR=standardised uptake value ratio; CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; HT=hormone therapy

**Supplementary table 42.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age, with cumulative adjustments for life course covariables.

		M0 ß 95% CL P			M1		M2		M3			M3+HT				
Outcome	N	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р	β	95% CI	Р
Hippocampal	124	0.004	-0.013,	0.651	0.004	-0.013,	0.618	0.006	-0.012,	0.505	0.004	-0.014,	0.688	0.004	-0.014,	0.634
volume*			0.020			0.021			0.024			0.022			0.022	
NAWM FA	109	-0.012	-0.028,	0.123	-0.012	-0.028,	0.161	-0.013	-0.030,	0.147	-0.011	-0.028,	0.230	-0.011	-0.029,	0.217
			0.003			0.005			0.005			0.007			0.007	
CT: Frontal	124	-0.001	-0.006,	0.714	-0.001	-0.006,	0.739	0.000	-0.005,	0.917	0.001	-0.005,	0.756	0.001	-0.005,	0.703
			0.004			0.004			0.006			0.007			0.007	

M0:age at scan, [\*TIV]; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure, APOE-ε4 status

CI=confidence interval; OR=odds ratio; SEP=socioeconomic position; HT=hormone therapy; TIV=total intracranial volume; NAWM=normal appearing white matter; FA=fractional anisotropy; CT=cortical thickness

**Supplementary table 43.** Model estimates and 95% confidence intervals demonstrating the predicted linear change in neuroimaging outcomes with each 1-year increase in menopause age among APOE-ε4 carriers and non-carriers, with cumulative adjustments for life course covariables.

			<u>Μ0</u> N β 95% Cl p		M1		M2		M3			M3+HT					
ΑΡΟΕ-ε4	Outcome	Ν	β	95% CI	р	β	95% CI	р	β	95% CI	р	β	95% CI	р	β	95% CI	р
Carriers	WMHV*	35	-	-0.233, -	0.032	-	-0.245, -	0.029	-	-0.265,	0.062	-	-0.243,	0.341	-	-0.240,	0.382
			0.122	0.011		0.129	0.013		0.129	0.007		0.080	0.084		0.074	0.092	
	TBV*	36	5.715	-0.070,	0.053	5.869	-0.393,	0.065	5.418	-1.379,	0.113	4.012	-3.636,	0.284	4.087	-3.798,	0.289
				11.499			12.131			12.216			11.660			11.971	
	NAWM MD	31	-	-0.067,	0.202	-	-0.067,	0.342	-	-0.081,	0.347	-	-0.092,	0.299	-	-0.095,	0.309
			0.026	0.015		0.022	0.024		0.026	0.030		0.031	0.030		0.031	0.033	
	CT: Harvard	36	0.009	0.002,	0.018	0.009	0.001,	0.037	0.007	-0.002,	0.112	0.007	-0.004,	0.193	0.007	-0.004,	0.219
	ADsig			0.017			0.017			0.016			0.018			0.018	
	CT:	36	0.006	-0.004,	0.236	0007	-0.004,	0.207	0.006	-0.006,	0.300	0.005	-0.010,	0.475	0.005	-0.010,	0.511
	Occipital			0.017			0.019			0.019			0.020			0.020	
	CT: Parietal	36	0.011	0.004,	0.003	0.011	0.003,	0.009	0.010	0.001,	0.024	0.009	-0.001,	0.076	0.009	-0.002,	0.089
				0.019			0.018			0.019			0.019			0.019	
	CT:	36	0.010	-0.001,	0.064	0.011	0.000,	0.049	0.010	-0.002,	0.113	0.008	-0.006,	0.241	0.008	-0.007,	0.265
	Temporal			0.020			0.022			0.022			0.022			0.023	
Non-	WMHV*	86	0.045	-0.027,	0.217	0.049	-0.027,	0.204	0.034	-0.046,	0.408	0.014	-0.061,	0.715	0.014	-0.061,	0.711
carriers				0.117			0.126			0.114			0.089			0.090	
	TBV*	88	1.550	-1.677,	0.342	1.751	-1.588,	0.300	0.482	-3.120,	0.791	-	-3.623,	0.997	-	-3.663,	0.992
				4.776			5.091			4.084		0.006	3.611		0.019	3.626	
	NAWM MD	78	0.027	0.000,	0.050	0.027	-0.002,	0.067	0.028	-0.004,	0.083	0.029	-0.004,	0.088	0.029	-0.004,	0.086
				0.054			0.055			0.060			0.061			0.062	
	CT: Harvard	88	-	-0.010,	0.104	-	-0.010,	0.097	-	-0.010,	0.205	-	-0.009,	0.255	-	-0.009,	0.191
	ADsig		0.004	0.001		0.005	0.001		0.004	0.002		0.003	0.003		0.004	0.002	
	CT:	88	-	-0.011,	0.112	-	-0.012,	0.092	-	-0.012,	0.157	-	-0.012,	0.138	-	-0.012,	0.088
	Occipital		0.005	0.001		0.005	0.001		0.005	0.002		0.005	0.002		0.006	0.001	
	CT: Parietal	88	-	-0.009,	0.189	-	-0.009,	0.249	-	-0.008,	0.514	-	-0.008,	0.565	-	-0.008,	0.492
			0.003	0.002		0.003	0.002		0.002	0.004		0.002	0.004		0.002	0.004	
	CT:	88	-	-0.010,	0.423	-	-0.009,	0.652	-	-0.010,	0.725	-	-0.009,	0.737	-	-0.009,	0.633
	Temporal		0.003	0.004		0.002	0.006		0.001	0.007		0.001	0.007		0.002	0.006	

M0:age at scan, [\*TIV]; M1:M0+childhood cognition, education, SEP; M2:M1+puberty, parity; M3:M2+smoking, BMI, blood pressure

ICI=lower confidence interval; uCI=upper confidence interval; SEP=socioeconomic position; HT=hormone therapy; TIV=total intracranial volume; WMHV=white matter hyperintensity volume; TBV=total brain volume; NAWM=normal appearing white matter; MD=mean diffusivity; CT=cortical thickness; Harvard ADsig=Harvard Alzheimer's Disease signature regions